

Modularity in supersymmetric gauge theory

Elias Furrer

School of Mathematics Trinity College Dublin

This thesis is submitted for the degree of *Doctor of Philosophy*.

2023

Supervisor: Prof. Dr. Jan Manschot

Declaration

I declare that this thesis has not been submitted as an exercise for a degree at this or any other university and it is entirely my own work. I agree to deposit this thesis in the University's open access institutional repository or allow the Library to do so on my behalf, subject to Irish Copyright Legislation and Trinity College Library conditions of use and acknowledgement. I consent to the examiner retaining a copy of the thesis beyond the examining period, should they so wish.

Parts of this thesis are based on the publications [1–4] and the preprint [5], which contains work in collaboration with Johannes Aspman, Georgios Korpas, Jan Manschot, Zhi-Cong Ong and Meng-Chwan Tan.

Elias Furrer July 2022

Publications

This thesis is based on the following publications [1-5], as well as unpublished work:

- Elliptic Loci of SU(3) Vacua Annales Henri Poincaré 22 (2021) 2775–2830, [hep-th/2010.06598] with J. Aspman and J. Manschot.
- [2] Cutting and gluing with running couplings in $\mathcal{N} = 2$ QCD *Phys. Rev. D* **105** (Jan, 2022) 025021, [hep-th/2107.04600] with J. Aspman and J. Manschot.
- [3] Four flavours, triality and bimodular forms *Phys. Rev. D* 105 (Jan, 2022) 025017, [hep-th/2110.11969] with J. Aspman and J. Manschot.
- [4] The u-plane integral, mock modularity and enumerative geometry Letters in Mathematical Physics 112 (2022) 30, [hep-th/2109.04302] with J. Aspman, G. Korpas, Z.-C. Ong and M.-C. Tan.
- [5] Topological twists of massive SQCD, Part I [hep-th/2206.08943] with J. Aspman and J. Manschot.

Summary

Supersymmetry provides a rich ground for qualitative and quantitative analyses in both quantum field theory and gravity. Many supersymmetric theories exhibit dualities such as the strong-weak coupling duality, which manifests itself through the equivalence of the dynamics of a theory for distinct values of its coupling constants. The mathematical formulation of dualities in many cases involves modular and automorphic forms, whose number theoretic properties are often as intriguing as the dualities themselves. It is a natural problem to determine a domain for the coupling constant parametrising inequivalent theories, and finding the relations between the couplings giving rise to the same dynamics.

In this thesis, we obtain several results on this question, particularly regarding the duality and modularity of Coulomb branches for four-dimensional $\mathcal{N} = 2$ supersymmetric quantum field theories. For pure $\mathcal{N} = 2$ super Yang– Mills theory (SYM) with gauge group SU(2), the Coulomb branch (CB) can be identified with the modular fundamental domain for the duality group of the theory. We extend this result for higher rank gauge groups and for the inclusion of matter.

There are several applications and further directions of study that our results enable. First, our exact results on fundamental domains allow to determine BPS spectra for a larger class of $\mathcal{N} = 2$ supersymmetric theories. Furthermore, our precise statements of duality allow to study the strongly coupled regions in great detail. Finally, they provide simple and exact formulas for the computation of correlation functions of topological theories. We discuss these results in more detail in the following.

To begin with, we study the asymptotically free $\mathcal{N} = 2$ SU(2) theories with fundamental matter. The duality groups were only known for $N_f = 2$ and 3 massless flavours, whereas the generic mass case including the peculiar role of massless $N_f = 1$ have remained elusive. We find a completely general description for arbitrary masses m_i and number N_f of flavours: The space of inequivalent couplings is a fundamental domain $\mathcal{F}(m_i)$, endowed with a collection of branch cuts and branch points. For special choices of the masses, the branch points annihilate and $\mathcal{F}(m_i)$ is a modular fundamental domain. The CB parameter u is the root of a sextic polynomial, and for certain mass configurations can be determined explicitly as a function of the effective coupling. The description incorporates all possible mass limits, such as decoupling of hypermultiplets and merging of local as well as non-local singularities, giving an analytic handle on the superconformal Argyres-Douglas theories.

The superconformal $N_f = 4$ theory with gauge group SU(2) distinguishes itself from the asymptotically free theories through a nontrivial dependence of the theory on an ultraviolet coupling $\tau_{\rm UV}$, and is a building block for fourdimensional $\mathcal{N} = 2$ superconformal field theories. We prove that for special mass configurations, the order parameter u transforms as a modular form for the infrared coupling τ as well as the ultraviolet coupling $\tau_{\rm UV}$. A simultaneous transformation on τ and $\tau_{\rm UV}$ permutes the moduli spaces according to the triality group of the flavour symmetry SO(8), and we show that the characteristic CB functions organise into vector-valued *bimodular forms*. These results allow to derive the S-duality transformations of the topologically twisted four flavour theory.

In a different direction, we study the duality of pure $\mathcal{N} = 2$ SYM with gauge group SU(3), where the two-dimensional space of vacua parametrises an intricate family of genus two Seiberg-Witten curves. We prove that two natural complex one-dimensional loci of the CB are parametrised each by elliptic (i.e. genus one) curves, such that these *elliptic loci* allow a modular parametrisation. The locus where mutually local dyons become massless provides a natural generalisation of the pure SU(2) fundamental domain, while the locus containing the superconformal Argyres-Douglas points is a fundamental domain for a Fricke group.

We furthermore consider topological twists of four-dimensional $\mathcal{N} = 2$ supersymmetric quantum chromodynamics with gauge group SU(2) and $N_f \leq 3$ fundamental hypermultiplets. The twists are labelled by a choice of background fluxes for the flavour group, which provides an infinite family of topological partition functions. We demonstrate that in the presence of such fluxes the theories can be formulated for arbitrary gauge bundles on a compact fourmanifold. Moreover, we consider arbitrary masses for the hypermultiplets, which introduce new intricacies for the evaluation of the low-energy path integral on the Coulomb branch. We develop techniques for the evaluation of these path integrals.

When the topological version of pure $\mathcal{N} = 2$ SU(2) SYM is formulated on a simply connected four-manifold X, it is well-known that a physical correlation function computes the celebrated Donaldson invariants of X, which are smooth four-manifold invariants. Based on earlier work by Mariño and Moore, we generalise recent results by Korpas and Manschot to the case where X is non-simply connected, and show that correlation functions on such manifolds can be evaluated using mock modular forms. For a product ruled surface $X = \Sigma_g \times \mathbb{CP}^1$, by shrinking the genus g Riemann surface Σ_g , this allows to make a connection between mock modular forms and the topological A-model with target space the moduli space of flat connections on Σ_g .

Acknowledgements

First and foremost, I would like to thank my supervisor Jan Manschot for giving me the opportunity to pursue my studies in theoretical physics. I have greatly enjoyed and benefited from our countless discussions and meetings. I would like to thank you for your great guidance and supervision, productive collaboration and kind support during the last four years.

I would like to thank Johannes Aspman for being a fantastic collaborator in many projects during the last four years. I believe that our mutual disagreement in many computational and conceptual aspects has been extremely fruitful for our work. Many thanks also for comments on a draft of this thesis.

The articles on which this thesis is based on are written in collaboration with Johannes Aspman, George Korpas, Jan Manschot, Zhi-Cong Ong and Meng-Chwan Tan. Many thanks to all of you for a great collaboration and for sharing your ideas and knowledge with me. It has been a great pleasure working with you all. I am also grateful to my other collaborators: Cyril Closset, Horia Magureanu and Gregory Moore. I have benefited significantly from our discussions.

I am happy to have been able to do several trips to conferences and schools during the last four years. These include: The spring school on Superstring Theory at ICTP Trieste 2019, the Young Researchers Integrability School at DESY Hamburg 2020, the KITP workshop on Modularity in Quantum Systems 2020, Eurostrings at ENS Lyon 2022, the Spring School at ICTP Trieste in 2022, and many other online conferences and schools. Many thanks to the organisers for letting me present some of my work on these occasions. This thesis has benefited from discussions with many people, some of which I met at these places. They are in particular: Boris Pioline, Edward Witten, Kaiwen Sun, Ken Ono, Kento Osuga, Kimyeong Lee, Lakshya Bhardwaj, Mario Martone, Philip Argyres, Samson Shatashvili, Shehryar Sikander and Thorsten Schimannek.

I am thankful to all the members of the School of Mathematics at Trinity College Dublin for providing a stimulating environment. Thanks in particular to my fellow students and postdocs Ana Retore, Andrea Orta, Anne Spiering, Antoine Vincenti, Chiara Paletta, George Korpas, Gim Seng Ng, Guy Jehu, Johannes Aspman, Martijn Hidding, Nico "1" Lang, Nico "2" Batelli, Paul Ryan, Petar Tadić, Raul Pereira, Riccardo Gonzo, Robin Karlsson, Sam Valach and Swapno Mondal for many great chats, coffee and lunch breaks, and Friday Fika. With my office colleagues Nico 1 & 2, Robin and Johannes I have enjoyed many Mousse Mondays, Film Fridays and Sightseeing Saturdays. A special thanks to Robin for daily Swedish coffee, climbing trips, chess games, and occasionally reminding me that it is important to take some time off work and go bouldering. I also thoroughly enjoyed all the seminars, journal clubs, colloquia and talks at the School of Maths.

Aside from the people mentioned above, I also wish to thank my friends

Cédric, Felix, Florian, Jeremy, Luca, Luke and Zan. Thank you for zoom calls during home office times, visits, trips and just staying in touch during my time abroad.

Ein ganz besonderer Dank gilt auch meinen Eltern, die mich seit vielen Jahren unterstützen. Ohne die tonnenweise Schokolade wäre ich wohl verhungert. Einen lieben Dank auch an meine Geschwister Selina und Josua für die vielen Besuche.

Last but not least, I would like to thank Theodor Tobler for his invention of the Toblerone, without which this thesis would not have been possible. Finally, I would like to thank Ireland for increasing my appreciation of the weather everywhere else.

My PhD was funded by the Provost's PhD Project Award of Trinity College Dublin.

Dedicated to my father, in loving memory.

Contents

1	Intr	roduction 1			
	1.1	Supersymmetry			
	1.2	Duality			
	1.3	Modular forms			
	1.4	Elliptic curves			
	1.5	Seiberg-Witten theory			
	1.6	Fundamental domains			
	1.7	Topological QFT			
2	Asymptotically free $\mathcal{N} = 2$ QCD 21				
	2.1	Introduction			
	2.2	Fundamental domains			
	2.3	Matone's relation			
	2.4	The $N_f = 1$ curve $\ldots \ldots 43$			
	2.5	The $N_f = 2$ curve			
	2.6	The $N_f = 3$ curve $\ldots \ldots \ldots$			
	2.7	Discussion			
3	Fou	r flavours, triality and bimodular forms 68			
	3.1	Introduction			
	3.2	Four flavours and triality			
	3.3	Order parameters and bimodular forms			
	3.4	Vector-valued bimodular forms			
	3.5	Discussion			
4	Elliptic loci of SU(3) vacua				
_	4.1	Introduction			
	4.2	The SU(3) Coulomb branch $\dots \dots \dots$			
	4.3	Locus \mathcal{E}_{v} : $v = 0$			
	4.4	Locus \mathcal{E}_{v} : $u = 0$			
	4.5	Elliptic curves for \mathcal{E}_n and \mathcal{E}_n			
	4.6	$SU(3)$ theory with matter $\ldots \ldots \ldots$			
	4.7	Discussion			
5	Ton	pological twists of massive SQCD 124			
0	p	Introduction			
	5.2	Special geometry and SW theories 126			
	5.3	The UV theory on a four-manifold 131			
	5.0	The effective theory on a four-manifold			
	5.5	The u-plane integral 140			
	5.6	Integration over fundamental domains			

6	The	u-plane integral for non-simply connected manifolds	149
	6.1	Introduction	. 149
	6.2	The effective theory for $b_1 > 0$. 150
	6.3	\mathcal{Q} -exact operators	. 157
	6.4	The <i>u</i> -plane integral for $b_1 > 0$. 161
7	Disc	cussion and conclusion	167
A	Aut	omorphic forms and elliptic curves	170
A	Aut A.1	omorphic forms and elliptic curves Elliptic modular forms	170 . 170
A	Aut A.1 A.2	omorphic forms and elliptic curves Elliptic modular forms	170 . 170 . 173
A	Aut A.1 A.2 A.3	omorphic forms and elliptic curvesElliptic modular formsSiegel modular formsFundamental domains	170 . 170 . 173 . 174
Α	Aut A.1 A.2 A.3 A.4	omorphic forms and elliptic curvesElliptic modular formsSiegel modular formsFundamental domainsMcKay-Thompson series	170 . 170 . 173 . 174 . 177
Α	Aut A.1 A.2 A.3 A.4 A.5	omorphic forms and elliptic curves Elliptic modular forms Siegel modular forms Fundamental domains McKay-Thompson series Algebraic modular functions	170 . 170 . 173 . 174 . 177 . 179
Α	Aut A.1 A.2 A.3 A.4 A.5 A.6	omorphic forms and elliptic curves Elliptic modular forms Siegel modular forms Fundamental domains McKay-Thompson series Algebraic modular functions Kodaira classification	170 . 170 . 173 . 174 . 177 . 179 . 182

1 Introduction

Theoretical physics has been enormously successful in predicting the behaviour of nature. In the last century, two main fundamental theories have emerged. On one hand, general relativity describes the fundamental interaction between objects with mass or energy. On the other hand, quantum field theory combines special relativity and quantum mechanics into a framework that describes subatomic particles, which underlie the electromagnetic, strong and weak interactions. The combination of both theories is necessary in order to understand aspects of the natural world, such as the nature of black holes and the formation of the universe. Since there are many conceptual and technical obstacles to the problem of combining both theories, it is of utmost interest to study the intrinsic properties of candidate unified theories in great detail.

One area of such theories which is poorly understood is the problem of strong coupling. The couplings of a theory measure the strength of the interactions of its constituents. When a coupling is known to be small, we can use the powerful method of perturbation theory to study the interactions. In many cases, when the interactions are turned off the theory becomes rather simple and can be solved exactly. Perturbation theory then allows to express the effect of a small coupling as a small correction to the exact solution. This only provides a good solution if the coupling is small compared to the quantities involved.

When the coupling on the other hand is large, the correction to the exact solution is large compared to the original solution, and thus does not provide a good approximation. Therefore, perturbation theory typically becomes obsolete when studying strong interactions. For most relevant physical theories such as the Standard Model, studying the physics of strong coupling in full generality has remained an important open problem.

One way to gain access to the properties of strong coupling is through supersymmetry. Supersymmetry is a natural feature of many unified theories. It allows us to study theories in a wider range of parameters (such as the couplings), as well as to calculate interactions exactly rather than approximatively. The broad motivation for this thesis is thus to study exact methods to access the physics of strong coupling and to demonstrate that these analyses allow to compute supersymmetric quantities exactly and in full detail.

This thesis considers the relation between supersymmetry, dualities, modular forms and topological field theory. The remainder of this first section is devoted to giving an introduction to some aspects of these theories, which are necessary to understand the following sections. We hopefully provide enough references in order for the reader to fill in gaps, and also refer to the Appendix A for more references.

1.1 Supersymmetry

Supersymmetry is a symmetry between two basic classes of particles in quantum field theory: bosons, which have integer-valued spin, and fermions, which have half-integer spin (see [6–8] for excellent introductions to supersymmetry). They have very different thermodynamic properties. While bosons follow the Bose-Einstein statistics, fermions obey the Pauli exclusion principle and follow the Fermi-Dirac statistics. In a supersymmetric theory, each particle from one class has a *superpartner* from the other class, whose spin differs by a halfinteger number. Moreover, the supersymmetric partners share the same mass and internal quantum numbers other than spin.

Supersymmetry is motivated by solutions to several problems in quantum field theory which are difficult to study in non-supersymmetric theories. Due to the constraints it gives on the dynamics, supersymmetric theories are often simpler to analyse. In many cases, quantities such as path integrals, correlation functions etc. can be determined exactly rather than only perturbatively, which gives an analytic handle on the physics. For instance, many supersymmetric theories contain interesting *dualities*, which manifests itself through the equivalence of the dynamics of a theory for distinct values of its coupling constants.

One instance of a duality is the strong-weak coupling duality, relating the dynamics for large and small values of the coupling constants. This allows in many cases to study strongly coupled phases of quantum field theory, which are not accessible through traditional perturbative calculations. A particularly interesting example of such a theory is quantum chromodynamics (QCD), as it is the theory of the strong interaction between quarks and gluons and thus an important part of the Standard Model of particle physics. QCD exhibits many interesting features, such as asymptotic freedom and colour confinement. Asymptotic freedom causes interactions between fields to become weaker as the energy scale increases. In particular, quarks interact weakly at high energies, and therefore can be studied through perturbation theory. At low energies however, the interactions become strong, which leads to the *confinement* of quarks and gluons. Colour confinement prohibits quarks and gluons to be isolated from each other. The supersymmetric version of QCD is asymptotically free as well, however the emergent strong-weak coupling duality allows to study the perturbative as well as non-perturbative dynamics at both weak and strong coupling in great detail.

Supersymmetry can be incorporated into a theory by extending the familiar Poincaré algebra by the supersymmetry algebra. The supersymmetry algebra is generated by spinors Q^I_{α} and $(Q^I_{\alpha})^{\dagger}$ called *supercharges*. Here, α are spinor indices, while $I = 1, \ldots, \mathcal{N}$ are the supersymmetry indices. The integer \mathcal{N} counts the number of independent supersymmetries of the algebra. Representations of the supersymmetry algebra are called *supermultiplets*, or *multiplets* for short. They consist of a collection of fields which are superpartners. In four dimensions, the possible amounts of supersymmetry for a quantum field theory are $\mathcal{N} = 0, 1, 2, 3$ and 4. For $\mathcal{N} = 1$ supersymmetry, the most commonly used supermultiplets are the $\mathcal{N} = 1$ vector multiplet and the chiral multiplet. In the vector multiplet, the highest component is the gauge field, while that of a chiral multiplet is a spinor. In $\mathcal{N} = 2$, common multiplets are the $\mathcal{N} = 2$ vector multiplet, containing the gauge field, and the hypermultiplet, containing a spinor as the highest component. In $\mathcal{N} = 4$, the vector multiplet contains again all the superpartners of the gauge field.

The space of theories with \mathcal{N} supersymmetries depends in an interesting way on \mathcal{N} :

- $\mathcal{N} = 0$ theories do not contain any supercharges. They are constrained only by physical principles, and a classification is currently out of reach.
- $\mathcal{N} = 1$ theories form an enormously big class of theories, and are believed to not be exactly solvable. However, certain subsectors are dictated by supersymmetry, such as the chiral superpotential. This allows to obtain many exact results on the low-energy dynamics [9–12].
- $\mathcal{N} = 2$ supersymmetry is a rich class of theories capable of describing many interesting phenomena, while allowing full solutions of many aspects of the theories. For instance, the low-energy effective action is governed by the so-called *prepotential*. The $\mathcal{N} = 2$ supersymmetry forces the prepotential to be a holomorphic function of the moduli, and the perturbative quantum corrections can be found exactly. This is analogous to the power of holomorphicity of conformal field theories in two dimensions. This thesis will be almost exclusively about aspects of $\mathcal{N} = 2$ supersymmetry.
- The number $\mathcal{N} = 3$ is interesting in its own right, for the following reason. If we consider a Lagrangian $\mathcal{N} = 3$ theory, it might not by itself be simultaneously charge conjugate, parity and time reversal (CPT) invariant. CPT is a possible symmetry of quantum field theories, and generally holds for all physical phenomena. When a Lagrangian $\mathcal{N} = 3$ theory is not CPT invariant, we can take instead the direct sum with its CPT conjugate. If the theory is assumed to contain particles of helicities bounded by 1, for $\mathcal{N} = 3$ and $\mathcal{N} = 4$ the particle spectra coincide precisely.

If we drop the requirement that the theory is Lagrangian, then it can be shown that genuine $\mathcal{N} = 3$ theories are necessarily isolated superconformal field theories (SCFT) [13]. However, it was shown recently that purely $\mathcal{N} = 3$ theories exist with an infinite family of them realised as the worldvolume of stacks of D3-branes probing so-called S-folds [14–17].

• $\mathcal{N} = 4$ theories finally are highly constrained. The most prominent example is maximally supersymmetric $\mathcal{N} = 4$ Yang–Mills theory (SYM),

which is featured in the first example of the conjectured AdS/CFT holography [18]. It is believed to be self-dual [19], that is, invariant under the exchange of electric and magnetic charges. The physics of an $\mathcal{N} = 4$ theory is generally simpler than that of an $\mathcal{N} = 2$ theory. For instance, the low energy dynamics obtained from the Seiberg-Witten (SW) solution is rather trivial.

A particular feature of supersymmetry is that it can be broken to lower supersymmetry. For instance, maximally supersymmetric $\mathcal{N} = 4$ SYM in four dimensions can be broken to the so-called $\mathcal{N} = 2^*$ theory, which is an $\mathcal{N} = 2$ supersymmetry-preserving mass deformation of $\mathcal{N} = 4$. This gives an analytic handle on structures of $\mathcal{N} = 4$ supersymmetric theories that are difficult to understand without breaking the supersymmetry. For instance, any $\mathcal{N} = 2$ theory, such as $\mathcal{N} = 4$ SYM, contains a *Coulomb branch* of inequivalent vacua. The mass terms of the $\mathcal{N} = 2^*$ deformation break the conformal symmetry of $\mathcal{N} = 4$, and give rise to a richer structure of the Coulomb branch. This allows for instance to study correlation functions in much greater detail than without any deformation.

Supersymmetry is an integral part of string theory, which aims to provide a unified description of gravity and quantum field theory. Indeed, a major motivation to study $\mathcal{N} = 2$ theories comes from string theory: Many $\mathcal{N} = 2$ field theories in four dimensions can be obtained from geometric engineering in Type IIA string theory on Calabi-Yau manifolds [20–23]. While all the theories studied in this thesis have a natural origin in string theory, our results mostly concern ordinary quantum field theory, i.e. without gravity. We will discuss relations to string theory sparsely throughout the text, but will expand on those in the conclusions in section 7.

1.2 Duality

In maximally supersymmetric $\mathcal{N} = 4$ SYM, one obtains a physically equivalent theory if the gauge coupling constant $g_{\rm YM}$ is replaced by its inverse $2\pi/g_{\rm YM}$. To describe this duality, it is convenient to combine the theta angle θ and the gauge coupling $g_{\rm YM}$ into the complexified gauge coupling,

$$\tau = \frac{\theta}{2\pi} + \frac{2\pi i}{g_{\rm YM}^2}.\tag{1.1}$$

It takes values in the upper half-plane

$$\mathbb{H} = \{ \tau \in \mathbb{C} \, | \, \operatorname{Im}(\tau) > 0 \}, \tag{1.2}$$

since the imaginary part of τ is positive. When the theta angle is included, the *Montonen–Olive* duality acts by $\tau \mapsto -1/\tau$ [19, 24]. Furthermore, the periodicity of the theta angle $\theta \mapsto \theta + 2\pi$ is translated to $\tau \mapsto \tau + 1$. These two transformations $\tau \mapsto -1/\tau$ and $\tau \mapsto \tau + 1$ can be considered as linear fractional transformations

$$\tau \longmapsto \gamma \cdot \tau \coloneqq \frac{a\tau + b}{c\tau + d}, \qquad \gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{SL}(2, \mathbb{Z}),$$
(1.3)

where $\operatorname{SL}(2,\mathbb{Z})$ is the group of unit determinant 2×2 matrices with integer coefficients. In particular, for $S = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$ and $T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$, we have that $S \cdot \tau = -1/\tau$ and $T \cdot \tau = \tau + 1$. The elements S and T generate $\operatorname{SL}(2,\mathbb{Z})$, i.e. any element of $\operatorname{SL}(2,\mathbb{Z})$ can be written as a word in S and T. Put differently, the $\mathcal{N} = 4$ theory contains an $\operatorname{SL}(2,\mathbb{Z})$ symmetry acting on both g_{YM} and θ nontrivially. If the theta angle is set to zero, the S-duality exchanges weak coupling $(g_{\mathrm{YM}} \ll 1)$ with strong coupling $(g_{\mathrm{YM}} \gg 1)$, which is why Montonen– Olive duality is an example of a strong-weak coupling duality (see [25] for a review). This demonstrates the power of dualities: Strong-weak dualities allow to gain insight into the strongly coupled phases, which are not accessible through traditional perturbative calculations.

The strong-weak coupling duality of $\mathcal{N} = 4$ SYM is of course not the only example of a duality in a physical theory. In fact, dualities are ubiquitous in quantum field theories, string theory and gravity. The most basic one is perhaps the electric-magnetic duality of electrodynamics, which is characterised by an equivalence of the theory under an exchange of electric and magnetic fields. One of the best-understood examples is the gauge/gravity duality, which relates a strongly coupled quantum field theory to a weakly coupled gravitational theory, and vice versa [18]. Another example is mirror symmetry, which gives a relation between the topological A-model and B-model [26–29]. Finally, there is a web of string dualities in string theory, M-theory and F-theory, such as the T-duality, S-duality and U-duality. Dualities also have far-reaching consequences and relations to pure mathematics, such as the study of enumerative invariants or the geometric Langlands program [24, 30–32].

We can schematically formalise a class of dualities as follows. Let us assume that a theory \mathcal{T} has a space \mathcal{M} of couplings $\tau \in \mathcal{M}$. These couplings can be considered as external parameters of \mathcal{T} , on which observables such as the partition function, correlation functions etc. depend explicitly. We can denote this dependence by $\mathcal{T}[\tau]$, which indicates the theory with a coupling τ . The symbol \mathcal{T} can denote a theory as e.g. a collection of correlation functions, or a sector of a specific theory such as the infrared.¹ Let us make this idea more precise in two examples.

One example carried throughout this introduction is $\mathcal{N} = 4$ SYM, for instance with gauge group SU(N). This theory enjoys superconformal invariance, and has a complex one-dimensional $\mathcal{N} = 4$ supersymmetry-preserving conformal manifold. A conformal manifold is the space of couplings of a conformal field theory (CFT) to exactly marginal operators, and thus can be understood

¹We want to stress that the notions introduced below should be understood as schematic, rather than precise.

as the space that parametrises a CFT. For $\mathcal{N} = 4$ SYM, this conformal manifold may be parametrised by the complexified gauge coupling τ (1.1). Thus an observable \mathcal{O} in $\mathcal{N} = 4$ for fixed rank N depends only on τ , and as such is a function $\mathcal{O}(\tau)$. Then by $\mathcal{T}_{\mathcal{N}=4}[\tau]$ we mean the collection of observables $\mathcal{O}(\tau)$.

The main focus of this thesis is on $\mathcal{N} = 2$ supersymmetry, which is not superconformal unless it is coupled to matter in specific representations. As will be discussed in detail below, $\mathcal{N} = 2$ gauge theories have a moduli space of inequivalent vacua, a component of which is the Coulomb branch. The Coulomb branch can be parametrised by order parameters u_i , which are functions of the low-energy effective coupling τ . As an example, in this context, by $\mathcal{T}_{\mathcal{N}=2}[\tau]$ we could mean such a function $u_i(\tau)$.

By comparing $\mathcal{T}[\tau_1]$ and $\mathcal{T}[\tau_2]$, we can use the "map" \mathcal{T} to test whether two couplings τ_1 and τ_2 generate two different theories or two equivalent theories. For instance, in $\mathcal{N} = 4$ SYM "equivalent" means that correlation functions take the same values, and thus give rise to equivalent dynamics of the theory. This idea naturally leads to the

Definition 1 (Fundamental domain). For a theory \mathcal{T} with space of couplings \mathcal{M} , a fundamental domain $\mathcal{F}[\mathcal{T}]$ is a subset of \mathcal{M} with the property that

- for any coupling $\tau \notin \mathcal{F}[\mathcal{T}]$, there exists a $\tau' \in \mathcal{M}$ such that $\mathcal{T}[\tau] = \mathcal{T}[\tau']$, and
- for any two couplings $\tau, \tau' \in \mathcal{F}[\mathcal{T}]$ with $\tau \neq \tau'$ we have that $\mathcal{T}[\tau] \neq \mathcal{T}[\tau']$.

Let us now assume that a group $G \ni g$ acts on the coupling τ , which we denote by $g \cdot \tau$. If

$$\mathcal{T}[g \cdot \tau] = \mathcal{T}[\tau] \tag{1.4}$$

for all $g \in G$ and for all $\tau \in \mathcal{M}$, then we say that G is the duality group of the theory \mathcal{T} for the action on the coupling $\tau \in \mathcal{M}$. When the theory has such a duality group G, there is different way to formulate a fundamental domain, as two couplings in \mathcal{M} can be compared as follows:

Definition 2 (Fundamental domain). Let \mathcal{T} be a theory with space \mathcal{M} of couplings and duality group G acting on \mathcal{M} . Then $\widetilde{\mathcal{F}}[\mathcal{T}]$ is a fundamental domain, if

- for any $\tau \notin \widetilde{\mathcal{F}}[\mathcal{T}]$, there exists a $g \in G$ such that $g \cdot \tau \in \widetilde{\mathcal{F}}[\mathcal{T}]$, and
- for any two distinct $\tau, \tau' \in \widetilde{\mathcal{F}}[\mathcal{T}]$, there is no $g \in G$ such that $\tau' = g \cdot \tau$.

Fundamental domains are in general not unique, and there is no canonical choice. For families of theories that are related by deformations, such as supersymmetric quantum chromodynamics (SQCD), fixing a frame for the choice of fundamental domains becomes important, since also the fundamental domains must be related by certain deformations. An important class of dualities are those which have a nontrivial action on the gauge group. For instance, Montonen–Olive duality of $\mathcal{N} = 4$ SYM in general is a symmetry which under a generalised S-duality $\tau \mapsto -\frac{1}{n_G \tau}$ replaces the gauge group G with its so-called Landlands dual group LG , where $n_G = 1, 2, 3$ is an integer that depends on G. For instance, for $G = \mathrm{SU}(N)$, the Langlands dual is ${}^L\mathrm{SU}(N) = \mathrm{SU}(N)/\mathbb{Z}_N$, while ${}^L\mathrm{U}(N) = \mathrm{U}(N)$ is Langlands self-dual. When the gauge group G is not Langlands self-dual, the S-transformation is interesting because it explains the strong coupling dynamics in terms of a weakly coupled description of a different theory with gauge group LG . Dualities affecting the gauge groups are not restricted to $\mathcal{N} = 4$, they are also found in $\mathcal{N} = 2$ supersymmetric theories [21, 33, 34]. In our schematic setup introduced above, we will only consider the cases where the duality keeps the gauge group fixed.

In this thesis, we will mainly focus on the case where \mathcal{M} is a subset of the upper half-plane \mathbb{H} , (1.2). On \mathbb{H} there is a natural group action of $\mathrm{SL}(2,\mathbb{Z})$ or any subgroup thereof, as defined in (1.3). For such subgroups, the definition 2 for a fundamental domain is widely used in the literature of modular groups and automorphic forms. From the generators of a group, there is an algorithm that draws a certain choice of fundamental domain (see Appendix A.3).

We find many examples where $g \cdot \tau$ is not a linear fractional transformation for $G \ni g$ an honest group, and the action needs to be properly generalised. Then definition 1 can be used instead of definition 2. This distinction will be made clear in this thesis.

Let us briefly return to the above example of $\mathcal{N} = 4$ SYM, whose duality group is SL(2, \mathbb{Z}) acting on the coupling $\tau \in \mathcal{M} = \mathbb{H}$ through the transformation (1.3).² The duality can be encoded in the equation

$$\mathcal{T}_{\mathcal{N}=4}[\gamma \cdot \tau] = \mathcal{T}_{\mathcal{N}=4}[\tau], \qquad (1.5)$$

which is a formulation of the Montonen–Olive duality. This allows to extract the physical consequences of the duality based on $SL(2, \mathbb{Z})$ spectral theory [35]. For the duality group $SL(2, \mathbb{Z})$, there is a standard choice of fundamental domain \mathcal{F} , which is given in Fig. 1.

From definition 2, it can be understood as follows. For some $\tau = x + iy \in \mathbb{H}$, $x \in \mathbb{R}$ is a real number. As such, there exists a unique $n \in \mathbb{Z}$ such that $\tau + n \in [-\frac{1}{2}, \frac{1}{2})$ lies in the unit interval between $\pm \frac{1}{2}$. Using the action (1.3), we can write $\tau + n = T^n \tau$, and the SL(2, \mathbb{Z}) element T^n maps τ inside this strip. This implies that a fundamental domain \mathcal{F} for SL(2, \mathbb{Z}) must be contained in the strip $-\frac{1}{2} \leq x < \frac{1}{2}$. On the other hand, if $|\tau| < 1$, then $|-1/\tau| > 1$. This can be seen by writing a complex number as an absolute value times a phase. Consequently, \mathcal{F} must be contained in the region defined by $|\tau| \geq 1$. The

²We ignore several important aspects (such as modular weights, multiplier systems, 't Hooft fluxes, spin structures, non-holomorphicity, modular anomaly, line operators etc.) in order to give a pedagogical example.



Figure 1: The key-hole fundamental domain \mathcal{F} of $SL(2,\mathbb{Z})$. The vertical sides are identified, as well as the two halfs of the boundary arc on the unit circle.

intersection of both constraints from the T and S transformations precisely gives the fundamental region in Fig. 1.

This explanation follows the construction of a fundamental domain as given in Appendix A.3: First, we write down a list of generators for the group. The modular group $SL(2, \mathbb{Z})$ is generated by T and S. The generator fixing $\tau = i\infty$ is T, and determines a vertical strip of width 1, say from $-\frac{1}{2}$ to $+\frac{1}{2}$. The Stransformation then reflects the argument on the half-circle $\mathbb{S}^1 \cap \mathbb{H}$ around $\tau = 0$ with radius 1. The half-circle defines an "interior" and "exterior", depending on whether a number is contained in the disk bounded by the unit circle or not. A choice of fundamental domain is then given by the intersection of all exterior regions with the strip defined by the element fixing $\tau = i\infty$. For $SL(2, \mathbb{Z})$, this gives precisely Figure 1.

1.3 Modular forms

A powerful tool to study dualities are modular forms. Modular or automorphic forms are functions that satisfy a functional equation such as (1.4) or generalisations thereof. A part of their importance comes from the fact that if certain growth conditions are included in the definition, they live in finite-dimensional vector spaces. This gives powerful constraints on their behaviour, while still being rich enough to encode nontrivial information. In this section, we present basic aspects of modular forms and modular functions. For a pedagogical introduction, we refer to the references in Appendix A.

Modular forms for $SL(2, \mathbb{Z})$ are defined as follows:

Definition 3 (Modular form). Let k be an integer. A function $f : \mathbb{H} \to \mathbb{C}$ is a modular form of weight k, if

- f is holomorphic in \mathbb{H} ,
- $f(\gamma \cdot \tau) = (c\tau + d)^k f(\tau)$ for all $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL(2, \mathbb{Z})$ and all $\tau \in \mathbb{H}$,
- f is holomorphic at ∞ .

The space of modular forms for a fixed weight k is a vector space over \mathbb{H} , and the third condition makes it finite-dimensional. For $\gamma = T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$, the functional equation becomes $f(\tau + 1) = f(\tau)$. This shows that f admits a Fourier expansion

$$f(\tau) = \sum_{n=0}^{\infty} a_n q^n, \qquad (1.6)$$

where $q = e^{2\pi i\tau}$. The Fourier coefficients $a_n \in \mathbb{C}$ encode the same information as the function f itself, and in many interesting cases are in fact integers. The expansion (1.6) does not contain terms with nonzero a_n for negative n, since for $\tau \to i\infty$ we have $q \to 0$, which would give a pole for f as $\tau \to i\infty$.

When $\gamma = -I$ in Definition 3, then $f = (-1)^k f$ implies that the only odd weight modular form for $SL(2,\mathbb{Z})$ is the zero function. For even k, a class of examples are the Eisenstein series. They can be defined as the q-series

$$E_k(\tau) = 1 - \frac{2k}{B_k} \sum_{n=1}^{\infty} \sigma_{k-1}(n) q^n, \qquad (1.7)$$

with $\sigma_k(n) = \sum_{d|n} d^k$ the divisor sum and B_k the Bernouilli numbers. For E_4 and E_6 , the Fourier series reads

$$E_4(\tau) = 1 + 240q + 2160q^2 + 6720q^3 + 17520q^4 + \mathcal{O}(q^5),$$

$$E_6(\tau) = 1 - 504q - 16632q^2 - 122976q^3 - 532728q^4 + \mathcal{O}(q^5).$$
(1.8)

It can be shown that for $k \ge 4$ even, E_k is a modular form of weight k for $SL(2,\mathbb{Z})$. For k = 2, E_2 is a so-called quasi-modular form, which means that the $SL(2,\mathbb{Z})$ transformation includes a shift in addition to the weight factor (see Appendix A.1).

If we denote the finite-dimensional vector space of modular forms of weight k for $\Gamma = \mathrm{SL}(2,\mathbb{Z})$ by $M_k(\Gamma)$, the direct sum

$$M_*(\Gamma) = \bigoplus_k M_k(\Gamma) \tag{1.9}$$

can be shown to be an algebra over \mathbb{C} : A modular form can be multiplied by a complex number, the sum of two modular forms of the same weight is a modular form, and the product of two modular forms of weight k and k' is a modular form of weight k + k'. This gives a simple way to construct new modular forms. One can show that for $\Gamma = \mathrm{SL}(2,\mathbb{Z})$, $M_*(\Gamma)$ is freely generated by E_4 and E_6 . This implies in particular that the Eisenstein series of weight 8 and larger can always be written as sums of products of E_4 and E_6 . For instance, it is easy to confirm that $E_4^2 = E_8$, $E_4E_6 = E_{10}$, etc. A proof of these identities is given by the insight that the vector spaces $M_k(\Gamma)$ are finitedimensional, and as such it is enough to compare a finite number of Fourier coefficients on both sides of the equation.

These relations furthermore predict an infinite number of functional identities among the Fourier coefficients of E_k , which are given by the divisor sums $\sigma_k(n)$. In this way, many problems in number theory can be cast into problems involving identities of modular forms. For instance, the divisor sum enjoys a multiplicative property: whenever a and b are coprime, then $\sigma_k(ab) = \sigma_k(a)\sigma_k(b)$.

A natural question suggested by (1.9) is whether the ring structure also admits a division. In particular, if two functions f and f' have weight k, then f/f' has weight 0 under SL(2, \mathbb{Z}). However, since for a nonzero holomorphic function $f : \mathbb{H} \to \mathbb{C}$, the inverse 1/f is generally *meromorphic* (i.e. it contains poles), the definition 3 needs to be adapted.

Definition 4 (Modular function). A function $f : \mathbb{H} \to \mathbb{C}$ is called a modular function, if

- f is meromorphic in \mathbb{H} ,
- $f(\gamma \cdot \tau) = f(\tau)$ for all $\gamma \in SL(2,\mathbb{Z})$ and all $\tau \in \mathbb{H}$,
- f is meromorphic at ∞ , i.e. the Fourier series is of the form $f(\tau) = \sum_{n=-m}^{\infty} a_n q^n$ for some $m \in \mathbb{Z}$.

From the generators E_4 and E_6 of $M_*(\Gamma)$, the simplest modular function that can be constructed is a quotient of two modular forms of weight 12, which is the least common multiple of 4 and 6. One such quotient is the modular *j*-invariant,

$$j = \frac{12^3 E_4^3}{E_4^3 - E_6^2},\tag{1.10}$$

which is a modular function for $SL(2,\mathbb{Z})$,

$$j(\gamma \cdot \tau) = j(\tau), \qquad \gamma \in \mathrm{SL}(2,\mathbb{Z}).$$
 (1.11)

Its Fourier coefficients

$$j(\tau) = q^{-1} + 744 + 196884q + 21493760q^2 + \mathcal{O}(q^3)$$
(1.12)

enjoy a remarkable relation to group theory, a phenomenon called *monstrous* moonshine [36]. It can be proven that the Fourier coefficients are the dimensions of the graded part of an infinite-dimensional algebra representation of the monster group \mathbb{M} , the largest sporadic simple group of order ~ 10⁵⁴. The Monster vertex algebra can be constructed as a conformal field theory describing 24 free bosons compactified on the torus induced by the 24-dimensional self-dual Leech lattice, followed by a \mathbb{Z}_2 orbifold [37].

The *j*-invariant has many other important properties. First of all, it generates modular functions for $SL(2,\mathbb{Z})$. In particular, every modular function

for $SL(2,\mathbb{Z})$ is a rational function in j, and conversely, every rational function in j is a modular function. This furthermore enhances (1.9) for weight 0 functions to an algebraic field: If g is a nonzero modular function, then f/gis a modular function. For $SL(2,\mathbb{Z})$, it can be shown that this *function field* can be generated by a single transcendental function, j being an example. Up to Möbius transformations and normalisations, such a function is unique and traditionally called *Hauptmodul*.

Another property of the *j*-function is that it knows about the fundamental domain \mathcal{F} of its invariance group $\mathrm{SL}(2,\mathbb{Z})$: Its fundamental domain obtained from Def. 1 coincides with the one from Def. 2. In other words, $j : \mathcal{F} \to \mathbb{C}$ is an isomorphism. This illustrates that in the presence of an invariance group, for a good choice of test function the definitions 1 and 2 can be equivalent.

For both modular forms and modular functions there is a variety of generalisations that have been studied in great detail. The most immediate generalisation is to subgroups of $SL(2, \mathbb{Z})$, in particular the *congruence* subgroups. The most common examples are the groups denoted by $\Gamma(n)$, $\Gamma_0(n)$ and $\Gamma^0(n)$, with $n \in \mathbb{N}$. They are defined through congruences of entries of the 2 × 2 matrices, for instance

$$\Gamma^{0}(n) = \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{SL}(2, \mathbb{Z}) \middle| b \equiv 0 \mod n \right\}.$$
(1.13)

The group $\Gamma_0(n)$ is similarly defined as the set of such matrices with $c \equiv 0 \mod n$, while an element γ of $\Gamma(n)$ satisfies $\gamma \equiv 1 \mod n$ for all components. The importance of those subgroups comes from the fact that certain elementary functions transform precisely under elements of those groups, rather than under all of $SL(2,\mathbb{Z})$. The simplest class of modular forms for congruence subgroups of $SL(2,\mathbb{Z})$ are arguably the Jacobi theta functions, which are defined as

$$\vartheta_{2}(\tau) = \sum_{r \in \mathbb{Z} + \frac{1}{2}} q^{r^{2}/2},
\vartheta_{3}(\tau) = \sum_{n \in \mathbb{Z}} q^{n^{2}/2},
\vartheta_{4}(\tau) = \sum_{n \in \mathbb{Z}} (-1)^{n} q^{n^{2}/2}.$$
(1.14)

They transform into each other under the action of $SL(2, \mathbb{Z})$, however they are invariant themselves under proper subgroups. See Appendix A.1 for the transformations and further properties.

1.4 Elliptic curves

Modular forms are intimately related to the theory of elliptic curves. An elliptic curve is a plane algebraic curve, which can be defined as the solution space $\{(x, y)\}$ of a cubic equation

$$E: y^2 = 4x^3 - g_2x - g_3, \tag{1.15}$$

where g_2 and g_3 are called Weierstraß invariants. The curve is required to be non-singular, which means that it should not have cusps, self-intersections or isolated points. Algebraically, this holds if and only if the discriminant

$$\Delta = g_2^3 - 27g_3^2 \tag{1.16}$$

is nonzero. The formulation of elliptic curves over the complex numbers corresponds to an embedding of the torus into the complex projective plane. This follows from the functional equation $\wp'(z)^2 = 4\wp(z)^3 - g_2\wp(z) - g_3$ for the Weierstraß elliptic function \wp . The Weierstraß function is doubly periodic, i.e. periodic with respect to a lattice Λ , and as such can be considered to be defined on a torus $T = \mathbb{C}/\Lambda$. The torus can be embedded in the complex projective plane using the map $z \mapsto [1 : \wp(z) : \frac{1}{2}\wp'(z)]$. This map is an isomorphism of Riemann surfaces from the torus to the cubic curve (1.15), and thus topologically an elliptic curve is a torus. The lattice Λ generating the torus can be understood as the linear span of a pair of nonzero complex numbers α_1 and α_2 . This pair defines a smallest cell, which is called the fundamental domain³ of the torus. Geometrically, the torus is then obtained by identifying the opposite edges of the fundamental domain.

The shape of the domain and thus of the torus is described by the ratio $\tau = \alpha_2/\alpha_1$, and is the *complex structure* or the *modular parameter* of the torus. It is clear that a different choice β_1 , β_2 of complex numbers can describe the same torus. Indeed, this is true if

$$\begin{pmatrix} \beta_1 \\ \beta_2 \end{pmatrix} = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \begin{pmatrix} \alpha_1 \\ \alpha_2 \end{pmatrix}, \tag{1.17}$$

where $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{SL}(2,\mathbb{Z})$. Since the ratio $\tau = \alpha_2/\alpha_1$ furthermore cancels overall signs of the vector $(\alpha_1, \alpha_2)^\mathsf{T}$, we can ignore overall signs also of γ and identify $\gamma \sim -\gamma$. In this way, we find the modular group of the torus as the isometry group $\mathrm{PSL}(2,\mathbb{Z})$ acting on the modular parameter τ .

From the curve (1.15) we can compute the \mathcal{J} -function

$$\mathcal{J} = 12^3 \frac{g_2^3}{\Delta}.\tag{1.18}$$

This quantity is an invariant for isomorphism classes of elliptic curves, meaning that two elliptic curves have the same \mathcal{J} -invariant if and only if they are isomorphic.

The types of elliptic curves relevant for this thesis are the ones that are parametrised by a complex variable. A particular class of 1-parameter families of elliptic curves are the *elliptic surfaces*. These are obtained by promoting the Weierstraß invariants to functions on a base space \mathbb{CP}^1 . Then the elliptic fibration

$$E \longrightarrow \mathcal{S} \longrightarrow \mathbb{CP}^1$$
 (1.19)

³Not to be confused with definition 1 or 2.

is called an elliptic surface, where \mathbb{CP}^1 can be understood as a compactification of the complex plane \mathbb{C} by adding a point ∞ . When g_2 and g_3 are polynomials on \mathbb{CP}^1 , then $\mathcal{J} : \mathbb{CP}^1 \to \mathbb{C}$ is a rational function. Let u denote a coordinate on \mathbb{CP}^1 . Then (1.18) is a rational function of u, and by the above arguments can be identified with the modular j-invariant, which is a function of the complex structure of the curve,

$$\mathcal{J}(u) = j(\tau). \tag{1.20}$$

This gives interesting relations between the coordinate u and the modular parameter τ , which are of central importance for this thesis. See [38–44] for more details on elliptic surfaces.

1.5 Seiberg-Witten theory

A beautiful synthesis of all the above introduced concepts is the full nonperturbative solution for the low-energy dynamics of four-dimensional $\mathcal{N} = 2$ supersymmetric Yang–Mills theory with gauge group SU(2), due to Seiberg and Witten [45, 46]. See [8, 47–54] for reviews.

Seiberg-Witten duality can be regarded as a lift of the electric-magnetic duality to $\mathcal{N} = 2$ supersymmetric quantum field theory. It however cannot be a duality analogous to Montonen–Olive duality in $\mathcal{N} = 4$, for several reasons. The maximally supersymmetric $\mathcal{N} = 4$ SYM theory is superconformal, and as such the gauge coupling is fixed. The pure $\mathcal{N} = 2$ supersymmetric Yang–Mills theory has the rather different feature that it is *asymptotically free*, such as the non-supersymmetric quantum chromodynamics for example is. Asymptotic freedom is the property of a theory that causes interactions between fields to become weaker as the energy scale increases. Correspondingly, the couplings should be considered as a function of the energy scale Λ . If duality exchanged the coupling $g_{\rm YM}$ with its inverse $1/g_{\rm YM}$ measured at a certain scale Λ , this would in general not be true at another scale Λ' . Thus rather than $\mathcal{N} = 2$ SYM theory itself, we can study its Wilsonian low-energy effective action. Seiberg-Witten duality is then an exact duality that acts on the *effective* coupling, rather than of the microscopic SU(2) theory.

The Seiberg-Witten geometry underlies the Coulomb branch of $\mathcal{N} = 2$ gauge theory. The $\mathcal{N} = 2$ vector multiplet consists of a gauge field A, a complex scalar field ϕ , and two Weyl fermions λ and ψ . They are all in the adjoint representation of the gauge group G, which we fix to G = SU(2). The Coulomb branch is the phase of the theory where SU(2) is broken to U(1) by a vacuum expectation value (vev) of the vector multiplet scalar ϕ . The potential of the theory is $V(\phi) = \frac{1}{g^2} \text{Tr}[\phi, \phi^{\dagger}]^2$, and we are interested in the moduli space of flat directions. These are found by setting

$$\phi = \begin{pmatrix} a & 0\\ 0 & -a \end{pmatrix}, \tag{1.21}$$

with a complex parameter. However, note that the Weyl group of SU(2) acts

on a by $a \mapsto -a$. The gauge invariant order parameter is the Coulomb branch expectation value of the theory in \mathbb{R}^4 ,

$$u = \frac{1}{16\pi} \left\langle \operatorname{Tr}(\phi^2) \right\rangle_{\mathbb{R}^4}, \qquad (1.22)$$

which parametrises the moduli space of inequivalent vacua. The renormalisation group flow generates a quantum scale Λ , at which the gauge coupling becomes strong.

The effective action then describes the physics of the remaining massless U(1) supersymmetric multiplets in terms of a holomorphic function, the *prepotential* F. By differentiating the prepotential with respect to the coordinate a, one finds the magnetic dual period $a_D = \frac{\partial F}{\partial a}$. A further differentiation results in the low-energy effective gauge coupling

$$\tau = \frac{\partial^2 F}{\partial a^2},\tag{1.23}$$

which becomes a combination of the effective coupling constant and the effective theta angle, similar to (1.1). The spectrum of the theory contains so-called *dyonic* states or *dyons*, which are states carrying electric as well as magnetic charges. The central charge of such a dyonic state with electric and magnetic charges $\gamma = (n_m, n_e)$ is given by

$$Z = \gamma \cdot \pi = n_m a_D + n_e a, \tag{1.24}$$

where $\pi = (a_D, a)^{\mathsf{T}}$ is the period vector.

The Seiberg-Witten (SW) solution provides a family of elliptic curves parametrised by the order parameter u, whose complex structure corresponds to the running coupling $\tau = \frac{\theta}{\pi} + \frac{8\pi i}{g^2}$. For the pure SU(2) theory, it is given by

$$y^2 = x^3 - ux^2 + \frac{1}{4}\Lambda^4 x.$$
 (1.25)

The curve provides exact results for the vevs of the scalar a and its dual a_D as period integrals. We have explicitly,

$$a = \int_{A} \lambda, \qquad a_D = \int_{B} \lambda,$$
 (1.26)

where λ is a meromorphic 1-form with $\frac{d\lambda}{du} = \frac{\sqrt{2}}{4\pi} \frac{dx}{y}$, and A and B are one-cycles which form a symplectic basis of the SW curve. The A-cycle corresponds to a straight line from 0 to 1, while the B-cycle corresponds to a straight line from 0 to τ .

As a and a_D are given by period integrals of a meromorphic 1-form over the elliptic curve, they satisfy a system of solutions to a set of Picard-Fuchs equations. This allows to express the periods in terms of hypergeometric functions [55]⁴,

$$a_D(u) = \frac{i}{2}(u-1) {}_2F_1(\frac{1}{2}, \frac{1}{2}, 2; \frac{1-u}{2}),$$

$$a(u) = \sqrt{\frac{(u+1)}{2}} {}_2F_1(-\frac{1}{2}, \frac{1}{2}, 1; \frac{2}{1+u}).$$
(1.27)

⁴We momentarily set $\Lambda = 1$.

At the strong coupling points, the periods $\pi(u) = (a_D(u), a(u))$ become $\pi(1) = (0, \frac{2}{\pi})$ and $\pi(-1) = (-\frac{4i}{\pi}, -\frac{2i}{\pi})$. According to the central charge formula (1.24), these values confirm that for u = 1, the monopole $\gamma = (1, 0)$ becomes massless, while for u = -1 the dyon $\gamma = (-1, 2)$ becomes massless. The limits $\lim_{u\to\pm 1} a(u)$ depend on the direction from which ± 1 are approached, which is due to the branch cut in the hypergeometric function [55].

The curve (1.25) is not in Weierstraß form (1.15) yet, however by rescaling $x \to x + \frac{u}{3}$ and $y \to \frac{y}{2}$, we find

$$y^{2} = 4x^{3} - \left(\frac{4}{3}u^{2} - \Lambda^{2}\right)x - \left(\frac{8}{27}u^{3} + \frac{1}{3}u\Lambda^{4}\right), \qquad (1.28)$$

from which we can read off the Weierstraß invariants

$$g_2 = \frac{4}{3}u^2 - \Lambda^4, \qquad g_3 = \frac{8}{27}u^3 + \frac{1}{3}u\Lambda^4.$$
 (1.29)

Using (1.18), we can then compute the \mathcal{J} -invariant

$$\mathcal{J}(\mathbf{u}) = 64 \frac{(4\mathbf{u}^2 - 3)^3}{\mathbf{u}^2 - 1},\tag{1.30}$$

which is a rational function of the dimensionless quantity $\mathbf{u} = \frac{u}{\Lambda^2}$, as a consequence of the SW curve (1.25) being a rational elliptic surface.

1.6 Fundamental domains

By identifying the \mathcal{J} -invariant of the SW curve with the modular *j*-invariant as in (1.20), the order parameter u is related to the coupling τ through

$$\mathcal{J}(\mathbf{u}(\tau)) = j(\tau). \tag{1.31}$$

Naively, $\mathbf{u}(\tau)$ should be invariant under $\mathrm{SL}(2,\mathbb{Z})$, since $j(\tau)$ is (see (1.11)). Below, we will show explicitly that this is not true. Acting with $\mathrm{SL}(2,\mathbb{Z})$ on (1.31) leaves the rhs invariant, and so must be the lhs. Since $\mathcal{J}(\mathbf{u})$ is a rational function, this implies that for any $\gamma \in \mathrm{SL}(2,\mathbb{Z})$ there exists a function h_{γ} such that $\mathbf{u}(\gamma \cdot \tau) = h_{\gamma}(\mathbf{u}(\tau))$, which is subject to the constraint that $\mathcal{J}(\mathbf{u}) = \mathcal{J}(h_{\gamma}(\mathbf{u}))$. In order to see this, let us solve the relation (1.31) for \mathbf{u} . Using (1.30), we can write (1.31) as a sextic polynomial equation

$$(\mathbf{u}^2 - 1)j - 64(4\mathbf{u}^2 - 3)^3 = 0, \qquad (1.32)$$

where we suppress the τ dependence of **u** and *j*. Since the sextic equation contains only even powers of **u**, it is a cubic equation in \mathbf{u}^2 and can thus be solved exactly. One particular solution is

$$\mathbf{u} = -\frac{1}{8\sqrt{3}} \sqrt[2]{\frac{\sqrt[3]{9j}}{\sqrt[3]{\left(\sqrt{3}\sqrt{12^3 - j} - 72\right)j}}} + \sqrt[3]{3}\sqrt[3]{\left(\sqrt{3}\sqrt{12^3 - j} - 72\right)j}} + 12^3,$$
(1.33)

such that u contains square and cube roots of modular functions. The other five solutions take a similar form. Generally, roots of modular functions are not modular, since roots induce branch points at zeros of the modular functions, thus spoiling the holomorphicity. However, here this is not the case. From the q-expansion of the j-invariant (1.12), we can easily compute the Fourier series

$$\mathbf{u}(\tau) = -\frac{1}{8}(q^{-1/4} + 20q^{1/4} - 62q^{3/4} + 216q^{5/4} + \mathcal{O}(q^{7/4})).$$
(1.34)

It appears that up to a simple normalisation, all Fourier coefficients are integers. Considering (1.33) with convoluted roots which generally induce rational coefficients with growing denominators, this might be rather surprising. In fact, the *q*-series (1.34) is known in the mathematics literature as the McKay-Thompson series of class 4C for the Monster group [36, 56-58]. As such, it is an example of a *replicable* function (see Appendix A.4 for a definition). It can be checked to arbitrary order that the *q*-expansion (1.34) agrees with that of

$$\mathbf{u}(\tau) = -\frac{1}{2} \frac{\vartheta_2(\tau)^4 + \vartheta_3(\tau)^4}{\vartheta_2(\tau)^2 \vartheta_3(\tau)^2},\tag{1.35}$$

where the Jacobi theta functions are defined in (1.14). Using the formulas given in Appendix A.1, one may check that this function is not invariant under the generators T and S of SL(2, \mathbb{Z}). For instance, under S the functions ϑ_2 and ϑ_4 are interchanged, while ϑ_3 is invariant (ignoring the modular weights, which cancel in this expression). Using the Jacobi identity $\vartheta_2^4 + \vartheta_4^4 = \vartheta_3^4$ we can then eliminate one theta function to obtain the action of the S-transformation on (1.35),

$$h_S(\mathbf{u}) = \frac{2\mathbf{u}\left(\mathbf{u} - \sqrt{\mathbf{u}^2 - 1}\right) - 3}{2\sqrt{2\mathbf{u}\left(\sqrt{\mathbf{u}^2 - 1} - \mathbf{u}\right) + 2}}.$$
(1.36)

We notice that h_{γ} does not necessarily need to be rational. We can furthermore confirm that $\mathcal{J}(h_S(\mathbf{u})) = \mathcal{J}(\mathbf{u})$, which is expected from (1.31). This shows that \mathbf{u} is not a modular function for SL(2, \mathbb{Z}), but solving (1.31) for \mathbf{u} instead breaks SL(2, \mathbb{Z}) into a subgroup.

Indeed, **u** can be shown to be a modular function for the congruence subgroup $\Gamma^{0}(4)$ of SL(2, Z), which we have defined in (1.13). The group $\Gamma^{0}(4)$ is generated by T^{4} and $ST^{-1}S$, and it can be checked explicitly using the formulas given in Appendix A.1 that **u** is invariant under these transformations. Furthermore, it is known that **u** is a Hauptmodul for $\Gamma^{0}(4)$, and as such it is an isomorphism $\mathbf{u} : \mathbb{H} \setminus \Gamma^{0}(4) \to \mathbb{C}$. The quotient $\mathcal{F}_{0} := \mathbb{H} \setminus \Gamma^{0}(4)$ is the fundamental domain for $\Gamma^{0}(4)$, and a particular choice can be drawn as explained in Appendix A.3, and we plot it in Fig. 2. The function (1.35) makes precise in which way its inverse $\tau(\mathbf{u})$ is multi-valued: For any given $\tau \in \mathcal{F}_{0}$, we can compute the infinite discrete set of couplings $\Gamma^{0}(4) \cdot \tau$, which all give rise to the same vacuum $\mathbf{u}(\tau)$ and thus to the same dynamics.

Since $u : \mathcal{F}_0 \to \mathbb{C}$ is an isomorphism, the Coulomb branch can equivalently be understood from the fundamental domain. From (1.30) it is clear that the



Figure 2: Fundamental domain of $\Gamma^0(4)$. This is the duality group of the pure SU(2) theory. The two cusps on the real line correspond to the strong coupling singularities of the gauge theory, while the cusp at $\tau = i\infty$ corresponds to weak coupling.

theory becomes singular at precisely three points: $\mathbf{u} = \infty$ and $\mathbf{u} = \pm 1$. At large τ , \mathbf{u} is large and describes the weakly coupled region. The asymptotics and monodromy can be found from the one-loop prepotential. At $\tau = 0$ and $\tau = 2$, the low-energy theory becomes singular since either the monopole or dyon, which were integrated out in the effective theory, becomes massless. These are precisely the two points where the elliptic curve (1.30) degenerates, and they correspond to $\mathbf{u} = -1$ and $\mathbf{u} = 1$. This can be confirmed by studying the q-expansions of the S and T^2S transformations of (1.35).

The duality group $\Gamma^0(4)$ can be equivalently found by studying the action of the monodromies on the Coulomb branch. Combining the periods a and a_D to the period vector $\pi = (a_D, a)^{\mathsf{T}}$ has an important property: They constitute a holomorphic section of a flat $\mathrm{SL}(2, \mathbb{Z})$ bundle, which are characterised by the monodromies around the singular points $\mathbf{u} = \infty$, $\mathbf{u} = \pm 1$:

$$\mathbf{M}_{\infty} = \begin{pmatrix} -1 & 4\\ 0 & -1 \end{pmatrix}, \quad \mathbf{M}_{-1} = \begin{pmatrix} 1 & 0\\ -1 & 1 \end{pmatrix}, \quad \mathbf{M}_{+1} \begin{pmatrix} -1 & 4\\ -1 & 3 \end{pmatrix}.$$
(1.37)

They satisfy the important consistency property $\mathbf{M}_{\infty} = \mathbf{M}_{-1}\mathbf{M}_{+1}$. Since the start and end point of a loop around a singular point coincide, the monodromies must leave u invariant, and the same must be true for the action of (1.37) on the coupling τ . Indeed, the monodromies (1.37) are in $\Gamma^{0}(4)$, under which u is invariant.

In view of the fact that the monodromy matrices (1.37) generate a subgroup of SL(2, Z), it is not surprising that **u** should be a modular function of τ . However, from the derivation of (1.35) from (1.33) this is not obvious. In fact, the presentation here has been chosen to demonstrate that this property is quite special and should be appreciated. One result of this thesis is that when the theory is generalised in various directions, such a correspondence

pure
$$\mathcal{N} = 2 \text{ SU}(2) \text{ SYM} \quad \longleftrightarrow \quad \Gamma^0(4)$$
 (1.38)

of the theory and its duality group can not be found in general. The arguably simplest generalisation is the inclusion of a massless hypermultiplet in the fundamental representation of SU(2), for which the SW curve was found in [46]. In section 2.4, we give at least five explanations why such a correspondence (1.38) does not exist for that theory, and thus does not exist in general. Rather, we generalise aspects of the notion of a duality group to fundamental domains, and then the correspondence (1.38) can be made precise.

Let us instead study why the surface (1.25) is special, and gives rise to a modular function. One reason is that the elliptic surface for pure SU(2) SYM is a modular elliptic surface. Such surfaces were studied in [40] and classified in [38]. In fact, all rational elliptic surfaces are classified through their configuration of singular fibres. Singular fibres are the loci of elliptic surfaces where the discriminant vanishes. As discussed above, for the SW curves these are $\mathbf{u} = \infty$ and $\mathbf{u} = \pm 1$. Every singular fibre has an associated Kodaira type, which is characterised by the order of vanishing of the Weierstraß invariants (see Appendix A.6). The singularities $\mathbf{u} = \pm 1$ correspond to an I_1 singularity, while the singularity $\mathbf{u} = \infty$ corresponds to an I_4^* singularity. The Kodaira configuration of the pure SU(2) theory is thus $(I_4^*, 2I_1)$, which is known to be a modular elliptic surface [38].

The singularity configuration $(I_4^*, 2I_1)$ is closely related to properties of the fundamental domain as plotted in Fig. 2: The cusp $\tau = i\infty$ has width 4, while there are two cusps $\tau = 0$ and $\tau = 2$ of width 1. The precise definition of a cusp and its width is given in Appendix A.3. From a fundamental domain such as the one in Fig. 2, the widths can be read off from the number of copies of \mathcal{F} that taper to a cusp. Furthermore, when a Hauptmodul for the subgroup is known, then the width is given by the smallest τ -periodicity of the q-expansion around a cusp. From (1.34) it is clear that at $\tau = i\infty$ this is 4.

Finally, there is an algebraic explanation, which is slightly more technical. Let us assume that f is a modular function for a congruence subgroup Γ of $PSL(2,\mathbb{Z})$. Then it is known that f is the root of a polynomial over the algebraic field $\mathbb{C}(PSL(2,\mathbb{Z}))$ of modular functions on $PSL(2,\mathbb{Z})$. A proof of this statement can for instance be found in [59], and is based on the following observation: If Γ is a congruence subgroup of $PSL(2,\mathbb{Z})$, then $PSL(2,\mathbb{Z}) = \bigcup_{j=1}^{n} \alpha_j \Gamma$, where $\alpha_j \in PSL(2,\mathbb{Z})$ for $j = 1, \ldots, n$ are coset representatives, with n the index of Γ in $PSL(2,\mathbb{Z})$. We can construct a function $g(\tau) = \prod_{j=1}^{n} (f(\alpha_j \tau) - X)$, with some constant X. Since $PSL(2,\mathbb{Z})$ transformations merely permute the coset representatives α_j , g is a modular function for $PSL(2,\mathbb{Z})$. If we consider this expression as a polynomial $P(X) = \prod_{j=1}^{n} (f \circ \alpha_j - X) \in \mathbb{C}(PSL(2,\mathbb{Z}))[X]$, then every coefficient must be a modular function for $PSL(2,\mathbb{Z})$. This shows that f is a root of the polynomial P(X) whose coefficients are modular function. tions for $PSL(2,\mathbb{Z})$. We discuss this aspect in more detail in Appendix A.5. What is clear from the above discussion is that the sextic equation (1.32) constructed from the SW curve is one example of such a polynomial whose roots are $PSL(2,\mathbb{Z})$ -transformations of the modular function u. We study generalisations of this aspect in great detail in section 2.

1.7 Topological QFT

Topological quantum field theory has been one of the most important contributions to mathematical physics in the 20th century, with a fruitful interplay between ideas from physics and mathematics [29, 60, 61]. It provides many insights into non-perturbative aspects of quantum field theory as well as low-dimensional topology [52, 62–69]. One example is Donaldson-Witten (DW) theory, which is a topological formulation of the $\mathcal{N} = 2$ supersymmetric Yang–Mills theory on an oriented smooth four-manifold X (see [52] for an introduction). Its equivalent IR (long distance) counterpart is an abelian theory, where Seiberg-Witten geometry dictates the physics [61]. One of the most important insights is that the classical Donaldson invariants can be derived from the Seiberg-Witten solution to $\mathcal{N} = 2$ supersymmetric Yang–Mills theory with gauge group SU(2) [61, 70–73]. In the seminal paper [73], the path integral of DW theory was found as

$$Z_{\rm DW} = Z_u + Z_{\rm SW},\tag{1.39}$$

where Z_{SW} denotes the generating function of SW invariants of the fourmanifold [71], while Z_u denotes the contribution to Z_{DW} from the Coulomb branch of the low-energy effective U(1) theory, the so-called *u*-plane. The *u*plane and its contribution to the path integral were studied in detail in [73–76]. The *u*-plane integral Z_u is of particular interest since it is non-vanishing only for four-manifolds with $b_2^+(X) \in \{0,1\}$. In turn, such four-manifolds are of particular interest since they are candidate topologies that probe the Coulomb branch of the theory.

Recently, interest in DW theory, and in particular the *u*-plane integral, was revived due to observations relating the latter for special four-manifolds to the theory of mock modular forms and harmonic Maass forms [77, 78]. For more generic compact four-manifolds, it was later reformulated in terms of the modular completion of a mock modular form [79-86].

One motivation of this thesis is to study correlation functions for topological theories other than the pure $\mathcal{N} = 2$ SYM. Many results were already obtained in the 1990s, for instance the generalisation to higher rank gauge groups [87], massless QCD [88], superconformal theories [66,89], extensions to non-simply connected four-manifolds [76, 90], etc. The formulation and computation of such *u*-plane integrals has remained a challenge, and an explicit evaluation for arbitrary four-manifolds and general 4d $\mathcal{N} = 2$ theories has not yet been achieved.

The rest of the thesis is structured as follows. In the following section 2, we discuss the generalisation of the aspects introduced in section 1.6 to the asymptotically free $\mathcal{N} = 2$ supersymmetric QCD with gauge group SU(2). We argue that the correspondence (1.38) does not exist when hypermultiplets are introduced, and instead propose an alternative formulation in terms of fundamental domains that do not correspond to subgroups of SL(2, Z). The obstruction is described precisely by a locus of branch points, which renders $u(\tau)$ non-holomorphic and thus non-modular. The branch points are closely related to other Coulomb branch functions such as $\frac{da}{du}$ or the discriminant through a generalisation of Matone's relation. We demonstrate our claims in a plethora of examples that we work out explicitly.

In section 3, we discuss the four flavour SU(2) theory, which is superconformal up to mass terms. While often it is studied in the massless limit, allowing for generic masses gives a more intricate structure of the Coulomb branch, where the symmetry group provides an interesting permutation of special mass configurations. This allows us to find nontrivial examples of bimodular forms and vector-valued bimodular forms.

In section 4, we then move on to the SU(3) theory, where the Coulomb branch is complex two-dimensional. For pure $\mathcal{N} = 2$ SYM, we demonstrate that there is a map from the genus two SW curve to elliptic (i.e. genus one) curves, which allows to study the Coulomb branch geometry using elliptic modular forms. For the locus where mutually local dyons become massless, we find a natural generalisation of the domain 2 for the pure SU(2) theory. For the locus containing the superconformal Argyres-Douglas vacua on the other hand, we find a fundamental domain for an Atkin-Lehner group, which is a subgroup of PSL(2, \mathbb{R}) rather than PSL(2, \mathbb{Z}). This is due to the fact that this locus gives rise to a non-rational elliptic surface.

In section 5, we consider the topological twists of the theories studied in section 2 and 3, mainly focusing on the asymptotically free theories. The fundamental domains for the effective gauge coupling can be understood as integration domains for topological correlation functions. One of the key ideas of the map from the *a*-plane to the fundamental domains is that the integrand becomes a modular function of τ . This allows us to prove that the integrand is single-valued under modular transformations, which is an important and nontrivial consistency check of the topological theory. We also present a formula for the evaluation of such correlation functions.

Finally, in section 6 we study the Donaldson-Witten theory on non-simply connected four-manifolds X. By adding a certain sum of Q-exact operators, we show that correlation functions can be computed using mock modular forms. This generalises recent results on the case where $b_1(X) = 0$.

We conclude in section 7, with giving directions for future research and suggesting extensions and applications of our results to other theories. In Appendix A, we finally collect properties and definitions of modular forms and elliptic curves, which are of fundamental importance for this thesis.

2 Asymptotically free $\mathcal{N} = 2$ QCD

In this section, we study the duality and modularity of $\mathcal{N} = 2$ SYM with $N_f = 0, 1, 2, 3$ hypermultiplets in the fundamental representation. These theories have a negative one-loop beta function and thus are asymptotically free. This section is mainly based on [2], while parts of section 2.2 are based on [5] and parts of section 2.6 are based on unpublished work.

2.1 Introduction

A manifestation of S-duality or strong-weak coupling duality is the equivalent dynamics of a quantum field theory at distinct values of its coupling constant [19, 45, 91-93]. A natural question for such a quantum field theory is the determination of a domain for the coupling constant parametrizing inequivalent quantum field theories. In this section, we address this question for asymptotically free $\mathcal{N} = 2$ Yang–Mills theories with gauge group SU(2) and $N_f \leq 3$ fundamental hypermultiplets. To this end, we consider the order parameter for the Coulomb branch, which is a function of the running coupling τ invariant under S-duality [45,46,94–97]. We put forward a fundamental domain \mathcal{F}_{N_f} for τ such that the function is one-to-one. Part of our motivation is the u-plane integral [73, 74], which is a physical approach to Donaldson invariants and other topological gauge-theoretic invariants of smooth compact four-manifolds. This approach involves an integral over the Coulomb branch of the theory. Recently, the change of variables from u to τ has been instrumental for the evaluation of the integral for generic four-manifolds [76–86, 98, 99], which suggests a potential fundamental role for this parametrisation of the Coulomb branch.

The Coulomb branches of the rank 1 theories mentioned above are complex one-dimensional, and parametrised by the Higgs vacuum expectation value $u = \frac{1}{16\pi^2} \langle \text{Tr}\phi^2 \rangle$, ϕ being the complex scalar of the $\mathcal{N} = 2$ vector multiplet [46] (see [8,47] for a review). In general, these order parameters are functions of the running coupling τ , the masses m_i of the hypermultiplets and the dynamical scales Λ_{N_f} generated by the renormalisation group flow.

Compared to the pure SU(2) case, in the massive theories with $N_f \leq 3$, we find a number of new phenomena. To study these theories, we consider their order parameters as roots of certain degree six polynomials constructed from the Seiberg-Witten (SW) curves. These polynomials in turn encode many of the interesting structures of the Coulomb branches. For example, their ramification loci include the Argyres-Douglas (AD) theories, where the curves degenerate, as well as branch points. We show that the fundamental domain of u can be described as six or less copies of the corresponding fundamental domain of the full modular group SL(2, Z) as displayed in Figure 1. The cusps of these domains correspond to the singularities of the physical theory and the width of each cusp to the number of hypermultiplets becoming massless there. The images of the $SL(2,\mathbb{Z})$ fundamental domains under the map $u(\tau)$ provide intriguing partitions of the *u*-plane. See for example Figures 5, 6 and 9

Since the polynomials are order six in u it is in general not possible to find the roots, and solve for u in terms of the coefficients. Only for special configurations of the masses, e.g., equal masses in $N_f = 2$ and one non-zero mass in $N_f = 3$, the polynomial splits over the field of modular functions for a congruence subgroup of $SL(2, \mathbb{Z})$, and we can thus find explicit closed expressions for u in terms of known modular forms, reproducing and extending previous results [88, 100–103].

For generic choices of the masses, the function $u(\tau)$ gives rise to branch points $\tau_{\rm bp}$, where $u - u(\tau_{\rm bp}) \sim \sqrt{\tau - \tau_{\rm bp}} + \ldots$ does not return to itself as τ encircles $\tau_{\rm bp}$. While the branch points, and the inevitable branch cuts, obstruct the identification of \mathcal{F}_{N_f} as a quotient $\Gamma \setminus \mathbb{H}$ with Γ a congruence subgroup, they provide a mechanism for \mathcal{F}_{N_f} to evolve as function of the mass. More precisely, the branch points move in the domain \mathcal{F}_{N_f} upon varying the masses, and the domain \mathcal{F}_{N_f} is literally cut and glued along the branch cuts. This provides a way to analyse how the domain evolves as function of the masses. We have studied this phenomena in detail in the following limits:

• Decoupling of a hypermultiplet:

A hypermultiplet decouples in the limit that its mass goes to infinity, $m \to \infty$. We demonstrate that in this situation, a branch cut disconnects (or cuts) the strong coupling cusp associated to this hypermultiplet from the rest of the domain. At the same time, the sides of the branch cut are identified to the sides of another branch cut. In this way, the strong coupling cusp is glued back to the weakly coupled cusp, near $i\infty$, where these branch points and cuts disappear in the limit $m \to \infty$. As a result, the periodicity at $i\infty$ increases by 1 in the limit, while the cusp has disappeared from the strongly coupled region. This is displayed for $N_f = 1$ in Figures 10.

• Merging of local singularities:

For a generic choice of masses, the theory with N_f hypermultiplets has $N_f + 2$ distinct strong coupling singularities in the *u*-plane, where dyons become massless and the effective field theory breaks down. By tuning the masses to special values, the singularities for l mutually local dyons can merge in the *u*-plane. We demonstrate that such cases give rise to a cusp with width l > 1 in \mathcal{F}_{N_f} . Moreover, when perturbing away from such a special value of the masses, we find that two branch cuts develop from the cusp, which disconnect the singularity in \mathcal{F}_{N_f} . This is displayed for $N_f = 2$ in Figure 12.

• Merging of non-local singularities (AD theories):

The dynamics is quite different if we tune the masses to special values where singularities corresponding to non-local dyons collide in the *u*- plane. Such singularities give rise to superconformal Argyres-Douglas (AD) field theories [104–109]. In such a situation, we find that two branch points in \mathcal{F}_{N_f} typically come together and annihilate at the preimage τ_{AD} of the AD singularity u_{AD} . The two branch cuts join at τ_{AD} in the interior of \mathcal{F}_{N_f} , and disconnect a region from \mathcal{F}_{N_f} with the "non-local" cusps. Thus \mathcal{F}_{N_f} consists then of < 6 copies of the SL(2, Z) fundamental domain, and the AD point is in a sense a remnant of the disconnected region. On the other hand if we take the appropriate scaling limit near the CFT point [105], we find that the disconnected region is a fundamental domain for the order parameter of the AD theory. If no other branch points remain in \mathcal{F}_{N_f} , the order parameters become modular functions for a congruence subgroup.

Let us briefly return to the *u*-plane integral. The change of variables from u to τ , gives rise to the factor $du/d\tau$ in the integrand. Interestingly, $du/d\tau$ can be expressed in case of the $N_f = 0$ theory in terms of the discriminant Δ and du/da [73], which is a consequence of a relation between the prepotential and u [100, 110]. Up to numerical constants, Δ and du/da are precisely the two gravitational couplings of the topological theory [72, 111], such that $du/d\tau$ is naturally included. Extending previous work on the massless $N_f \leq 3$ [88], we derive a further generalisation for all cases $N_f \leq 4$ with generic masses. We also discuss how this relation encodes interesting information on the special points of the Coulomb branch, specifically the branch points.

2.2 Fundamental domains

In this section, we develop techniques to determine a fundamental domain for the effective coupling of asymptotically free $\mathcal{N} = 2$ SU(2) SQCD.

2.2.1 The SW solutions

We recall a few essential aspects of the SW solutions for these theories, which we use to analyse u as function of τ . The gauge group SU(2) is spontaneously broken to U(1) on the Coulomb branch. The order parameter for this branch is the vev u, defined as

$$u = \frac{1}{16\pi^2} \left\langle \operatorname{Tr} \phi^2 \right\rangle_{\mathbb{R}^4} \in \mathcal{B}_{N_f}, \qquad (2.1)$$

where the trace is in the 2-dimensional representation of SU(2). Topologically, \mathcal{B}_{N_f} is the complex plane \mathbb{C} minus $2 + N_f$ singular points (for generic masses).

The scalar field related to the photon in the low energy effective field theory is a, while a_D is related to the dual photon. The SW solution identifies these fields as periods of a specific differential λ over two dual cycles, A and B, of an elliptic curve with complex structure τ ,

$$a = \int_{\gamma} \lambda, \qquad a_D = \int_{\gamma_D} \lambda.$$
 (2.2)

To list the SW curves of the theories with $N_f \leq 3$ hypermultiplets, let Λ_{N_f} be the scale of the theory with N_f hypermultiplets, and m_j , $j = 1, \ldots, N_f$ be the masses of the hypermultiplets. The SW curves of the theories are given by $[46]^5$

$$N_{f} = 0: \quad y^{2} = x^{3} - ux^{2} + \frac{1}{4}\Lambda_{0}^{4}x,$$

$$N_{f} = 1: \quad y^{2} = x^{2}(x - u) + \frac{1}{4}m\Lambda_{1}^{3}x - \frac{1}{64}\Lambda_{1}^{6},$$

$$N_{f} = 2: \quad y^{2} = (x^{2} - \frac{1}{64}\Lambda_{2}^{4})(x - u) + \frac{1}{4}m_{1}m_{2}\Lambda_{2}^{2}x - \frac{1}{64}(m_{1}^{2} + m_{2}^{2})\Lambda_{2}^{4},$$

$$N_{f} = 3: \quad y^{2} = x^{2}(x - u) - \frac{1}{64}\Lambda_{3}^{2}(x - u)^{2} - \frac{1}{64}(m_{1}^{2} + m_{2}^{2} + m_{3}^{2})\Lambda_{3}^{2}(x - u) + \frac{1}{4}m_{1}m_{2}m_{3}\Lambda_{3}x - \frac{1}{64}(m_{1}^{2}m_{2}^{2} + m_{2}^{2}m_{3}^{2} + m_{1}^{2}m_{3}^{2})\Lambda_{3}^{2}.$$

$$(2.3)$$

The family of SW curves are Jacobian rational elliptic surfaces with singular fibres [39, 40, 114, 115]. Rational in this context means that g_2 and g_3 are polynomials in u of degree at most 4 and 6, respectively [116].

Decoupling a hypermultiplet corresponds to the following double scaling limit [117]

$$m_j \to \infty, \quad \Lambda_{N_f} \to 0, \quad m_j \Lambda_{N_f}^{4-N_f} = \Lambda_{N_f-1}^{4-(N_f-1)}$$
(2.4)

One can directly decouple more than one hypermultiplet, where the scales of the low energy theories are defined as

$$\Lambda_0^2 = m\Lambda_2, \qquad \Lambda_0^4 = m^3\Lambda_3, \qquad \Lambda_1^3 = m^2\Lambda_3, \tag{2.5}$$

and m is the equal mass of the hypermultiplets being decoupled. These curves are constructed in such a way that their mathematical discriminants will, up to an overall normalisation, correspond to the *physical discriminant*. This we define as the monic polynomial,

$$\Delta_{N_f} \coloneqq \prod_{i=1}^{N_f+2} (u-u_i), \qquad (2.6)$$

with u_i being singular points of the effective theory, where hypermultiplets become massless. It is a polynomial of degree deg $\Delta_{N_f} = N_f + 2$ in u.⁶ To see this, we bring the SW curves (2.3) into Weierstraß form by shifting $x \to x + \frac{u}{3} + \frac{\Lambda_3^2}{192} \delta_{3,N_f}$, and rescaling $y \to y/2$,

$$\mathcal{W}: \quad y^2 = 4 \, x^3 - g_2 \, x - g_3, \tag{2.7}$$

⁵There are other formulations of the SW curve. For example the class S form is $x^2 = p_{N_f}(z, u, \Lambda_{N_f}, \boldsymbol{m})$ [20,112,113]. This has the advantage that the SW differential is canonically determined as $\lambda = x \, dz$. The analysis in the present section still holds for these formulations.

⁶One important note is that in [45] another convention is used for the curve of the pure theory. This gives the duality group $\Gamma(2)$ rather than $\Gamma^0(4)$ as in the above. The $\Gamma(2)$ -convention, however, turns out to not be suitable for the discussion in this section due to multiplicities of the singularities of the curve.

where $g_2 = g_2(u, \boldsymbol{m}, \Lambda_{N_f})$ and $g_3 = g_3(u, \boldsymbol{m}, \Lambda_{N_f})$ are polynomials in u, $\boldsymbol{m} = (m_1, \ldots, m_{N_f})$ and the scale Λ_{N_f} . The discriminant Δ_{N_f} is unchanged for this change of variables, and equals

$$\Delta_{N_f} = (-1)^{N_f} \Lambda_{N_f}^{2N_f - 8} (g_2^3 - 27 g_3^2), \qquad (2.8)$$

where the last factor is the "mathematical" discriminant. The functions g_2 and g_3 can be combined to an absolute invariant \mathcal{J} ,

$$\mathcal{J} = 12^3 \frac{g_2^3}{g_2^3 - 27g_3^2}.$$
 (2.9)

As opposed to g_2 and g_3 , \mathcal{J} is invariant under admissible changes of variables. Two curves are isomorphic if and only if they have the same absolute invariant \mathcal{J} . Since $g_2(u, \boldsymbol{m}, \Lambda)$ and $g_3(u, \boldsymbol{m}, \Lambda)$ are polynomial functions of u, \boldsymbol{m} and Λ for the SW curves, \mathcal{J} is naturally a rational function $\mathcal{J}(u, \boldsymbol{m}, \Lambda)$ of these variables. On the other hand, the modular Weierstraß form expresses \mathcal{J} in terms of the complex structure τ , namely as the modular *j*-invariant $j(\tau)$ (see (A.9) for a definition).

$$\mathcal{J}(u, \boldsymbol{m}, \Lambda) = j(\tau). \tag{2.10}$$

This allows to obtain u as function of τ , which is physically the effective coupling constant. Cusps are points where $j(\tau) = \infty$, which correspond to $\tau \in \{i\infty\} \cup \mathbb{Q}$. The *j*-function has fundamental domain $\mathcal{F} = \mathrm{SL}(2,\mathbb{Z}) \setminus \mathbb{H}$, which is typically taken to be the key-hole fundamental domain displayed in Figure 1. In other words, the function $j : \mathcal{F} \to \mathbb{C}$ is a bijective map.

2.2.2 Partitioning the upper half-plane

We are interested in determining the fundamental domains \mathcal{F}_{N_f} for the effective coupling τ for a theory with $0 \leq N_f < 4$. Let us consider u as a function,

$$u: \mathbb{H} \longrightarrow \mathcal{B}_{N_f},\tag{2.11}$$

and study the analytic properties of this map. We will discuss later the dependence of \mathcal{F}_{N_f} on the masses \boldsymbol{m} , which we will make manifest in the notation as $\mathcal{F}_{N_f}(\boldsymbol{m})$ or more compactly $\mathcal{F}(\boldsymbol{m})$. We find that for $N_f \geq 1$ and generic masses the duality group does not act on τ by fractional linear transformations. This prevents us from defining a fundamental domain as is customary for a congruence subgroup Γ of $SL(2,\mathbb{Z})$: For any point $\tau \in \mathbb{H}$ there exists a $g \in \Gamma$ such that $g \cdot \tau \in \Gamma \setminus \mathbb{H}$, and no two distinct points τ, τ' in $\Gamma \setminus \mathbb{H}$ are equivalent to each other under Γ . Rather, we can compare if points τ, τ' are equivalent under (2.11): If we define the equivalence relation

$$\tau \sim \tau' \iff u(\tau) = u(\tau'),$$
 (2.12)

then the quotient set \mathbb{H}/\sim is a fundamental domain \mathcal{F}_{N_f} for the function u. Upon plotting \mathcal{F}_{N_f} as a domain in \mathbb{H} , we will have to introduce identifications along co-dimension 1 segments as for \mathcal{F} in Figure 1. To determine \mathcal{F}_{N_f} , we bring (2.9) into a more convenient form. We multiply (2.9) by Δ_{N_f} and bring all terms to one side. This gives the sextic polynomial,

$$P_{N_f}(X) \coloneqq \left(g_2(X, \boldsymbol{m}, \Lambda)^3 - 27g_3(X, \boldsymbol{m}, \Lambda)^2\right) j - 12^3 g_2(X)^3$$

= $a_6 X^6 + a_5 X^5 + \ldots + a_1 X + a_0,$ (2.13)

where the coefficients $a_i = a_i(\boldsymbol{m}, \Lambda, j)$ are polynomial functions of \boldsymbol{m}, Λ , and the *j* function, $a_i(\boldsymbol{m}, \Lambda, j) \in \mathbb{C}[\boldsymbol{m}, \Lambda, j]$. The polynomials (2.13) can thus be viewed as polynomials over the field $\mathbb{C}[\boldsymbol{m}, \Lambda, j]$. As discussed in Appendix A.5, roots to such polynomials can be considered as algebraic modular forms, generalising the well-studied class of classical modular forms for congruence subgroups of PSL(2, \mathbb{Z}).

We see that (2.9) is equivalent to $P_{N_f}(u) = 0$ for $\Delta_{N_f} \neq 0$, or in other words, away from the singular locus of the theory. The roots of P_{N_f} can therefore be identified with the order parameter of the corresponding SW curve. Recall that we can assign $U(1)_{\mathcal{R}}$ charges $[u : m_i : x : y] = [4 : 2 : 4 : 6]$ to the quantities of the Seiberg-Witten curves [46]. Since g_2 and g_3 are polynomials in u by construction, by bringing the SW curves to the Weierstraß form and using that [u] = 4 we have that the degrees of g_2 and g_3 as polynomials in umust be $\deg(g_2) = 2$ and $\deg(g_3) = 3$. Therefore, P_{N_f} is a sextic polynomial in X.

For generic masses \boldsymbol{m} , the sextic equation $P_{N_f} = 0$ gives rise to n = 6different solutions as functions of j, while for special choices of \boldsymbol{m} , such as those giving rise to superconformal (AD) theories, we have $2 \leq n \leq 4$ different j-dependent solutions and 6-n j-independent solutions. Since $j : \mathcal{F} \to \mathbb{C}$ is an isomorphism, the $n \leq 6$ solutions provide a multi-valued (n-valued) function over \mathcal{F} .

To obtain u as a single-valued function of the effective coupling, we choose a different copy of \mathcal{F} for each of the $n \leq 6$ branches, and appropriately identify the boundaries of these domains. These are related to \mathcal{F} by an element of $SL(2,\mathbb{Z})$, and their union is

$$\mathcal{F}_{N_f} = \bigcup_{j=1}^n \alpha_j \mathcal{F},\tag{2.14}$$

with $\alpha_j \in \mathrm{SL}(2,\mathbb{Z})$. A priori, there is no canonical choice for the α_j , they are determined up to the action of the duality group of the theory. However, some choices are more natural than others. If we demand that \mathcal{F}_{N_f} is connected and take $\alpha_1 = \mathbb{1} \in \mathrm{SL}(2,\mathbb{Z})$, there is only a finite number of choices for \mathcal{F}_{N_f} . In some cases, \mathcal{F}_{N_f} is a modular curve $\Gamma \setminus \mathbb{H}$ for a congruence subgroup $\Gamma \subseteq \mathrm{SL}(2,\mathbb{Z})$. In such cases, n equals the index of Γ in $\mathrm{SL}(2,\mathbb{Z})$ [118] (see also Appendix A.3 for the corresponding definitions for modular curves). For later use, we define the set of α_j as $\mathcal{C}_{N_f} = \{\alpha_j, j = 1, \ldots, n\}$.

For generic masses, n = 6 and \mathcal{F}_{N_f} has $3 + N_f$ cusps, corresponding to weak coupling $\tau \to i\infty$ and the $2 + N_f$ singularities of the theory. We find the widths
of the cusps by expanding $j(\tau) = \mathcal{J}(u, \boldsymbol{m}, \Lambda_{N_f})$ for τ near the cusp. For general $N_f \in \{0, 1, 2, 3\}$, the cusp at infinity has width $h_{\infty} = 4 - N_f$. This is because $q^{-1} \sim j(\tau) = \mathcal{J} \sim u^{4-N_f}$, which implies $u(\tau) \sim q^{-\frac{1}{4-N_f}}$ (where $q = e^{2\pi i \tau}$). Thus for large τ , $u(\tau)$ is invariant under T^{4-N_f} , where $T : \tau \mapsto \tau + 1$. Near any singularity u_s , it is clear that $q^{-1} \sim \frac{1}{(u-u_s)^{h_s}}$, where n_s is the multiplicity of the singularity. Similarly, near u_s one finds $u(\tau) - u_s \sim q^{\frac{1}{h_s}}$. Locally, the function $u(\tau)$ has period h_s , giving the width h_s of the cusp. The widths of all cusps then add up to 6,

$$h_{\infty} + \sum_{s} h_s = 6. \tag{2.15}$$

As mentioned above, the equation $P_{N_f} = 0$ gives six different solutions for u. A natural question that then arises is which of these six to use as our u. In some sense this is of course arbitrary, all of them correspond to the order parameter u simply expressed in different duality frames. On the other hand, the most natural solution is the one corresponding to the weak coupling duality frame where |u| is large for $\tau \to i\infty$. Since the width of the cusp at infinity is $4-N_f$ we see that there is still some ambiguity in this choice as long as $N_f < 3$, but for $N_f = 3$ there is exactly one choice. We show in section 2.6 that this has $u \to -\infty$ for $\tau \to i\infty$, and it turns out that this choice can be taken for all $N_f \leq 3$ theories, and is preserved under the decoupling of hypermultiplets, we therefore make this choice throughout. Note that this sign differs from the conventional choice in the literature [46, 88, 102].

Different mass configurations can give different decompositions of 6. When singularities merge, their cusps are identified under the duality group and their widths add up. Moreover a cusp moves from the real axis to infinity upon decoupling of a matter multiplet.

For special choices of the masses, not all singularities correspond to cusps $i\infty$ or the real line; also singularities in the interior of the upper half-plane can occur. The theories at these points are of superconformal or Argyres-Douglas type, and the widths of all cusps add up to n.

Yet another aspect of the parametrisation by τ is that for special values of τ in the interior of \mathcal{F} , otherwise distinct solutions can coincide. These are branch points of the solutions, where the function $u(\tau)$ ceases to be meromorphic in τ . The branch points in \mathcal{F}_{N_f} emanate a branch cut. We will discuss these aspects in more detail in section 2.2.3.

For generic masses the equation $P_{N_f}(X) = 0$ furthermore defines a Riemann surface, which is a 6-fold ramified covering over the classical modular curve $SL(2,\mathbb{Z})\setminus\mathbb{H}$ [119]. On this Riemann surface, any root u forms a *meromorphic* map to the Coulomb branch. It would be interesting to study the topology of these surfaces in more detail. See also Appendix A.5 and [120].

Even if the duality group is not a congruence subgroup of $SL(2,\mathbb{Z})$, there is a procedure to find closed expressions for the order parameters in special cases. The sextic equation (2.13) for fixed masses \boldsymbol{m} and scale Λ can be viewed as a polynomial over the algebraic field $\mathbb{C}(\Gamma)$ of modular functions on $\Gamma =$ $\mathrm{SL}(2,\mathbb{Z})$. Such nontrivial polynomials define field extensions over $\mathbb{C}(\Gamma)$. By the fundamental theorem of Galois theory, there is a one-to-one correspondence between the Galois group of the field extension and its intermediate fields. Intermediate fields can be obtained by adjoining roots of the polynomial to the base field. Since $P_{N_f}(X)$ is a sextic polynomial, for generic masses \boldsymbol{m} it is not possible to find exact expressions for the roots. However, if one of the intermediate fields is known, the polynomial factors over the intermediate field into products of polynomials of lower degree. If the resulting degree is less than or equal to 4, there are closed formulas for the roots.

We find below that in many cases, such as massive $N_f = 2$ and 3 with one mass parameter, $\mathbb{C}(\Gamma(2))$ for the principal congruence subgroup $\Gamma(2)$ (see Appendix A.1) is an intermediate field. Since the function $\lambda = \frac{\vartheta_2^4}{\vartheta_3^4}$ is a Hauptmodul for the genus 0 congruence subgroup $\Gamma(2)$, it is the root of a polynomial of degree $[\Gamma : \Gamma(2)] = 6$ over $\mathbb{C}(\Gamma)$. More precisely, there exists a rational function \mathcal{R} with the property that $\mathcal{R}(\lambda(\tau)) = j(\tau)$. It is given by

$$\mathcal{R}(p) = 2^8 \frac{(1 + (p-1)p)^3}{(p-1)^2 p^2}.$$
(2.16)

Instead of solving $\mathcal{J}(u, \boldsymbol{m}, \Lambda) = j(\tau)$ we can then rather solve $\mathcal{J}(u, \boldsymbol{m}, \Lambda) = \mathcal{R}(\lambda(\tau))$. If $\mathbb{C}(\Gamma(2))$ is an intermediate field, the sextic equation corresponding to this equation factors over $\mathbb{C}(\Gamma(2))$. In massive $N_f = 2, 3$ we find that it factors into three quadratic polynomials with coefficients depending on λ , which can be easily solved analytically. Such rational relations between the *j*-invariant and Hauptmoduln exist for any genus 0 congruence subgroup, which are classified. They allow to invert the equation $\mathcal{J}(u, \boldsymbol{m}, \Lambda) = j(\tau)$ for a large class of mass parameters, as we demonstrate in the following sections. See also [121–123].

2.2.3 Ramification locus

The covering $\mathcal{F}_{N_f}(\boldsymbol{m}) \to \mathcal{B}_{N_f}$ is not 1-to-1 on a discrete subset, namely at points of $\mathcal{F}_{N_f}(\boldsymbol{m})$ where the discriminant $D(P_{N_f})$ vanishes.⁷ In all cases, $N_f = 0, 1, 2, 3$, we find that the discriminant of P_{N_f} factorises as

$$D(P_{N_f}) = j^4 (j - 1728)^3 (D_{N_f}^{\rm AD})^3 D_{N_f}^{\rm bp}.$$
 (2.17)

We discuss each of the three factors:

The m-independent factor

⁷The discriminant of a polynomial $p(X) = X^n + a_{n-1}X^{n-1} + \ldots + a_1X + a_0 = \prod_{j=1}^n (X - r_j)$ is defined as $D(p) = \prod_{i < j} (r_i - r_j)^2$, in particular it vanishes if and only if two roots coincide. Since we are interested in finding the zeros of D(p), we are not careful about overall normalisation factors.

The factor $j^4 (j-1728)^3$ is independent of the masses \boldsymbol{m} and can be understood from (2.13). It is immediate that when $j = 12^3$, every root of P_{N_f} has multiplicity at least 2, and if j = 0 every root has multiplicity at least 3. On \mathbb{H} this occurs whenever $\tau \in \mathrm{SL}(2,\mathbb{Z}) \cdot i$ or $\tau \in \mathrm{SL}(2,\mathbb{Z}) \cdot \omega_3$, with $\omega_j = e^{2\pi i/j}$. On the modular curve $\mathrm{SL}(2,\mathbb{Z}) \setminus \mathbb{H}$, these orbits collapse to a point and in fact the covering π is ramified only over $\{i\infty, i, \alpha\}$, or $j \in \{0, 1728, \infty\}$, respectively. This resembles the *Belyi functions*, which are holomorphic maps from a compact Riemann surface to $\mathbb{P}^1(\mathbb{C})$ ramified over precisely these three points [39, 124]. They can be described combinatorially by so-called *dessins d'enfants*. Such dessins have also appeared in the context of SW theory [125–127]. For generic masses, the SW family of curves do not satisfy this definition, as there are additional ramification points.

The polynomial $D_{N_f}^{\text{AD}}$

The factor $D_{N_f}^{\text{AD}}$ corresponds to Argyres-Douglas (AD) loci, where two or more singularities coincide [104, 105]. More precisely, the zero locus of $D_{N_f}^{\text{AD}}$ corresponds to the masses for which the Coulomb branch contains AD points. To see this, recall that the AD points correspond to

$$g_2(u, \boldsymbol{m}, \Lambda) = g_3(u, \boldsymbol{m}, \Lambda) = 0.$$
(2.18)

Since g_2 and g_3 are polynomials in u of degrees 2 and 3, respectively, we can eliminate u from the above equations and characterise $\mathcal{L}_{N_f}^{\text{AD}}$ as the zero locus of a polynomial $D_{N_f}^{\text{AD}}$ in \boldsymbol{m} ,

$$\mathcal{L}_{N_f}^{\text{AD}} = \{ \boldsymbol{m} \in \mathbb{C}^{N_f} | D_{N_f}^{\text{AD}}(\boldsymbol{m}) = 0 \}.$$
(2.19)

These are precisely the polynomials appearing in (2.17). From the SW curves we can easily find that they are given by

$$\begin{split} D_0^{\rm AD} &= 1, \\ D_1^{\rm AD} &= 27\Lambda_1^3 - 64m^3, \\ D_2^{\rm AD} &= \Lambda_2^6 - 12m_1m_2\Lambda_2^4 + 3(9m_1^4 + 9m_2^4 - 2m_1^2m_2^2)\Lambda_2^2 - 64m_1^3m_2^3, \\ D_3^{\rm AD} &= \Lambda_3^9 - 12\widetilde{M}_2\Lambda_3^7 + 168\widetilde{M}_3\Lambda_3^6 - 174\widetilde{M}_4'\Lambda_3^5 + 48\widetilde{M}_4\Lambda_3^5 \\ &+ 168\widetilde{M}_2\widetilde{M}_3\Lambda_3^4 - 372\widetilde{M}_3^2\Lambda_3^3 + 24\widetilde{M}_6'\Lambda_3^3 - 64\widetilde{M}_6\Lambda_3^3 \\ &- 24\widetilde{M}_3\widetilde{M}_4'\Lambda_3^2 + 96\widetilde{M}_3\widetilde{M}_4\Lambda_3^2 + 6\widetilde{M}_2\widetilde{M}_3^2\Lambda_3 - 27\widetilde{M}_8'\Lambda_3 + 8\widetilde{M}_3^3, \end{split}$$
(2.20)

where for $N_f = 3$ we have defined the symmetric combinations

$$\widetilde{M}_{2k} = 2^{6k} \sum_{j=1}^{3} m_j^{2k}, \qquad \widetilde{M}_3 = 2^9 \prod_{j=1}^{3} m_j,$$

$$\widetilde{M}'_4 = 2^{12} \sum_{i < j} m_i^2 m_j^2, \qquad \widetilde{M}'_6 = 2^{18} \sum_{i \neq j} m_i^2 m_j^4, \qquad \widetilde{M}'_8 = 2^{24} \sum_{i < j} m_i^4 m_j^4.$$
(2.21)

The type of singularity that appears for specific masses on these loci are found by studying the order of vanishing of g_2 , g_3 and Δ according to the Kodaira classification,

$$II: \text{ ord}(g_2, g_3, \Delta) = (1, 1, 2) \text{ or } (2, 1, 2),$$

$$III: \text{ ord}(g_2, g_3, \Delta) = (1, 2, 3),$$

$$IV: \text{ ord}(g_2, g_3, \Delta) = (2, 2, 4).$$

(2.22)

See Appendix A.6 for more details. The zero loci of the AD polynomials can be understood as codimension 1 loci in the space $\mathbb{C}^{N_f} \ni \boldsymbol{m}$ [105]. For $N_f = 3$ such a locus is shown in Fig. 3. Argyres-Douglas loci are studied for a more general class of SW theories in [108].



Figure 3: The AD locus \mathcal{L}_3^{AD} for $N_f = 3$ with masses $\boldsymbol{m} = (m, m, \mu)$ in the real (m, μ) -plane, with units $\Lambda_3 = 1$. It is a union of three smooth lines, two of them generically describing type *II* AD points and the third one type *III*. The two *II* lines touch at a *III* point, while both *II* lines touch the *III* line in a type *IV* AD point.

In section 2.2.2, we argued that the widths of the different cusps of the SU(2) theories always add up to $n \leq 6$. We will now argue that n < 6 if and only if m is a zero m_{AD} of $D_{N_f}^{AD}$. It is possible that some zero of Δ is also a zero of g_2 . Then the index is given by the degree of the numerator of j, which can be smaller than 6. In Sections 2.4.2, 2.5.3 and 2.6.4-2.6.6 we study a few examples of AD theories appearing in the $N_f = 1, 2, 3$ theories, and demonstrate that the curve degenerates to Kodaira types II, III or IV. Each singularity type is not exclusive to a specific number of flavours, but appears on the discriminant divisor of the higher N_f theories as well [105]. See Sections 2.6.5 and 2.6.6 for two explicit examples of this. The three types of AD theories corresponds to 2, 3 or 4 mutually non-local states becoming massless at the AD point. The cusps corresponding to the non-local states are disconnected from the rest of the domain, and the branch points collide at an

elliptic point of the duality group. As a result, the index is reduced by ord Δ , which equals the number of mutually non-local states becoming massless, i.e., 2, 3, and 4 for the theories *II*, *III* and *IV*, respectively. Note that the order of vanishing of the discriminant may be larger than zero for ordinary singularities as well, so it is not enough to simply subtract ord Δ from six to get the index right but rather we should subtract the number of mutually non-local states becoming massless at each cusp,

$$n = 6 - \#$$
 (mutually non-local massless dyons). (2.23)

This is because for the index to reduce it is necessary for g_2 and Δ to have a common root, such that due to (2.8) it is also a root of g_3 and because of (2.18) therefore an AD point. In the limit $\mathbf{m} \to \mathbf{m}_{AD}$, the 6 - n copies of $\mathcal{F}_{N_f}(\mathbf{m})$ corresponding to the regular singularities are removed from the fundamental domain. We have also found mass configurations whose corresponding Coulomb branch contains *two* (type II) AD points. The correspondence (2.23) nevertheless holds, for a similar argument as presented above.

The polynomial $D_{N_f}^{\mathrm{bp}}$

The last factor $D_{N_f}^{\text{bp}}$ corresponds to branch points. These are values of j for which two solutions of $P_{N_f}(X) = 0$ coincide, such that the map $u : \mathcal{F}_{N_f}(\boldsymbol{m}) \to \mathcal{B}_{N_f}$ is not 1-to-1 on these points. The identifications are different from the multiple images of \mathcal{F} in \mathcal{B}_{N_f} , which identify the images of the boundary of \mathcal{F} , $\alpha_j(\partial \mathcal{F})$, in $\mathcal{F}_{N_f}(\boldsymbol{m})$.

The $D_{N_f}^{\mathrm{bp}}$ are explicitly given by

$$D_0^{\rm bp} = 1,$$

$$D_1^{\rm bp} = 27j\Lambda_1^6 - 27 \cdot 2^{14}m^3\Lambda_1^3 + 2^{20}m^6,$$

$$D_2^{\rm bp} = (m_1^2 - m_2^2)^2 j^2 \Lambda_2^8 - 128\Lambda_2^4 (216(m_1^8 + m_2^8) - 288m_1^2m_2^2(m_1^4 + m_2^4) + 16m_1^4m_2^4 + 240m_1^3m_2^3\Lambda_2^2 - 72m_1m_2(m_1^4 + m_2^4)\Lambda_2^2 + 9(m_1^4 + m_2^4)\Lambda_2^4 - 42m_1^2m_2^2\Lambda_2^4 - 2m_1m_2\Lambda_2^6)j + 2^{12}(16m_1m_2 - \Lambda_2^2)^3P_2^{\rm AD},$$

(2.24)

and we define $\mathcal{L}_{N_f}^{\text{bp}}$ as the zero locus of $D_{N_f}^{\text{bp}}$. The expression for D_3^{bp} for generic masses is very long so we do not write it out here, but we can note that it is has degree three in j. For later reference we write it out for two special mass configurations

$$D_3^{\rm bp}(m,m,m) = 432m^4 \Lambda_3^2 j + (8m - \Lambda_3)^2 (16m + \Lambda_3)^3 (64m + \Lambda_3),$$

$$D_3^{\rm bp}(m,0,0) = 16m^4 \Lambda_3^2 j + (8m - \Lambda_3)^3 (8m + \Lambda_3)^3.$$
(2.25)

To show that the zero locus of these polynomials really correspond to branch points we will need some specific details of the corresponding theory and we therefore hold off on this discussion until the respective sections below. We can, however, note that by solving $D_{N_f}^{\text{bp}} = 0$ for j and plugging it into (2.13) we get the corresponding solutions for u. For example, in $N_f = 1$ we find $u = \frac{4}{3}m^2$ and as we will see, away from $m = m_{AD} = \frac{3}{4}\Lambda_1$, this is not part of the discriminant of the curve and therefore does not correspond to a physical singularity of the theory. We denote a branch point of u in \mathcal{F}_{N_f} by τ_{bp} , and its image in \mathcal{B}_{N_f} as u_{bp} . As explained in Section 2.3.3, for generic masses there are two branch points τ_{bp} and τ'_{bp} with image $u_{bp} = u(\tau_{bp}) = u(\tau'_{bp})$. Since their image in \mathcal{B}_{N_f} is the same, the points τ_{bp} and τ'_{bp} are identified in \mathcal{F}_{N_f} , even though they appear as distinct points in plots of \mathcal{F}_{N_f} in \mathbb{H} . A branch cut emanates from each branch point; there can be a single cut connecting both branch points, or two separate cuts which go to either $i\infty$ or to the real axis.

Mutually local singularities

As the masses are tuned, some of the singularities on the Coulomb branch can collide. If we consider Δ_{N_f} as a polynomial in u, its discriminant $D(\Delta_{N_f})$ vanishes if and only if two roots coincide. It is straightforward to show that for $N_f \leq 3$,

$$D(\Delta_{N_f}) = \left(D_{N_f}^{\text{AD}}\right)^3 \prod_{i < j} (m_i - m_j)^2 (m_i + m_j)^2.$$
(2.26)

This factorises the locus in mass space where singularities collide into two orthogonal components: The first component is the Argyres-Douglas (AD) locus given by the polynomial equation $D_{N_f}^{AD} = 0$, where mutually *non-local* singularities collide [105, 128]. The other component is characterised by the equations $m_i = \pm m_j$, and one can check that this gives rise to mutually *local* singularities colliding. Here, the flavour symmetry gets enhanced and a Higgs branch opens up [46].

Given a mass configuration $\mathbf{m} = (m_1, \ldots, m_{N_f})$, we can denote by k_l the weight (or multiplicity) of the *l*-th singularity, and by $\mathbf{k}(\mathbf{m}) = (k_1, k_2, \ldots)$ the vector of those weights. Since the Coulomb branch $\mathcal{B}_{N_f}(\mathbf{m})$ contains $2 + N_f$ singularities aside from weak coupling $u = \infty$, it is clear that $\mathbf{k}(\mathbf{m})$ provides a partition of $2 + N_f$. This in turn partitions the mass space $\mathbb{C}^{N_f} \ni \mathbf{m}$ into finitely many regions where $\mathbf{k}(\mathbf{m})$ is locally constant. As an example, in Fig. 4 we plot the contours of (2.26) for $N_f = 2$ in the real $\mathbf{m} = (m_1, m_2)$ plane.

The possible singularity structures of the rational elliptic curves (2.3) are classified in Persson's list of allowed configurations of singular fibres [43, 44]. From Kodaira's classification, it follows that any solution to (2.26) gives rise to a singularity on the Coulomb branch of Kodaira type I_k , II, III, or IV. As described in [2,129], the solutions to $0 = D_{N_f}^{AD}$ give rise to AD points of Kodaira type II, III and IV. The second component $0 = \prod_{i < j} (m_i - m_j)(m_i + m_j)$ can be studied in more detail. These are $2(N_f - 1)$ independent equations. Whenever one of the factors vanishes, the SW surface contains an I_k singularity with $k \ge 2$. For $N_f = 2$, the only possibility is I_2 , while for $N_f = 3$ singularities of type I_2 , I_3 and I_4 are possible. The point in the Coulomb branch \mathcal{B}_{N_f} corresponding to an I_k singularity with $k \ge 2$ intersects with a Higgs branch of quaternionic dimension $k - 1 \ge 1$ [46]. Further merging these I_k singularities with a mutually non-local singularity does not affect the



Figure 4: Partitioning of the real $\boldsymbol{m} = (m_1, m_2)$ plane in $N_f = 2$, in units of $\Lambda_2 = 1$. On the AD component the Coulomb branch $\mathcal{B}_2(\boldsymbol{m})$ contains an AD point of Kodaira type II (blue) or III (green). On the other component (orange), mutually local singularities collide. If \boldsymbol{m} is varied along a continuous path that does not cross the partitioning $0 = D(\Delta_{N_f})$, the weight vector $\boldsymbol{k}(\boldsymbol{m})$ is constant.

Higgs branch, such that the points with a *III* or *IV* singularity also intersect with a Higgs branch of quaternionic dimension one or two, respectively, while the points with AD theories of type *II* do not intersect with any Higgs branch.

The genus of $\mathcal{F}_{N_f}(\boldsymbol{m})$

For special choices of the masses \boldsymbol{m} , $\mathcal{F}(\boldsymbol{m})$ coincides with the modular curve $X(\Gamma)$ for a congruence subgroup $\Gamma \in \mathrm{SL}(2,\mathbb{Z})$. Then the genus of $\mathcal{F}(\boldsymbol{m})$ is given by that of $X(\Gamma)$, for which there is the formula (A.40) in terms of the index n, the number of elliptic points ε_2 and ε_3 and the number of cusps ε_{∞} . In all examples of such masses \boldsymbol{m} discussed below, we find that $\mathcal{F}(\boldsymbol{m})$ is a genus zero Riemann surface. In the presence of branch points in $\mathcal{F}(\boldsymbol{m})$, Equation (A.40) needs to be modified. First, we note that for an AD theory, τ_{AD} corresponds to an elliptic point. In fact, in all AD cases studied here, (2.23) can be expressed as⁸

$$n = 6 - 2\varepsilon_3 - 3\varepsilon_2, \tag{2.27}$$

For ε_{∞} there is no simple formula since for example it is not unique in some limit $\mathbf{m} \to \mathbf{m}_{AD}$, but rather depends on the direction in mass space from which \mathbf{m}_{AD} is approached. As the map $u : \mathcal{F}(\mathbf{m}) \to \mathcal{B}_{N_f}$ is between Riemann surfaces $\mathcal{F}(\mathbf{m})$ and \mathcal{B}_{N_f} , we can consider the Riemann-Hurwitz formula (A.39), which relates their genera g. The inverse map $\tau : \mathcal{B}_{N_f} \to \mathcal{F}(\mathbf{m})$ can be defined

 $^{^{8}}$ The type IV AD point can be viewed as a collision of two elliptic fixed points of period 3.

through $\tau = \frac{da_D}{da} = \frac{da_D}{du} / \frac{da}{du}$, with the periods a, a_D given by (2.2). The dependence of τ on u is holomorphic everywhere [41, 42]. Then (A.39) for the inverse map implies that $0 = g_{\mathcal{B}_{N_f}} \geq g_{\mathcal{F}(\boldsymbol{m})}$, such that

$$g_{\mathcal{F}(\boldsymbol{m})} = 0. \tag{2.28}$$

Applying this to the Riemann-Hurwitz formula for the ramified covering $\mathcal{F}(\boldsymbol{m}) \rightarrow \mathcal{F}$, we find the number of distinct branch points on $\mathcal{F}(\boldsymbol{m})$ for arbitrary \boldsymbol{m} as

$$\sum_{\tau_{\rm bp}\in\mathcal{F}(\boldsymbol{m})} (e_{\tau_{\rm bp}} - 1) = \varepsilon_{\infty} - 3 + \varepsilon_2 + \varepsilon_3.$$
(2.29)

This shows that $\mathcal{F}(\boldsymbol{m})$ is a Riemann sphere with ε_{∞} cusps, ε_2 , ε_3 elliptic points of periods 2 and 3 and $\varepsilon_{\infty} - 3 + \varepsilon_2 + \varepsilon_3$ branch points. As an example, in massless $N_f = 1$ (see section 2.4.1) we have $\epsilon_{\infty} = 1 + 3 = 4$, while all singularities are on \mathbb{Q} and thus $\varepsilon_2 = \varepsilon_3 = 0$. There is one branch point in $\mathcal{F}(0)$, which agrees with (2.29).

2.2.4 Partitioning the *u*-plane

An approach to better understand the *u*-plane geometry is to study the partitions that the map $u: \mathcal{F}_{N_f} \to \mathcal{B}_{N_f}$ produces on the *u*-plane \mathcal{B}_{N_f} . Let us study the union (2.14). Now since $u(\mathcal{F}_{N_f}) = \mathcal{B}_{N_f}$, it is natural to ask what

$$\mathcal{T}_{\boldsymbol{m}} = u\left(\bigcup_{j=1}^{n} \alpha_j \partial \mathcal{F}\right) \subseteq \mathcal{B}_{N_f}$$
(2.30)

describes. The insight is that while $j : \mathcal{F} \to \mathbb{C}$ is an isomorphism, it surjects the boundary onto a half-line,

$$j(\partial \mathcal{F}) = (-\infty, 12^3] \subseteq \mathbb{R} \subseteq \mathbb{C}.$$
 (2.31)

This is straightforward to prove. On the half-lines $i[\frac{\sqrt{3}}{2},\infty)$ the q-series of j is an alternating series with the same Fourier coefficients as j and therefore real. On the arc $\{e^{\varphi i} | \varphi \in (\frac{\pi}{3}, \frac{2\pi}{3})\}$ the complex conjugate of $j(e^{\varphi i})$ is equal to the value of j at the S-transform of $e^{\varphi i}$ and therefore equal to $j(e^{\varphi i})$.

The only other region in \mathcal{F} where j is real are the $SL(2,\mathbb{Z})$ images of the half-line $i[1,\infty)$ on the imaginary axis. We can directly apply this to the SW curves, whose j-invariant $\mathcal{J}(u, \boldsymbol{m}, \Lambda)$ is identified with $j(\tau)$. The partitioning is then

$$\mathcal{T}_{\boldsymbol{m}} = \{ u \in \mathcal{B}_{N_f} \mid \mathcal{J}(u, \boldsymbol{m}, \Lambda_{N_f}) \in (-\infty, 12^3] \}.$$
(2.32)

It is included in the level set $\operatorname{Im} \mathcal{J} = 0$. Let us therefore study the curves

$$\operatorname{Im} \mathcal{J}(u, \boldsymbol{m}, \Lambda_{N_f}) = 0, \qquad (2.33)$$

which contrary to (2.32) are algebraic curves. It turns out that some of the components of this equation do not belong to the partitioning (2.32), and it



Figure 5: Identification of the components of the partitioning \mathcal{T} in the pure theory. The *u*-plane \mathcal{B}_0 is partitioned into 6 regions $u(\alpha \mathcal{F})$, with the $\alpha \in SL(2, \mathbb{Z})$ given in both pictures.

is clear that they correspond to components of curves with $j > 12^3$. Due to the imaginary part, it is instructive to choose coordinates $u/\Lambda_{N_f}^2 = x + iy$. The equations (2.33) are straightforward to compute in terms of zero-loci of polynomials in x and y. For fixed \boldsymbol{m} , they define algebraic varieties

$$T_{m}(x,y) = 0.$$
 (2.34)

More specifically, they are an N_f -parameter family of affine algebraic plane curves. For the pure $N_f = 0$ theory, one finds

$$T_0 = xy(81 - 288x^2 + 336x^4 - 128x^6 + 288y^2 - 352x^2y^2 - 128x^4y^2 + 336y^4 + 128x^2y^4 + 128y^6).$$
(2.35)

The identification of this partitioning of the *u*-plane for the pure theory is shown in Figure 5. The defining equations can be computed in full generality for any N_f , but they are rather lengthy: The polynomials T_m for generic masses have total degree $8 + N_f$. For generic real masses, the polynomials T_m have 30, 131, and 1081 terms in $N_f = 1$, 2 and 3, respectively. If we allow the masses to be complex, we can decompose $m_i = \text{Re } m_i + i\text{Im } m_i$ and the T_m are then polynomials in x, y, $\text{Re } m_i$ and $\text{Im } m_i$. For generic (complex) masses in $N_f = 1$, 2 and 3, T_m has 93, 1310 and 48754 terms, respectively.

The polynomials T_m are in general reducible. For instance, for $\mathbf{m} = (m, m)$ and $\mathbf{m} = (m, 0, 0)$, T_m factors into multiple nontrivial polynomials. It is straightforward to check that T_m for given N_f flows into T_m for $N_f - 1$ by decoupling one hypermultiplet. This allows to study the decoupling procedure of the fundamental domains in detail.

The partitioning \mathcal{T}_m is a finite union of smooth curves that intersect. The tessellation of \mathbb{H} in $\mathrm{SL}(2,\mathbb{Z})$ images of \mathcal{F} ,

$$\mathcal{T}_{\mathbb{H}} = \bigcup_{\alpha \in \mathrm{SL}(2,\mathbb{Z})} \alpha(\partial \mathcal{F}) = \left\{ \tau \in \mathbb{H} \, | \, j(\tau) \le 12^3 \right\},\tag{2.36}$$

has intersection points $\tau \in \mathrm{SL}(2,\mathbb{Z}) \cdot e^{\frac{\pi i}{3}}$, where $j(\tau) = 0$. From (2.10) we see that these intersection points correspond to $\mathcal{J}(u, \boldsymbol{m}, \Lambda) = 0$, whose only solutions are given by $g_2(u, \boldsymbol{m}, \Lambda) = 0$ (see (2.9)). Since g_2 is a polynomial in uof degree 2 for all curves (2.3), there are at most two intersection points in \mathcal{T}_m corresponding to $\mathcal{J} = 0$. As g_2 is strictly quadratic, there is also always at least one such point. We find below that when the branch points (as introduced in section 2.2.3) belongs to \mathcal{T}_m , they give further intersection points of \mathcal{T}_m .

One can study how the partitioning is deformed upon varying the masses. For the cases where the branch points belong to \mathcal{T}_m , the complex *u*-plane is generically partitioned into 6 regions. When going to the AD locus two or more of these regions shrink to a point together with at least one branch point. At precisely $m = m_{AD}$, the *u*-plane is then partitioned into ≤ 4 regions, giving an explanation for the discontinuous decrease in the index in the limit $m \rightarrow m_{AD}$. This can also be understood directly from the polynomials $T_m(x, y)$. For instance, at the point $m = m_{AD} = \frac{3}{4}\Lambda_1$ in $N_f = 1$, the polynomial $T_{m_{AD}}(x, y)$ contains a factor $9 - 24x + 16x^2 + 16y^2$. Its zero locus in \mathbb{R}^2 is just a point $x + iy = \frac{3}{4} = u_{AD}/\Lambda_1^2$, while the massive deformation away from m_{AD} describes a curve that encloses a region. For $u_{bp} \notin \mathcal{T}_m$ one needs to cut and glue interior points of different regions and the *u*-plane is therefore partitioned into less than 6 regions. See for example Fig. 11.

2.3 Matone's relation

In pure $\mathcal{N} = 2$ supersymmetric gauge theories, there is a relation between instanton corrections of the prepotential F to instanton corrections of $u \sim \langle \text{Tr}\phi^2 \rangle$ (2.1). For the pure SU(2) gauge theory, it has been found in [110] that

$$u = \pi i (2F - a\partial_a F). \tag{2.37}$$

In combination with the Picard-Fuchs differential equations for the periods a and a_D (see [96, (2.5)] or [110, (30)]), this gives an explicit recursion relation [100, (8)]

$$\partial_a^3 F = \frac{\pi^2}{4} \frac{(a\partial_a^2 F - \partial_a F)^3}{(1 + \pi^2 (2F - a\partial_a F)^2)}$$
(2.38)

for the prepotential F. Using (2.37), one may write this relation as

$$\frac{du}{d\tau} = -4\pi i \Delta \left(\frac{da}{du}\right)^2,\tag{2.39}$$

where $\Delta = u^2 - \Lambda_0^4$ is the physical discriminant. In order to see this, one can differentiate (2.37) w.r.t. *a* to find $\frac{du}{da} = \pi i(a_D - a\tau)$. In the following, we will call relations such as (2.39) *Matone's relations* rather than (2.37), (2.38) or equivalent equations.

One can furthermore differentiate (2.37) w.r.t. u to find $aa'_D - a'a_D = \frac{i}{\pi}$, where $' = \partial_u$. By computing $\frac{d\tau}{du} = \partial_u \partial_a^2 F$, this identity together with the Picard-Fuchs equations implies (2.39). Due to its central importance,

generalisations of (2.37) to many theories have been found. For SU(N) SQCD it reads

$$2F = \left(\Lambda \partial_{\Lambda} + \sum_{j} m_{j} \partial_{m_{j}} + \sum_{j} a_{j} \partial_{a_{j}}\right) F, \qquad (2.40)$$

which can be interpreted as an RG flow of the prepotential, $\Lambda \partial_{\Lambda} F \sim u$ [130– 132]. By combining $\pi = (\Lambda, \boldsymbol{m}, \boldsymbol{a})^{\mathsf{T}}$ into a vector, the prepotential satisfies the differential equation $\pi^{\mathsf{T}} \partial_{\pi} F = 2F$. This Euler equation characterises the homogeneity of the prepotential. Its relations to supergravity [131, 133], the Ω -background [134, 135], integrability [136–138] and 5 dimensions [139, 140] are well-known.

In $\mathcal{N} = 2$ SU(2) SYM, all ingredients of (2.39) are classically modular, while the prepotential F is not. This is easy to see from the above relations: It involves the period a_D , which unlike a is not modular since it depends explicitly non-modularly on τ . Picard-Fuchs solutions of massive SU(2) SQCD have been studied in [141–143], where generalisations of (2.39) for the massless $N_f =$ 1,2,3 theories have been obtained. By including massless hypermultiplets, (2.39) receives interesting corrections [88]: The physical discriminant Δ is divided by another polynomial in u. For massless $N_f = 2,3$ it divides Δ , while it does not for $N_f = 1$. In this Section, we derive a generalisation of (2.39) for $N_f = 1, 2, 3, 4$ with generic masses and give an explanation of these denominators.

Another motivation comes from the topologically twisted theory [61], where correlation functions of the topological theory on a four-manifold X can be computed as integrals over the u-plane [73]. The coupling of the low energy effective theory to topological invariants of the background gravitational field is encoded by a holomorphic function $A^{\chi}B^{\sigma}$, where χ and σ are the the Euler characteristic and the signature of X. For pure SU(2) SYM, the functions A and B are $A = \left(\Delta \frac{d\tau}{du}\right)^{\frac{1}{4}}$ and $B = \Delta^{\frac{1}{8}}$ [72].⁹ The function B^8 can in fact be viewed as a definition of the physical discriminant [144]. Due to the pure SU(2) Matone relation (2.39), it was realised in [73] that $A = \left(\frac{du}{da}\right)^{\frac{1}{2}}$. If we include hypermultiplets, (2.39) is modified and this argument does not work anymore. Instead, it was argued [73] that $A = \left(\frac{du}{da}\right)^{\frac{1}{2}}$ is in fact correct when including hypermultiplets. However, this bases on the assumption that A^{χ} is a holomorphic modular form of weight $-\frac{\chi}{2}$, and has neither zeros nor poles anywhere on the *u*-plane. In $\mathcal{N} = 2$ SQCD, these statements are not quite true. We will discuss these issues in section 5.

In this Section, we derive a generalisation of (2.39) for massive $N_f = 1, 2, 3$. Section 2.3.1 derives expressions for da/du and Δ_{N_f} as functions of τ . Section 2.3.2 derives Matone's relation (2.52) for generic $N_f \leq 3$.

⁹The prefactors were determined in [73] for pure SU(2) and in [111] for the asymptotically free SU(2) theories. We ignore them in the following.

2.3.1 Periods and Weierstraß form

We proceed by deriving an expression for da/du. To this end, recall that a is given as a period integral (2.2), and that the derivative of the SW differential λ to u is holomorphic [46]. Therefore, we can express da/du in terms of the variables x and y of (2.7)

$$\frac{da}{du} = \frac{\sqrt{2}}{4\pi} \int_{\gamma} \frac{dx}{y},\tag{2.41}$$

where γ is one of the cycles of the elliptic curve. To determine this quantity for the theories with $N_f \leq 3$, we map the curve \mathcal{W} to the modular Weierstraß form $\widetilde{\mathcal{W}}, A: \mathcal{W} \to \widetilde{\mathcal{W}}$. See for example [145, Section 7.1]. The curve $\widetilde{\mathcal{W}}$ reads

$$\widetilde{\mathcal{W}}: \quad \widetilde{y}^2 = 4\widetilde{x}^3 - \widetilde{g}_2\widetilde{x} - \widetilde{g}_3, \qquad (2.42)$$

with the variables related by the map A as

$$A: \begin{cases} \tilde{x} = \alpha^{2} x = \wp(z), \\ \tilde{y} = \alpha^{3} y = \wp'(z), \\ \tilde{g}_{2} = \alpha^{4} g_{2}, \\ \tilde{g}_{3} = \alpha^{6} g_{3}, \end{cases}$$
(2.43)

where \wp is the Weierstraß function and $z \in \mathbb{C}$ a coordinate on the curve. Since $\widetilde{\mathcal{W}}$ is the modular Weierstraß curve, the variables \tilde{g}_2 and \tilde{g}_3 equal

$$\tilde{g}_2 = \frac{4\pi^4}{3} E_4, \qquad \tilde{g}_3 = \frac{8\pi^6}{27} E_6,$$
(2.44)

with E_k the Eisenstein series defined in (1.7). We note that the variables for \mathcal{W} (2.7) have weight 0 under modular transformations, while in (2.42) the weights are wt $(\alpha, \tilde{y}, \tilde{x}, \tilde{g}_2, \tilde{g}_3) = (1, 3, 2, 4, 6)$. Using the two equations for \tilde{g}_2 and \tilde{g}_3 , we can solve for u and α . The relation

$$\alpha = \frac{\sqrt{2\pi}}{3} \sqrt{\frac{g_2}{g_3} \frac{E_6}{E_4}},$$
(2.45)

will be particularly useful for us in the next subsection. This relation can also be derived using Picard-Fuchs equations [146].

Now it is straightforward to determine da/du (2.41) using the Weierstraß representation of (\tilde{x}, \tilde{y}) ,

$$\frac{da}{du} = \frac{\sqrt{2}\,\alpha}{4\pi} \int_{\tilde{\gamma}} \frac{d\tilde{x}}{\tilde{y}} = \frac{\sqrt{2}\,\alpha}{4\pi},\tag{2.46}$$

where $\tilde{\gamma}$ is the image of the γ under the map A, with the variable z of $\tilde{x}(z)$ changing from 0 to 1.

We continue by studying the discriminants of \mathcal{W} and $\widetilde{\mathcal{W}}$. Using $E_4^3 - E_6^2 = 12^3 \eta^{24}$ with η as in (A.18), we find for the discriminant of $\widetilde{\mathcal{W}}$, $\tilde{\Delta} = (2\pi)^{12} \eta^{24}$. The discriminant of \mathcal{W} , Δ_{N_f} (2.8), on the other hand is a polynomial in u, m and Λ and therefore has weight 0. The two discriminants are related by

$$\tilde{\Delta} = \alpha^{12} (-1)^{N_f} \Lambda_{N_f}^{8-2N_f} \Delta_{N_f}, \qquad (2.47)$$

or substituting α in terms of da/du (2.46),

$$\eta^{24} = 2^6 (-1)^{N_f} \Lambda_{N_f}^{2(4-N_f)} \left(\frac{da}{du}\right)^{12} \Delta_{N_f}, \qquad (2.48)$$

which holds for $0 \le N_f \le 3$. Similar expression exist for $N_f = 4$ and $\mathcal{N} = 2^*$ [85].

Let us consider the case that \mathcal{W} or $\widetilde{\mathcal{W}}$ is singular. The curve $\widetilde{\mathcal{W}}$ is only singular at the cusps $\tau \in \{i\infty\} \cup \mathbb{Q}$, since $\tilde{\Delta} \sim \eta^{24}$ vanishes at the cusps and is non-vanishing for τ in the interior of \mathbb{H} . From (2.47) we see that, at the cusps of $\widetilde{\mathcal{W}}$ either da/du or Δ_{N_f} must vanish. On the other hand, for τ in the interior of \mathbb{H} , $\tilde{\Delta}$ is non-vanishing. This means that, if \mathcal{W} is singular ($\Delta_{N_f} = 0$) for such values of τ , da/du should diverge. This is exactly what happens at the AD points,

$$\frac{du}{da}(\tau_{\rm AD}) = 0, \qquad \Delta_{N_f}(u(\tau_{\rm AD})) = 0, \qquad \tau_{\rm AD} \in \mathbb{H}.$$
(2.49)

We can further note that $\frac{du}{da}(\tau) = 0$ is true also for singularities that are cusps and not elliptic points, i.e., $\Delta_{N_f} = 0$ for $\tau \in \mathbb{Q}$. This is because if u is not an elliptic point then $g_2 \neq 0$ and $g_3 \neq 0$, since otherwise, from $\Delta = g_2^3 - 27g_3^2$, both would be zero, giving an elliptic point. Then, from (2.46) we have that $\left(\frac{du}{da}\right)^2$ is proportional to $\frac{E_4}{E_6}$. This is a meromorphic modular form of weight -2for SL(2, \mathbb{Z}), and one can show using modular transformations that it vanishes on \mathbb{Q} . Therefore, we have that $\Delta_{N_f} = 0$ implies $\frac{du}{da} = 0$.

2.3.2 Matone's relation

We will now give a generalisation of (2.39) that holds also for the massive $N_f = 1, 2, 3$ theories. Let us denote by ' the derivative with respect to u keeping \boldsymbol{m} and Λ_{N_f} fixed. The derivative with respect to τ is always given explicitly. From the explicit expression for j as function of τ (A.9), it is easy to check that $\frac{d}{d\tau}j = -2\pi i \frac{E_6}{E_4}j$. Using the chain rule and (2.10), we can express this as $\frac{d}{d\tau}\mathcal{J} = \mathcal{J}' \frac{du}{d\tau}$. This gives the first important identity,

$$\frac{du}{d\tau} = -2\pi i \frac{E_6}{E_4} \frac{\mathcal{J}}{\mathcal{J}'},\tag{2.50}$$

which holds for any SW curve. From (2.9) we can compute \mathcal{J}' in terms of g'_2 and g'_3 . Using relations (2.45) and (2.46), we can substitute E_6/E_4 in terms of g_2 , g_3 and da/du. This gives the exact relation

$$\frac{du}{d\tau} = -72\pi i \frac{g_3}{g_2} \frac{\mathcal{J}}{\mathcal{J}'} \left(\frac{da}{du}\right)^2 = -\frac{8\pi i}{3} \frac{g_2^3 - 27g_3^2}{2g_2g_3' - 3g_2'g_3} \left(\frac{da}{du}\right)^2.$$
(2.51)

An analogous formula for five-dimensional gauge theories was derived from the Picard-Fuchs perspective in [147, Eq. (4.23)]. Both factors on the rhs are only relative invariants, but their product is an absolute invariant of the curve \mathcal{W} .

The numerator on the rhs is proportional to the physical discriminant. The equation has modular weight 2, since both $\frac{du}{d\tau}$ and $\left(\frac{da}{du}\right)^2$ are of weight 2.

For $0 \le N_f \le 3$, we can compute the corresponding g_i , and one can rewrite (2.51) as¹⁰

$$\frac{du}{d\tau} = -\frac{16\pi i}{4 - N_f} \frac{\Delta_{N_f}}{P_{N_f}^{\rm M}} \left(\frac{da}{du}\right)^2,\tag{2.52}$$

where we substituted (2.8) for Δ_{N_f} , and defined the polynomial $P_{N_f}^{\rm M}$,

$$P_{N_f}^{\rm M} = \frac{6}{4 - N_f} \, (-1)^{N_f} \, \Lambda_{N_f}^{2N_f - 8} \, (2g_2g_3' - 3g_2'g_3). \tag{2.53}$$

The normalisation is chosen such that $P_{N_f}^{\mathrm{M}}$ is a monic polynomial. Explicit computation gives,

$$P_0^{M} = 1,$$

$$P_1^{M} = u - \frac{4}{3}m_1^2,$$

$$P_2^{M} = u^2 - \frac{3}{2}(m_1^2 + m_2^2)u + 2m_1^2m_2^2 + \frac{1}{8}m_1m_2\Lambda_2^2 - \frac{1}{64}\Lambda_2^4,$$

$$P_3^{M} = u^3 - 2M_2u^2 + (3M_4' + \frac{3}{4}M_3\Lambda_3 - \frac{1}{64}M_2\Lambda_3^2)u + \frac{1}{256}M_3\Lambda_3^3 - \frac{1}{4}M_2M_3\Lambda_3 + \frac{1}{32}(M_4 - M_4')\Lambda_3^2 - 4M_3^2,$$
(2.54)

where we defined

$$M_{2} = m_{1}^{2} + m_{2}^{2} + m_{3}^{2}, \qquad M_{3} = m_{1}m_{2}m_{3},$$

$$M_{4} = m_{1}^{4} + m_{2}^{4} + m_{3}^{4}, \qquad M_{4}' = \sum_{i \leq i} m_{i}^{2}m_{j}^{2}.$$
(2.55)

We note that these polynomials appear in the Picard-Fuchs equations for the periods of these theories and their zeros give regular singular points of the differential equations [141, 143].¹¹

2.3.3 Branch points

An important difference between $N_f = 0$ and $N_f > 0$ are the poles where $P_{N_f}^{\mathrm{M}}$ vanishes. To understand these poles as well as zeros of $du/d\tau$, note that at such points the change of variables between u and τ is ill-defined. We have seen earlier that the change of variables is ill-defined at the points where the discriminant $D(P_{N_f})$ (2.17) vanishes. Indeed if we substitute $\mathcal{J}(u, \boldsymbol{m}, \Lambda_{N_f})$ for $j(\tau)$ in $D_{N_f}^{\mathrm{bp}}$, $P_{N_f}^{\mathrm{M}}$ factors out.

The reason for this is the following. The discriminant of a polynomial p vanishes if and only p has a double root. It can be computed as the *resultant* of

¹⁰We can in fact perform the same computations in the case of $N_f = 4$, leading to a similar formula.

¹¹The identity (2.52) does in fact not depend on the specific form of the SW curves. Given a Jacobian rational elliptic surface, let $\omega = \int_{\gamma} \frac{dx}{y}$ be the period of the Néron differential on the elliptic curve. Then $\frac{du}{d\tau} = \frac{1}{3\pi i}\omega^2 \Delta/(2g_2g'_3 - 3g'_2g_3)$, with u a coordinate on $\mathbb{P}^1(\mathbb{C})$.

the polynomial and its formal derivative, $D(p) \sim \operatorname{Res}_X(p, p')$ (see also [106]).¹² The zero locus $D(P_{N_f}) = 0$ of $P_{N_f}(X)$ is then given by the solutions to the two equations $P_{N_f}(X) = 0$ and $P_{N_f}'(X) = 0$. Since $\Delta_{N_f} \neq 0$, all solutions can be found by solving the former for j and inserting into the latter. It is straightforward to show that this gives

$$\frac{g_2^2 g_3}{\Delta_{N_f}} P_{N_f}^{\rm M} = 0, \qquad (2.56)$$

which provides the decomposition (2.17): If $g_2 = 0$ but $g_3 \neq 0$, then j = 0. If $g_3 = 0$ but $g_2 \neq 0$, then $j = 12^3$. If both $g_2 = g_3 = 0$, we are in $\mathcal{L}_{N_f}^{\mathrm{AD}} \subseteq \mathcal{L}_{N_f}^{\Delta}$. Now since the sextic equation is only well-defined away from the physical discriminant locus $\mathcal{L}_{N_f}^{\Delta}$ where $\Delta_{N_f} = 0$, the true branch point locus $\mathcal{L}^{\mathrm{bp}}$ is the difference of the Matone locus $\mathcal{L}_{N_f}^{\mathrm{M}} = \{u \in \mathcal{B}_{N_f} | P_{N_f}^{\mathrm{M}} = 0\}$ and the discriminant locus,

$$\mathcal{L}_{N_f}^{\rm bp} = \mathcal{L}_{N_f}^{\rm M} \setminus \mathcal{L}_{N_f}^{\Delta}.$$
 (2.57)

On the Coulomb branch with N_f hypermultiplets there are generically $2 + N_f$ distinct singular points. For special mass configurations \boldsymbol{m} , some singularities can collide. Then Δ_{N_f} has a double root. From above it is clear that this is equivalent to $D(\Delta_{N_f}) = 0$, which in turn is equivalent to $\Delta_{N_f} = 0$ and $\Delta'_{N_f} = 0$. We can again solve the former for g_2 and g_3 and insert into the latter to obtain $P_{N_f}^M \sim \frac{g'_2}{g_3} \Delta_{N_f} = 0$. This implies that whenever Δ_{N_f} has a double root, it is also a root of $P_{N_f}^M$. It is also observed in all examples below. To be more precise, if Δ_{N_f} contains a root of d > 1-th order, then Δ'_{N_f} has the same root but with multiplicity d-1. The excess factors can be extracted by the operation $gcd(\Delta_{N_f}, \Delta'_{N_f})$, where gcd is the polynomial greatest common divisor. The multiple roots are removed from the discriminant by the square-free factorisation¹³

$$\widehat{\Delta}_{N_f} = \frac{\Delta_{N_f}}{\gcd(\Delta_{N_f}, \Delta'_{N_f})}.$$
(2.58)

This reduced discriminant $\widehat{\Delta}_{N_f}$ has single roots only, concretely we map $\prod_s (u-u_s)^{n_s}$ to $\prod_s (u-u_s)$. This quantity is also of importance for determining gravitational couplings to Seiberg-Witten theory [148]. One can show that $gcd(\Delta_{N_f}, \Delta'_{N_f})$ always divides $P^{\rm M}_{N_f}$, such that

$$\widehat{P}_{N_f}^{\mathrm{M}} \coloneqq \frac{\widehat{\Delta}_{N_f}}{\Delta_{N_f}} P_{N_f}^{\mathrm{M}}$$
(2.59)

¹²The resultant of two polynomials over a commutative ring is a polynomial of their coefficients which vanishes if and only if the polynomials have a common root. It can be computed as the determinant of their Sylvester matrix.

¹³The polynomial gcd is unique only up to multiplication with invertible constants, we choose it such that $\widehat{\Delta}_{N_f}$ is again monic.

is in fact a polynomial. The branch point equation (2.56) is then equivalent to $\widehat{P}_{N_f}^{\mathrm{M}}/\widehat{\Delta}_{N_f} = 0$, which reduces to

$$\widehat{P}_{N_f}^{\rm M} = 0. \tag{2.60}$$

The Matone relation thus always takes the form

$$\frac{du}{d\tau} = -\frac{16\pi i}{4 - N_f} \frac{\widehat{\Delta}_{N_f}}{\widehat{P}_{N_f}^{\mathrm{M}}} \left(\frac{da}{du}\right)^2, \qquad (2.61)$$

where both $\widehat{\Delta}_{N_f}$ and $\widehat{P}_{N_f}^{\mathrm{M}}$ are polynomials. In the subsequent sections we show explicitly that the roots of the denominator (2.60) are precisely the branch points. We note that for generic masses the form (2.61) does not differ from (2.52), because $\widehat{}$ is trivial when all roots are distinct.

As argued above, AD points correspond to points τ_{AD} in the upper halfplane. Since they lie on the discriminant locus, we exclude them to define the sextic polynomial P_{N_f} . We will discuss in more detail below that, if the masses approach the AD locus, a branch point in the *u*-plane collides with two mutually non-local singularities forming the AD point. The branch point under consideration lifts, while the $N_f - 1$ other branch points remain for a generic point on the AD mass locus $\mathcal{L}_{N_f}^{AD}$. Thus for a generic point on the AD mass locus, AD points are not branch points of $u(\tau)$. A non-generic example is the most symmetric AD theory, the *IV* fibre in $N_f = 3$, discussed in more detail in Section 2.6.4. For this theory, τ_{AD} corresponds to a singular point of the theory as well as a branch point. As a result, the domain for τ does not correspond to that of a congruence subgroup of SL(2, Z).

Since any branch point $\tau_{\rm bp}$ induces a non-trivial monodromy, u does not have a regular Taylor series at such a point. For instance, if the u-plane contains one branch point $u_{\rm bp} = u(\tau_{\rm bp})$, then we have $u(\tau)-u_{\rm bp} = \mathcal{O}(\sqrt{\tau - \tau_{\rm bp}})$ as $\tau \to \tau_{\rm bp}$. If the leading coefficient is nonzero, then $\frac{du}{d\tau}$ diverges at $\tau_{\rm bp}$. Away from the discriminant locus, this can be understood from (2.61): From (2.46) we see that $\frac{da}{du}$ is regular and nonzero at a branch point, since none of g_2 , g_3 , E_4 and E_6 diverge or vanish. Thus the zeros of the denominator $\widehat{P}_{N_f}^{\rm M}$ correspond to the singular points of $\frac{du}{d\tau}$, as observed.

This can also be seen directly from the \mathcal{J} -invariant of the SW curve. It is easy to show that

$$\mathcal{J}' = 36^3 \frac{g_2^2 g_3}{\Delta_{N_f}^2} P_{N_f}^{\mathrm{M}}, \qquad (2.62)$$

which due to (2.56) vanishes at any branch point $u_{\rm bp}$. Since for fixed mass and scale $\mathcal{J}(u)$ is rational in u, it is a meromorphic function on \mathcal{B}_{N_f} . Away from the discriminant locus it thus has a Taylor series around $u_{\rm bp}$, where the linear coefficient is missing. We therefore find

$$\mathcal{J}(u) - \mathcal{J}(u_{\rm bp}) = \mathcal{O}\left((u - u_{\rm bp})^{n_{\rm bp}}\right), \qquad (2.63)$$

with $n_{\rm bp} \geq 2$. Now we identify $\mathcal{J}(u) = j(\tau)$, which relates the power series of u and τ . For a generic $\tau \in \mathbb{H}$, j has a regular Taylor series at τ with non-zero linear coefficient. However if τ is in the SL(2, Z)-orbit of i or $e^{\frac{\pi i}{3}}$, j has a zero of order 2 or 3. Let $n_{\tau_{\rm bp}} \in \{1, 2, 3\}$ be this number for a given branch point $\tau_{\rm bp} \in \mathbb{H}$. Then $\mathcal{J}(u) - \mathcal{J}(u_{\rm bp}) = \mathcal{O}((\tau - \tau_{\rm bp})^{n_{\tau_{\rm bp}}})$, such that from (2.63) we conclude

$$u(\tau) - u_{\rm bp} = \mathcal{O}\left((\tau - \tau_{\rm bp})^{n_{\tau_{\rm bp}}/n_{\rm bp}}\right), \qquad (2.64)$$

where the leading coefficient is strictly non-zero. From this we see that the branch point $\tau_{\rm bp}$ does not necessarily correspond to an $n_{\rm bp}$ -th root, but since the ratio can cancel $\tau_{\rm bp}$ rather corresponds to a branch point of order

$$\frac{n_{\rm bp}}{\gcd(n_{\rm bp}, n_{\tau_{\rm bp}})}.$$
(2.65)

It is difficult to compute this integer for a generic branch point, however in all examples discussed below it is equal to 2, which corresponds to a *square* root.

If the number $\frac{n_{\tau_{\rm bp}}}{n_{\rm bp}} \in \mathbb{Q} \setminus \mathbb{Z}$ is larger than 1, then it is clear that $\frac{du}{d\tau}(\tau_{\rm bp}) = 0$. Conversely, if $\frac{n_{\tau_{\rm bp}}}{n_{\rm bp}} < 1$ then $\frac{du}{d\tau}(\tau_{\rm bp}) = \infty$. We thus see that any branch point has the property that $\frac{du}{d\tau}$ diverges or vanishes, such that the change of variables from the *u*-plane to the τ -plane is not well-defined.

The branch point locus also allows to find the effective coupling at the AD points. In the limit where the masses approach the AD locus, $\mathbf{m} \to \mathbf{m}_{AD}$, the AD point u_{AD} is the point where branch points u_{bp} in the *u*-plane merges with mutually non-local singularities. While away from \mathbf{m}_{AD} the effective coupling τ of the singularities remain as distinct cusps on the real line, the branch points move along certain paths inside \mathbb{H} . In an AD limit $\mathbf{m} \to \mathbf{m}_{AD}$, a number of pairs of branch points, τ_{bp} and τ'_{bp} , coincide at the intersection of copies of \mathcal{F} , and the branch cut will then disconnect regions from \mathcal{F}_{N_f} . The effective coupling of the AD point τ_{AD} is therefore given by that of the merged branch points. This is an efficient way to determine τ_{AD} , which otherwise can only be found by inverting modular functions. Moreover, if the duality group is a congruence subgroup of $SL(2,\mathbb{Z})$, τ_{AD} corresponds to an elliptic point of the duality group.

2.4 The $N_f = 1$ curve

To make the above discussions more concrete we will now go on to study some specific examples. We will start by including one hypermultiplet. The $N_f = 1$ theory has been discussed in some detail in [103, 149–152].

In the massive $N_f = 1$ theory, there are three (in general) distinct strong coupling singularities where a hypermultiplet becomes massless. These remain at distinct points in the massless limit, while for special values of the mass two of them can merge into AD points. To analyse the $N_f = 1$ theory we will start by restricting to the massless case and then go to an AD mass. Here we can find closed expressions for u in terms of well-known modular forms. Only in the AD case does the theory turn out to be modular. In the end we can use the knowledge gained from these cases to draw some conclusions of the general massive case.

2.4.1 The massless theory

Let us begin with the massless $N_f = 1$ theory. Using the procedure outlined in Sec. 2.2.2 we find [101]

$$\frac{u(\tau)}{\Lambda_1^2} = -\frac{3}{2^{\frac{7}{3}}} \frac{\sqrt{E_4(\tau)}}{\sqrt[3]{E_4(\tau)^{\frac{3}{2}} - E_6(\tau)}} = -\frac{1}{16} (q^{-1/3} + 104q^{2/3} - 7396q^{5/3} + \mathcal{O}(q^{8/3})),$$
(2.66)

where we again have made the choice of solution consistent with our convention, such that $u \to -\infty$ for $\tau \to i\infty$. This function also appears as an order parameter in pure SU(3) SW theory (see section 4) as well as in the description of certain elliptically fibred Calabi-Yau spaces [153]. The singularities of the curve are $\frac{u^3}{\Lambda_1^6} = -\frac{3^3}{2^8}$. They are associated with states of charges (1,0), (1,1) and (1,2) becoming massless. The global \mathbb{Z}_3 symmetry acts as $T^{-1}: u(\tau - 1) = \omega_3 u(\tau)$, with $\omega_j = e^{\frac{2\pi i}{j}}$.

By restricting to the imaginary axis, we can perform the S-transformation. For this, let $\tau = i\beta$ with $\beta > 0$. We have that $E_4(i/\beta) = (i\beta)^4 E_4(i\beta) = \beta^4 E_4(i\beta)$. Taking the square root is unambiguous since E_4 is real on the imaginary axis and β^4 is positive. This gives $E_4^{\frac{1}{2}}(i/\beta) = \beta^2 E_4^{\frac{1}{2}}(i\beta)$. On the other hand for E_6 we have $E_6(\tau) = (i\beta)^6 E_6(i\beta) = -\beta^6 E_6(i\beta)$. This implies that the relative sign of E_6 flips, and it holds for $\tau_D \in i\mathbb{R}_{>0}$ that

$$\frac{u_D(\tau_D)}{\Lambda_1^2} = -\frac{3}{2^{\frac{7}{3}}} \frac{E_4(\tau_D)^{\frac{1}{2}}}{\left(E_4(\tau_D)^{\frac{3}{2}} + E_6(\tau_D)\right)^{\frac{1}{3}}} = \frac{3}{2^{\frac{8}{3}}} \left(1 + 144q - 3456q^2 + 596160q^3 + \mathcal{O}(q^4)\right).$$
(2.67)

With the \mathbb{Z}_3 symmetry $u(\tau - 1) = \omega_3 u(\tau)$ this confirms the strong coupling singularities given above.

The monodromies on the massless $N_f = 1$ *u*-plane are [46]

$$M_{1} = \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix} = STS^{-1},$$

$$M_{2} = \begin{pmatrix} 0 & 1 \\ -1 & 2 \end{pmatrix} = (TS)T(TS)^{-1},$$

$$M_{3} = \begin{pmatrix} -1 & 4 \\ -1 & 3 \end{pmatrix} = (T^{2}S)T(T^{2}S)^{-1},$$

$$M_{\infty} = \begin{pmatrix} -1 & 3 \\ 0 & -1 \end{pmatrix} = PT^{-3},$$

(2.68)

where $P = S^2 = -1$. Note that these matrices generate the full $SL(2,\mathbb{Z})$ modular group rather than a (congruence) subgroup. Indeed, as fractional linear transformations acting on the complex structure through their matrix

representations, they do not leave u invariant. However, we can consider these matrices as compositions of paths in the fundamental domain, and as such they do leave u invariant. To make the connection to the discussion in [1] more direct we can note that by using another choice of homology basis in the present case we can construct a different set of monodromies, see for example [117], which exactly coincides with the ones listed for the SU(3) case of [1].

Since E_4 has a simple zero at $\tau = \omega_3$ (and $\operatorname{SL}(2,\mathbb{Z})$ -images), $u(\tau)$ has a branch point at $\tau_{\mathrm{bp}} = \omega_3$. The function $u(\tau)$ does not possess a Taylor series at τ_{bp} and is therefore not holomorphic at τ_{bp} . Since $u(\tau_{\mathrm{bp}}) = 0$, the branch point of $u(\tau)$ indeed agrees with what is found in (2.24). Since u is not holomorphic on \mathbb{H} , it can also not be classically modular. Another reason why u is not modular is the following. If we define $x := -16\frac{u}{\Lambda_1^2} = q^{-\frac{1}{3}} + \mathcal{O}(q^{\frac{2}{3}})$, then one can read off from the curve that $\mathcal{J} = x^6/(x^3 - 432)$. This implies that u should be a Hauptmodul of an index 6 subgroup of $\operatorname{SL}(2,\mathbb{Z})$ with width $h(\infty) = 3$ and width decomposition 6 = 3 + 1 + 1 + 1 (see (A.37)). From the classification of index 6 groups in Table 4 we see that such a subgroup of $\operatorname{SL}(2,\mathbb{Z})$ does not exist. In fact, no index 6 subgroup of $\operatorname{SL}(2,\mathbb{Z})$ with 4 cusps exists. This distinguishes massless $N_f = 1$ from massless $N_f = 0, 2, 3$, where the duality groups are congruence subgroups isomorphic to $\Gamma^0(4)$ [101].

From (2.66) one finds

$$\frac{du}{d\tau} = \frac{\pi i \Lambda_1^2}{2^{\frac{7}{3}}} \frac{E_4^{\frac{3}{2}} + E_6}{E_4^{\frac{1}{2}} \left(E_4^{\frac{3}{2}} - E_6 \right)^{\frac{1}{3}}}, \qquad \frac{da}{du} = \frac{i \left(E_4^{\frac{3}{2}} - E_6 \right)^{\frac{5}{6}}}{2^{\frac{1}{3}} \sqrt{3} \Lambda_1}.$$
 (2.69)

We can explicitly check that these satisfy Matone's relation, (2.52), for massless $N_f = 1$,

$$\frac{du}{d\tau} = -\frac{16\pi i}{3} \frac{\Delta}{u} \left(\frac{da}{du}\right)^2. \tag{2.70}$$

1

The fundamental region

$$\mathcal{F}_1(0) = \bigcup_{\ell=0}^2 T^\ell \mathcal{F} \cup T^\ell S \mathcal{F}$$
(2.71)

as in (2.14) was obtained in [1]. It is shown in Fig. 6, together with its image under u to the u-plane. We stress that (2.71) can not be written as $G \setminus \mathbb{H}$ for any subgroup $G \subseteq SL(2,\mathbb{Z})$.

In the massless $N_f = 1$ theory, the partitioning (2.32) is contained in the algebraic plane curve $T_{(0)}(x, y) = 0$, where $\frac{u}{\Lambda_1^2} = x + iy$ and

$$T_{(0)}(x,y) = y(3x^2 - y^2)(27x^3 + 128x^6 - 81xy^2 + 384x^4y^2 + 384x^2y^4 + 128y^6).$$
(2.72)

The first two factors of $T_{(0)}(x, y)$ contains also values which correspond to $j > 12^3$ and they need to be sufficiently truncated. The identification of the algebraic curve with the partitioning of \mathbb{H} is immediate from Fig. 6.



Figure 6: Left: Proposal of a fundamental region for massless $N_f = 1$. It is clear that it can not be a fundamental domain of any $SL(2, \mathbb{Z})$ subgroup if we identify the sides, $\tau \sim \tau + 3$. This is because the lower boundary is not given by a union of half-disks. If we do *not* identify the sides then the picture is in fact a fundamental domain for $\Gamma^0(4)$. This is because there is an element in $\Gamma^0(4)$ which maps $i\infty \mapsto 1$. Right: Plot of the massless $N_f = 1$ *u*-plane as the union of the images of *u* under the six $SL(2, \mathbb{Z})$ images of \mathcal{F} as in (2.71), in units of $\Lambda_1^2 = 1$. In particular, the function *u* is surjective onto \mathbb{C} on this domain. The singular points sit neatly in the interior of the strong coupling regions $u(S^{\ell}\mathcal{F})$, $\ell = 0, 1, 2$. The origin u = 0 is $\tau = \omega_3 + \mathbb{Z}$ as discussed above, and bounds all six domains as is clear from the left figure.

2.4.2 Type *II* AD mass

In section 2.2.3 we saw that the $N_f = 1$ theory has AD fixed points in its moduli space. To study these, we fix the mass to be one of the AD values, $m \to m_{\rm AD} = \frac{3}{4}\Lambda_1$, specified by the zero locus of the AD polynomials (2.20). Two mutually non-local singularities now collide at the AD point $u = u_{\rm AD} = \frac{3}{4}\Lambda_1^2$ while the third one simplifies to $u_0 = -\frac{15}{16}\Lambda_1^2$, such that the discriminant reads

$$\Delta = (u - u_{\rm AD})^2 (u - u_0). \tag{2.73}$$

From the curve we now find

$$\frac{u(\tau)}{\Lambda_1^2} = -\frac{f_{3B}(\tau) + 15}{16},$$
(2.74)

where f_{3B} is the McKay-Thompson series of class 3B for the Monster group [36] (see Appendix A.4 for an overview of McKay-Thompson series),

$$f_{3B}(\tau) = \left(\frac{\eta(\frac{\tau}{3})}{\eta(\tau)}\right)^{12} = 27 \left(\frac{b_{3,0}(\frac{\tau}{3})}{b_{3,1}(\frac{\tau}{3})}\right)^3 - 27$$

= $q^{-\frac{1}{3}} - 12 + 54 q^{\frac{1}{3}} - 76 q^{\frac{2}{3}} - 243 q + 1188 q^{\frac{4}{3}} - 1384 q^{\frac{5}{3}} + \mathcal{O}(q^2).$
(2.75)

The Dedekind η -function is defined in (A.18), while the functions $b_{3,j}$ are defined in (A.14). We are again careful to choose the solution for u with the consistent asymptotics. Substitution of the q series (2.75) in (2.74) reproduces the q-series based on [103, Eq. (4.93)].

Using Theorem 1 in Appendix A.1, we find that $\tau \mapsto f_{3B}(3\tau)$ is a classical modular function for $\Gamma_0(3)$ and therefore u is a modular function for $\Gamma^0(3)$. A



Figure 7: Left: Fundamental domain of $\Gamma^0(3)$, the duality group of $N_f = 1$ with $m = m_{AD}$. The AD point $\tau_{AD} = \sqrt{3}\omega_{12}$ is highlighted. Right: Plot of the $N_f = 1$ *u*-plane with AD mass as the union of the images of *u* under the 6 - 2 = 4 SL $(2, \mathbb{Z})$ images of \mathcal{F} forming $\Gamma^0(3) \setminus \mathbb{H}$, as in the left figure. The complex plane can clearly be covered by 4 triangles. There is only one strong coupling region, which is the circular region. It contains u_0 in its interior. The AD point (orange) lies on the boundary of $T^{\pm 1}\mathcal{F}$, as is clear from the left figure. The areas with the same colours are mapped to each other in the two figures.

fundamental domain of $\Gamma^0(3)$ is

$$\mathcal{F}_1(m_{\rm AD}) = \bigcup_{\ell=0}^2 T^\ell \mathcal{F} \cup S \mathcal{F}.$$
 (2.76)

This is shown in Fig. 7 together with the map to the *u*-plane. The cusps are $i\infty$ and 0, with widths 3 and 1, respectively. We take from [154, Table 4.1] that $\Gamma^0(3)$ has an elliptic fixed point of order 3.

Using the transformation properties of the η -function (A.19) it is straightforward to show that the locations of the singularities of (2.73) in the τ -plane are given by ($\omega_j = e^{2\pi i/j}$)

$$u(i\infty) = \infty, \qquad u(0) = u_0, \qquad u(\sqrt{3}\omega_{12}) = u_{AD}, \qquad (2.77)$$

where the proper limits are understood. The AD point $\sqrt{3}\omega_{12}$ is stabilised by the order 3 element $\begin{pmatrix} -1 & 3 \\ -1 & 2 \end{pmatrix} \in \Gamma^0(3)$, and it is therefore the order 3 elliptic fixed point of $\Gamma^0(3)$. Comparing the locations to the massless case we see that the regular singularity u_0 has stayed on $\tau = 0$, while, contrary to the massless case, the cusps with the two mutually non-local singularities are disconnected (or cut) from the domain for massless $N_f = 1$, and leaves as remnant the point τ_{AD} into the interior of \mathbb{H} . This procedure also reduces the index of the solution: Indeed, from (A.32) we compute that ind $\Gamma^0(3) = 4$, where the 6 - 4 = 2 AD points do not contribute since they are not cusps. This can also be seen from the fact that

$$j = \frac{(f_{3B} + 3)^3 (f_{3B} + 27)}{f_{3B}}.$$
 (2.78)

Indeed, since $\mathcal{J} = 1728 \frac{g_2^3}{\Delta}$, a common factor $(f_{3B} + 27)^2$ of g_2^3 and Δ has cancelled. The last factor $f_{3B} + 3 = 0$ in (2.78) implies $j(\tau) = 0$ and therefore

 $\tau = \omega_3 \mod \mathrm{SL}(2,\mathbb{Z})$. In fact, it corresponds to $u(\omega_3) = u(\omega_3 + 1) = -\frac{3}{4}\Lambda_1^2$ and it is just a regular point in the *u*-plane. We can also read off from this that $\operatorname{ord}(g_2, g_3, \Delta) = (1, 1, 2)$ and therefore the AD theory in $N_f = 1$ is according to Table 5 a type II singular fibre [155].

We can also study more characteristic functions of the theory with the AD mass. Using Appendix A.1, we can differentiate (2.74) to find

$$\frac{du}{d\tau}(\tau) = \frac{\pi i \Lambda_1^2}{24} f_{3B}(\tau) b_{3,0}(\frac{\tau}{3})^2.$$
(2.79)

This implies that $\frac{du}{d\tau}$ is a modular form of weight 2 for $\Gamma^0(3)$, without phases. One can also show that

$$\frac{da}{du}(\tau) = \frac{\sqrt{2}i}{\sqrt{27}\Lambda_1} \sqrt{\frac{b_{3,1}(\frac{\tau}{3})^3}{b_{3,0}(\frac{\tau}{3})}}.$$
(2.80)

An expression for da/du in terms of $_2F_1$ was given in [156, Eq. (4.13)].

The q-expansion of $\frac{da}{du}$ has growing denominators, and therefore $\frac{da}{du}$ is not a modular form of weight 1 for $\Gamma^0(3)$. However, it is straightforward to check that $\left(\frac{da}{du}\right)^2$ is a modular form of weight 2 for $\Gamma^0(3)$. We thus find the Matone relation

$$\frac{du}{d\tau} = -\frac{16\pi i}{3}\widehat{\Delta}\left(\frac{da}{du}\right)^2,\tag{2.81}$$

where $\widehat{\Delta}$ denotes the reduced discriminant. This is consistent with (2.52).

The monodromies can be found from the ones of the massless theory (2.68),

$$M_1 = STS^{-1} = \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix}, \quad M_{\rm AD} = M_2M_3 = T^2(ST)^{-1}T^{-2} = \begin{pmatrix} -1 & 3 \\ -1 & 2 \end{pmatrix}.$$
(2.82)

They generate the duality group $\Gamma^0(3)$ and give the large u monodromy $M_1M_{\rm AD} = PT^{-3}$. Furthermore, M_1 stabilises $\tau = 0$ and $M_{\rm AD}$ stabilises the AD point $\tau_{\rm AD} = \sqrt{3}\omega_{12}$. We have that $M_{\rm AD}^6 = \mathbb{1}$ and therefore $\tau_{\rm AD}$ is indeed an elliptic fixed point. The AD monodromy is conjugate to $(ST)^{-1}$, which fixes $\tau = \omega_3$. Since $\tau_{\rm AD} = \omega_3 + 2$, this gives a path in τ -space.

2.4.3 Generic real mass

By turning on a generic real mass, the singularities do not split compared to the massless case since there are already $N_f + 2 = 3$ singularities. Therefore, the fundamental domain of the massive theory should look similar to the massless one of Fig. 6, but we now need to consider the presence of branch points and cuts in more detail. We will discuss this and the limits to the pure theory as well as the the theory with the AD mass now.

For generic mass we have not been able to find a closed expression for u as a function of τ . By expanding $\mathcal{J}(u, m, \Lambda_1)$ and inverting the series we can,

however, get an expansion of u for the general massive theory near any cusp. For example, the expansion near $\tau = i\infty$ reads $(\mu = \frac{m}{\Lambda_1})$

$$\frac{u(\tau)}{\Lambda_1^2} = -\frac{1}{16}q^{-\frac{1}{3}} - \frac{1}{3}\mu^2 + \left(\frac{32}{9}\mu^4 - 6\mu\right)q^{\frac{1}{3}} - \left(\frac{5120}{81}\mu^6 - \frac{160}{3}\mu^3 + \frac{13}{2}\right)q^{\frac{2}{3}} + \mathcal{O}(q),$$
(2.83)

where we are careful to choose the expansion such that $u \to -\infty$ for consistency with our conventions. It is easy to see that this reproduces the earlier expansions, (2.66) and (2.74), in the respective limits.

The branch point locus is given by the zero locus of (2.24). By calculating $\mathcal{J}(u, m, \Lambda_1)$ from the curve and plugging it into the polynomial D_1^{bp} we find that the zero of the linear polynomial is $u = u_{\text{bp}} = \frac{4}{3}m^2$, and we recognise that this is the polynomial appearing in the denominator of the generalised Matone relation, (2.54), such that $\frac{du}{d\tau}$ diverges here. In the massive $N_f = 2, 3$ theories, where the theories can be studied in detail, we argue that it corresponds to two branch points in the closure of the fundamental domain, which are connected by a branch cut. Motivated by these analyses, we can draw the two branch point loci for positive mass. It is given in Fig. 8. For m = 0, the branch point is located at the origin u = 0. At the AD point, they collide, the branch cut vanishes and the order parameter becomes holomorphic, and even modular. In the $m \to \infty$ limit, the branch points also move to infinity.



Figure 8: Conjectured paths of the branch points in the fundamental domain of the massive $N_f = 1$ theory.

We can also confirm this from the analysis in section 2.3.3. By expanding $\mathcal{J}(u) - \mathcal{J}(u_{\rm bp})$ around $u_{\rm bp}$ for generic mass m, the linear coefficient is zero. The $(u - u_{\rm bp})^2$ coefficient vanishes if and only if either m = 0 or $m = m_i := \frac{3}{4\sqrt[3]{2}}$. For m = 0 we have $\tau_{\rm bp} \sim \omega_3$, such that $n_{\omega_3} = 3$ in the notation of section 2.3.3. Furthermore, $n_{\rm bp} = 6$, such that the order of the branch point (2.65) is the denominator of the reduced fraction $\frac{3}{6}$, namely 2. Since E_4 has a simple zero at $\tau_{\rm bp}$, this agrees with (2.66) having a square root.

From Fig. 8 we see that the branch point loci pass through $\tau_{\rm bp} = 1 + i$ where $m = m_i$, such that $n_i = 2$. Furthermore we find $n_{\rm bp} = 4$, and thus the order of the branch point is 2. Since $\frac{2}{4} = \frac{1}{2}$, it is indeed again the branch point of a square root.

For any other mass $m \in \mathbb{R}_{\geq 0} \setminus \{0, m_i, m_{AD}\}$ we have $n_{bp} = 2$ while $n_{\tau_{bp}} = 1$, such that the branch point is again of order 2. This demonstrates that the loci in Fig. 8 are complete: there is a single branch point on the Coulomb branch \mathcal{B}_1 , and for any mass there are two branch points of a square root in \mathbb{H} , which are connected by a single branch cut. It also implies that if an expression such as (2.66) existed for generic mass, while it could contain higher roots of modular forms, they can never have zeros in \mathbb{H} (as is the case also for m = 0).

We can study the partition of the *u*-plane provided by (2.32) in detail. For $m \neq m_{\rm AD}$ the *u*-plane is partitioned into six regions, whose union of boundary pieces is included in the algebraic curve given by the zero locus (2.34) of T_1 , where $T_1 = y\tilde{T}_1$ and

$$\begin{split} \tilde{T}_{1} &= 972\mu^{4} + 8192\mu^{2}x^{7} - 21504\mu^{3}x^{5} - 12096\mu^{2}x^{4} + 18432\mu^{4}x^{3} + 1296\mu^{3}x^{2} \\ &+ 16128\mu x^{6} + 1944\mu x^{3} + 8192\mu^{2}x^{5}y^{2} + 22528\mu^{3}x^{3}y^{2} - 8192\mu^{2}x^{3}y^{4} - 3456\mu^{2}x^{2}y^{2} \\ &+ 2304\mu x^{4}y^{2} - 6912\mu x^{2}y^{4} - 16384x^{6}y^{2} - 12288x^{4}y^{4} + 4320x^{3}y^{2} - 6144x^{8} \\ &- 1296x^{5} - 5184\mu^{5}x - 729\mu^{2}x - 18432\mu^{4}xy^{2} - 21504\mu^{3}xy^{4} - 8192\mu^{2}xy^{6} \\ &- 1944\mu xy^{2} - 1296xy^{4} + 4752\mu^{3}y^{2} + 8640\mu^{2}y^{4} + 6912\mu y^{6} + 2048y^{8}. \end{split}$$

$$(2.84)$$

Since the AD point $m = m_{AD}$ corresponds to a phase transition, we have to study the two cases $m < m_{AD}$ and $m > m_{AD}$ separately.

The case $m < m_{AD}$

From Fig. 8 we can take the location of the branch points. There is one singularity u_1 on the negative real line, and the other two are complex conjugates (as Δ_1 is a real polynomial). Using the definition (2.32), it is straightforward to show that not all of y = 0 lies in \mathcal{T}_1 , but rather only the real half-line with $u \ge u_1$. Furthermore, the lines truncate at the singularities. On the upper-half plane, the branch points can be viewed as endpoints of branch cuts coming from $\tau = \frac{1}{2} + \frac{\sqrt{3}}{2}i$ and $\tau = \frac{5}{2} + \frac{\sqrt{3}}{2}i$. See Fig. 9. From this it is straightforward to see how taking the massless limit gives back Fig. 6.

The case $m > m_{AD}$

At $m = m_{\rm AD}$ two singularities collide, and Δ_1 has a double root. Since Δ_1 is a real polynomial and depends smoothly on m, the two singularities which are complex for $m < m_{\rm AD}$ are real for $m > m_{\rm AD}$. There is no meaningful identification of the singular points when going through $m = m_{\rm AD}$, however for large m there is a distinguished singularity u_* that diverges. We can make



Figure 9: Identification of the components of the partitioning \mathcal{T}_m in $N_f = 1$ for $\mu < \mu_{AD}$, here for the choice $\mu = \frac{1}{10}$. The *u*-plane \mathcal{B}_1 is partitioned into 6 regions $u(\alpha \mathcal{F})$, with the $\alpha \in SL(2,\mathbb{Z})$ given in both pictures. The branch point (purple) identifies two points on $\partial \mathcal{F}_1(m)$. The green/orange arcs are separated by a branch cut, and the opposite ends are glued with the corresponding identical colour on the other side. The blue points correspond to $j = \mathcal{J} = 0$ and form the intersection points of \mathcal{T}_1 .

the choice of \mathcal{F}_m suitable for the limit $m \to \infty$, where we should obtain Fig. 2. By studying the dependence of the partition of the *u*-plane on the mass, one finds that u_* is bounded by a region whose area grows as $m \to \infty$. It squeezes into $T\mathcal{F}$ and $T^3\mathcal{F}$ and becomes $T^2\mathcal{F}$ in the limit $m \to \infty$. However, as we want to put the singularities on the real line we need it to touch this axis for finite $m > m_{\rm AD}$. In order to find the corresponding fundamental domain, we can glue parts of the boundary $\partial \mathcal{F}_1$, such that it not only agrees with the geometry of the partition of the *u*-plane, but also the decoupling procedure is inherent. See Fig. 10.

2.4.4 Generic complex mass

We can also consider a generic complex mass. The locus of AD masses (2.20) is then real codimension 2. In fact, it is just $\omega_3 m_{AD}$, with ω_3 a cube root of unity. If *m* is not any of these three values, the corresponding Coulomb branch has three distinct singularities.

We can decompose m = a + ib, and T_m is then a polynomial in a, b, xand y. From (2.24) we see that if m is complex, then $j(\tau_{\rm bp}) = \mathcal{J}(u_{\rm bp})$ is also complex, such that $\tau_{\rm bp}$ is generically an interior point of \mathcal{F} or an SL(2, \mathbb{Z}) copy thereof. The branch cuts most conveniently run from such branch points to the intersection points of the curves, where $\mathcal{J}(u) = j(\tau) = 0$. From (2.9) it is clear that they correspond to the two solutions of $g_2(u) = 0$. We plot the partitioning of the u-plane with the branch cuts for an imaginary mass in Fig. 11.

Due to the fact that $u_{\rm bp} \notin \mathcal{T}_1$, the branch cuts run to the interior of the



Figure 10: Identification of the components of the partitioning \mathcal{T}_m in $N_f = 1$ for $\mu > \mu_{AD}$, here for the choice $\mu = \frac{6}{5}$. The *u*-plane \mathcal{B}_1 is partitioned into 6 regions $u(\alpha_j \mathcal{F})$, with the $\alpha_j \in SL(2,\mathbb{Z})$ given in both pictures. The branch point (purple) identifies four points on $\partial \mathcal{F}_1(m)$.



Figure 11: Identification of the components of the partitioning \mathcal{T}_m in $N_f = 1$ for a complex mass $\mu \in \mathbb{C}$, here for the choice $\mu = \frac{i}{10}$. The *u*-plane \mathcal{B}_1 is partitioned into 5 regions, which is due to the fact that $u(\mathcal{F})$ and $u(T^2S\mathcal{F})$ are glued at the branch cuts (dashed). The purple dot is the branch point in $N_f = 1$.

 $\alpha_j \mathcal{F}$. The four sides of the two cuts are pairwise identified, which makes points on the branch cut smooth points on the Riemann surface. This identification glues different regions $u(\alpha_j \mathcal{F})$ together, in this case $u(\mathcal{F})$ and $u(T^2S\mathcal{F})$. This is clearly visible in Fig. 11, where the dashed lines in the *u*-planes are the images of the branch cuts in \mathcal{F}_m , and they do *not* belong to the partitioning \mathcal{T}_1 . As a result, the *u*-plane is partitioned into five and not six components. This is not in contradiction with section 2.2 because the fundamental domain $\mathcal{F}_1(m)$ is still a union of six copies of \mathcal{F} : the cutting and glueing along the branch cuts is an additional feature of the domain.

2.5 The $N_f = 2$ curve

Let us now move on to discuss the theory with two hypermultiplets. This theory has four strong coupling singularities where massless hypermultiplets appear. For general masses they are distinct points while for special mass configurations one or more singularities can collide.

The equal mass case $\mathbf{m} = (m, m)$ is has been studied in [157]. Its modular properties, fundamental domains, and mass limits are studied in great detail in [2]. It is also conceptually and computationally equivalent to the $N_f = 3$ theory with $\mathbf{m} = (m, 0, 0)$, which we discuss in section 2.6.1. For these reasons, we omit the discussion here and refer the reader to section 5.1 of [2].

2.5.1 Two distinct masses

In the generic case, the two masses are distinct. As in $N_f = 1$, we can expand and invert the \mathcal{J} -invariant for large u to find the series $(\mu_i = \frac{m_i}{\Lambda_2})$

$$\frac{u(\tau)}{\Lambda_2^2} = -\frac{1}{64}q^{-\frac{1}{2}} - \frac{1}{2}(\mu_1^2 + \mu_2^2) + \left(24(\mu_1^4 + \mu_2^4) + 16\mu_1^2\mu_2^2 - 32\mu_1\mu_2 - \frac{5}{16}\right)q^{\frac{1}{2}} - 128\left(\mu_1^2 + \mu_2^2\right)\left(16(\mu_1^4 + \mu_2^4) - 14\mu_1\mu_2 + 1\right)q + \mathcal{O}(q^{\frac{3}{2}}).$$
(2.85)

The double singularity u_* in the equal mass case now splits into two distinct singularities, u_*^{\pm} . Due to the locus of masses giving rise to *u*-planes with AD points, it is difficult to give a fundamental domain $\mathcal{F}_2(\boldsymbol{m})$ for any choice of $\boldsymbol{m} = (m_1, m_2)$. From (2.24) it is clear that there are two distinct branch points in \mathcal{B}_2 . When both m_1 and m_2 are real and small, i.e. have not made a phase transition compared to $\boldsymbol{m} = 0$, one branch point $u_{\text{bp},1}$ belongs to $\mathcal{T}_{\boldsymbol{m}}$, while the other $u_{\text{bp},2}$ does not. However, $\mathcal{J}(u_{\text{bp},2}) = j(\tau_{\text{bp},2}) \in \mathbb{R}$ is also real but larger than 12³. A natural choice of branch cuts is along the tessellation $\{\tau \in \mathbb{H} \mid j(\tau) \in \mathbb{R}\}$, which aside from (2.36) contains the SL(2, \mathbb{Z}) images of the positive imaginary axis. The plot of the partitioning $\mathcal{T}_{\boldsymbol{m}}$ shows a feature found already in $N_f = 1$ with a complex mass (see section 2.4.4): The *u*-plane is partitioned into only 5 regions, which is due to two regions $u(\alpha_j \mathcal{F})$ being glued along pairs of branch cuts (see Fig. 12). The splitting of u_* into two distinct singularities in this case does not require the two regions $TS\mathcal{F}$ and



Figure 12: Identification of the components of the partitioning $\mathcal{T}_{(m_1,m_2)}$ in $N_f = 2$ for the particular choice $\mu_1 = \frac{1}{10}$ and $\mu_2 = \frac{1}{4}$. The *u*-plane \mathcal{B}_2 is naively partitioned into six regions $u(\alpha \mathcal{F})$, with the $\alpha \in \mathrm{SL}(2,\mathbb{Z})$ given in both pictures. Two regions $u(TS\mathcal{F})$ and $u(TST^{-1}\mathcal{F})$ are however glued along the pairs of branch cuts (dotted), running from the two singular points u^{\pm}_* (orange, square) to the branch point $\tau_{\mathrm{bp},2}$ (purple, square). They do not belong to the partitioning \mathcal{T}_m . A natural choice for the branch cut is along the lines where $j(\tau)$ is real.

 $TST^{-1}\mathcal{F}$ to taper to distinct cusps, as we have that both $TS, TST^{-1}: i\infty \mapsto 1$. The two singularities are rather split due to the branch cut, and the limit of $u(\tau)$ as $\tau \to 1$ depends on the path from which $\tau = 1$ is approached. This is different from $u_+ \neq u_-$, where the boundary pieces near the cusps are not identified.

2.5.2 The massless theory

When we go to the massless theory we now find

$$\frac{u(\tau)}{\Lambda_2^2} = -\frac{1}{8} \frac{\vartheta_3(\tau)^4 + \vartheta_4(\tau)^4}{\vartheta_2(\tau)^4} = -\frac{1}{8} - \frac{1}{64} \left(\frac{\eta(\frac{\tau}{2})}{\eta(2\tau)}\right)^8$$

$$= -\frac{1}{64} (q^{-1/2} + 20q^{1/2} - 62q^{3/2} + 216q^{5/2} + \mathcal{O}(q^{7/2})).$$
(2.86)

This function is the completely replicable function of class 4C and is a Hauptmodul for $\Gamma(2)$ [56–58]. The physical discriminant becomes $\Delta = (u + \frac{\Lambda_2^2}{8})^2(u - \frac{\Lambda_2^2}{8})^2$. The two cusps correspond to $u(0) = -\frac{\Lambda_2^2}{8}$ and $u(1) = +\frac{\Lambda_2^2}{8}$. They are associated with the particles of charges (1,0) and (1,1) becoming massless.

A fundamental domain for $\Gamma(2)$ is given by

$$\mathcal{F}_2(0,0) = \mathcal{F} \cup T\mathcal{F} \cup S\mathcal{F} \cup TS\mathcal{F} \cup ST^{-1}\mathcal{F} \cup TST\mathcal{F}$$
(2.87)

and is plotted in Fig. 13 together with the map to the *u*-plane. This picture gives rise to the *dessin d'enfant* of the *j*-invariant [158, Fig. 6], as *u* is a linear function of the modular λ -invariant, which has critical points $\lambda = 0, 1, \infty$.



Figure 13: Left: Fundamental domain of $\Gamma(2)$. This is the duality group of massless $N_f = 2$. All three cusps $\{i\infty, 0, 1\}$ have width 2. Right: Plot of the massless $N_f = 2$ *u*-plane as the union of the images of *u* under the ind $\Gamma(2) = 6 \operatorname{SL}(2,\mathbb{Z})$ images of \mathcal{F} . Here, we use the decomposition $\Gamma(2) \setminus \mathbb{H} = \bigcup_{k,\ell=0}^{1} T^{\ell} S^{k} \mathcal{F} \cup ST^{-1} \mathcal{F} \cup TST \mathcal{F}$. There is a \mathbb{Z}_{2} symmetry which acts by $u \mapsto -u$. The singularities $\tau = 0, 1$ are both touched by two triangles each.

2.5.3 Type III AD mass

If we choose $m_1 = m_2 = m_{AD} = \frac{1}{2}\Lambda_2$, we find a *u*-plane with an AD theory of type *III* located at $u = u_{AD} = \frac{3}{8}\Lambda_2^2$ [105]. Three singularities collide in this point, while one remains at $u_0 = -\frac{5}{8}\Lambda_2^2$. The discriminant now takes the form

$$\Delta = (u - u_{\rm AD})^3 (u - u_0). \tag{2.88}$$

Using $\Gamma(2)$ as an intermediate field of the sextic equation, we can show that

$$\frac{u(\tau)}{\Lambda_2^2} = -\frac{f_{2B}\left(\frac{\tau}{2}\right) + 40}{64} = -\frac{1}{64} \left(q^{-1/2} + 16 + 276q^{1/2} - 2048q + \mathcal{O}(q^{3/2})\right),$$
(2.89)

where f_{2B} is defined as

$$f_{2B}(\tau) = \left(\frac{\eta(\tau)}{\eta(2\tau)}\right)^{24} = 256 \frac{\vartheta_3(\tau)^4 \vartheta_4(\tau)^4}{\vartheta_2(\tau)^8},$$
 (2.90)

and it is the McKay-Thompson series of class 2B [56–58]. It is a Hauptmodul for $\Gamma_0(2)$. Therefore, u is a modular function for $\Gamma^0(2)$. A fundamental domain of $\Gamma^0(2)$ is

$$\mathcal{F}_2(\boldsymbol{m}_{\rm AD}) = \mathcal{F} \cup T\mathcal{F} \cup S\mathcal{F} \tag{2.91}$$

and is shown in Fig. 14. It has index 3 in $PSL(2,\mathbb{Z})$, since three mutually non-local singularities have collided. This can also be seen from the fact that the curve reads

$$j(\tau) = \frac{(f_{2B}(\frac{\tau}{2}) + 16)^3}{f_{2B}(\frac{\tau}{2})}.$$
(2.92)

One has that $u(\tau_{AD}) = u_{AD}$ whenever $f_{2B}(\frac{\tau_{AD}}{2}) = -64$, whose solution locus intersects with our choice of $\Gamma^0(2) \setminus \mathbb{H}$ in $\tau_{AD} = 1 + i$. This can be proven from the S-transformation of the Dedekind η function. It is also easy to check



Figure 14: Fundamental domain of $\Gamma^0(2)$. This is the duality group of $N_f = 2$ with masses $\boldsymbol{m} = \frac{1}{2}(\Lambda_2, \Lambda_2)$. The AD point corresponds to the elliptic fixed point $\tau_{AD} = 1 + i$.

that $u(0) = u_0$. Taking the proper limits from $N_f = 2$ [2] we directly find $\frac{du}{d\tau}$ as well as $\frac{da}{du}$ and we can check that they satisfy the Matone relation

$$\frac{du}{d\tau} = -\frac{16\pi i}{2}\widehat{\Delta}\left(\frac{da}{du}\right)^2,\tag{2.93}$$

consistent with (2.52). Both branch points of the $N_f = 2$ theory have collided along with the singularities where mutually non-local states become massless. The monodromies are

$$M_0 = STS^{-1} = \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix}, \quad M_{\rm AD} = TS^{-1}T^{-1} = \begin{pmatrix} -1 & 2 \\ -1 & 1 \end{pmatrix}, \quad (2.94)$$

and they satisfy $M_0M_{\rm AD} = M_{\infty}$ with $M_{\infty} = PT^{-2}$. Furthermore, $M_{\rm AD}^4 = 1$, such that $\tau_{\rm AD}$ indeed is an elliptic fixed point of $\Gamma^0(2)$. The AD monodromy is conjugate to S^{-1} , which fixes $\tau = i$. Since $\tau_{\rm AD} = i + 1$, this gives a path in τ -space.

2.6 The $N_f = 3$ curve

We will start by discussing the $N_f = 3$ theory with one non-zero mass, $\boldsymbol{m} = (m, 0, 0)$, where we can find an explicit expression for u in terms of Jacobi theta functions. After this we discuss the generic mass case, the massless theory and a number of theories with specific AD masses.

2.6.1 One non-zero mass

For the general theory it turns out to be complicated to find closed expressions for u, but if we only keep one non-zero mass, $\boldsymbol{m} = (m, 0, 0)$, we can make more progress. Four of the strong coupling singularities now merge in pairs of two and the physical discriminant becomes

$$\Delta = (u - u_{+})^{2} (u - u_{-})^{2} (u - u_{*}), \qquad (2.95)$$

with

$$u_{\pm} = \pm \frac{m\Lambda_3}{8}, \qquad u_* = \frac{\Lambda_3^2}{2^8} + m^2.$$
 (2.96)

There are two AD points at $m = m_{AD} = \pm \frac{1}{16}\Lambda_3$ and $u = u_{AD} = \frac{1}{128}\Lambda_3^2$ where either u_+ or u_- merges with u_* to give a type III singular fibre. We now find that the sextic equation for u again splits over the intermediate field $\Gamma(2)$. In this case there is only one solution that has $|u| \to \infty$ for $\tau \to i\infty$, and as mentioned before this has $u \to -\infty$. This is then the reason why we have persistently chosen this convention in all other cases, to make the decoupling limits from $N_f = 3$ consistent. We find that

$$\frac{u}{\Lambda_3^2} = -\frac{2\vartheta_3^4\vartheta_4^4 + (\vartheta_3^4 + \vartheta_4^4)\sqrt{f_3}}{64\vartheta_2^8} = -\frac{1}{2^{12}} \left(\frac{1}{q} + (-8 + 4096\mu^2) + 4(5 + 32768\mu^2 - 4194304\mu^4)q + \mathcal{O}(q^2)\right),$$
(2.97)

where we have defined $f_3 = \frac{64m^2}{\Lambda_3^2} \vartheta_2^8 + \vartheta_3^4 \vartheta_4^4$ and $\mu = \frac{m}{\Lambda_3}$. Due to the appearance of the square root in (2.97) u is not holomorphic, and similarly to the $N_f = 1$ case there will be branch points in the fundamental domain. They are given by

$$j^{\rm bp}(m) = \frac{(\Lambda_3 - 8m)^3(\Lambda_3 + 8m)^3}{16m^4\Lambda_3^2}.$$
 (2.98)

By plugging in the expression for \mathcal{J} in terms of u we find that the branch point lies at $u_{bp} = 2m^2$ in the u-plane, as is also found by studying the Matone polynomial (2.54). We can also use known relations between the j-invariant and theta functions to check that (2.98) coincides with $f_3 = 0$, such that the branch point of u is that of the square root in (2.97).

We see that the branch point of the square root is $f_3(\tau_0) = 0$. Near τ_0 , the expansion of f_3 reads $f_3(\tau) = (\tau - \tau_0)h(\tau)$, where $h(\tau)$ is holomorphic near τ_0 and $h(\tau_0) \neq 0$. Then one branch of the square root reads $\sqrt{f_3(\tau)} = \sqrt{\tau - \tau_0}\sqrt{h(\tau)}$. Now since $h(\tau_0) \neq 0$, we have that $\tau \mapsto \sqrt{h(\tau)}$ is nonzero and in fact holomorphic in a neighbourhood of τ_0 . However, $\tau \mapsto \sqrt{\tau - \tau_0}$ is strictly non-holomorphic at τ_0 . This proves that u is not holomorphic at τ_0 .

It is straightforward to calculate other interesting quantities explicitly from (2.97), including

$$\frac{da}{du} = \frac{2\sqrt{2}i}{\Lambda_3} \frac{\vartheta_2^4}{\sqrt{\vartheta_3^4 + \vartheta_4^4 + 2\sqrt{f_3}}} \\
\frac{du}{d\tau} = \pi i \Lambda_3^2 \vartheta_3^4 \vartheta_4^4 \frac{2^8 \mu^2 \vartheta_2^8 + \vartheta_3^8 + 6\vartheta_3^4 \vartheta_4^4 + \vartheta_4^8 + 4(\vartheta_3^4 + \vartheta_4^4)\sqrt{f_3}}{2^7 \vartheta_2^8 \sqrt{f_3}}.$$
(2.99)

We can explicitly check that they satisfy Matone's relation (2.52),

$$\frac{du}{d\tau} = -16\pi i \frac{\widehat{\Delta}}{u - u_{\rm bp}} \left(\frac{da}{du}\right)^2. \tag{2.100}$$

On the rhs, the double singularities u_+ and u_- have cancelled, while, as discussed in section 2.3, the branch point $u_{\rm bp} = 2m^2$ remains in the denominator.

Fundamental domain

A fundamental domain can be found in the following way. The six roots of the sextic equation give the six cusp expansions. They can be canonically normalised to match the form of the expansion (2.97) at ∞ . Then instead of studying which transformations give the right values at the cusps, we can take the cusp expansions and try to find maps $\alpha_j \in SL(2,\mathbb{Z})$ that take $u(\tau)$ to the functions under study. Due to the square root, this is quite subtle. We furthermore need to take m as generic, such that the square root does not resolve. This allows to find the maps $\alpha_j \in SL(2,\mathbb{Z})$, which give the fundamental domain

$$\mathcal{F}_3(m,0,0) = \mathcal{F} \cup S\mathcal{F} \cup ST^{-1}\mathcal{F} \cup TST\mathcal{F} \cup TST^2\mathcal{F} \cup TST^2S\mathcal{F}, \quad (2.101)$$

shown in Fig. 15. It is valid for all masses m that do not allow the square root to resolve. We prove below that this does not happen unless m = 0 or $m = m_{\rm AD} = \frac{1}{16}\Lambda_3$.

Let us also study the paths of branch points in the fundamental domain. Similarly as in massive $N_f = 1, 2$ [2], we analyse the critical values of (2.98). We have that $\lim_{m \to 0} j^{\text{bp}}(m) = +\infty$, $j^{\text{bp}}(m_{\text{AD}}) = 12^3$, $j^{\text{bp}}(\frac{\Lambda_3}{8}) = 0$ and $\lim_{m \to \infty} j^{\text{bp}}(m) = -\infty$. It is easy to show that $j^{\text{bp}} : (0, \infty) \to \mathbb{R}$ is monotonically decreasing and therefore injective.

Since $u_{\rm bp} = 2m^2$, we have that at m = 0 the branch points coincides with u_+ and u_- . At the AD point, $m = \frac{\Lambda_3}{16}$, it collides along with u_* and u_+ . Finally, for $m \to \infty$ it diverges, just as u_* does. This fixes the points $\tau = 0$ and $\tau = 1$ for m = 0, $\tau = \tau_{\rm AD} = \frac{1}{2} + \frac{i}{2}$ for $m = m_{\rm AD}$ and $\tau = \frac{1}{2}$ or $i\infty$ for $m = \infty$. The simplest curves connecting these three points are quarter-circles with radius $\frac{1}{2}$ around $\tau = \frac{1}{2}$ starting from either $\tau = 0$ or $\tau = 1$, followed by a vertical path from $\frac{1}{2} + \frac{i}{2}$ to either $\frac{1}{2}$ or $i\infty$.

The fundamental domain together with the path of the branch points found from the above considerations is shown in Fig. 15.

The various checks of the branch points paths are analogous to $N_f = 1, 2$. We can plot $j(\tau)$ along these curves and find that it has the same global properties and critical points as (2.98). The intermediate value $j(\tau) = 0$ corresponds to $m = \frac{\Lambda_3}{8}$ and $\tau = \frac{1}{2} + \frac{1}{2\sqrt{2}}i$, which is in the SL(2, Z)-orbit of ω_3 . Along the branch point locus, u simplifies to

$$u(\tau_{\rm bp}) = -\frac{f_{\rm 2B}(\tau_{\rm bp})}{2^{13}},$$
 (2.102)

with f_{2B} given by (2.90). On the paths in Fig. 15 this function behaves precisely as $u_{\rm bp}(m) = 2m^2$.

Partitioning of the *u*-plane

Finally, we can study the partitioning that the domain (2.101) induces on the u-plane under the map (2.97). As studied in section 2.2.4, the partitioning



Figure 15: Fundamental domain for $N_f = 3$ with $\mathbf{m} = (m, 0, 0)$. The dashed lines corresponds to the path of the branch points on the positive ray from massless to infinite mass.

is contained in a real algebraic plane curve, which is given by the equation $\operatorname{Im} \mathcal{J}(u, \boldsymbol{m}, \Lambda_3) = 0$. For generic $\mu = \frac{m}{\Lambda_3}$ where $\boldsymbol{m} = (m, 0, 0)$, we can compute it as the zero locus of the polynomial

$$\begin{split} T_{(m,0,0)} =& y \left(2\mu^2 + 128x^2 - 256\mu^2 x - x + 128y^2 \right) \\ & \times \left(10616832\mu^8 + 147456\mu^6 + 672\mu^4 + \mu^2 + 2199023255552x^8 - 137438953472\mu^2 x^6 + 8796093022208x^6y^2 - 25769803776x^6 + 51539607552\mu^2 x^5 + 1275068416x^5 - 22548578304\mu^4 x^4 - 2617245696\mu^2 x^4 + 13194139533312x^4 y^4 - 137438953472\mu^2 x^4 y^2 - 77309411328x^4 y^2 - 12976128x^4 + 1610612736\mu^4 x^3 + 2097152\mu^2 x^3 + 103079215104\mu^2 x^3 y^2 + 2550136832x^3 y^2 + 49152x^3 - 301989888\mu^6 x^2 + 37748736\mu^4 x^2 + 159744\mu^2 x^2 + 8796093022208x^2 y^6 + 137438953472\mu^2 x^2 y^4 - 77309411328x^2 y^4 - 53687091200\mu^4 x^2 y^2 - 2550136832\mu^2 x^2 y^2 - 9175040x^2 y^2 - 64x^2 - 37748736\mu^6 x - 344064\mu^4 x - 768\mu^2 x + 51539607552\mu^2 xy^4 + 1275068416xy^4 + 536870912\mu^4 xy^2 - 18874368\mu^2 xy^2 - 16384xy^2 + 2199023255552y^8 + 137438953472\mu^2 y^6 - 25769803776y^6 - 22548578304\mu^4 y^4 + 67108864\mu^2 y^4 + 3801088y^4 + 301989888\mu^6 y^2 + 16777216\mu^4 y^2 + 102400\mu^2 y^2 + 64y^2). \end{split}$$

The second factor on the rhs gives a circle on the $x + iy = \frac{u}{\Lambda_3^2}$ -plane with radius $|\mu^2 - \frac{1}{2^8}|$ and centre $(x, y) = (\mu^2 + \frac{1}{2^8}, 0)$. By tuning the mass μ from 0 to ∞ , one passes through the AD point $\mu = \frac{1}{2^4}$ where the radius of the circle shrinks to 0. For this mass, three regions defined through $T_{(m,0,0)} = 0$ collapse to a point $x + iy = \frac{u_{AD}}{\Lambda_3^2} = \frac{1}{128}$, which is the only root over \mathbb{R}^2 of the quadratic polynomial. This gives further evidence that the domain 15 is in fact correct for all $\mu \in (0, \infty) \setminus \{\frac{1}{16}\}$.

Limits to zero, AD and infinite mass

As in the $\mathbf{m} = (m, m) N_f = 2$ theory, there are three interesting limits: $m \to 0, m \to \infty$ and $m \to m_{AD}$. In the massless limit, we aim to recover $\Gamma_0(4) \setminus \mathbb{H}$. This is not difficult to see: Under $\Gamma_0(4)$, we can identify $TST^2S\mathcal{F}$ with $ST^{-2}S\mathcal{F}$, and similarly $TST^2\mathcal{F}$ with $ST^{-2}\mathcal{F}$, since the transition maps are in $\Gamma_0(4)$. This gives precisely Fig. 17.

By decoupling the massive hypermultiplet, the theory flows to massless $N_f = 2$. We find that $u_* \to \infty$, while $u_{\pm} \to \pm \frac{\Lambda_2^2}{8}$. From section 2.5.2 it is clear that the singularities u_{\pm} do not move in τ -space. The cusp region $TST^2S\mathcal{F}$ is identified with $T\mathcal{F}$ under the duality group $\Gamma(2)$ of massless $N_f = 2$. Moreover, the remaining differing triangle $TST^2\mathcal{F}$ can be mapped to $TS\mathcal{F}$ using $\Gamma(2)$. This then gives precisely $\Gamma(2) \setminus \mathbb{H}$ as in Fig. 13.

Finally, in the limit $m \to m_{\rm AD} = \frac{1}{16}\Lambda_3$ the singularities u_* and u_+ collide. Since they drop out of the curve, we should remove all regions near those cusps. In Fig. 15 we can remove the triangles $TST\mathcal{F}$, $TST^2\mathcal{F}$ and $TST^2S\mathcal{F}$, after which the index 3 group $\Gamma_0(2)$ remains. This is precisely what is found as the duality group of the $\mathbf{m} = (m_{\rm AD}, 0, 0)$ theory, as shown in Fig. 14. The pre-image of the merged non-local singularities $u_{\rm AD}$ is the point $\tau_{\rm AD}$, which lies in the interior of \mathbb{H} and corresponds to the point where the branch points have collided.

2.6.2 Generic masses

For generic masses $\boldsymbol{m} = (m_1, m_2, m_3)$, the order parameter reads

$$\frac{u}{\Lambda_3^2} = -\frac{1}{2^{12}} \left(\frac{1}{q} + (-8 + 4096M_2) + 4 \left(5 + 32768M_2 + 3670016M_3 - 4194304M_4 - 4194304M_4' \right) q + \mathcal{O}(q^2) \right),$$
(2.104)

where the coefficients M_i are the symmetric polynomials defined in (2.55) for the variables $\frac{m_i}{\Lambda_3}$. There are five generally distinct singular points.

Due to the $N_f = 3$ distinct branch points on the *u*-plane, the fundamental domain for a given mass \boldsymbol{m} has an intricate web of branch cuts. Furthermore, the fundamental domains $\mathcal{F}_3(\boldsymbol{m})$ change as \boldsymbol{m} passes through \mathcal{L}_3^{AD} (see Fig. 3). A fundamental domain $\mathcal{F}_3(\boldsymbol{m})$ can also change when \boldsymbol{m} is varied such that Δ_3 has any double root, and when branch points in \mathbb{H} pass through the SL(2, \mathbb{Z}) tessellation $\mathcal{T}_{\mathbb{H}}$ (2.36).

For any given mass \boldsymbol{m} one easily computes T_3 from (2.33), and truncates the plot of the level set to the region where $\mathcal{J}(u, \boldsymbol{m}, \Lambda_3) \leq 12^3$. The branch points $u_{\rm bp}$ are the zeros of $P_3^{\rm M}$ (2.54). On the upper half-plane \mathbb{H} , a branch point $\tau_{\rm bp}$ is any of the SL(2, \mathbb{Z})-images of $(j|_{\mathcal{F}})^{-1}(\mathcal{J}(u_{\rm bp}))$. When $\mathcal{J}(u_{\rm bp}) \leq 12^3$, then obviously $\tau_{\rm bp} \in \mathcal{T}_{\mathbb{H}}$. If $\mathcal{J}(u_{\rm bp}) > 12^3$, then $\tau_{\rm bp} \in \mathrm{SL}(2, \mathbb{Z}) \cdot i\mathbb{R}_{>0}$. Lastly, if $\mathcal{J}(u_{\rm bp}) \notin \mathbb{R}$, then $\tau_{\rm bp}$ is an interior point of an SL(2, \mathbb{Z}) copy of \mathcal{F} .

In Fig. 16 we plot the *u*-plane and corresponding fundamental domain for three distinct masses. The five distinct singular points are partitioned into five regions $u(\alpha \mathcal{F})$, where two of them are glued by branch cuts.



Figure 16: Identification of the components of the partitioning $\mathcal{T}_{(m_1,m_2,m_3)}$ in $N_f = 3$ for the particular choice $\mu_1 = \frac{1}{10}$, $\mu_2 = \frac{3}{10}$ and $\mu_3 = \frac{5}{10}$. The *u*-plane \mathcal{B}_3 is partitioned into five regions $u(\alpha \mathcal{F})$, as those for ST^{-1} and ST are glued by two branch cuts. The fundamental domain is given by six copies of \mathcal{F} , with three pairs of branch points (purple). Two branch points (triangle and disk) lie on \mathcal{T}_3 . The third (square) lies in the interior and glues the copies $ST\mathcal{F}$ and $ST^{-1}\mathcal{F}$. A natural choice for the branch cuts not lying in \mathcal{T}_3 (dashed) is on the real axis in the *u*-plane, for which the path in \mathbb{H} is along the tessellation $SL(2, \mathbb{Z}) \cdot i\mathbb{R}_{>0}$. The singularity at $\tau = 0$ does not have width 3 as apparent, since due to the branch cuts $u(\tau)$ assumes three different values depending on which side of the branch cut $\tau = 0$ is approached from.

2.6.3 The massless theory

When sending $m \to 0$ from above we find

$$\frac{u(\tau)}{\Lambda^2} = -\frac{1}{64} \frac{\vartheta_3(\tau)^2 \vartheta_4(\tau)^2}{(\vartheta_3(\tau)^2 - \vartheta_4(\tau)^2)^2} = -\frac{1}{2^{12}} \left(\frac{\eta(\tau)}{\eta(4\tau)}\right)^8$$

$$= -\frac{1}{2^{12}} (q^{-1} - 8 + 20q - 62q^3 + \mathcal{O}(q^{9/2})).$$
(2.105)

It is the completely replicable function of class 4C and a Hauptmodul for $\Gamma_0(4)$ [56–58]. The physical discriminant is $\Delta = u^4(u - \frac{\Lambda_3^2}{2^8})$, and one finds that the singularities are located at u(0) = 0 and $u(\frac{1}{2}) = \frac{\Lambda_3^2}{2^8}$. At $\tau = 0$ a dyon with charge (1,0) becomes massless, while at $\tau = \frac{1}{2}$ one finds instead that a dyon with charge (2,1) becomes massless. The massless $N_f = 3$ *u*-plane has no global symmetries.

A choice of fundamental domain for $\Gamma_0(4)$ is

$$\mathcal{F}_3(0,0,0) = \mathcal{F} \cup S\mathcal{F} \cup ST\mathcal{F} \cup ST^{-1}\mathcal{F} \cup ST^{-2}\mathcal{F} \cup ST^{-2}S\mathcal{F}, \qquad (2.106)$$

and is shown in Fig. 17.

2.6.4 Type IV AD mass

As illustrated in Fig. 3, on the generic mass $N_f = 3 u$ -plane, there is not only the *IV* AD point but also a variety of *III* and *II* points. We will give a few



Figure 17: Fundamental domain of $\Gamma_0(4)$, the duality group of massless $N_f = 3$. The cusp at $\tau = 0$ has width four, while the cusp at $\tau = \frac{1}{2}$ has width one.

explicit examples of the u-plane of the theories with masses tuned to these specific values, starting with the most symmetric case.

For $\boldsymbol{m} = (m, m, m)$ and $m = \frac{1}{8}\Lambda_3$, four mutually non-local singularities collide in $u_{AD} = \frac{1}{32}\Lambda_3^2$. The remaining singularity is $u_0 = -\frac{19}{2^8}\Lambda_3^2$ and never collides with the other four. The physical discriminant is $\Delta = (u - u_0)(u - u_{AD})^4$. One finds that

$$\frac{u}{\Lambda_3^2} = -\frac{j^* + 304}{2^{12}},\tag{2.107}$$

where

$$j^* = 432 \frac{\sqrt{j} + \sqrt{j - 1728}}{\sqrt{j} - \sqrt{j - 1728}} = 432 \frac{E_4^{\frac{3}{2}} + E_6}{E_4^{\frac{3}{2}} - E_6}$$

$$= \frac{1}{q} - 120 + 10260q - 901120q^2 + 91676610q^3 + O\left(q^4\right)$$
(2.108)

is the Ramanujan-Sato series of level 1 [159–161].¹⁴

Inverting (2.108) we find

$$j = \frac{(j^* + 432)^2}{j^*}.$$
(2.109)

Using this and a discussion similar to the massless $N_f = 1$ case for the transformations of j^* we find that the singularities are located at $(\omega_j = e^{2\pi i/j})$

$$u(i\infty) = \infty, \quad u(\omega_3) = u_{AD}, \quad u(0) = u_0.$$
 (2.110)

¹⁴The Ramanujan-Sato series generalise Ramanujan's formula for $\frac{1}{\pi}$ as a series of quotients of modular forms. They exist for level 1 up to 11 and beyond. The level 1 series is the only one whose generating function can not be expressed by an η -quotient [162].


Figure 18: Left: Proposed fundamental domain of the duality group Γ_{j^*} of the $N_f = 3$ $\boldsymbol{m} = \frac{1}{8}(\Lambda_3, \Lambda_3, \Lambda_3)$ theory. It is a region with index 6 - 4 = 1 + 1, elliptic fixed point ω_3 and cusp at 0. This uniquely fixes it. Right: Plot of the *u*-plane for the corresponding theory as the union of the images of *u* under the 6 - 4 = 2 SL $(2, \mathbb{Z})$ images of \mathcal{F} as in the left figure. The complex plane can clearly be covered by 2 triangles. This demonstrates that the proposal of the left figure is indeed a fundamental domain for (2.107). The weak coupling region is covered by $u(\mathcal{F})$. The strong coupling region $u(S\mathcal{F})$ contains the singular point u_0 in its interior. The AD point (orange) lies on the boundary.

We can read off from (2.109) that $\operatorname{ord}(g_2, g_3, \Delta) = (2, 2, 4)$ at the AD point, such that according to Table 5 we indeed have a singular fibre of Kodaira type IV [155].

From (2.109) we read off that the duality group Γ_{j^*} has index 2, which is consistent with the previous cases in $N_f = 1, 2$ in that a factor of $(u - u_{AD})^4$ has cancelled from g_2^3 and Δ , and therefore does not contribute to the index. The fundamental region of u is therefore of index 2 with 0, ω_3 and $i\infty$ on its boundary. However, there is no index 2 subgroup of SL(2, Z) with two distinct cusps [154, Table 4.1].¹⁵ This agrees with the fact that (2.107) is not a classical modular form and the monodromy group does not promote to a modular group since its action on u is not associative (see section 2.4.1). We can nevertheless propose a fundamental region

$$\mathcal{F}_3(\boldsymbol{m}_{\rm AD}) = \mathcal{F} \cup S\mathcal{F}, \qquad (2.111)$$

see Figure 18.

The monodromies are found by consistency,

$$m_0 = STS^{-1} = \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix}, \quad m_{\rm AD} = TS^{-1} = \begin{pmatrix} -1 & 1 \\ -1 & 0 \end{pmatrix},$$
 (2.112)

and are unique in SL(2, Z).¹⁶ They fix $\tau = 0$ and $\tau_{AD} = e^{\pi i/3} = \frac{1}{2} + \frac{\sqrt{3}}{2}i$, respectively, and produce the large u monodromy $m_0 m_{AD} = PT^{-1}$. Just as in the massless $N_f = 1$ case, we note that as matrices they do not form a

¹⁵In fact, there is exactly one index 2 subgroup of SL(2, \mathbb{Z}) and it has only one cusp of width two. This group is sometimes referred to as $\Gamma_0(1)^*$ and is generated by TS and T^2 , and the Hauptmodul is given by $\sqrt{j-1728} = 8 \frac{(\vartheta_2^4 + \vartheta_3^4)(\vartheta_3^4 + \vartheta_4^4)(\vartheta_4^4 - \vartheta_2^4)}{\vartheta_2^4 \vartheta_3^4 \vartheta_4^4}$.

¹⁶The overall signs are fixed in the following way. The large u monodromy is PT^{-1} . The monodromy at m_0 is oriented such that it conjugates to T. This fixes the sign of $m_{\rm AD}$ from the below relation.

congruence subgroup but instead generate the whole of $SL(2,\mathbb{Z})$, since $m_{\infty} = PT^{-1}$ and $Tm_0T = S$. However, u is not invariant under S.

2.6.5 Type III AD mass

In the single mass case $\mathbf{m} = (m, 0, 0)$ with $m_{\rm AD} = \frac{1}{16}\Lambda_3$, the $N_f = 3$ curve has an AD point at $u_{\rm AD} = \frac{1}{128}\Lambda_3^2$. The physical discriminant is $\Delta = (u - u_{\rm AD})^3(u - u_0)^2$ with $u_0 = -\frac{1}{128}\Lambda_3^2$, which is $-u_{\rm AD}$ by coincidence. One easily finds

$$\frac{u(\tau)}{\Lambda_3^2} = -\frac{f_{2B}(\tau) + 32}{2^{12}},$$
(2.113)

with f_{2B} defined in (2.90). This fits nicely into the description as ind $\Gamma_0(2) = 3$ is equal to 6 - 3 = 3, being the number of mutually non-local singularities collided at u_{AD} . We find that $u(\tau) = u_0$ is equivalent to $f_{2B}(\tau) = 0$, and one can easily show that f_{2B} vanishes at the cusp $\tau = 0$. Using the S-transformation of η , we can show that $\tau_{AD} = \frac{1}{2} + \frac{i}{2}$. In terms of the Hauptmodul, the curve reads $j = \frac{(f_{2B}+256)^3}{f_{2B}^2}$. This shows that the AD point is indeed a type *III* singularity. It also follows that $j(\tau_{AD}) = 12^3$ and that τ_{AD} is in the SL(2, Z) orbit of *i*. The duality group $\Gamma_0(2)$ is generated by $g_1 = T$ and $g_2 = ST^2S$. The AD point is stabilised by $g_1g_2 \in \Gamma_0(2)$, which makes it an elliptic fixed point. A fundamental domain for *u* is given in Figure 19. The effective coupling at the AD point is explained through the fact that the $N_f = 3$ branch point collides along with the three mutually non-local singularities in u_{AD} , and the two branch points on the upper half-plane as drawn in Fig. 15 collide at τ_{AD} for $m = m_{AD}$.



Figure 19: Fundamental domain of $\Gamma_0(2)$. This is the duality group of $N_f = 3$ with mass $\boldsymbol{m} = \frac{1}{16}(\Lambda_3, 0, 0)$. The AD point is the elliptic fixed point of the domain, located at $\tau_{AD} = \frac{1}{2} + \frac{i}{2}$.

The monodromies are

$$M_0 = ST^2 S^{-1} = \begin{pmatrix} 1 & 0 \\ -2 & 1 \end{pmatrix}, \quad M_{\rm AD} = (TS)^{-1} S^{-1} TS = \begin{pmatrix} -1 & 1 \\ -2 & 1 \end{pmatrix}. \quad (2.114)$$

The first one describes the path around the cusp $\tau = 0$, which has width 2. The AD monodromy is conjugate to S^{-1} , which fixes $\tau = i$. The path is then given by the map $(TS)^{-1} : i \mapsto \tau_{AD}$. The matrices (2.114) satisfy $M_0M_{AD} = M_{\infty}$ with $M_{\infty} = PT^{-1}$ and $M_{AD}^2 = 1$, such that τ_{AD} is indeed an elliptic fixed point for $\Gamma_0(2)$ of order 2. They are also related to (2.94) by conjugation with diag(2, 1), which induces the isomorphism between the $\Gamma_0(2)$ and $\Gamma^0(2)$ curves.

2.6.6 Type *II* AD mass

On the equal mass $\boldsymbol{m} = (m, m, m)$ curve we can also tune the mass to $m = m_{\rm AD} = -\frac{1}{64}\Lambda_3$ to find a type II AD theory at $u_{\rm AD} = \frac{5}{2^{10}}\Lambda_3^2$. By fixing the mass to this value we find

$$\frac{u(\tau)}{\Lambda_3^2} = -\frac{f_{3B}(3\tau) + 7}{2^{12}} \tag{2.115}$$

with f_{3B} given in (2.75). At u_{AD} two mutually non-local singularities collide, while the other three reside at $u_0 = -\frac{7}{2^{12}}\Lambda_3^2$. The physical discriminant is therefore $\Delta = (u - u_{AD})^2(u - u_0)^3$. We know from section 2.4.2 that $\tau \mapsto f_{3B}(3\tau)$ is a Hauptmodul for $\Gamma_0(3)$, and in fact the fundamental domain is just given by the one for $\Gamma^0(3)$ as in Fig. 7, with every point divided by 3. It also decomposes into SL(2, Z) images of \mathcal{F} , see Fig. 20.



Figure 20: Fundamental domain of $\Gamma_0(3)$, the duality group of $N_f = 3$ with mass $\boldsymbol{m} = -\frac{1}{64}(\Lambda_3, \Lambda_3, \Lambda_3)$. It has index 4 in PSL(2, \mathbb{Z}) and a type *II* AD point located at the elliptic fixed point $\tau_{\rm AD} = \frac{\omega_{12}}{\sqrt{3}} = \frac{1}{2} + \frac{1}{2\sqrt{3}}i$. The width of the cusp $\tau = 0$ is 3.

The AD point $u(\tau_{AD}) = u_{AD}$ translates to $f_{3B}(3\tau_{AD}) = -27$ which has $\tau_{AD} = \frac{1}{\sqrt{3}}\omega_{12}$ as a solution (where $\omega_j = e^{2\pi i/j}$). The other singularity satisfies

 $f_{3B}(3\tau) = 0$ and therefore $\tau = 0$. In terms of the Hauptmodul of $\Gamma_0(3)$ the *j*-invariant of the curve with above given mass \boldsymbol{m}_{AD} reads $j = (f_{3B} + 27)(f_{3B} + 243)^3/f_{3B}^3$. This proves that the AD singularity is Kodaira type II and therefore indeed equivalent to the II theory in $N_f = 1$, see section 2.4.2. It is interesting that both curves are parametrised by the same Hauptmodul, as the number of singularities on the curves are different.

The monodromies are given by

$$M_0 = ST^3 S^{-1} = \begin{pmatrix} 1 & 0 \\ -3 & 1 \end{pmatrix}, \quad M_{\rm AD} = (TS)^{-1} (ST)^{-1} (TS) = \begin{pmatrix} -1 & 1 \\ -3 & 2 \end{pmatrix},$$
(2.116)

which are just (2.82) conjugated by diag(3, 1). They furthermore satisfy $M_0 M_{AD}$ = M_{∞} with $M_{\infty} = PT^{-1}$. Since $M_{AD}^6 = \mathbb{1}$ in PSL(2, \mathbb{Z}), the AD point τ_{AD} is an elliptic fixed point in $\Gamma_0(3)$. Its stabiliser M_{AD} decomposes into the monodromy $(ST)^{-1}$ around $\tau = \omega_3$, and the path $(TS)^{-1} : \omega_3 \mapsto \tau_{AD}$.

2.7 Discussion

We have studied the Coulomb branches \mathcal{B}_{N_f} of SU(2) $\mathcal{N} = 2$ Yang–Mills theories with $N_f \leq 3$ massive hypermultiplets in the fundamental representation. In particular, we have considered the order parameter u as function of the effective coupling, and derive domains \mathcal{F}_{N_f} such that $u : \mathcal{F}_{N_f} \to \mathcal{B}_{N_f}$ is 1-to-1. We find that generically the function u has square roots appearing in the expressions for u, such that \mathcal{F}_{N_f} is not isomorphic to a domain $\Gamma \setminus \mathbb{H}$ for a congruence subgroup of SL(2, \mathbb{Z}). Nevertheless, exact expressions can be determined, such as for $N_f = 2$ with 2 equal masses, and $N_f = 3$ with one non-vanishing mass. For other special values, branch points and cuts can be absent and the fundamental domain is that of a modular curve for a congruence subgroups of SL(2, \mathbb{Z}), as also encountered in cases in the literature [45, 100–102].

We described how the order parameters are naturally expressed as roots of certain degree six polynomials with modular functions as coefficients. Many interesting aspects of the theories can be read off from these polynomials:

- The degree of the polynomial tells us that the fundamental domains of the order parameters can be described as six copies of the ordinary SL(2,ℤ) domain.
- For the modular theories this further implies that the duality group needs to be at most index six in SL(2, Z).
- The discriminant of the sextic polynomials includes the branch points as well as the superconformal AD fixed points of the theories.
- We further discussed how one can explicitly construct fundamental regions of order parameters as images of fundamental domains in \mathcal{F}_{N_f} . The partitioning of the fundamental regions of the order parameters seem to

generalise aspects of the dessins d'enfants [39, 124–127, 158, 163, 164] to the case of non-modular elliptic surfaces.

Physically, the branch points and cuts provide a mechanism for \mathcal{F}_{N_f} to evolve as function of the mass. This is most apparent in the limits where a hypermultiplet decouples or multiple singularities coincide, where branch cuts appear to "cut" and "glue" regions of \mathcal{F}_{N_f} . In particular near an AD point, regions with non-local cusps are disconnected from \mathcal{F}_{N_f} . This makes it manifest that on the *u*-plane, not only non-local singularities become coincident, but also branch points, which ceases to be branch points in the limit because also the pre-images in \mathcal{F}_{N_f} have collided.

We believe that our methods can be adapted to many other rank one theories, such as those of class S [21, 113]. The present analyses could perhaps also be used to draw lessons about moduli spaces of other theories, such as $\mathcal{N} = 2$ SYM with gauge group SU(N) for N > 2 or Calabi-Yau compactifications in string theory, where in many cases similar structures should arise. Remnants of which could perhaps be seen in [1, 153]. Lastly, we hope our methods find applications in similar geometries such as F-theory [165] and 5d SCFTs [166]. Moreover, our findings benefit the evaluation of the *u*-plane integral [71, 73, 81, 82, 84], which we discuss in section 5.

We would further like to mention to explore potential physical consequences of the branch points. It is known that the AD points correspond to critical points of a second order phase transition [157, 167, 168]. It might then be natural to think of the branch cuts in \mathcal{F}_{N_f} , as in for example Fig. 8, as boundaries over which a first order phase transition takes place. Since branch points and cuts seem to be a generic feature, it would suggest that similar points appear in all theories with these kinds of superconformal fixed points. It would of course be very interesting to study this further and we leave that for future work.

3 Four flavours, triality and bimodular forms

In this Section, we study the $\mathcal{N} = 2$ supersymmetric SYM theory with $N_f = 4$ fundamental hypermultiplets. Together with the asymptotically free theories investigated in section 2, it completes the study of SU(2) theories with fundamental hypermultiplets and non-positive one-loop beta functions. This section is based on [3].

3.1 Introduction

The $\mathcal{N} = 2$ supersymmetric Yang–Mills field theory with gauge group SU(2) and $N_f = 4$ fundamental hypermultiplets is distinguished for various reasons [46], including:

- The theory is superconformal up to mass terms for the hypermultiplets, and is a benchmark for four-dimensional SCFT's with $\mathcal{N} = 2$ supersymmetry [105, 169–171].
- The theory is a building block for other four-dimensional $\mathcal{N} = 2$ SCFT's and the 2d/4d correspondence [21, 172].
- The theory exhibits an intriguing electric-magnetic duality group including triality [46]. This duality group acts on the UV coupling $\tau_{\rm UV}$ and running coupling constant τ , and contains elements which act simultaneously on the two couplings as well as separately.
- The theory is a "parent" theory from which the asymptotically free $\mathcal{N} = 2$, SU(2) theories with $N_f \leq 3$ hypermultiplets can be obtained by decoupling one or more hypermultiplets [45, 46].

The focus of the present section is on the third bullet point. We analyse duality groups for the couplings $\tau_{\rm UV}$ and τ of the $N_f = 4$ theory as function of the masses. To this end, explicit expressions for the order parameter u = $\langle \operatorname{Tr} \phi^2 \rangle$, with ϕ the complex scalar of the vector multiplet, are determined as function of both $\tau_{\rm uv}$ and τ . We identify several loci in the space of masses where u transforms as a modular form for both $\tau_{\rm UV}$ and τ . This extends section 2 on theories with $N_f \leq 3$ to $N_f = 4$. In section 2, we determined fundamental domains for the running coupling τ for the asymptotically free theories by analysing in detail the order parameter u as function of the effective coupling $\tau \in \mathbb{H}$. We have demonstrated that for generic masses this function has branch points, with the consequence that the fundamental domain for τ is in general not of the form $\Gamma \setminus \mathbb{H}$ for a congruence subgroup $\Gamma \subset SL(2,\mathbb{Z})$. Only for specific values of the masses, such as those giving rise to Argyres-Douglas (AD) points, the order parameter is (weakly) holomorphic as function of τ , and the fundamental domain is that of a congruence subgroup. In this section, we find that these features are present as well for the $N_f = 4$ theory, but with an additional dependence on $\tau_{\rm uv}$.

At special modular loci, some properties of the $N_f = 4$ order parameters are similar to that of the $\mathcal{N} = 2^*$ SU(2) theory, i.e. the superconformal theory with a single adjoint hypermultiplet. The $\mathcal{N} = 2^*$ order parameter transforms as a modular form under the group $\Gamma(2) \times \Gamma(2)$ with the first factor acting on τ and the second on τ_{UV} , while it also transforms as a modular form under simultaneous SL(2, Z) transformations of τ and τ_{UV} [148, 173, 174]. It was later clarified that $u(\tau, \tau_{\text{UV}})$ is an example of a meromorphic *bimodular* form [85]. Such functions have appeared, although sporadically, in the mathematical literature [175–177].

The SU(2) $N_f = 4$ theory exhibits a richer structure: It has four mass parameters that give rise to seven singular vacua on the *u*-plane. For special choices of the masses, the u-planes contain any of the three SU(2) Argyres-Douglas superconformal points (A_1, A_2) , (A_1, A_3) and (A_1, D_4) , while in the massless case there is a non-abelian Coulomb point with a five quaternionicdimensional Higgs branch [104, 105]. For generic masses, the singularities are roots of a sextic polynomial, for which there is no known expression. The flavour symmetry SO(8) becomes the universal cover Spin(8) in the quantum theory. It has a *triality* group Out(Spin(8)) of outer automorphisms, which is isomorphic to the symmetric group S_3 on three letters. The full symmetry group of the $N_f = 4$ curve is then the semidirect product $\text{Spin}(8) \rtimes_{\varphi} \text{SL}(2, \mathbb{Z})$, which is induced by the group homomorphism $\varphi : \mathrm{SL}(2,\mathbb{Z}) \to \mathrm{Out}(\mathrm{Spin}(8)).$ As the triality group is of order $|S_3| = 6$, the orbits of the group action on mass space \mathbb{C}^4 generally have six elements. However, there are specific mass configurations with enhanced global symmetry that are invariant under subgroups of the triality group, for which the orbits collapse, either to three elements or to a single element.

We study four such configurations in detail, and show that their order parameters, periods and discriminants are bimodular forms for subgroups of $SL(2,\mathbb{Z})$. For the triality invariant case $(m_1, m_2, m_3, m_4) = (m, m, 0, 0)$ we find that the order parameter is a bimodular form of weight (0, 2) with $\Gamma(2)$ acting on both τ and τ_{UV} individually, while it is also bimodular for $SL(2,\mathbb{Z})$ acting on τ and τ_{UV} simultaneously. If all four hypermultiplets are rather given an equal mass, the triality orbit has three elements. The *u*-planes for these three mass configurations are modular curves for the three subgroups of $SL(2,\mathbb{Z})$ conjugate to $\Gamma^0(4)$. The order parameters, periods and discriminants are permuted by triality, and can thus be organised into vectors to form one-parameter families of vector-valued bimodular forms for $SL(2,\mathbb{Z})$. We further give some examples of exact expressions for order parameters of more complicated theories with two independent mass parameters. These theories then include both AD points and branch cuts.

3.2 Four flavours and triality

The one-loop beta function of $\mathcal{N} = 2$ supersymmetric Yang–Mills theory with $N_f \leq 4$ hypermultiplets in the fundamental representation is $\beta_{N_f}(g_{\rm YM}) = -\frac{g_{\rm YM}^3}{16\pi^2}(4-N_f)$. The gauge coupling $g_{\rm YM}$ is combined with the theta angle θ in the Lagrangian as $\tau = \frac{\theta}{\pi} + \frac{8\pi i}{g_{\rm YM}^2}$. This complexified gauge coupling can be considered as the expectation value of a background chiral superfield. In the renormalisation scheme where the superpotential remains a holomorphic function of all chiral superfields, the one-loop running coupling at the energy scale E can be expressed as [10]

$$\tau(E) = \tau_{\rm\scriptscriptstyle UV} - \frac{4 - N_f}{2\pi i} \log \frac{E}{\Lambda_{\rm\scriptscriptstyle UV}}.$$
(3.1)

It is one-loop exact in the holomorphic scheme, and thus for $N_f < 4$ the combination

$$\Lambda_{N_f}^{4-N_f} \coloneqq \Lambda_{\rm UV}^{4-N_f} e^{2\pi i \tau_{\rm UV}} \tag{3.2}$$

of the scale $\Lambda_{\rm UV}$ and the coupling $\tau_{\rm UV}$ is invariant to all orders in perturbation theory. This complexified dynamical scale Λ_{N_f} sets the overall scale of the theory. For $N_f = 4$ on the other hand, there is a distinguished dimensionless parameter $\tau_{\rm UV}$, on which the theory depends nontrivially. To shorten the notation, we will also set $\tau_0 := \tau_{\rm UV}$ and $q_0 := e^{2\pi i \tau_0}$ in the following.

3.2.1 The curve

The low-energy physics of $\mathcal{N} = 2$ SYM with $N_f = 4$ massive hypermultiplets has been determined in [46, 148, 173, 178–180]. Similar to the asymptotically free ($N_f \leq 3$) cases, the physics is encoded in an elliptic curve which depends holomorphically on the Coulomb branch parameter $u \in \mathcal{B}_4$. This coordinate uparametrises the Coulomb branch \mathcal{B}_4 of the $N_f = 4$ theory. Let us first define the symmetric mass combinations

$$\begin{bmatrix} m_1^k \end{bmatrix} = \sum_{i=1}^4 m_i^k, \qquad \begin{bmatrix} m_1^2 m_2^2 \end{bmatrix} = \sum_{i < j} m_i^2 m_j^2$$
$$\begin{bmatrix} m_1^4 m_2^2 \end{bmatrix} = \sum_{i \neq j} m_i^4 m_j^2, \qquad \begin{bmatrix} m_1^2 m_2^2 m_3^2 \end{bmatrix} = \sum_{i < j < k} m_i^2 m_j^2 m_k^2, \qquad (3.3)$$
$$Pf(\boldsymbol{m}) = m_1 m_2 m_3 m_4.$$

The $N_f = 4$ curve for generic masses is then [46]

$$y^{2} = W_{1}W_{2}W_{3} + A\left(W_{1}T_{1}\left(e_{2} - e_{3}\right) + W_{2}T_{2}\left(e_{3} - e_{1}\right) + W_{3}T_{3}\left(e_{1} - e_{2}\right)\right) - A^{2}N,$$
(3.4)

where

$$W_{i} = x - e_{i}u - e_{i}^{2}R,$$

$$A = (e_{1} - e_{2})(e_{2} - e_{3})(e_{3} - e_{1}),$$

$$R = \frac{1}{2} [m_{1}^{2}],$$

$$T_{1} = \frac{1}{12} [m_{1}^{2}m_{2}^{2}] - \frac{1}{24} [m_{1}^{4}],$$

$$T_{2,3} = \mp \frac{1}{2} Pf(\boldsymbol{m}) - \frac{1}{24} [m_{1}^{2}m_{2}^{2}] + \frac{1}{48} [m_{1}^{4}],$$

$$N = \frac{3}{16} [m_{1}^{2}m_{2}^{2}m_{3}^{2}] - \frac{1}{96} [m_{1}^{4}m_{2}^{2}] + \frac{1}{96} [m_{1}^{6}],$$
(3.5)

and the half periods

$$e_1 = \frac{1}{3}(\vartheta_3^4 + \vartheta_4^4), \quad e_2 = -\frac{1}{3}(\vartheta_2^4 + \vartheta_3^4) \quad e_3 = \frac{1}{3}(\vartheta_2^4 - \vartheta_4^4)$$
(3.6)

are functions of $\tau_0 = \tau_{\text{UV}}$, with $e_1 + e_2 + e_3 = 0$. The Jacobi theta functions ϑ_i are defined in Appendix A. Since the rhs of (3.4) is a cubic polynomial in x, it is indeed an elliptic curve. We obtain the low energy theory with $N_f = 3$ flavours by taking the limit $\tau_0 \to i\infty$ (or, equivalently, $q_0 \to 0$) and $m_4 \to \infty$ while holdingtuals2z $\Lambda_3 = 64q_0^{\frac{1}{2}}m_4$ fixed. The order parameters are then related as [46]

$$u_{N_f=4} + \frac{1}{4} e_1 \left[\!\left[m_1^2\right]\!\right] \to u_{N_f=3}.$$
 (3.7)

See (2.3) for the corresponding curves.

Let us study the singularity structure of the Coulomb branch. For generic masses $\mathbf{m} = (m_1, m_2, m_3, m_4)$, there are six distinct strong coupling singularities. By tuning the mass, some of those singularities can collide. If we weight each singularity by the number of massless hypermultiplets at that point, the total weighted number of singularities on the *u*-plane is thus always 6. Denote by k_l the weight of the *l*-th singularity, and by $\mathbf{k}(\mathbf{m}) = (k_1, k_2, ...)$ the vector of those weights. In Table 1, we list a selection of specifically symmetric mass configurations. One notices that certain a priori unrelated cases have the same weight vector \mathbf{k} and global symmetries, such as the cases {B, C, D} and {E, F, G}. This will be explained in the next subsection. It is also clear that $\mathbf{k}(\mathbf{m})$ gives a partition of 6, the total number of singularities on \mathcal{B}_4 .

3.2.2 Triality

Let us study the symmetries of the $N_f = 4$ curve (3.4) with mass $\boldsymbol{m} = (m_1, m_2, m_3, m_4)$. Scale invariance, the U(1)_R R-symmetry and the SL(2, Z) symmetry acting on τ_0 are explicitly broken by the masses. There is a remnant scale invariance on the Coulomb branch, which manifests itself in the \mathcal{J} -invariant $\mathcal{B}_4 \times \mathbb{C}^4 \times \mathbb{H} \to \mathbb{C}$ of the curve being a quasi-homogeneous rational function of degree 0 and type (2, 1, 0),

$$\mathcal{J}(s^2 u, s \, \boldsymbol{m}, \tau_0) = \mathcal{J}(u, \boldsymbol{m}, \tau_0), \quad s \in \mathbb{C}^*.$$
(3.8)

Name	m	$oldsymbol{k}(oldsymbol{m})$	global symmetry
А	(m,m,0,0)	(2, 2, 2)	$SU(2) \times SU(2) \times SU(2) \times U(1)$
В	(m,m,m,m)	(4, 1, 1)	$SU(4) \times U(1)$
C	$\left(2m,0,0,0\right)$	(4, 1, 1)	$SU(4) \times U(1)$
D	(m,m,m,-m)	(4, 1, 1)	$SU(4) \times U(1)$
E	(m,m,μ,μ)	(2, 2, 1, 1)	$SU(2) \times SU(2) \times U(1) \times U(1)$
F	$(m+\mu, m-\mu, 0, 0)$	(2, 2, 1, 1)	$SU(2) \times SU(2) \times U(1) \times U(1)$
G	$(m,m,\mu,-\mu)$	(2, 2, 1, 1)	$SU(2) \times SU(2) \times U(1) \times U(1)$

Table 1: List of some mass cases with enhanced flavour symmetry in $N_f = 4$, with $\mu \neq m$. The vector $\mathbf{k}(\mathbf{m})$ lists the multiplicities of all singularities on the Coulomb branch \mathcal{B}_4 with mass \mathbf{m} .

The $N_f = 4$ theory has an SO(8) flavour symmetry, which becomes the universal double cover Spin(8) in the quantum theory. In particular, there exists a short exact sequence

$$1 \to \mathbb{Z}_2 \to \operatorname{Spin}(8) \to \operatorname{SO}(8) \to 1$$
 (3.9)

of Lie groups. The cover Spin(8) has an order 6 group Out(Spin(8)) of outer automorphisms, which is isomorphic to S_3 [181, 182].¹⁷

This group of outer automorphisms acts on the $N_f = 4$ theory as follows. The states with $(n_m, n_e) = (0, 1)$ are the elementary hypermultiplets, which transform in the fundamental vector representation of Spin(8). The magnetic monopole (1, 0) transforms as one spinor representation, and the dyon (1, 1)transforms as the conjugate spinor representation [46]. By an accidental isomorphism, these three representations are all 8-dimensional and irreducible, and they are permuted by the outer automorphism group $Out(Spin(8)) \cong S_3$. It is generated by

which act on the column vector $\mathbf{m} \in \mathcal{M} \coloneqq \mathbb{C}^4$ from the left [46,183,184]. The map \mathcal{T} exchanges the two spinors keeping the vector fixed, while \mathcal{S} exchanges the vector with the spinor, keeping the conjugate spinor fixed. This is depicted in Fig. 21.

The generators (3.10) satisfy the algebra

$$\mathcal{T}^2 = \mathcal{S}^2 = (\mathcal{S}\mathcal{T})^3 = \mathcal{S}\mathcal{T}^2\mathcal{S} = \mathbb{1}, \qquad (3.11)$$

¹⁷For any Lie group G, there are three associated groups. Aut(G) is the Lie group consisting of all automorphisms of G (i.e. group isomorphisms $G \to G$), Inn(G) is a normal subgroup of Aut(G) consisting of inner automorphisms given by $\alpha_g(h) \coloneqq ghg^{-1}$ for any $g \in G$, and Out(G) = Aut(G)/Inn(G) is the quotient group. The automorphism group of Spin(8) is Aut(SO(8)) = PSO(8) $\rtimes S_3$ [181].



Figure 21: Dynkin diagram of $\mathfrak{d}_4 = \text{Lie}(\text{Spin}(8))$. The group $\mathscr{T} \cong S_3$ of outer isomorphisms acts by permutations on the three conjugacy classes of irreducible representations v, s and \bar{s} attached to the nodes of the diagram. The 28-dimensional adjoint representation is left invariant by \mathscr{T} .

which is a presentation of the symmetric group S_3 . Since $\mathcal{T}^{\mathsf{T}}\mathcal{T} = \mathcal{S}^{\mathsf{T}}\mathcal{S} = \mathbb{1}$ but det $\mathcal{T} = \det \mathcal{S} = -1$, the matrices \mathcal{T} and \mathcal{S} generate a subgroup

$$\mathscr{T} = \langle \mathcal{T}, \mathcal{S} \rangle \tag{3.12}$$

of the orthogonal group O(4, \mathbb{C}), isomorphic to S_3 .¹⁸ As a consequence, they leave the inner product $[m_1^2]$ (3.3) invariant.

The flavour symmetry mixes with the $SL(2,\mathbb{Z})$ -symmetry acting on the UV-coupling τ_0 in an interesting way. To see this, notice that the reduction $\mathbb{Z} \to \mathbb{Z}_2$ modulo 2 induces a homomorphism $SL(2,\mathbb{Z}) \to SL(2,\mathbb{Z}/2\mathbb{Z})$. Since $SL(2,\mathbb{Z}/2\mathbb{Z}) \cong S_3$ are isomorphic, by transitivity we have a group homomorphism

$$\varphi: \mathrm{SL}(2,\mathbb{Z}) \longrightarrow \mathrm{Out}(\mathrm{Spin}(8)). \tag{3.13}$$

The full symmetry group of the $N_f = 4$ theory is the semidirect product [46]¹⁹

$$\mathfrak{T} \coloneqq \operatorname{Spin}(8) \rtimes_{\varphi} \operatorname{SL}(2, \mathbb{Z}). \tag{3.14}$$

The group (\mathfrak{T}, \bullet) consists of elements $(A, \gamma) \in \text{Spin}(8) \times \text{SL}(2, \mathbb{Z})$, with group operation

$$(A,\gamma) \bullet (\tilde{A},\tilde{\gamma}) \coloneqq (A\,\varphi(\gamma)(\tilde{A}),\gamma \circ \tilde{\gamma}). \tag{3.15}$$

The action of (3.10) is thus accompanied with an action of $SL(2, \mathbb{Z})$ on τ and τ_0 . From (3.11) we find that \mathcal{T}^2 and $\mathcal{ST}^2\mathcal{S}$ leave any mass configuration invariant. This implies that the theory should also be invariant under the simultaneous action of T^2 and ST^2S on the two couplings. These two matrices in $SL(2,\mathbb{Z})$ generate the principal congruence subgroup $\Gamma(2)$. From this it is also clear that

$$\operatorname{SL}(2,\mathbb{Z})/\Gamma(2) = \{I, T, S, TS, ST, TST\} \cong S_3,$$
(3.16)

which is another way to see that the group of outer isomorphisms is S_3 [184]. This action is depicted in Fig. 22. The subgroup $\Gamma(2)$ is the kernel of the

¹⁸They actually form a subgroup of $O(4, \mathbb{Q})$, but act on $\boldsymbol{m} \in \mathbb{C}^4$.

¹⁹Recall that for two groups G and H, a group homomorphism $\varphi : G \to \operatorname{Aut}(H)$ defines a semi-direct product $H \rtimes_{\varphi} G \subset H \times G$ with the multiplication $(h_1, g_1)(h_2, g_2) := (h_1\varphi(g_1)(h_2), g_1g_2)$. For $(h, g) \in H \rtimes_{\varphi} G$, the inverse is found as $(\varphi(g^{-1})(h^{-1}), g^{-1})$.



Figure 22: Action of $SL(2,\mathbb{Z})$ on $SL(2,\mathbb{Z})/\Gamma(2) \cong S_3$

above group homomorphism $SL(2,\mathbb{Z}) \to SL(2,\mathbb{Z}/2\mathbb{Z})$, such that it is in fact a normal subgroup $\Gamma(2) \triangleleft SL(2,\mathbb{Z})$.

The moduli spaces of the cases A–G of Table 1 are related by \mathscr{T} in the following way. We have that m_A, m_C, m_F are invariant under \mathcal{T} . Case A is invariant under both \mathcal{T} and \mathcal{S} . The \mathcal{S} -transformation relates cases B and C, as well as E and F, while leaving cases D and G invariant. We depict the relation among cases B, C and D in Fig. 23. For the cases E, F and G, there is an analogous diagram. An instance of these relations is that the weights of



Figure 23: Relation among $m_{\rm B} = (m, m, m, m)$, $m_{\rm C} = (2m, 0, 0, 0)$ and $m_{\rm D} = (m, m, m, -m)$.

the singular structure on the Coulomb branch are invariant under those spaces that are related by triality,

$$\boldsymbol{k}(\mathscr{T}\boldsymbol{m}) = \boldsymbol{k}(\boldsymbol{m}). \tag{3.17}$$

Using the action of the SO(8) flavour group, a large range of masses with equivalent duality diagrams can be reached. For example, the mass $\boldsymbol{m} = (2m, 0, 0, 0)$ is related to $\boldsymbol{m} = (0, 0, 0, 2m)$ by an SO(8) rotation. The first one is invariant under \mathcal{T} while the second one is not. The orbit under \mathcal{T} and \mathcal{S} for the case $\boldsymbol{m} = \boldsymbol{m}_{\rm B} = (2m, 0, 0, 0)$ is, as we have just discussed, given by Fig. 23, while that of $\boldsymbol{m} = (0, 0, 0, 2m)$ is given in Fig. 24. We see that it is of order six, and includes different relative signs compared to $\boldsymbol{m}_{\rm A}$ and $\boldsymbol{m}_{\rm D}$. On closer inspection, we note that the mass vectors come in pairs differing by an overall sign, which is an element of SO(8). Thus identifying the mass vectors related by SO(8) in diagram 24, we find that it is equivalent to diagram 23.



Figure 24: Orbit of the mass vector $\boldsymbol{m} = (0, 0, 0, 2m)$ under \mathcal{T} and \mathcal{S} .

3.2.3 Group action

The action

$$\begin{array}{ccc} \mathscr{T} \times \mathscr{M} \longrightarrow \mathscr{M} \\ (g, \boldsymbol{m}) \longmapsto g \cdot \boldsymbol{m} \end{array} \tag{3.18}$$

of the triality group \mathscr{T} on mass space \mathscr{M} can be studied in great detail. It is easy to check that the action is faithful²⁰, but neither free²¹ nor transitive²².

Up to conjugation, $S_3 \cong \mathscr{T}$ has four subgroups. They are: the trivial group \mathbb{Z}_1 , the symmetric group $S_2 \cong \mathbb{Z}_2$, the alternating group $A_3 \cong \mathbb{Z}_3$, and S_3 itself. They have order 1, 2, 3, and 6, respectively. All three proper subgroups are abelian. For a given \boldsymbol{m} , triality thus not always acts by the full S_3 but rather by a subgroup. For every $\boldsymbol{m} \in \mathscr{M}$ we can study the orbit $\mathscr{T} \cdot \boldsymbol{m} = \{g \cdot \boldsymbol{m} \mid g \in \mathscr{T}\}$. The sets of orbits of \mathscr{M} then give a partition of \mathscr{M} under the action (3.18).

First, notice that since \mathscr{T} is a finite group, all elements have finite order. In particular, $\mathcal{T}^2 = \mathscr{S}^2 = (\mathcal{T}\mathcal{S}\mathcal{T})^2 = \mathbb{1}$ and $(\mathscr{S}\mathcal{T})^3 = (\mathcal{T}\mathcal{S})^3 = \mathbb{1}$. The stabiliser subgroup of a mass $\mathbf{m} \in \mathscr{M}$ is defined as $\mathscr{T}_{\mathbf{m}} = \{g \in \mathscr{T} \mid g \cdot \mathbf{m} = \mathbf{m}\}$. By the orbit-stabiliser theorem

$$|\mathscr{T} \cdot \boldsymbol{m}| = |\mathscr{T}| / |\mathscr{T}_{\boldsymbol{m}}|, \qquad (3.19)$$

it suffices to study the fixed point equations in order to identify the stabiliser subgroups $\{\mathbb{Z}_1, S_2, A_3, S_3\}$ with the subgroups of \mathscr{T} . It is straightforward to identify the fixed point loci

$$\mathcal{L}_{\mathcal{T}} = \{ \boldsymbol{m} \in \mathscr{M} \mid m_4 = 0 \},$$

$$\mathcal{L}_{\mathcal{S}} = \{ \boldsymbol{m} \in \mathscr{M} \mid m_1 = m_2 + m_3 + m_4 \},$$

$$\mathcal{L}_{\mathcal{STS}} = \{ \boldsymbol{m} \in \mathscr{M} \mid m_1 = m_2 + m_3 - m_4 \},$$

$$\mathcal{L}_{\mathcal{ST}} = \mathcal{L}_{\mathcal{TS}} = \{ \boldsymbol{m} \in \mathscr{M} \mid m_1 = m_2 + m_3 \text{ and } m_4 = 0 \},$$
(3.20)

²⁰For every $g \neq h \in \mathscr{T}$ there exists an $m \in \mathscr{M}$ such that $g \cdot m \neq h \cdot m$.

²¹A group action is free if it has no fixed points, but $\boldsymbol{m} = 0$ is a fixed point for any $g \in \mathscr{T}$. ²²For each pair $\boldsymbol{m}, \tilde{\boldsymbol{m}} \in \mathscr{M}$ there exists $g \in \mathscr{T}$ such that a $g \cdot \boldsymbol{m} = \tilde{\boldsymbol{m}}$. A counterexample would be $\boldsymbol{m} = 0$ and $\tilde{\boldsymbol{m}} \neq 0$.

where $\mathcal{L}_g = \{ \boldsymbol{m} \in \mathscr{M} \mid g \cdot \boldsymbol{m} = \boldsymbol{m} \}$. For \boldsymbol{m} in precisely one of $\mathcal{L}_{\mathcal{T}}, \mathcal{L}_{\mathcal{S}}$ or $\mathcal{L}_{\mathcal{STS}}$, one finds that $|\mathscr{T} \cdot \boldsymbol{m}| = 3$. From (3.19) it then follows that $|\mathscr{T}_{\boldsymbol{m}}| = 2$, such that $\mathscr{T}_{\boldsymbol{m}} \cong S_2$. In fact, since \mathcal{T}, \mathcal{S} and \mathcal{STS} are all order 2 elements of \mathscr{T} , the stabiliser groups $\mathscr{T}_{\boldsymbol{m}}$ for \boldsymbol{m} in either of the three loci are precisely the three order 2 conjugate subgroups of $\mathscr{T} \cong S_3$.²³

The intersection

$$\mathcal{L}_1 = \mathcal{L}_{\mathcal{T}} \cap \mathcal{L}_{\mathcal{S}} = \{ \boldsymbol{m} \in \mathscr{M} \mid m_1 = m_2 + m_3 \text{ and } m_4 = 0 \}$$
(3.21)

is the locus of triality invariant masses, $\mathscr{T} \cdot \mathbf{m} = \mathbf{m}$. Thus, according to (3.19) we have $|\mathscr{T}_{\mathbf{m}}| = 6$ for such masses, such that indeed $\mathscr{T}_{\mathbf{m}} = \mathscr{T}$. For the last locus in (3.20), we see immediately that $\mathcal{L}_{S\mathcal{T}} = \mathcal{L}_{\mathcal{T}S} = \mathcal{L}_{\mathcal{T}} \cap \mathcal{L}_{S}$ contains precisely the invariant masses. Therefore, if \mathbf{m} is kept fixed by either \mathcal{TS} or $S\mathcal{T}$ then it is also fixed by both \mathcal{T} and \mathcal{S} and therefore by all of \mathscr{T} . Since $S\mathcal{T}$ and \mathcal{TS} are the only elements of \mathscr{T} of order 3, there is actually no mass \mathbf{m} such that $\mathscr{T} \cdot \mathbf{m}$ has 2 elements, and so there is no stabiliser subgroup isomorphic to A_3 . By case analysis, it is also easy to prove that the set $\mathscr{T} \cdot \mathbf{m}$ has 1, 3 or 6 elements.

Let us summarise. If $m \in \mathcal{L}_1$, it is invariant under \mathscr{T} . If m is in any of $\mathcal{L}_{\mathcal{T}}$, $\mathcal{L}_{\mathcal{S}}$ or $\mathcal{L}_{\mathcal{STS}}$, it could be in the intersection of any two of them. These intersections are however all equal to \mathcal{L}_1 , which is of course because any two elements of $\{\mathcal{T}, \mathcal{S}, \mathcal{STS}\}$ generate \mathscr{T} . This is depicted in Fig. 25.



Figure 25: The loci (3.20) with nontrivial stabiliser groups on the subspace $m_2 = m_3 = 0$ in \mathcal{M} . They all mutually intersect in the locus \mathcal{L}_1 of triality invariant masses.

If \boldsymbol{m} is then an element of

$$\mathcal{L}_3 = \mathcal{L}_{\mathcal{T}} \cup \mathcal{L}_{\mathcal{S}} \cup \mathcal{L}_{\mathcal{STS}} \setminus \mathcal{L}_1, \qquad (3.22)$$

then the stabiliser group of \boldsymbol{m} is isomorphic to S_2 . If \boldsymbol{m} does not lie in either \mathcal{L}_1 or \mathcal{L}_3 , then there is no remaining symmetry. It lies in

$$\mathcal{L}_6 = \mathscr{M} \setminus \mathcal{L}_1 \cup \mathcal{L}_3, \tag{3.23}$$

²³If we represent S_3 in cycle notation of permutations of $\{1, 2, 3\}$, the three order 2 conjugate subgroups of S_3 are $\{(), (1, 2)\}, \{(), (1, 3)\}$ and $\{(), (2, 3)\}$.

and its stabiliser group is trivial.

3.3 Order parameters and bimodular forms

For the $N_f = 4$ SW theory, there are other curves than the one introduced by Seiberg and Witten [20,21,111,113,148,179,185–187]. In this section, we focus on the modularity of the original SW curve (3.4). We proceed by studying the mass configurations with the largest flavour symmetry groups, A, B, C and D. In all these cases, u is a weight (0, 2) bimodular form, which we define below, for a triple of groups related to the duality group of the decoupling theory where the mass of the hypermultiplets is infinitely large, and the stabiliser group of the mass under the triality action. Since case A is triality invariant, u in that case transforms under the full $SL(2, \mathbb{Z})$ group. The other cases B, C and D are permuted by triality, and furnish a vector-valued bimodular form.

The massless case where $\mathbf{m}_0 = (0, 0, 0, 0)$ is very simple, as $j(\tau) = \mathcal{J}(u, 0, \tau_0)$ = $j(\tau_0)$ and therefore

$$\tau(u) = \tau_0 \tag{3.24}$$

is constant over the whole Coulomb branch $\mathcal{B}_4 \ni u$. In other words, the coupling τ is fixed and thus does not run, which is a consequence of the massless $N_f = 4$ theory being exactly superconformal. There are six singularities, which all sit at the origin u = 0 and form the non-abelian Coulomb point with a five quaternionic-dimensional Higgs branch [105].

3.3.1 Case A

For the mass $\mathbf{m}_{\rm A} = (m, m, 0, 0)$, this allows to express u as a rational function in Jacobi theta functions of τ_0 and τ . There are in fact six solutions to the correspondence $\mathcal{J}(u(\tau), \mathbf{m}, \tau_0) = j(\tau)$. A consistent way of choosing which solution to use, which we will employ throughout, is to take the one that has the right decoupling limit when decoupling the massive hypermultiplets, i.e., the one that decouples to the order parameter of massless $N_f = 2$, Eq. (2.86).

In view of the more complicated mass cases, we can further simplify the rather lengthy expression. The dependence on τ is in fact only through $\lambda = \frac{\vartheta_2^4}{\vartheta_3^4}$. This is not quite true for τ_0 , for which u has weight 2 [148]. This weight factor can be extracted by eliminating $\vartheta_4(\tau_0)$ through the Jacobi identity (A.12) and $\vartheta_2(\tau_0)$ through the definition of $\lambda(\tau_0)$. This gives

$$u_{\rm A}(\tau,\tau_0) = -\frac{m^2}{3}\vartheta_3(\tau_0)^4 \frac{\lambda(\tau_0)^2 + 2(\lambda(\tau) - 1)\lambda(\tau_0) - \lambda(\tau)}{\lambda(\tau_0) - \lambda(\tau)}.$$
 (3.25)

The simple mass dependence of u_A is a consequence of the scaling symmetry (3.8). The second prefactor $\vartheta_3(\tau_0)^4$ gives the weight 2. The remaining quotient is written in a manifestly invariant fashion. Let us denote by Γ_{τ} (Γ_{τ_0}) a group acting by linear fractional transformations on τ (τ_0). As $\vartheta_3(\tau_0)^4$ is a modular form of weight 2 and $\lambda(\tau_0)$ a modular function (of weight 0) for $\Gamma(2)_{\tau_0}$, one can easily see that $u_{\rm A}(\tau, \tau_0)$ is a weight 2 modular form for $\Gamma(2)_{\tau_0}$ for fixed τ , and a modular function for $\Gamma(2)_{\tau}$ for fixed τ_0 . We thus have that

$$u_{\mathcal{A}}(\gamma_{1}\tau,\gamma_{2}\tau_{0}) = (c_{2}\tau_{0} + d_{2})^{2}u_{\mathcal{A}}(\tau,\tau_{0}), \quad \gamma_{i} = \begin{pmatrix} a_{i} & b_{i} \\ c_{i} & d_{i} \end{pmatrix} \in \Gamma(2)$$
(3.26)

for i = 1, 2. We call u_A modular for $\Gamma(2)_{\tau} \times \Gamma(2)_{\tau_0}$, where the occurrence of two groups indicates that they act on both variables τ and τ_0 separately.

The mass $\boldsymbol{m}_{\mathrm{A}}$ is invariant under the triality group (3.12). As triality acts on τ and τ_0 together, this suggests that u_{A} transforms under a simultaneous transformation of SL(2, \mathbb{Z}). Indeed, if one acts simultaneously on τ and τ_0 with SL(2, \mathbb{Z}), it is easy to check from $T : \lambda \mapsto \frac{\lambda}{\lambda-1}$ and $S : \lambda \mapsto 1 - \lambda$ that $u_{\mathrm{A}}(\tau, \tau_0)$ transforms as

$$u_{\mathcal{A}}(\gamma\tau,\gamma\tau_0) = (c\tau_0 + d)^2 u_{\mathcal{A}}(\tau,\tau_0), \quad \gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{SL}(2,\mathbb{Z}).$$
(3.27)

We call u_A modular for $SL(2,\mathbb{Z})_{(\tau,\tau_0)}$, where the notation indicates that the single group $SL(2,\mathbb{Z})$ acts on both τ and τ_0 simultaneously. The two transformations (3.26) and (3.27) are characteristic properties for functions known as "bimodular forms" [175–177]. For our application to $N_f = 4$ SQCD, we will adopt the following definition:

Definition 5 (Bimodular form). Let $(\Gamma_1, \Gamma_2; \Gamma)$ be a triple of subgroups of $SL(2, \mathbb{R})$ commensurable with $SL(2, \mathbb{Z})$.²⁴ A two-variable meromorphic function $F : \mathbb{H} \times \mathbb{H} \to \mathbb{C}$ is called a bimodular form of weight (k_1, k_2) for the triple $(\Gamma_1, \Gamma_2; \Gamma)$ if it satisfies both Condition 1 & 2:

• Condition 1: For all $\gamma_i = \begin{pmatrix} a_i & b_i \\ c_i & d_i \end{pmatrix} \in \Gamma_i$, i = 1, 2, F transforms as

$$F(\gamma_1\tau_1,\gamma_2\tau_2) = \chi(\gamma_1,\gamma_2) \left(c_1\tau_1 + d_1\right)^{k_1} (c_2\tau_2 + d_2)^{k_2} F(\tau_1,\tau_2), \quad (3.28)$$

for a certain multiplier $\chi : \Gamma_1 \times \Gamma_2 \to \mathbb{C}^*$. We call this the <u>separate</u> transformation of F under (Γ_1, Γ_2) , and denote it by $(\Gamma_1)_{\tau_1} \times (\overline{\Gamma_2})_{\tau_2}$.

• Condition 2: For all $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma$, F transforms as

$$F(\gamma \tau_1, \gamma \tau_2) = \phi(\gamma) \left(c\tau_1 + d \right)^{k_1} (c\tau_2 + d)^{k_2} F(\tau_1, \tau_2), \tag{3.29}$$

for a multiplier $\phi : \Gamma \to \mathbb{C}^*$. We call this the <u>simultaneous</u> transformation of F under Γ , and denote it by $\Gamma_{(\tau_1, \tau_2)}$.

Note that condition 2 follows from condition 1 if Γ is the intersection of Γ_1 and Γ_2 , $\Gamma = \Gamma_1 \cap \Gamma_2$ with $\phi(\gamma) = \chi(\gamma, \gamma), \gamma \in \Gamma$.

²⁴A subgroup $\Gamma \subset SL(2,\mathbb{R})$ is commensurable with $SL(2,\mathbb{Z})$ if $\Gamma \cap SL(2,\mathbb{Z})$ has finite index in both $SL(2,\mathbb{Z})$ and $SL(2,\mathbb{R})$. This includes in particular all congruence subgroups of $SL(2,\mathbb{Z})$.

This definition contains the main aspects of other definitions of bimodular forms in the literature [85, 175-177].

The definition above for the triple $(\Gamma_1, \Gamma_2; \Gamma_1 \cap \Gamma_2)$ is equivalent to the definition in [176]. For the triple $(\Gamma_1, \Gamma_1; SL(2, \mathbb{Z}))$, our definition is equivalent to the one of [85]. Finally, for $k_1 = k_2$ and the triple $(\Gamma_1, \Gamma_1; \Gamma)$, our definition is equivalent with [177]. Finally, the definition of Stienstra and Zagier [175], as cited in [176], does require Condition 2 without requiring Condition 1.

From definition 5, we find that $u_{\rm A} : \mathbb{H} \times \mathbb{H} \to \mathbb{C}$ in (3.25) is a bimodular form of weight (0,2) for the triple

$$(\Gamma(2), \Gamma(2); \operatorname{SL}(2, \mathbb{Z})), \tag{3.30}$$

with trivial multipliers χ and ϕ . In fact, $m \mapsto u_A$ is a 1-parameter family of such bimodular forms.

The function (3.25) can be easily expanded in either $q = e^{2\pi i \tau}$ or $q_0 = e^{2\pi i \tau_0}$. When expanding u_A around $q_0 = 0$, every coefficient is a modular function for $\Gamma(2)_{\tau}$. If we denote the vector space of holomorphic modular forms of weight k for $\Gamma \subseteq \mathrm{SL}(2,\mathbb{Z})$ by $\mathcal{M}_k(\Gamma)$, then $u_A \in \mathcal{M}_0(\Gamma(2))[\![q_0^{\frac{1}{4}}]\!]$. Conversely, we have that $u_A \in \mathcal{M}_2(\Gamma(2))[\![q^{\frac{1}{4}}]\!]$.

Recall that $\Gamma(2)$ is a genus zero congruence subgroup. As such, its Hauptmodul λ is the single transcendental generator of the function field of $\Gamma(2) \setminus \mathbb{H}^*$. Since u_A is modular in τ as well as τ_0 for $\Gamma(2)$ and no larger subgroup of $SL(2,\mathbb{Z})$, the transcendence of λ then implies that (3.25) cannot be simplified further.

The Coulomb branch \mathcal{B}_4 for the mass $\boldsymbol{m} = \boldsymbol{m}_A$ has six singularities that come in three pairs of two. By expanding $\lambda(\tau)$ around the cusps, one easily finds

$$u_{A}(\frac{1}{2},\tau_{0}) = -\frac{m^{2}}{3}\vartheta_{3}(\tau_{0})^{4}(\lambda(\tau_{0})-2),$$

$$u_{A}(0,\tau_{0}) = -\frac{m^{2}}{3}\vartheta_{3}(\tau_{0})^{4}(\lambda(\tau_{0})+1),$$

$$u_{A}(1,\tau_{0}) = -\frac{m^{2}}{3}\vartheta_{3}(\tau_{0})^{4}(-2\lambda(\tau_{0})+1).$$

(3.31)

Notice that the singularities are holomorphic modular forms of weight 2 for $\Gamma(2)_{\tau_0}$, and are permuted by elements of $(\mathrm{SL}(2,\mathbb{Z}))/\Gamma(2))_{\tau_0}$. The reason for $u_{\mathrm{A}}(\frac{1}{2},\tau_0) = u_{\mathrm{A}}(i\infty,\tau_0)$ is explained in section 3.4.2. Since λ is a Hauptmodul for $\Gamma(2)$, for given $\tau_0 \in \Gamma(2) \setminus \mathbb{H}$ there is exactly one $\tau \in \Gamma(2) \setminus \mathbb{H}$ where u has a pole. It is where τ approaches τ_0 , $u_{\mathrm{A}}(\tau_0,\tau_0) = \infty$.

We can furthermore compute the period $\frac{da}{du}$. Actually, $\frac{da}{du}$ is not invariant under the monodromy around ∞ , but multiplied by -1. Instead of $\frac{da}{du}$, we may consider $\left(\frac{da}{du}\right)^2$, which is monodromy invariant [2, 146]. In the pure $(N_f = 0)$ SU(2) case, it is a modular form of weight 2 for $\Gamma^0(4)$. The weight is the same in $N_f = 4$, however it also transforms well under fractional linear transformations of τ_0 . More specifically, we find that

$$\left(\frac{da}{du}\right)_{A}^{2}(\tau,\tau_{0}) = \frac{1}{8m^{2}}\frac{\vartheta_{3}(\tau)^{4}}{\vartheta_{3}(\tau_{0})^{8}}\frac{\lambda(\tau) - \lambda(\tau_{0})}{\lambda(\tau_{0})(\lambda(\tau_{0}) - 1)}.$$
(3.32)

The normalisation may be checked from the fact that $\frac{da}{du} \sim \frac{1}{\sqrt{8u}}$ for $u \to \infty$, which due to (3.25) corresponds to $\tau \to \tau_0$. Since ϑ_3^4 is a modular form of weight 2 for $\Gamma(2)$, it follows that $\left(\frac{da}{du}\right)_A^2$ satisfies condition 1 of definition 5 with weight (2, -4) for $\Gamma(2)_{\tau} \times \Gamma(2)_{\tau_0}$. As \mathbf{m}_A is left invariant by triality, $\left(\frac{da}{du}\right)_A^2$ must also be modular for SL(2, Z). Indeed, one easily finds that $\left(\frac{da}{du}\right)_A^2$ also satisfies condition 2, such that it is a bimodular form of weight (2, -4) for the triple $(\Gamma(2), \Gamma(2); SL(2, Z))$.

We can also compute the physical discriminant, which for the case A reads

$$\Delta_{\rm A} = (u - u_{\rm A}(\frac{1}{2}, \tau_0))^2 (u - u_{\rm A}(0, \tau_0))^2 (u - u_{\rm A}(1, \tau_0))^2.$$
(3.33)

Since the singularities (3.31) themselves are modular forms for τ_0 , it is again a bimodular form. One easily computes

$$\Delta_{\rm A}(\tau,\tau_0) = m^{12} \vartheta_3(\tau_0)^{24} \frac{\lambda(\tau)^2 (\lambda(\tau) - 1)^2 \lambda(\tau_0)^4 (\lambda(\tau_0) - 1)^4}{(\lambda(\tau) - \lambda(\tau_0))^6}.$$
 (3.34)

As ϑ_3^{24} is a modular form of weight 12 for $\Gamma(2)$, this shows that Δ_A has modular weight (0, 12) under $\Gamma(2)_{\tau} \times \Gamma(2)_{\tau_0}$. With the same reasoning as above, we find that Δ_A is a bimodular form of weight (0, 12) for the same triple (3.30).

The SW curve for the $\mathcal{N} = 2^*$ theory with mass m is identical to the one of $N_f = 4$ (3.4) with $\frac{1}{2} \mathbf{m}_A = (\frac{m}{2}, \frac{m}{2}, 0, 0)$ [46]. A reason for this is that both theories have three singularities each with monodromy being conjugate to T^2 . This was in fact the ansatz of Seiberg and Witten to determine the curve with generic masses. This allows to easily find the order parameter for $\mathcal{N} = 2^*$,

$$u_{\mathcal{N}=2^*}(\tau,\tau_0) = \frac{1}{4} u_{\mathcal{A}}(\tau,\tau_0).$$
(3.35)

In particular, it is a bimodular form of weight (0, 2) for $(\Gamma(2), \Gamma(2); SL(2, \mathbb{Z}))$. The derivative $\frac{da}{du}$ also only receives an overall normalisation from $N_f = 4$. In $\mathcal{N} = 2^*$ the singularities each have degeneracy 1 and not 2 as in $N_f = 4$ case A. Therefore, we have that $\Delta_{\mathcal{N}=2^*} = \sqrt{\Delta_A}$, which is a polynomial of degree 3 in u and a bimodular form of weight (0, 6) [85].

The expressions for $u_{\mathcal{N}=2^*}$ in the literature [85,148,173,188,189] are related to $u_A(\tau,\tau_0)$ by a transformation in (3.16), which corresponds to the choice of a different solution of the sextic equation associated with the $\mathcal{N}=2^*$ theory [2]. The different choices can be absorbed in the double scaling limit. The counting of the number of poles of u_A is immediate from our expression (3.25) (see comment on the transcendence of λ in section 3.3.1).

3.3.2 Case B

The equal mass case $\mathbf{m}_{\rm B} = (m, m, m, m)$ can be treated with the same technique as in the previous subsection. Since $N_f = 4$ with four equal masses flows to $N_f = 0$ for $m \to \infty$, we can express the τ dependence through the Hauptmodul $f \coloneqq \frac{\vartheta_2^4 + \vartheta_3^4}{\vartheta_2^2 \vartheta_3^2}$ of $\Gamma^0(4)$. In fact, the $N_f = 0$ order parameter (1.35) is just $\frac{u}{\Lambda_0^2} = -\frac{1}{2}f$. The order parameter $u_{\rm B}$ reads

$$u_{\rm B}(\tau,\tau_0) = -\frac{m^2}{3}\vartheta_2(\tau_0)^2\vartheta_3(\tau_0)^2\frac{2f(\tau_0)^2 + f(\tau)f(\tau_0) - 12}{f(\tau_0) - f(\tau)},$$
(3.36)

which thus does not involve ϑ_4 . Since $\vartheta_2(\tau_0)^2 \vartheta_3(\tau_0)^2$ is a holomorphic modular form of weight 2 for $\Gamma^0(4)_{\tau_0}$, we find that $u_{\rm B}(\tau, \tau_0)$ has bimodular weight (0, 2)for the separate transformations under $\Gamma^0(4)_{\tau} \times \Gamma^0(4)_{\tau_0}$.

As $\mathcal{T} : \mathbf{m}_{\mathrm{B}} \mapsto \mathbf{m}_{\mathrm{D}}$, there is no simultaneous action of T on τ and τ_{0} leaving u_{B} invariant. Also, since $\mathcal{S} : \mathbf{m}_{\mathrm{B}} \mapsto \mathbf{m}_{\mathrm{C}}$, S does not leave u_{B} invariant. However, a subgroup $\mathscr{T}_{\mathbf{m}_{\mathrm{B}}}$ of \mathscr{T} leaves \mathbf{m}_{B} invariant: Out of the six elements of \mathscr{T} , \mathbf{m}_{B} is left invariant by 1 and \mathcal{TST} . As the action of \mathscr{T} is combined in (3.14) with a simultaneous action on τ and τ_{0} , we find that u_{B} is expected to be invariant under a simultaneous transformation of $TST \in \mathrm{SL}(2,\mathbb{Z})$. However, due to the algebra (3.11), the same holds for T^{2} . These two matrices generate the congruence subgroup $\Gamma^{0}(2)$ of $\mathrm{SL}(2,\mathbb{Z})$. It is straightforward to check from the explicit expression (3.36) that u_{B} transforms with weight (0, 2) under a simultaneous transformation on τ and τ_{0} of $\Gamma^{0}(2)_{(\tau,\tau_{0})}$. This proves that u_{B} is an example of a bimodular form of weight (0, 2) for the triple

$$(\Gamma^{0}(4), \Gamma^{0}(4); \Gamma^{0}(2)). \tag{3.37}$$

As classified in section 3.2.3, the stabiliser subgroup $\mathscr{T}_{m_{\mathrm{B}}} = \{\mathbb{1}, \mathcal{TST}\}$ for the mass m_{B} is isomorphic to the group $S_2 \cong \mathbb{Z}_2$ of order 2. This agrees with the fact that $\mathrm{SL}(2,\mathbb{Z})/\Gamma^0(2) \cong S_2$.

The singularities are

$$u_{\rm B}(1,\tau_0) = -\frac{m^2}{3}\vartheta_2(\tau_0)^2\vartheta_3(\tau_0)^2(-f(\tau_0)),$$

$$u_{\rm B}(0,\tau_0) = -\frac{m^2}{3}\vartheta_2(\tau_0)^2\vartheta_3(\tau_0)^2(2f(\tau_0)+6),$$

$$u_{\rm B}(2,\tau_0) = -\frac{m^2}{3}\vartheta_2(\tau_0)^2\vartheta_3(\tau_0)^2(2f(\tau_0)-6),$$

(3.38)

which again are holomorphic modular forms of weight 2 for $\Gamma^0(4)_{\tau_0}$. Due to the duality group $\Gamma^0(4)_{\tau}$, we have that $u_B(1,\tau_0) = u_B(i\infty,\tau_0)$. This singularity has degeneracy 4, and flows to ∞ for $m \to \infty$. The singularity in the interior is $u_B(\tau_0,\tau_0) = \infty$. One can also check that the singularities (3.38) never collide: The conditions $u_B(1,\tau_0) = u_B(0,\tau_0)$ or $u_B(1,\tau_0) = u_B(2,\tau_0)$ are equivalent to $f(\tau_0) = \pm 2$, whose only solutions are the two cusps $\tau_0^+ = 0$ and $\tau_0^- = 2$ of $\Gamma(2)$. Since the SW curve is singular for those values of τ_0 , the singularities do not merge for any finite masses.

Similarly as before, one finds

$$\left(\frac{da}{du}\right)_{\rm B}^2(\tau,\tau_0) = \frac{1}{8m^2} \frac{\vartheta_2(\tau)^2 \vartheta_3(\tau)^2}{\vartheta_4(\tau_0)^8} \left(f(\tau) - f(\tau_0)\right). \tag{3.39}$$

Since f is a Hauptmodul, $\vartheta_2^2 \vartheta_3^2$ a modular form of weight 2 and ϑ_4^8 a modular form of weight 4 for $\Gamma^0(4)$, it follows that $\left(\frac{da}{du}\right)_{\rm B}^2$ is a bimodular form of weight (2, -4) for the triple (3.37). Finally, the physical discriminant reads

$$\Delta_{\rm B}(\tau,\tau_0) = m^{12}\vartheta_4(\tau_0)^{24} \frac{(f(\tau)^2 - 4)(f(\tau_0)^2 - 4)^2}{(f(\tau) - f(\tau_0))^6},\tag{3.40}$$

which is a bimodular form of weight (0, 12) for the triple (3.37).

3.3.3 Case C

Let us study the case where only one hypermultiplet is massive, that is, $\mathbf{m}_{\rm C} = (2m, 0, 0, 0)$. The particular normalisation of $\mathbf{m}_{\rm C}$ is chosen such that the diagram 23 holds without any prefactors. Since in the limit $m \to \infty$ we get massless $N_f = 3$, we can express the τ dependence through the Hauptmodul $\tilde{f} = \frac{\vartheta_3^2 \vartheta_4^2}{(\vartheta_3^2 - \vartheta_4^2)^2}$ of $\Gamma_0(4)$. The order parameter of the massless $N_f = 3$ theory reads $\frac{u}{\Lambda_3^2} = -\frac{1}{64}\tilde{f}$ (2.105), and the functions f and \tilde{f} are related by $f(4\tau) = 16\tilde{f}(\tau) + 2$. One finds for the order parameter $u_{\rm C}$,

$$u_{\rm C}(\tau,\tau_0) = -\frac{m^2}{3}\vartheta_3(\tau_0)^2\vartheta_4(\tau_0)^2\frac{2\tilde{f}(\tau_0)^2 + (10\tilde{f}(\tau) + 1)\tilde{f}(\tau_0) + 2\tilde{f}(\tau)}{\tilde{f}(\tau_0)(\tilde{f}(\tau_0) - \tilde{f}(\tau))}, \quad (3.41)$$

which is independent of $\vartheta_2(\tau_0)$. Again, the factor $\vartheta_3(\tau_0)^2 \vartheta_4(\tau_0)^2$ is a modular form of weight 2 for $\Gamma_0(4)_{\tau_0}$, and the quotient is a meromorphic modular function of $\Gamma_0(4)$ for both τ and τ_0 . Thus $u_{\rm C}$ satisfies condition 1 of definition 5 with weight (0, 2) for $\Gamma_0(4)_{\tau} \times \Gamma_0(4)_{\tau_0}$.

Since $\mathcal{T}: \boldsymbol{m}_{\mathrm{C}} \mapsto \boldsymbol{m}_{\mathrm{C}}$, there is a simultaneous *T*-duality

$$\mathcal{T}: u_{\rm C}(\tau+1, \tau_0+1) = u_{\rm C}(\tau, \tau_0), \qquad (3.42)$$

which is straightforward to check from (3.41). As $S : \mathbf{m}_{\mathrm{C}} \mapsto \mathbf{m}_{\mathrm{B}}$, this exchanges the order parameters

$$u_{\rm C}(-\frac{1}{\tau}, -\frac{1}{\tau_0}) = \tau_0^2 u_{\rm B}(\tau, \tau_0), \qquad (3.43)$$

which we can also explicitly check. We can again study the stabiliser subgroup of $\mathscr{T}_{m_{\rm C}}$ of \mathscr{T} . It is the group generated by \mathcal{T} and $\mathcal{ST}^2\mathcal{S}$, such that $u_{\rm C}$ is expected to transform simultaneously under T and ST^2S . These two matrices generate the congruence subgroup $\Gamma_0(2)$ of $\mathrm{SL}(2,\mathbb{Z})$, which is conjugate to $\Gamma^0(2)$. Thus we find that $u_{\rm C}$ is a bimodular form of weight (0,2) for the triple

$$(\Gamma_0(4), \Gamma_0(4); \Gamma_0(2)).$$
 (3.44)

Lastly, we can also study

$$\left(\frac{da}{du}\right)_{\mathcal{C}}^{2}(\tau,\tau_{0}) = \frac{1}{8m^{2}}\frac{\vartheta_{3}(\tau)^{2}\vartheta_{4}(\tau)^{2}}{\vartheta_{2}(\tau_{0})^{8}}\frac{\tilde{f}(\tau) - \tilde{f}(\tau_{0})}{\tilde{f}(\tau)\tilde{f}(\tau_{0})}.$$
(3.45)

It is straightforward to check that $\left(\frac{da}{du}\right)_{\rm C}^2$ is a bimodular form of weight (2, -4) for the triple (3.44). For the discriminant $\Delta_{\rm C}$ there exists a similar expression to (3.40), and it is a bimodular form of weight (0, 12) for (3.44).

3.3.4 Case D

Let us finally also study the case $\mathbf{m}_{\mathrm{D}} = (m, m, m, -m)$. It is related to cases B and C as in Fig. 23. We have that $\mathbf{m}_{\mathrm{D}} \in \mathcal{L}_{S}$, while $\mathbf{m}_{\mathrm{B}} \in \mathcal{L}_{STS}$ and $\mathbf{m}_{\mathrm{C}} \in \mathcal{L}_{T}$. From the SW curve one easily finds

$$u_{\rm D}(\tau,\tau_0) = -\frac{m^2}{3} i\vartheta_2(\tau_0)^2 \vartheta_4(\tau_0)^2 \frac{2\hat{f}(\tau_0)^2 + \hat{f}(\tau)\hat{f}(\tau_0) - 12}{\hat{f}(\tau_0) - \hat{f}(\tau)},\tag{3.46}$$

where

$$\hat{f}(\tau) = f(\tau+1) = i \frac{\vartheta_2(\tau)^4 - \vartheta_4(\tau)^4}{\vartheta_2(\tau)^2 \vartheta_4(\tau)^4}.$$
(3.47)

Since f is a Hauptmodul for $\Gamma^{0}(4)$, \hat{f} is a Hauptmodul for a subgroup of $SL(2,\mathbb{Z})$ conjugate to $\Gamma^{0}(4)$,

$$\widetilde{\Gamma^{0}(4)} = T \, \Gamma^{0}(4) \, T^{-1} = \langle T^{4}, ST^{2} \rangle.$$
(3.48)

A fundamental domain for $\widetilde{\Gamma^0(4)}$ is given by

$$\widetilde{\Gamma^{0}(4)} \setminus \mathbb{H} = \mathcal{F} \cup T\mathcal{F} \cup T^{2}\mathcal{F} \cup T^{3}\mathcal{F} \cup TS\mathcal{F} \cup T^{3}S\mathcal{F}, \qquad (3.49)$$

with $\mathcal{F} = \mathrm{SL}(2,\mathbb{Z}) \setminus \mathbb{H}$. It is straightforward to check that $u_{\mathrm{D}}(\tau,\tau_0)$ transforms with weight (0,2) under $\widetilde{\Gamma^0(4)}_{\tau} \times \widetilde{\Gamma^0(4)}_{\tau_0}$.

The subgroup $\mathscr{T}_{m_{\mathrm{D}}} \subset \mathscr{T}$ leaving invariant m_{D} is generated by \mathscr{S} and \mathcal{T}^2 . The two corresponding $\mathrm{SL}(2,\mathbb{Z})$ transformations S and T^2 generate the theta group $\widetilde{\Gamma^0(2)} \coloneqq \Gamma_{\theta}$ (A.2), which is a congruence subgroup of $\mathrm{SL}(2,\mathbb{Z})$ with index 3, conjugate to $\Gamma_0(2)$ and $\Gamma^0(2)$. Thus we find that u_{D} is a bimodular form of weight (0, 2) for the triple

$$(\widetilde{\Gamma^{0}(4)}, \widetilde{\Gamma^{0}(4)}; \widetilde{\Gamma^{0}(2)}).$$
(3.50)

The three groups $\{\Gamma_0(2), \Gamma^0(2), \overline{\Gamma^0(2)}\} \ni \Gamma$ are in fact the three groups $\mathrm{SL}(2, \mathbb{Z}) \supset \Gamma \supset \Gamma(2)$ with index 3 and 2 cusps, and they correspond to the three conjugate order 2 subgroups of S_3 .

3.4 Vector-valued bimodular forms

The analysis of the A, B, C and D theories may suggest that the order parameter $u_{\boldsymbol{m}}$ for a generic mass \boldsymbol{m} transforms with weight (0, 2) under $G_{\tau} \times G_{\tau_0}$ for some subgroup $G \subseteq \mathrm{SL}(2,\mathbb{Z})$. This is however not true in general, as for generic masses there are branch points and associated branch cuts, which spoil the modularity [2]. The discussion in section 2 for $N_f \leq 3$ suggests that for a fixed τ or fixed τ_0 , there is a natural choice of fundamental domain $\mathcal{F}(\boldsymbol{m}) \subseteq \mathbb{H}$ for $u_{\boldsymbol{m}}$, such that $u_{\boldsymbol{m}} : \mathcal{F}(\boldsymbol{m}) \to \mathcal{B}_4$ is one-to-one. For a generic choice of masses, monodromies on the u-plane give rise to monodromies of $\mathcal{F}(\boldsymbol{m})$, but these do not generate a congruence subgroup of $\mathrm{SL}(2,\mathbb{Z})$ for a generic mass. For special cases however, $\mathcal{F}(\boldsymbol{m})$ is equal to $\Gamma \setminus \mathbb{H}$ for some subgroup $\Gamma \subseteq \mathrm{SL}(2,\mathbb{Z})$, such as when \boldsymbol{m} is equal to $\boldsymbol{m}_{\mathrm{A}}$, $\boldsymbol{m}_{\mathrm{B}}$, $\boldsymbol{m}_{\mathrm{C}}$ or $\boldsymbol{m}_{\mathrm{D}}$, for which Γ is $\Gamma(2)$, $\Gamma^{0}(4)$, $\Gamma_{0}(4)$ or $\widetilde{\Gamma^{0}(4)}$. If the mass \boldsymbol{m} is such that \mathcal{B}_{4} contains a superconformal Argyres-Douglas point, $\Gamma \subseteq \mathrm{SL}(2,\mathbb{Z})$ can also be a subgroup of index smaller than 6 [2,3,129].

3.4.1 Generic masses

In the above discussed examples A–D, the duality groups Γ_1 of τ and Γ_2 of τ_0 are identical. This is not generally true, even if $u(\tau, \tau_0)$ is modular in τ and τ_0 [3]. However, we can demonstrate that $\Gamma_1 \subset \Gamma_2$. A common nonperturbative definition of the UV coupling constant is the low-energy effective coupling τ in the limit where the order parameter is large,

$$\tau_0 = \lim_{u \to \infty} \tau(u). \tag{3.51}$$

Since it is not associated with a singularity, it is neither a cusp nor an elliptic point and therefore an arbitrary interior point in the space of $\tau \in \mathbb{H}$. If $\Gamma_1 \subsetneq \Gamma_2$ is not a proper subgroup, then in general $\tau_0 \in \Gamma_2 \setminus \mathbb{H}$ is not an element of a choice of fundamental domain $\Gamma_1 \setminus \mathbb{H}$. However, there exists a $\gamma_1 \in \Gamma_1$ with the property that $\gamma_1 \tau_0 \in \Gamma_1 \setminus \mathbb{H}$. Since $u(\tau, \tau_0)$ has weight 0 in τ , we notice that

$$u(\gamma_1 \tau_0, \tau_0) = u(\tau_0, \tau_0) = \infty,$$
 (3.52)

which is the weak coupling region in \mathcal{B}_4 . If $\Gamma_2 \subsetneq \Gamma_1$ however, then there exist two points $\tau_0 \neq \tilde{\tau}_0$ in the fundamental $\Gamma_1 \setminus \mathbb{H}$, which are not related by any element $\gamma_1 \in \Gamma_1$. Then $u(\tau, \tau_0)$ and $u(\tau, \tilde{\tau}_0)$ are two distinct points in \mathcal{B}_4 . This contradicts the fact that the $N_f = 4$ Coulomb branch \mathcal{B}_4 only contains one such singularity. This shows that indeed $\Gamma_1 \subseteq \Gamma_2$.

The weight (0, 2) of u can be explained as follows. Monodromies on the u-plane act on the low-energy effective coupling τ and by definition leave u invariant. Thus $u(\tau, \tau_0)$ is required to have weight 0 in τ . For τ_0 , recall that the order parameter relates to the prepotential F of the theory by a logarithmic derivative with respect to the instanton counting parameter [110, 130–132, 190, 191]

$$u = 4\pi i q_0 \frac{\partial F}{\partial q_0} = 2 \frac{\partial F}{\partial \tau_0}.$$
(3.53)

As the prepotential F has weight 0 in τ_0 , this shows that $u(\tau, \tau_0)$, has weight 2 in τ_0 .

The other possible modular transformations are those involving the masses, which is the action of the triality group $\text{Spin}(8) \rtimes_{\varphi} \text{SL}(2,\mathbb{Z})$. From the above analysis, we expect that for generic mass \boldsymbol{m} the order parameter $u_{\boldsymbol{m}}$ transforms as

$$\begin{aligned} \mathcal{T} : & u_{m}(\tau + 1, \tau_{0} + 1) = u_{\mathcal{T}m}(\tau, \tau_{0}), \\ \mathcal{S} : & u_{m}(-\frac{1}{\tau}, -\frac{1}{\tau_{0}}) = \tau_{0}^{2} u_{\mathcal{S}m}(\tau, \tau_{0}). \end{aligned}$$
 (3.54)

Due to the branch points and cuts for generic masses, these transformations are again very subtle to perform. From (3.11) and in particular $\mathcal{T}^2 = \mathbb{1}$, (3.54)

implies

$$\mathcal{T}^2: \quad u_m(\tau+2,\tau_0+2) = u_m(\tau,\tau_0). \tag{3.55}$$

We can check explicitly that it is true for example for case B as in (3.36), which is not \mathcal{T} -invariant.

As discussed in section 3.2.3, the group action $\mathscr{T} \times \mathscr{M} \to \mathscr{M}$ partitions the mass space $\mathscr{M} \ni \mathbf{m}$ into three regions \mathcal{L}_1 , \mathcal{L}_3 and \mathcal{L}_6 , where the orbits $\mathscr{T} \cdot \mathbf{m}$ have length 1, 3 and 6. The stabiliser subgroups of \mathbf{m} are then subgroups of S_3 of order 6, 2 and 1, i.e. isomorphic to S_3 , S_2 or $S_1 = \{e\}$. The homomorphism φ (3.13) between SL(2, \mathbb{Z}) and $\mathscr{T} = \text{Out}(\text{Spin}(8))$ then dictates the subgroup

$$\varphi^{-1}[\mathscr{T}_{\boldsymbol{m}}] \tag{3.56}$$

under which u_m is simultaneously invariant. The preimage of the stabiliser subgroup under φ thus constitutes the third component Γ of the triple $(\Gamma_1, \Gamma_2; \Gamma)$ in definition 5.

The case $oldsymbol{m}\in\mathcal{L}_1$

When $\boldsymbol{m} \in \mathcal{L}_1$, then the stabiliser group of \boldsymbol{m} has six elements and the orbit $\mathscr{T} \cdot \boldsymbol{m}$ consists of \boldsymbol{m} only. Then there is only one function in (3.54), and $u_{\boldsymbol{m}}$ transforms with weight (0, 2) under SL(2, \mathbb{Z})_{(τ, τ_0)}, as in condition 2 of definition 5. An example is u_A as given in (3.25), and the transformation is checked in (3.27).

The case $\boldsymbol{m} \in \mathcal{L}_3$

The case $\boldsymbol{m} \in \mathcal{L}_3$ is most interesting, as it is not trivial $(\boldsymbol{m} \in \mathcal{L}_1)$ and not generic $(\boldsymbol{m} \in \mathcal{L}_6)$. Namely, when the orbit $\mathscr{T} \cdot \boldsymbol{m}$ contains three elements, the stabiliser group is isomorphic to the symmetric group S_2 with two elements. Then the three functions associated with the three elements of the orbit $\mathscr{T} \cdot \boldsymbol{m}$ form a vector that transforms under SL $(2,\mathbb{Z})$. An example for this are the functions $u_{\rm B}, u_{\rm C}, u_{\rm D}$ found in Sections 3.3.2–3.3.4. As is clear from Fig. 23, they are related to each other by triality. If we organise $\boldsymbol{u}_3 = (u_{\rm B}, u_{\rm C}, u_{\rm D})^{\mathsf{T}}$, using (3.36), (3.41) (3.46) one can prove that

$$\boldsymbol{u}_{3}(\tau+1,\tau_{0}+1) = \begin{pmatrix} 0 & 0 & 1\\ 0 & 1 & 0\\ 1 & 0 & 0 \end{pmatrix} \boldsymbol{u}_{3}(\tau,\tau_{0}),$$

$$\boldsymbol{u}_{3}(-1/\tau,-1/\tau_{0}) = \tau_{0}^{2} \begin{pmatrix} 0 & 1 & 0\\ 1 & 0 & 0\\ 0 & 0 & 1 \end{pmatrix} \boldsymbol{u}_{3}(\tau,\tau_{0}).$$
(3.57)

As the matrices are in $GL(3, \mathbb{C})$, there exists a 3-dimensional representation $SL(2,\mathbb{Z}) \to GL(3,\mathbb{C})$. This shows that $u_3(\tau, \tau_0)$ furnishes a vector-valued

bimodular form of weight (0, 2) for $SL(2, \mathbb{Z})$, agreeing with the following definition:²⁵

Definition 6 (Vector-valued bimodular form). Let

$$\boldsymbol{F} = \begin{pmatrix} F_1 \\ \vdots \\ F_p \end{pmatrix} : \mathbb{H} \times \mathbb{H} \to \mathbb{C}^p \tag{3.58}$$

be a p-tuple of two-variable meromorphic functions, $p \in \mathbb{N}$. Then \mathbf{F} is called a vector-valued bimodular form of weight (k_1, k_2) for $\Gamma \subset SL(2, \mathbb{Z})$, if

- each component F_j is a bimodular form of weight (k_1, k_2) for some triple $(\Gamma_1^j, \Gamma_2^j; \Gamma^j)$, as in definition 5, and
- there exists a p-dimensional complex representation $\rho : \Gamma \to GL(p, \mathbb{C})$ such that

$$\mathbf{F}(\gamma\tau_1,\gamma\tau_2) = (c\tau_1 + d)^{k_1}(c\tau_2 + d)^{k_2}\rho(\gamma)\mathbf{F}(\tau_1,\tau_2)$$
(3.59)

for all $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma$ and all $\tau_1, \tau_2 \in \mathbb{H}$.

Since u_3 is parametrised by the mass $m \in \mathbb{C}$, $m \mapsto u_3(m, \tau, \tau_0)$ is in fact a 1-parameter family of vector-valued bimodular forms of weight (0,2) for $\mathrm{SL}(2,\mathbb{Z})$. The triality action of $\mathrm{SL}(2,\mathbb{Z})$ permutes the triples $(\Gamma_1^j, \Gamma_2^j; \Gamma^j)$ in an interesting way. The action of the $\mathrm{SL}(2,\mathbb{Z})$ generators on u is given by (3.57). As $\Gamma_1^j = \Gamma_2^j$ for the cases B, C, D, both Γ_1^j and Γ_2^j are conjugated by the corresponding element of $\mathrm{SL}(2,\mathbb{Z})$. An instance of this is the group $\Gamma^0(4)$ (3.48), which is the set of elements of $\Gamma^0(4)$ conjugated by T. Similarly, we have that $\Gamma^0(4)$ is conjugate to $\Gamma_0(4)$ by conjugation with S. The same is true for the three groups $\Gamma^0(2)$, $\Gamma_0(2)$ and $\Gamma^0(2)$ that the cases B, C, D simultaneously transform under, these three conjugate subgroups are permuted under $\mathrm{SL}(2,\mathbb{Z})$ just as $\Gamma^0(4)$, $\Gamma_0(4)$ and $\Gamma^0(4)$ are.

The case $\boldsymbol{m} \in \mathcal{L}_6$

The remaining case is that $\boldsymbol{m} \in \mathcal{L}_6$, where $\mathscr{T} \cdot \boldsymbol{m}$ has six elements. Then we can organise $\boldsymbol{u}_6 = (u_{\boldsymbol{m}}, u_{\mathcal{T}\boldsymbol{m}}, u_{\mathcal{S}\boldsymbol{m}}, u_{\mathcal{T}\mathcal{S}\boldsymbol{m}}, u_{\mathcal{S}\mathcal{T}\boldsymbol{m}}, u_{\mathcal{T}\mathcal{S}\mathcal{T}\boldsymbol{m}})^{\mathsf{T}}$, which is a collection of six pairwise distinct functions. By studying the action of \mathcal{T} and \mathcal{S} on the

²⁵It is customary to define vector-valued modular forms for $SL(2,\mathbb{Z})$, however vector-valued modular forms for proper subgroups Γ of $SL(2,\mathbb{Z})$ are familiar in rational CFTs [192–194] and so we leave our definition more generic.

vector $(\boldsymbol{m}, \mathcal{T}\boldsymbol{m}, \mathcal{S}\boldsymbol{m}, \mathcal{T}\mathcal{S}\boldsymbol{m}, \mathcal{S}\mathcal{T}\boldsymbol{m}, \mathcal{T}\mathcal{S}\mathcal{T}\boldsymbol{m})^{\mathsf{T}}$, we find the transformations

The vector \boldsymbol{u}_6 is not a vector-valued bimodular form for $\mathrm{SL}(2,\mathbb{Z})$, because the components of \boldsymbol{u}_6 do not transform as modular forms under the separate action of $\Gamma_{1,2}^j$, $j = 1, \ldots, 6$ due to the branch cuts, as discussed for the $N_f \leq 3$ theories in section 2. The simultaneous action of $\mathrm{SL}(2,\mathbb{Z})$ on τ and τ_0 is not obstructed by the branch cuts of $u(\tau, \tau_0)$ [3].

The matrices in (3.57) and (3.60) are not only in $GL(n, \mathbb{C})$, but they are in fact permutation matrices: Because triality permutes the respective moduli spaces, the order parameters are merely permuted and there are no phases. Thus we have that

$$\boldsymbol{u}(\gamma\tau,\gamma\tau_0) = (c\tau_0 + d)^2 P_{\pi(\gamma)} \boldsymbol{u}(\tau,\tau_0), \qquad (3.61)$$

where $P_{\pi(\gamma)}$ is the permutation matrix for the permutation $\pi(\gamma) \in S_{|\mathscr{T} \cdot \boldsymbol{m}|}$, which can be found from the action of \mathscr{T} on \boldsymbol{m} .

For the period $\left(\frac{da}{du}\right)^2$, there are similar results. For instance, one can check that

$$\left(\left(\frac{da}{du}\right)_{\rm B}^2, \left(\frac{da}{du}\right)_{\rm C}^2, \left(\frac{da}{du}\right)_{\rm D}^2\right)^{\rm T}$$
(3.62)

is a vector-valued bimodular form of weight (2, -4) for $SL(2, \mathbb{Z})$. As u has weight (0, 2), it is not obvious how the discriminant Δ transforms since it is a polynomial in u. However, because triality acts on the 6 singularities as well, in general Δ is a vector-valued bimodular form of weight (0, 12) for $SL(2, \mathbb{Z})$. This can be checked explicitly for the cases B, C, D, where $\Delta_3 = (\Delta_B, \Delta_C, \Delta_D)^{\mathsf{T}}$ is a 1-parameter family of vector-valued bimodular forms of weight (0, 12) for $SL(2, \mathbb{Z})$.

3.4.2 Fundamental domains

In the asymptotically free theories we argued that the *u*-planes can be identified with fundamental domains $\mathcal{F}_{N_f}(\boldsymbol{m})$. For this we make the correspondence that

the number of singularities gives the number of rational cusps, the number of BPS states becoming massless at each singularity gives the width of each cusp, and the width at $i\infty$ is given by $4 - N_f$. Then, the sum of all cusps is RG invariant. By following the RG flow from $N_f = 3$ to $N_f < 3$ we find that gradually a singularity at strong coupling (a rational cusp) moves to infinity and is identified with the weak coupling region $(i\infty)$. Reversing this argument implies that for $N_f = 4$ there should be six rational cusps and the width at infinity vanishes. This is consistent with the fact [46] that $u = \infty$ does not correspond to a cusp of the curve anymore. Rather, it lies in the interior of $\mathbb{H} \ni \tau_0$.

It is found in the above subsections that depending on the mass configuration, the fundamental domain for an order parameter is related to the one of the underlying theory where all massive hypermultiplets are decoupled. We can depict those domains in an equivalent way that is more suitable to our description. For this, one chooses an equivalent fundamental domain with the property that the width at $i\infty$ is zero and the number of rational cusps is equal to the number of singularities, with according width. For instance, in case A where $\boldsymbol{m} = (m, m, 0, 0)$ the duality group is $\Gamma(2)$, whose cusps in the decoupling limit (with the same duality group) we choose as $\{i\infty, 0, 1\}$. In $N_f = 4$ it is more suitable to represent $i\infty$ by a rational number. For this we can use that $\Gamma(2) \ni ST^{-2}S : i\infty \mapsto \frac{1}{2}$, being a preferable representative of the third cusp. As it necessarily also has width 2, both \mathcal{F} and $T\mathcal{F}$ can be mapped to the region $\tau = \frac{1}{2}$. This is depicted in Fig. 26. The decoupling to massless $N_f = 2$ is illustrated in Fig. 27. The domains in this case are exactly equivalent, Fig. 26 merely allows to extend the $N_f \leq 3$ description of the cusps to $N_f = 4$.

We stress that the decoupling limit for $N_f = 4$, with the order parameter a bimodular form as (3.25), is quite different from the asymptotically free theories with $N_f \leq 3$. In the latter theories, $u(\tau)$ is not holomorphic and modular except for special points in mass space (complex co-dimension N_f) [2]. The decoupling of a hypermultiplet is in these theories accompanied by a branch point moving to infinity. In this way, a singularity merges with the weak coupling cusp. For cases A, B, C and D in $N_f = 4$ on the other hand, there is no branch point for any value of the mass m, and in particular there is also none for $m \to \infty$.

3.5 Discussion

In this section, we have studied in detail the Coulomb branch of the superconformal $N_f = 4$ theory with gauge group SU(2), which has remained of great interest throughout the years [21, 46, 50, 73, 77, 105, 111, 113, 148, 173, 178–180, 183–187, 195, 196]. For the mass configurations with the largest flavour symmetry group, such as when one, two and four hypermultiplets have an equal mass, we show that the Coulomb branch is parametrised by a function $u(\tau, \tau_0)$



Figure 26: Fundamental domain of $N_f = 4$ case A with $\boldsymbol{m} = (m, m, 0, 0)$.



Figure 27: Decoupling the two massive hypermultiplets in $N_f = 4$ case A gives the domain of massless $N_f = 2$ (blue). Two of the differing regions (gray) are the regions near $\tau = \frac{1}{2}$, which are mapped (orange) to $i\infty$. The remaining two are merely mapped to $\Gamma(2)$ equivalent regions near the same cusp such that the resulting domain is connected. Alternatively, the fundamental domain in this figure is the image of $ST^{-1}STS$ acting on the domain in Figure 26.

that is not only invariant under separate modular transformations of τ and τ_0 , but also exhibits invariance under a simultaneous transformation under τ and τ_0 . By restricting to the stabiliser subgroup of a given mass under the triality action, such order parameters constitute nontrivial examples of bimodular forms (see (3.25) for example). Furthermore, the moduli spaces are permuted under triality, and the order parameter, periods, discriminants etc. furnish vector-valued bimodular forms, which we also introduce (see definition 6).

The analysis of other mass configurations can be done using the techniques established in section 2. As more complicated mass configurations minevitably introduce branch points and cuts, in general u_m is not a bimodular form. A simultaneous transformation of τ and τ_0 is yet to be expected by triality, while the separate transformations are induced by monodromies and as such do not in general lie in SL(2, Z) [46]. However, even in such cases the action of the monodromy group of the *u*-plane can be understood as paths in the fundamental domain for τ . See section 4.3.3 for a discussion of these aspects for gauge group SU(3).

Our results allow to study the topologically twisted theory on a fourmanifold X [52, 61, 72–74], where the the path integral can be expressed as an integral over the fundamental domain for the effective coupling τ . In fact, a closed expression for the order parameter is enough to define the integrand. The modularity for τ allows to show that the integral measure is well-defined. The triality action then gives the S-duality orbit of the $N_f = 4$ theory on X [197].

It would also be interesting to apply our results to other theories with an IR moduli space of vacua as well as a non-trivial conformal manifold. Such theories may include subsectors with triality symmetry, such as F-theory [198], quiver gauge theories [21], the AGT correspondence [172], little string theory [199] and string/string/string triality [200].

4 Elliptic loci of SU(3) vacua

In this section, we investigate the modularity of the pure SU(3) SW theory. While in the SU(2) theories of Sections 2 and 3 the obstruction to modularity is due to the hypermultiplets, we show here that for rank two the modularity is broken in specific cases even without hypermultiplets. This section is mainly based on [1], and section 4.6 is based on unpublished work.

4.1 Introduction

Many observables in supersymmetric theories can be determined non-perturbatively in terms of hypergeometric, modular or other special functions. The best understood example is $\mathcal{N} = 2$ supersymmetric Yang–Mills theory with gauge group SU(2) [45, 46], which we introduced in detail in section 1.5. Its space of vacua is parametrised by the vacuum expectation value (vev) u = $\frac{1}{2}\langle \operatorname{Tr} \phi^2 \rangle$, where ϕ is the complex scalar in the $\mathcal{N}=2$ vector multiplet. The renormalisation group flow generates a quantum scale Λ , at which the gauge coupling becomes strong. In the weak-coupling region $|u| \gg \Lambda^2$, the semiclassical BPS spectrum consists of massive monopoles and dyons. The theory can be solved non-perturbatively in terms of the Seiberg-Witten (SW) curve [45]. This solution demonstrates that the effective abelian gauge theory breaks down at two special points, $u = \pm \Lambda^2$. The electric-magnetic duality group is generated by the monodromies around these singular points. It is a subgroup of $SL(2,\mathbb{Z})$, which acts by linear fractional transformations on the effective coupling constant τ . With the SW solution, various physical quantities can be exactly determined as functions of τ using modular functions [73, 74, 101, 103, 201].

Similar non-perturbative solutions have been developed for gauge theories with matter multiplets [46] and theories with other gauge groups [94–97,202– 204]. In pure Yang–Mills theory with compact gauge group G, the Coulomb branch has complex dimension $r = \operatorname{rank}(G)$. Classically, the moduli space is parametrised by the vevs $u_{I+1} \sim \langle \operatorname{Tr} \phi^{I+1} \rangle$, $I = 1, \ldots, r$. The r(r+1)/2couplings τ_{IJ} are determined by the r order parameters u_I . The electricmagnetic duality group is a subgroup of $\operatorname{Sp}(2r, \mathbb{Z})$, generated by monodromies around singular loci. While this also demonstrates a link to modularity, the connection has remained more elusive, and the connection is best established for the superconformal theories [178, 205–208].

One complication for asymptotically free theories is that the structure of the singular loci is in general quite intricate. This section focuses on the asymptotically free SU(3) theory without hypermultiplets, whose singular loci have a rich structure [104, 105, 128, 209, 210]. There are six singular (complex) lines which intersect in five points. A particularly interesting phenomenon occurs at two of these five vacua, namely those where three mutually non-local dyons become massless, such that there is no duality frame in which all of these states only carry electric charge. This indicates that the system is in a critical phase, which led to the discovery of new superconformal theories [104,105,128].

Another complication for SU(N > 2) is that the number of couplings exceeds the dimension of the Coulomb branch. The observables are therefore defined on a subspace of the genus N - 1 Siegel upper half-space \mathbb{H}_{N-1} . For the SU(3) theory, the Coulomb branch is parametrised by two order parameters which determine three coupling constants, τ_{11} , τ_{12} and τ_{22} . The curve and the SW differential for pure SU(N) gauge theory have first been proposed in [94]. As a first step to explore the modularity of the SU(3) theory, we relate the hyperelliptic Seiberg-Witten curve to the Rosenhain form, which is an algebraic expression in terms of Siegel theta series. To exactly match the Rosenhain curve and Seiberg-Witten curve, we use the fact that the complexified masses a_I and $a_{D,I} = \frac{\partial \mathcal{F}}{\partial a_I}$ are solutions of second order partial differential equations of Picard-Fuchs (PF) type. The solutions to such equations can be expressed in terms of the generalised hypergeometric function F_4 of Appell [96]. The Siegel theta series and their modular transformations can provide insights for the analytic continuation and monodromies of the solution in terms of F_4 .

The Rosenhain curve allows us to characterise the SU(3) Coulomb branch, parametrised by the two Casimirs $u = u_2$ and $v = u_3$, as the zero-locus of three equations inside a five-dimensional space. The structure of these equations simplifies on one-dimensional loci of the Coulomb branch. We study two of these loci in detail, namely \mathcal{E}_u where v = 0 and \mathcal{E}_v where u = 0. On each of these loci, the equations reduce to two algebraic relations of Siegel theta functions, relating the couplings τ_{IJ} to a single independent one. Interestingly, each of these loci in the space of genus two curves also parametrises a family of (genus 1) elliptic curves. Both loci interpolate between a weak-coupling regime with large order parameters and a strong-coupling regime where u/Λ^2 and v/Λ^3 are $\mathcal{O}(1)$. Locus \mathcal{E}_u contains three cusps where mutually local dyons become massless, while locus \mathcal{E}_v contains two special points where mutually non-local dyons becomes massless. The latter are the superconformal Argyres-Douglas points.

Since an elliptic locus parametrises a family of elliptic curves, there must be a coupling τ valued in a fundamental domain (or modular curve) for a discrete group in the upper half-plane \mathbb{H} . We derive the generators of the discrete subgroup from the monodromies of the SU(3) theory. We provide two solutions for the locus \mathcal{E}_u . The coupling for the first solution is $\tau_- = \tau_{11} - \tau_{12}$, while $\tau_{22} = \tau_{11}$. The order parameter u equals a modular form u_- for the congruence subgroup $\Gamma^0(9) \subset SL(2,\mathbb{Z})$ (4.49),

$$u = u_{-}(\tau_{-}). \tag{4.1}$$

The cusps of the fundamental domain of $\Gamma^0(9)$ map exactly to the singular points on this locus. The coupling for the second solution is $\tau_+ = \tau_{11} + \tau_{12}$. In terms of this coupling, Equation (4.59) expresses u as

$$u = u_+(\tau_+),$$
 (4.2)

where u_+ is expressed in terms of roots of modular forms, while it is not a modular function for a congruence subgroup of $SL(2,\mathbb{Z})$. We call it a *sextic* modular function since it is a solution to a sextic equation. The inverses of the identities (4.1) and (4.2) provide all order *u*-expansions for $\tau_{11} = \tau_{22}$ and τ_{12} on this locus. The function u_+ appeared earlier as the solutions for the order parameter on the Coulomb branch of the $\mathcal{N} = 2$, SU(2) theory with one massless hypermultiplet [101]. While this Coulomb branch and \mathcal{E}_u are isomorphic as four punctured spheres, it is striking that the solutions of the order parameters are identical.

We find another intriguing structure for the second locus \mathcal{E}_v where u = 0. We are able to demonstrate for this locus that v is left invariant by the action of the principal congruence subgroup $\Gamma(6) \subset \mathrm{SL}(2,\mathbb{Z})$. The fundamental domain $\Gamma(6) \setminus \mathbb{H}$ has 12 cusps, where v diverges. Surprisingly, this appears to imply the existence of strongly coupled vacua in the region where v is large, which is unexpected since large v is known to correspond to weak coupling. The paradox is resolved by realizing that v is invariant under a transformation which is not contained in $\mathrm{SL}(2,\mathbb{Z})$, namely a *Fricke involution* $\tau \mapsto -1/n\tau$ for integer $n \geq 2$. This transformation maps the putative cusps to $i\infty$. The result is that v is a modular function for a discrete subgroup $\Gamma_v \subset \mathrm{SL}(2,\mathbb{R})$ of Atkin-Lehner type, and we show that the non-trivial monodromies on this locus do generate this group.

We demonstrate furthermore that the elliptic curves underlying the two loci \mathcal{E}_u and \mathcal{E}_v are related to the genus two curve in a precise way. For a genus two curve Σ_2 , a holomorphic map $\varphi : \Sigma_2 \to \Sigma_1$ to an elliptic curve Σ_1 may exist. Such maps were studied in the classic works by Legendre and Jacobi, and more recently in [211, 212]. The existence of the map φ depends on the complex structure moduli τ_{IJ} . The family of such curves spans a complex co-dimension one locus \mathcal{L}_2 in the complex three-dimensional space of genus two curves. At the elliptic loci of the Coulomb branch of the SU(3) theory mentioned above, \mathcal{L}_2 intersects the SU(3) Coulomb branch, such that for any point on the elliptic loci, there is a degree two map from the genus two curve to an elliptic curve, or in other words the genus two curve is a double cover of the elliptic curve. Besides \mathcal{E}_u and \mathcal{E}_v , \mathcal{L}_2 also includes a third elliptic locus, \mathcal{E}_3 (4.91), which does not contain any of the singular points of the Coulomb branch.

Our work motivates a similar analysis for SU(N) gauge theories, whose Coulomb branch parametrises a curve of genus N - 1. The order parameters $u_I, I = 2, ..., N$, are expected to be given by higher genus modular functions of the coupling matrix τ_{IJ} . They should furthermore be invariant under a subgroup of $Sp(2r, \mathbb{Z})$ generated by the monodromies. The existence of maps to elliptic or lower genus curves is however more subtle for such theories [213,214].

4.2 The SU(3) Coulomb branch

We study in this section the SU(3) Coulomb branch. We first recall the Seiberg-Witten geometry in section 4.2.1 following [94, 96, 215]. Section 4.2.2 reviews the Picard-Fuchs solution for the complexified masses and couplings. Section 4.2.3 uses those results to write the curve in Rosenhain form.

4.2.1 Seiberg-Witten geometry

The vector multiplet scalar ϕ can be gauge rotated into the Cartan subalgebra of SU(3). Then, ϕ can be expanded in terms of the two Cartan generators H_I , I = 1, 2, as

$$\phi = a_1 H_1 + a_2 H_2. \tag{4.3}$$

Non-vanishing vevs of ϕ break the gauge group in general to U(1)². The central charges of the gauge bosons are then given by

$$Z_1 = 2a_1 - a_2,$$

$$Z_2 = 2a_2 - a_1,$$

$$Z_3 = a_1 + a_2.$$

(4.4)

We denote electric-magnetic charges under $U(1)^2$ as $\gamma = (m_1, m_2, n_1, n_2)$, where m_i are the magnetic and n_i the electric charges respectively, and the period vector as $\pi = (a_{D,1}, a_{D,2}, a_1, a_2)^{\mathsf{T}}$. The central charge for a generic γ is then given by $Z_{\gamma} = \gamma \cdot \pi$, where \cdot is the standard scalar product.

The Coulomb branch is parametrised by vevs of Casimirs of ϕ , $u_I \sim \langle \text{Tr} \phi^I \rangle$, I = 2, 3. Gauge invariant combinations for SU(3) are

$$u = u_{2} = \frac{1}{2} \langle \operatorname{Tr}(\phi^{2}) \rangle_{\mathbb{R}^{4}} = a_{1}^{2} + a_{2}^{2} - a_{1}a_{2},$$

$$v = u_{3} = \frac{1}{3} \langle \operatorname{Tr}(\phi^{3}) \rangle_{\mathbb{R}^{4}} = a_{1}a_{2}(a_{1} - a_{2}).$$
(4.5)

These relations can be rewritten in terms of two cubic equations for a_1 and a_2 as

$$a_1^3 - ua_1 - v = 0,$$

$$a_2^3 - ua_2 + v = 0.$$
(4.6)

There is a spontaneously broken global \mathbb{Z}_6 symmetry acting on u and v by $u \mapsto \alpha u$ and $v \mapsto -v$, with $\alpha = e^{2\pi i/3}$. Classically, the discriminant is the determinant $\Delta_{\text{classical}}$ of the matrix $B_{IJ} = \frac{\partial u_{I+1}}{\partial a_J}$. It reads

$$\Delta_{\text{classical}} = \det B_{IJ} = (a_1 - 2a_2)(2a_1 - a_2)(a_1 + a_2), \qquad (4.7)$$

and vanishes when one of the gauge bosons (4.4) becomes massless.

Let us denote the space parametrised by u and v by \mathcal{U} . We parametrise points on this space by $(\underline{u}, v) \in \mathcal{U}$, where \underline{u} is the normalised parameter, $\underline{u} = \sqrt[3]{\frac{4}{27}}u$. The moduli space \mathcal{U} parametrises a complex two-dimensional family of hyperelliptic curves of genus two [94, 215],

$$y^{2} = (x^{3} - ux - v)^{2} - \Lambda^{6}, \qquad (4.8)$$

which has discriminant

$$\Delta_{\Lambda} = \Lambda^{18} (4u^3 - 27(v + \Lambda^3)^2) (4u^3 - 27(v - \Lambda^3)^2).$$
(4.9)

This can be viewed as a product of the discriminants of two elliptic curves whose v parameters are separated by $2\Lambda^3$. Note that the \mathbb{Z}_6 global symmetry leaves the discriminant invariant. It vanishes if and only if $\underline{u}^3 = (v \pm \Lambda^3)^2$. We will frequently use units where the dynamical scale $\Lambda = 1$ and we note that it can always be restored from dimensional analysis.

If we restrict to $\operatorname{Im} v = 0$, the zero locus of the discriminant describes six singular curves which intersect in the following points. On the v = 0 plane, there are four singularities, namely $\underline{u} \in \{\infty, 1, \alpha, \alpha^2\}$. On the other hand for $\underline{u} = 0$, there are two singularities at $v = \pm 1$. These are the Argyres-Douglas points, where mutually non-local BPS states become massless and the theory becomes superconformal [104]. Figure 28 sketches the singular lines on the subset of \mathcal{U} where $\operatorname{Im} v = 0$. The singular lines represent regions in \mathcal{U} where the effective action of the pure $\mathcal{N} = 2$ theory becomes singular, and they are associated with vacua where hypermultiplets become massless.

Similarly to the SU(2) case, the periods transform under monodromies which generate the duality group of the theory. The classical part of the monodromy group is given by the Weyl group of the SU(3) root lattice, which acts as reflections on lines perpendicular to the positive roots. The perturbative quantum correction comes from the one-loop effective action. It contributes to the prepotential as

$$\mathcal{F}_{1-\text{loop}} = \frac{i}{2\pi} \sum_{\alpha} Z_{\alpha}^2 \log Z_{\alpha}, \qquad (4.10)$$

where the sum runs over all positive roots α_1 , α_2 and $\alpha_3 = \alpha_1 + \alpha_2$. Here, Z_{α} are the central charges (4.4) of the gauge bosons.

4.2.2 Picard-Fuchs solution

One way to find the non-perturbative solution is to notice that the periods satisfy second order partial differential equations of Picard-Fuchs (PF) type, whose solution space is spanned by the generalised hypergeometric function F_4 of Appell [96]. We review some aspects of the PF solution in the following, and more details can be found in [1]. We study two interesting regions, one where u is large and v small, and the other one where v is large and u is small.

In the limit of large u and small v, reference [96] determines the a_I and $a_{D,I}$ non-perturbatively in terms of the fourth Appell hypergeometric function



Figure 28: Singular lines $\Delta(u, v) = 0$ in the SU(3) moduli space with Im v = 0, associated to massless dyons [96]. The red dots represent the strong coupling points $(\underline{u}, v) = (1, 0)$, $(\alpha, 0)$ and $(\alpha^2, 0)$ on the v = 0 plane \mathcal{E}_u , where two singular lines intersect. The blue dots represent the AD points $(\underline{u}, v) = (0, 1)$ and (0, -1) respectively, where three singular lines intersect. They lie on \mathcal{E}_v , which is represented by the Re v axis here. The two loci \mathcal{E}_u and \mathcal{E}_v intersect in the origin (u, v) = (0, 0) (brown).

 $F_4(a, b, c, d; x, y)$. For $\sqrt{|x|} + \sqrt{|y|} < 1$, this function is given by

$$F_4(a, b, c, d; x, y) = \sum_{m,n \ge 0} \frac{(a)_{m+n} (b)_{m+n}}{m! \, n! \, (c)_m (d)_n} \, x^m \, y^n, \tag{4.11}$$

where $(a)_m = \frac{\Gamma(a+m)}{\Gamma(a)}$ is the Pochhammer symbol. We will also need expansions of F_4 for large y, which can be achieved by replacing the sum over n by the hypergeometric series ${}_2F_1$,

$$F_4(a, b, c, d; x, y) = \sum_{m \ge 0} \frac{(a)_m (b)_m}{m! (c)_m} {}_2F_1(a + m, b + m, d; y) x^m.$$
(4.12)

While analytic continuations are known for $_2F_1$, they are not well established for F_4 .

In order to match the Picard-Fuchs solutions with the periods, we need to expand the periods around the classical solutions in (4.6). We therefore need to find the roots of these two cubics.

The general formula for the roots of a depressed cubic equation, $ax^3 + bx + c = 0$, is given by

$$\xi_k = -\frac{1}{3a} \left(\alpha^k C + \frac{\Delta_0}{\alpha^k C} \right), \quad k \in \{0, 1, 2\},$$
(4.13)

where $\alpha = e^{2\pi i/3}$, $C^3 = \frac{\Delta_1 \pm \sqrt{\Delta_1^2 - 4\Delta_0}}{2}$, $\Delta_0 = -3ab$ and $\Delta_1 = 27a^2c$ [216]. The choice of sign in front of the square root in C is arbitrary, in the sense that it only corresponds to a permutation of the roots.

It is however important to fix the ambiguities in taking the square and cubic root. We fix the ambiguity in the square root by the following choice for the branch of the logarithm: For any complex number $z \in \mathbb{C}^*$, we set $\log(z) = \log|z| + i\operatorname{Arg}(z)$ with $-\pi < \operatorname{Arg}(z) \leq \pi$. The ambiguity in the cubic root of a complex number z is fixed by demanding that the real part of $\sqrt[3]{z}$ has the largest absolute value among the three solutions to $\rho^3 = z$. Thus $\sqrt[3]{1} = 1$ and $\sqrt[3]{-1} = -1$. Two of the cube roots of i and -i have equal real parts. We fix the remaining ambiguity by setting $\sqrt[3]{i} = e^{\pi i/6} = \frac{\sqrt{3}}{2} + \frac{i}{2}$ and $\sqrt[3]{-i} = e^{-\pi i/6} = \frac{\sqrt{3}}{2} - \frac{i}{2}$.

To list the roots of our two equations, we define

$$s_{\pm}(a,b) = \sqrt[3]{\frac{b}{2} \pm \sqrt{\frac{b^2}{4} - \frac{a^3}{27}}}.$$
 (4.14)

Using Eq. (4.13), we then find that the roots of (4.6) for a_1 are given by

$$\xi_{1}(u, v) = s_{+}(u, v) + s_{-}(u, v),$$

$$\xi_{2}(u, v) = \alpha s_{+}(u, v) + \alpha^{2} s_{-}(u, v),$$

$$\xi_{3}(u, v) = \alpha^{2} s_{+}(u, v) + \alpha s_{-}(u, v),$$

(4.15)

and the roots for a_2 by $-\xi_j(u, v)$. This gives the $3 \times 3 = 9$ solutions to the equations in (4.6). However, (4.5) is supposed to have only $2 \times 3 = 6$ solutions. Let us determine the 6 solutions in one of the regimes of interest for SU(3) Yang-Mills theory: we assume u is large and close to the positive axis: $u = \lambda - i\epsilon\lambda$ with λ real and very large and $0 < \epsilon \ll 1$. Note that in this regime

$$s_{\pm}(u,v) = \sqrt[3]{\frac{v}{2} \pm i\sqrt{\frac{u^3}{27} - \frac{v^2}{4}}}.$$
(4.16)

Furthermore, $s_+(u, v) s_-(u, v) = u/3$ and

$$s_{-}(u, -v) = e^{-\pi i/3} s_{+}(u, v) = -\alpha s_{+}(u, v)$$
(4.17)

hold. For v = 0, we have $s_+(u, 0) = e^{\pi i/6} \sqrt{u/3}$ and $s_-(u, 0) = e^{-\pi i/6} \sqrt{u/3}$, and thus

$$\xi_1(u,0) = \sqrt{u}, \xi_2(u,0) = -\sqrt{u}, \xi_3(u,0) = 0.$$
(4.18)

This demonstrates that the solutions to (4.5) for (a_1, a_2) are given by

 $(\xi_1, -\xi_2), \ (\xi_1, -\xi_3), \ (\xi_2, -\xi_1), \ (\xi_2, -\xi_3), \ (\xi_3, -\xi_1), \ (\xi_3, -\xi_2).$ (4.19)

The non-perturbative effective action is characterised by the holomorphic prepotential \mathcal{F} , which allows to define the dual periods $a_{D,I} = \frac{\partial \mathcal{F}}{\partial a_I}$. Both periods a_I and $a_{D,I}$ are given by linear combinations of Appell functions. The large u expansion reads [96]

$$a_{D,1}(u,v) = -\frac{i}{2\pi} \left(\sqrt{u} + \frac{3}{2}\frac{v}{u}\right) \log\left(\frac{27}{4u^3}\right) - \frac{1}{\pi} \left(\frac{i}{2} + 2\alpha_1\right) \sqrt{u} + \dots,$$

$$a_1(u,v) = \sqrt{u} + \frac{1}{2}\frac{v}{u} + \dots,$$

(4.20)

with $a_{D,2}(u,v) = a_{D,1}(u,-v)$, $a_2(u,v) = a_1(u,-v)$ and $\alpha_1 \in \mathbb{C}$ a constant. The coupling constants $\tau_{IJ} = \frac{\partial a_{D,I}}{\partial a_J}$ are determined using the chain rule,

$$\tau_{11}(u,v) = \tau_{22}(u,-v) = \frac{i}{\pi} \log(8u^3) + \frac{9iv}{2\pi} u^{-3/2} - \left(\frac{129i}{32\pi} + \frac{63iv^2}{8\pi}\right) u^{-3} + \dots,$$
(4.21)

The off-diagonal τ_{12} is given by the series

$$\tau_{12}(u,v) = -\frac{\tau_{11}(u,v) + \tau_{22}(u,v)}{4} - \frac{1}{2\pi i}\log(8) + \frac{1}{2\pi i}\frac{27}{4}f(u,v), \qquad (4.22)$$

where

$$f(u,v) = \frac{(1-4v^2)}{8}u^{-3} + \left(\frac{453}{1024} - 3v^2 - \frac{31}{16}v^4\right)u^{-6} + \dots$$
(4.23)

Similarly, we find that the large v expansion of the coupling matrix reads $(\omega = e^{\pi i/6})$

$$\tau_{11} \sim \frac{i}{\pi} \log(108v^2) - 1 + \frac{\omega}{\pi} uv^{-2/3} + \frac{\omega^5}{6\pi} u^2 v^{-4/3} - \left(\frac{11i}{18\pi} + \frac{4i}{27\pi} u^3\right) v^{-2} + \dots,$$
(4.24)

and τ_{12} and τ_{22} are given by similar series. At u = 0 we have $\tau_{11} = \tau_{22} + 1$ and $\tau_{12} = -\frac{\tau_{11}}{2} + 1$.

4.2.3 Seiberg-Witten curve in Rosenhain form

In this section, we will relate the SU(3) Seiberg-Witten curve to the curve in Rosenhain form, which is a degree 5 equation. Every genus two hyperelliptic curve can be brought to the Rosenhain form [217]

$$y^{2} = x(x-1)(x-\lambda_{1})(x-\lambda_{2})(x-\lambda_{3}).$$
(4.25)

The three roots λ_i of the polynomial are also referred to as *Rosenhain invariants*. These invariants are complementary to the Igusa invariants [218, 219].

By a lemma of Picard, the Rosenhain invariants can be expressed in terms of even theta constants as

$$\lambda_1 = \frac{\Theta_1^2 \Theta_3^2}{\Theta_2^2 \Theta_4^2}, \quad \lambda_2 = \frac{\Theta_3^2 \Theta_8^2}{\Theta_4^2 \Theta_{10}^2}, \quad \lambda_3 = \frac{\Theta_1^2 \Theta_8^2}{\Theta_2^2 \Theta_{10}^2}. \tag{4.26}$$
The functions Θ_j are instances of genus two Siegel modular forms,

$$\Theta\begin{bmatrix}a\\b\end{bmatrix}(\Omega) = \sum_{k\in\mathbb{Z}^2} \exp\left(\pi i(k+a)^{\mathsf{T}}\Omega(k+a) + 2\pi i(k+a)^{\mathsf{T}}b\right), \qquad (4.27)$$

where the entries of the column vectors a and b take values in the set $\{0, \frac{1}{2}\}$. The argument Ω is a 2 × 2-matrix

$$\Omega = \begin{pmatrix} \tau_{11} & \tau_{12} \\ \tau_{12} & \tau_{22} \end{pmatrix}, \tag{4.28}$$

valued in the Siegel upper half-plane \mathbb{H}_2 . We refer to Appendix A.2 for a precise definition and references. The moduli space of genus two curves \mathcal{M}_2 is complex three-dimensional. Since the SW order parameters u and v are two complex parameters, the SU(3) Coulomb branch maps out a complex two-dimensional space $\mathcal{U} \subset \mathcal{M}_2$ in the moduli space of genus two curves. In other words, \mathcal{U} is a divisor of \mathcal{M}_2 .

To relate the Rosenhain curve (4.25) to the Seiberg-Witten curve (4.8), note that a degree 5 polynomial as in (4.25) can be obtained by a linear fractional transformation of a degree 6 hyperelliptic equation $y^2 = \prod_{j=1}^{6} (x - r_j)$, which maps three of the roots to ∞ , 0 and 1. Linear fractional maps leave cross-ratios invariant, which is a convenient way to relate the λ_j to u and v. Let us define the cross-ratio of four points $z_i \in \mathbb{CP}^1$ as

$$C(z_1, z_2, z_3, z_j) = \frac{(z_1 - z_3)(z_2 - z_j)}{(z_1 - z_j)(z_2 - z_3)},$$
(4.29)

such that $C(\{\infty, 0, 1, \lambda_j\}) = \lambda_j$.

Note that we have 120 different possibilities to map three roots among the $\{r_j\}$ to $0, 1, \infty$, and another 3! possibilities to identify the three cross-ratios in the hyperelliptic setting with the λ_j . By studying the large u expansions of these for non-zero v, one can easily identify which cross-ratios, in terms of the r_i , correspond to which λ_j . To this end, let $\alpha = e^{2\pi i/3}$ as before. The roots of the rhs of (4.8) are then given by (with $\Lambda = 1$)

$$\begin{aligned} r_1 &= s_+(u,v+1) + s_-(u,v+1), & r_4 &= s_+(u,v-1) + s_-(u,v-1), \\ r_2 &= \alpha \, s_+(u,v+1) + \alpha^2 \, s_-(u,v+1), & r_5 &= \alpha \, s_+(u,v-1) + \alpha^2 \, s_-(u,v-1), \\ r_3 &= \alpha^2 \, s_+(u,v+1) + \alpha \, s_-(u,v+1), & r_6 &= \alpha^2 \, s_+(u,v-1) + \alpha \, s_-(u,v-1), \\ \end{aligned}$$

$$(4.30)$$

where

$$s_{\pm}(u,v) = \sqrt[3]{\frac{v}{2} \pm \sqrt{\frac{v^2}{4} - \frac{u^3}{27}}}.$$
 (4.31)

To simplify notation, let us set $s_{\pm\pm} \coloneqq s_{\pm}(u, v \pm 1)$. The large u, small v

expansions for the roots are

$$r_{1} = \sqrt{u} + \frac{1+v}{2u} + \dots, \qquad r_{4} = \sqrt{u} - \frac{1-v}{2u} + \dots,$$

$$r_{2} = -\sqrt{u} + \frac{1+v}{2u} + \dots, \qquad r_{5} = -\sqrt{u} - \frac{1-v}{2u} + \dots, \qquad (4.32)$$

$$r_{3} = -\frac{1+v}{u} + \dots, \qquad r_{6} = \frac{1-v}{u} + \dots$$

Plugging the weak-coupling expansions (4.21) into the Rosenhain invariants gives the leading behaviour for the λ_j . From this we can see that each invariant λ_j approaches 1 in the large u limit.

We continue by determining which of the 720 possible sets of cross-ratios matches with the theta constants. We have to determine which roots correspond to the first three points z_i , i = 1, 2, 3, in the cross-ratio (4.29). Since the three theta constants approach 1 in the large u limit, we should take for $\{z_1, z_2\}$ in (4.29) the roots which vanish in this limit, thus $\{r_3, r_6\}$. Together with the choice of z_2 , this reduces to 8 possible triplets. From a further comparison between the Rosenhain invariants and the cross-ratios, we determine that $z_1 = r_6$, $z_2 = r_3$ and $z_3 = r_2$. With $C_j := C(r_6, r_3, r_2, r_j)$ for j = 1, 4 and 5, we arrive at

$$\lambda_1 = C_5, \qquad \lambda_2 = C_1, \qquad \lambda_3 = C_4. \tag{4.33}$$

These are three equations for five unknowns, namely τ_{11} , τ_{12} , τ_{22} , u and v. To make it more manifest that the right hand side depends on only two variables, let us express the cross-ratios C_i in terms of $s_{\pm\pm}$,

$$C_{1} = \alpha^{2} \frac{[\alpha s_{+-} + s_{--} - s_{++} - \alpha s_{-+}] [s_{++} - \alpha s_{-+}]}{[\alpha^{2} s_{+-} + \alpha s_{--} - s_{++} - s_{-+}] [s_{-+} - s_{++}]},$$

$$C_{4} = -\frac{[\alpha s_{+-} + s_{--} - s_{++} - \alpha s_{-+}] [\alpha^{2} s_{++} + \alpha s_{-+} - s_{+-} - s_{--}]}{3[s_{+-} - \alpha s_{--}] [s_{-+} - s_{++}]},$$

$$C_{5} = -\alpha^{2} \frac{[\alpha s_{+-} + s_{--} - s_{++} - \alpha s_{-+}] [\alpha s_{++} + s_{-+} - s_{+-} - \alpha s_{--}]}{3[s_{--} - s_{+-}] [s_{-+} - s_{++}]}.$$
(4.34)

Note that these expressions are true on the full moduli space. For $u \neq 0$, we can define

$$X = \frac{s_{++}}{\sqrt{u/3}}, \qquad Y = \frac{s_{+-}}{\sqrt{u/3}}, \tag{4.35}$$

such that $X^{-1} = s_{-+}/\sqrt{u/3}$ and $Y^{-1} = s_{--}/\sqrt{u/3}$, since $s_{+\pm}s_{-\pm} = u/3$. The cross-ratios can then be expressed as

$$C_{1} = -\alpha^{2} \frac{X(X - \alpha Y)(X - Y^{-1})(X - \alpha X^{-1})}{(X^{2} - 1)(X - \alpha^{2}Y)(X - \alpha Y^{-1})},$$

$$C_{4} = -\frac{1}{3}\alpha^{2} \frac{(X - \alpha Y)^{2}(X - Y^{-1})(X - \alpha Y^{-1})}{X(X^{2} - 1)(Y - \alpha Y^{-1})},$$

$$C_{5} = -\frac{1}{3} \frac{(X - \alpha Y)(X - Y^{-1})^{2}(X - \alpha^{2}Y)}{X(X^{2} - 1)(Y - Y^{-1})}.$$
(4.36)

We thus see that the Coulomb branch can be identified with the zero-locus of the three equations (4.36) inside the space $(\lambda_1, \lambda_2, \lambda_3, X, Y)$. One may in principle eliminate X and Y to arrive at a single equation in terms of the λ_j . In the following two sections, we will restrict to the two one-dimensional sub-loci \mathcal{E}_u and \mathcal{E}_v of the solution space of (4.33), where v = 0 and u = 0 respectively.

4.3 Locus \mathcal{E}_u : v = 0

In this section we analyse the locus v = 0. We will demonstrate that the order parameter u can be expressed in terms of classical modular forms on this locus. In fact, we will arrive at two distinct expressions depending on a choice of effective coupling. In Section 4.5, we will discuss these aspects from the geometric point of view.

4.3.1 Algebraic relations

On the locus v = 0 we have that $\tau_{11}(u, 0) = \tau_{22}(u, 0)$ and $\tau_{12}(u, 0)$ is given by (4.22). Let us analyse these coupling constants, now from the perspective of Section 4.2.3. For u large and positive, s_{\pm} has a large magnitude and phase $e^{\pi i/6}$. Similarly, the phase of s_{\pm} is approximately given by $e^{-\pi i/6}$. This means that

$$s_{--} = -\alpha s_{++}, \qquad s_{+-} = -\alpha^2 s_{-+}, \qquad X = -\alpha^2 Y^{-1}.$$
 (4.37)

Using this and (4.35), we find that (4.36) now turns into

$$C_{1} = -\frac{(X + X^{-1})(X - \alpha X^{-1})}{(X - X^{-1})(X + \alpha X^{-1})},$$

$$C_{4} = -\frac{1}{3}\frac{(X + X^{-1})^{2}}{(X - X^{-1})^{2}},$$

$$C_{5} = +\frac{1}{3}\frac{(X + X^{-1})(X + \alpha X^{-1})}{(X - \alpha X^{-1})(X - X^{-1})}.$$
(4.38)

Since the rhs of (4.38) depends only on one variable X, the cross-ratios C_j satisfy two algebraic equations, which can be determined by solving the equations for X^2 . One finds

$$C_1 C_5 - C_4 = 0,$$

(3C₄ - C₁)² - C₄(C₁ + 1)² = 0. (4.39)

Using (4.33) and (4.26), the cross-ratios are identified with quotients of Siegel theta functions (see Appendix A.2), and the above equations take the form

$$0 = \Theta_3^4 - \Theta_4^4,$$

$$0 = \Theta_1^2 \Theta_2^2 \Theta_8^4 \Theta_3^4 - \Theta_2^4 \Theta_8^2 \Theta_{10}^2 \Theta_3^4 + 8 \Theta_1^2 \Theta_2^2 \Theta_4^2 \Theta_8^2 \Theta_{10}^2 \Theta_3^2 + \Theta_1^2 \Theta_2^2 \Theta_4^4 \Theta_{10}^4 - 9 \Theta_1^4 \Theta_4^4 \Theta_8^2 \Theta_{10}^2.$$
(4.40)

The two systems of equations above are equivalent given that none of the λ_j vanish or are infinite, which is an assumption of Picard's lemma (4.26). We

can use the second relation of (4.38) to solve for u,

$$u^{3} = \frac{\sqrt{27}}{2} \frac{(3C_{4}+1)^{3}}{\sqrt{C_{4}}(C_{4}-1)},$$
(4.41)

and in terms of theta constants this gives

$$u^{3} = \frac{\sqrt{27}}{2} \frac{(3\Theta_{1}^{2}\Theta_{8}^{2} + \Theta_{2}^{2}\Theta_{10}^{2})^{3}}{\Theta_{1}\Theta_{2}^{3}\Theta_{8}\Theta_{10}^{3}(\Theta_{1}^{2}\Theta_{8}^{2} - \Theta_{2}^{2}\Theta_{10}^{2})}.$$
(4.42)

This can be viewed as a generalisation of the rank 1 result (1.35), in the sense that we can write the parameter u as a rational function of theta series. It follows naively that u transforms as a weight 0 function under a subgroup of $Sp(4, \mathbb{Z})$.

4.3.2 A modular expression for u

The solutions to the algebraic relations (4.40) are not unique due to the periodicity in the τ_{IJ} . The first equation implies $\tau_{11} - \tau_{22} = 2k$ with $k \in \mathbb{Z}$, but we know from (4.21) that k = 0. From (4.22) we can make a power series expansion for τ_{12} in terms of $p = e^{2\pi i \tau_{11}}$. One finds

$$\tau_{12} = -\frac{1}{2}\tau_{11} - \frac{1}{2\pi i}\log(8) + \frac{1}{2\pi i}\frac{27}{4}h(p), \qquad (4.43)$$

with

$$h(p) = p^{\frac{1}{2}} - \frac{63}{16}p + \frac{1447}{64}p^{\frac{3}{2}} - \frac{307679}{2048}p^2 + \mathcal{O}(p^{\frac{5}{2}}), \qquad (4.44)$$

by satisfying the second relation in (4.40) order by order. Substitution of (4.43) in (4.41) gives the following *p*-expansion for *u*,

$$u = \frac{1}{2}p^{-\frac{1}{6}} + \frac{43}{8}p^{\frac{1}{3}} - \frac{2923}{128}p^{\frac{5}{6}} + \frac{1713}{16}p^{\frac{4}{3}} + \mathcal{O}(p^{\frac{11}{6}}).$$
(4.45)

One can verify agreement with the Picard-Fuchs approach by substituting this expansion in Eq. (4.21). As this series is only an expansion for small p, it is not very elucidating. To arrive at a closed expression, we aim to express u as a function of a "coupling constant" which transforms well under the duality transformations. This is not the case for τ_{11} .

However when $\tau_{11} = \tau_{22}$, the inversion $\mathcal{S} = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \in Sp(4, \mathbb{Z})$ acts naturally on the linear combinations $\tau_{\pm} = \tau_{11} \pm \tau_{12}$, which are in one-to-one correspondence with τ_{11} and τ_{12} . From (A.24), we deduce for the action of \mathcal{S} on τ_{\pm}

$$S: \tau_{11} \pm \tau_{12} \mapsto -\frac{1}{\tau_{11} \pm \tau_{12}}.$$
(4.46)

That is to say, it reduces to the ordinary S-transformation $\tau_{\pm} \mapsto -1/\tau_{\pm}$. Moreover, $\tau_{\pm} \in \mathbb{H}$ for both \pm . To see this note that since Im(Ω) is positive definite, we have that $y_{11} > 0$ and $y_{11}y_{22} - y_{12}^2 > 0$, where $y_{IJ} = \text{Im}(\tau_{IJ})$. Whenever $y_{11} = y_{22}$, the latter inequality implies that $y_{11}^2 > y_{12}^2$. Since $y_{11} > 0$, it implies $y_{11} > y_{12}$ and $y_{11} > -y_{12}$ simultaneously. From this we learn that $y_{11} - y_{12}$ and $y_{11} + y_{12}$ are both positive and therefore $\tau_{\pm} := \tau_{11} \pm \tau_{12} \in \mathbb{H}$.

We will proceed by considering $\tau_{-} \rightleftharpoons \tau$, leaving the discussion on τ_{+} for section 4.3.3. To determine u as function of τ , one can first find the series expansion for τ in terms of p, invert and substitute $p(\tau)$ in (4.45). Alternatively, one can revert to the Picard-Fuchs solution, by inverting the series (4.21) for v = 0,

$$q = e^{2\pi i(\tau_{11}(u) - \tau_{12}(u))} = U^3 + 45U^4 + 1512U^5 + 45672U^6 + \dots, \qquad U = \frac{1}{4u^3}.$$
(4.47)

Either method gives us the following series for u,

$$\sqrt[3]{4}u = q^{-\frac{1}{9}} + 5q^{\frac{2}{9}} - 7q^{\frac{5}{9}} + 3q^{\frac{8}{9}} + 15q^{\frac{11}{9}} - 32q^{\frac{14}{9}} + \mathcal{O}(q^{\frac{17}{9}}).$$
(4.48)

This expansion is also known as the McKay-Thompson series of class 9B for the Monster group [36, 56-58]. Thus similarly to the *u* for rank 1 (1.35), we find a McKay-Thompson series. We then have

$$u = u_{-}(\tau) = \sqrt[3]{\frac{27}{4}} \frac{b_{3,0}\left(\frac{\tau}{3}\right)}{b_{3,1}\left(\frac{\tau}{3}\right)},\tag{4.49}$$

where $b_{3,j}$ are theta series for the A_2 root lattice,

$$b_{3,j}(\tau) = \sum_{k_1,k_2 \in \mathbb{Z} + \frac{j}{3}} q^{k_1^2 + k_2^2 + k_1 k_2}, \quad j \in \{-1,0,1\}.$$
 (4.50)

The theta series $b_{3,i}$ transform under the generators of $SL(2,\mathbb{Z})$ as $(\alpha = e^{2\pi i/3})$

$$S: \quad b_{3,j}\left(-\frac{1}{\tau}\right) = -\frac{i\tau}{\sqrt{3}} \sum_{l \mod 3} \alpha^{2jl} b_{3,l}(\tau),$$

$$T: \quad b_{3,j}(\tau+1) = \alpha^{j^2} b_{3,j}(\tau).$$
(4.51)

The solution u_{-} can also be expressed in terms of the Dedekind η -function (A.18) as

$$u_{-}(\tau) = \sqrt[3]{\frac{27}{4}} \left(1 + \frac{1}{3} \frac{\eta \left(\frac{\tau}{9}\right)^3}{\eta(\tau)^3} \right).$$
(4.52)

Using Theorem 1 in Appendix A.1, one finds that $u_{-}(9\tau)$ is a modular function for the congruence subgroup $\Gamma_0(9)$ (also defined in Appendix A.1). This implies that u is a modular function for $\Gamma^0(9)$, which is generated by the matrices T^9 , STS and $(T^3S)T(T^3S)^{-1}$. In fact, it is easy to see from (4.51) that $u_{-}(\tau - 3) = \alpha u_{-}(\tau)$ for all $\tau \in \mathbb{H}$. Furthermore, u rotates as well under TST^{-2} , $u_{-}(\frac{\tau-3}{\tau-2}) = \alpha u_{-}(\tau)$. The two elements T^3 and TST^{-2} generate $\Gamma^0(3)$ and u can therefore be interpreted as a modular function for $\Gamma^0(3)$ with multipliers α^k . Let us analyse the strong coupling singularities $u^3 = \frac{27}{4}$ for v = 0 in terms of the variable τ . We will demonstrate that these correspond to $\tau \to 0, 3$ and -3. Using (4.51), one finds that the expansion around 0 takes the form

$$\sqrt[3]{\frac{4}{27}} u_{-,D}(\tau_D) = \frac{b_{3,0}(3\tau_D) + 2b_{3,1}(3\tau_D)}{b_{3,0}(3\tau_D) - b_{3,1}(3\tau_D)}$$

$$= 1 + 9 q_D + 27 q_D^2 + 81 q_D^3 + 198 q_D^4 + \mathcal{O}(q_D^5),$$
(4.53)

with $\tau_D = -1/\tau$, $q_D = e^{2\pi i \tau_D}$ and $u_{-,D}(\tau_D) := u_{-}(-1/\tau_D)$. In the same notation we can invert the series to find

$$q_D = \chi - 3\chi^2 + 9\chi^3 - 22\chi^4 + 21\chi^5 + 207\chi^6 + \mathcal{O}(\chi^7), \qquad (4.54)$$

where $\chi := (\sqrt[3]{4/27} u - 1)/9$. It follows that $q_D \to 0$ for $\sqrt[3]{4/27} u \to 1$ or $\chi \to 0$. This can be directly confirmed by analytically continuing the Picard-Fuchs expansion around $u = \sqrt[3]{27/4}$.

The expansion around ± 3 can then be obtained from the one around 0 by shifting the argument $\tau_{D,\pm} = -\frac{1}{\tau} \pm 3$, and one finds using the *T*-transformation (4.51) that

$$u_{-,D}(\tau_{D,\pm}) = \alpha^{\pm 1} \sqrt[3]{\frac{27}{4}} \frac{b_{3,0}(3\tau_D) + 2b_{3,1}(3\tau_D)}{b_{3,0}(3\tau_D) - b_{3,1}(3\tau_D)}$$
(4.55)

The expansions around the points 3 and -3 differ from the one around 0 only by the phases $\alpha^{-1} = \alpha^2$ and α . Together with (4.53), this proves that indeed $\tau \to \{0, -3, 3\}$ corresponds to the three singularities $\underline{u} \to \{1, \alpha, \alpha^2\}$. Due to the T^9 -invariance of the solution (4.49), there is an ambiguity in identifying the τ -parameter with $\tau + 9\mathbb{Z}$. These \mathbb{Z}_2 points are studied in detail in [215, 220]. They correspond to the 3 vacua of the $\mathcal{N} = 1$ theory after deforming the $\mathcal{N} = 2$ theory by relevant or marginal terms.

The modular analysis is completely analogous to the SU(2) theory, as reviewed in section 1.5: The cusps of $\Gamma^0(9)$ are $\{0, -3, 3, i\infty\}$, which is exactly where u assumes the \mathbb{Z}_2 vacua and the semi-classical limit. The fundamental domain of $\Gamma^0(9)$ is given in Figure 29 and is the union of 12 images of the SL(2, \mathbb{Z}) fundamental domain $\mathcal{F} = \text{SL}(2, \mathbb{Z}) \setminus \mathbb{H}$,

$$\Gamma^{0}(9) \backslash \mathbb{H} = \bigcup_{\ell=-4}^{4} T^{\ell} \mathcal{F} \cup S \mathcal{F} \cup T^{3} S \mathcal{F} \cup T^{-3} S \mathcal{F}.$$
(4.56)

Using (4.52), we can find the exact coupling at the origin of the moduli space. We have that $u(\tau_0) = 0$ for the $\Gamma^0(9)$ orbit of

$$\tau_0 = \sqrt{3}\,\omega = \frac{3}{2} + \frac{\sqrt{3}}{2}i,\tag{4.57}$$

with $\omega = e^{\pi i/6}$. The point τ_0 lies on the boundary of the fundamental domain, on the point where the boundary arcs from different cusps meet. The elements $(STS)^k \in \Gamma^0(9) \text{ map } \tau_0 \mapsto \tau_0 - 3k$ for integer k, which identifies the "corners"



Figure 29: Fundamental domain $\Gamma^{0}(9) \setminus \mathbb{H}$ of the congruence subgroup $\Gamma^{0}(9)$. It consists of 12 images of the key-hole fundamental domain \mathcal{F} .

in Figure 29. This is compatible with the global \mathbb{Z}_3 symmetry, which also acts by T^{-3} and leaves the origin invariant. It is in complete analogy to the SU(2) picture: We find the nice picture that the cusps of $\Gamma^0(9) \setminus \mathbb{H}$ are in one-to-one correspondence with the singularities $u^3 = \frac{27}{4}$ and $u = \infty$ and the origin is the symmetric point where the boundary arcs meet.

4.3.3 *u* as a sextic modular function

While we chose in the above the modular parameter $\tau_{-} = \tau_{11} - \tau_{12}$, Equation (4.46) shows that we could equally well consider $\tau_{+} = \tau_{11} + \tau_{12}$. We will consider the variable $\tau \coloneqq \tau_{+}$ in this subsection. We can determine the first terms in the *q*-expansion of *u*, which results in

$$u = u_{+}(\tau) = \frac{1}{4} \left(q^{-1/3} + 104 \, q^{2/3} - 7396 \, q^{5/3} + \mathcal{O}(q^{8/3}) \right). \tag{4.58}$$

This series can be recognised as the q-expansion of

$$u_{+}(\tau) = \sqrt[3]{\frac{27}{2}} \frac{E_4(\tau)^{1/2}}{(E_4(\tau)^{3/2} - E_6(\tau))^{1/3}},$$
(4.59)

where E_4 and E_6 are the Eisenstein series (1.7). We will derive this explicitly in section 4.5. The function u_+ is a root of the sextic equation

$$X^{6} - \frac{j(\tau)}{64} X^{3} + \frac{27 \, j(\tau)}{256} = 0, \qquad (4.60)$$

where j is the j-invariant (A.9). Since the coefficients of this sextic equation are modular functions for SL(2, Z), we call u_+ a sextic modular function (see also Appendix A.5). Due to the fractional powers in (4.59), u_+ is not a modular function for SL(2, Z). In fact, $E_4^{1/2}$ and u_+ are not invariant under any subgroup of SL(2, Z). One way to see this is that E_4 has a simple zero for $\tau = \alpha$, such that the square root introduces a branch cut. While the family of sextic modular functions thus includes functions which are not modular for $SL(2,\mathbb{Z})$, this family also includes functions which are modular for an index 6 congruence subgroup of $SL(2,\mathbb{Z})$. The order parameter for SU(2) (1.35) is an example of the latter. One can thus view the family of sextic modular functions as an extension of the family of modular functions for index 6 congruence subgroups.

Interestingly, u_+ is up to an overall factor the same function as the order parameter of the massless $N_f = 1$ theory with gauge group SU(2) [101,103,221], see (2.66). This aspect distinguishes massless $N_f = 1$ from $N_f = 0, 2, 3$, since for the latter theories the order parameters are modular functions for congruence subgroups isomorphic to $\Gamma^0(4)$ [101]. On the other hand, it is known since the time of Fricke and Klein that similar fractional powers of modular forms as in u_+ do appear in the context of Picard-Fuchs equations and hypergeometric functions [222, 223].

As mentioned before, the fractional powers in (4.59) are incompatable with any subgroup of $SL(2, \mathbb{Z})$. Nevertheless, if we choose a basepoint, we can show that u_+ is invariant under transformations of τ , which combine to a closed trajectory with starting and endpoint equal to the base point. We choose the base point τ_b with $\operatorname{Re}(\tau_b) = 0$ and $\operatorname{Im}(\tau_b) \gg 1$. First, using the modular transformation of E_4 and E_6 , we find for the expansion of τ near 0,

$$\tau \to 0: \qquad u_+(\tau) = u_{+,D}(-1/\tau),$$
(4.61)

with

$$u_{+,D}(\tau_D) = \sqrt[3]{\frac{27}{2}} \frac{E_4(\tau_D)^{1/2}}{(E_4(\tau_D)^{3/2} + E_6(\tau_D))^{1/3}}$$

$$= \sqrt[3]{\frac{27}{4}} \left(1 + 144 \, q_D - 3456 \, q_D^2 + 596160 \, q_D^3 + \dots \right).$$
(4.62)

The S-transform $u_{+,D}$ is also a solution to (4.60) and thus also a sextic modular function. From Eq. (4.58) we see that u_+ is invariant under $\tau \mapsto \tau + 3$ at weak coupling, $\text{Im}(\tau) \gg 1$. Let us introduce T_w for the translation at weak coupling. Moreover at strong coupling, $0 < \text{Im}(\tau) \ll 1$, u_+ is invariant under $\tau_D = -1/\tau \mapsto \tau_D + 1$. Let us introduce T_s for the translation at strong coupling. We can get the monodromies around the other cusps, $\tau = \pm 1$ from conjugation with T_w . We then find that u_+ is left invariant by

$$T_w^{3n}, \qquad (T_w^{\ell}S)T_s(T_w^{\ell}S)^{-1}, \qquad \ell, n \in \mathbb{Z},$$
 (4.63)

where S is the usual inversion $\tau \mapsto -1/\tau$, mapping τ from weak to strong coupling. These transformations are sketched in Figure 30 for n = 1 and $\ell = 0, \pm 1$.

We denote the invariance group of u_+ by Γ_{u_+} . It is generated by the elements in (4.63) with n = 1, and $\ell = 0, 1$. From the invariance under (4.63), one derives that a fundamental domain is given by

$$\bigcup_{\ell=-1}^{1} T^{\ell} \mathcal{F} \cup T^{\ell} S \mathcal{F}.$$
(4.64)

It consists of six copies of \mathcal{F} , which is directly related to u_+ being a sextic modular function. This fundamental domain is the grey area in Figure 30. The domain is clearly topologically equivalent to the fundamental domain in Figure 29. The expansions of u_+ and $u_{+,D}$ demonstrate that $u_+(i\infty) = \infty$,



Figure 30: Fundamental domain for u_+ . The vertical lines at $\tau = \pm 3/2$ are identified, as well as each pair of the two arcs meeting at a cusp -1, 0 or 1. The point τ_b is the base point for the monodromies, which are compositions of T_w , T_s and S. T_w is a shift $\tau \mapsto \tau + 1$ at weak coupling, T_s circles around a strong coupling cusp, and S maps τ from weak to strong coupling.

 $u_+(0) = \sqrt[3]{\frac{27}{4}}$ and $u_+(\pm 1) = \alpha^{\mp} \sqrt[3]{\frac{27}{4}}$. We will derive u_+ from the SW geometry in section 4.5.

Because u_+ is not a weakly holomorphic modular form, but involves fractional powers of modular forms, it is problematic to identify the transformations (4.63) with elements of SL(2, Z). One way to see that this identification is problematic is that the composition of S, T_w and T_s does not satisfy the relation $(ST)^3 = -1$, if we identify $T_w = T_s = T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$. To further study this aspect, let us list the $SL(2,\mathbb{Z})$ matrices corresponding to (4.63),

$$T^{3} = \begin{pmatrix} 1 & 3 \\ 0 & 1 \end{pmatrix},$$

$$STS^{-1} = \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix},$$

$$(TS)T(TS)^{-1} = \begin{pmatrix} 0 & 1 \\ -1 & 2 \end{pmatrix},$$

$$T^{-1}S)T(T^{-1}S)^{-1} = \begin{pmatrix} 2 & 1 \\ -1 & 0 \end{pmatrix}.$$
(4.65)

These matrices fix each of the cusps $\{\infty, 0, 1, -1\}$. On the other hand, u_+ is not invariant under the modular action of the matrices on τ , $\tau \mapsto (a\tau + b)/(c\tau + d)$ except for T^{3n} . For example, STS^{-1} would map $\tau = i\infty$ to -1. The values of u_+ are however different for these two arguments: $u_+(i\infty) = \infty$ and $u_+(-1) = \alpha \sqrt[3]{\frac{27}{4}}$. Furthermore, the matrices (4.65) generate the full modular group SL(2, \mathbb{Z}).

The origin $u_+(\tau_0) = 0$ of the moduli space is again given by the points where the boundary arcs meet: At $\tau_0 = \alpha$ we have that E_4 vanishes but E_6 does not. From (4.59) it is then clear that $\tau_0 + \mathbb{Z}$ are indeed the zeros of u_+ . This is also compatible with the \mathbb{Z}_3 global symmetry, which according to (4.58) acts as T^{-1} and leaves the origin invariant.

4.4 Locus \mathcal{E}_v : u = 0

(

We will now consider the second elliptic locus, namely where u = 0. By doing a similar analysis as in section 4.3 but now for large v, we find that the correct matching between the cross-ratios and the Rosenhain invariants for this limit is

$$\lambda_1 = C_5, \qquad \lambda_2 = C_4, \qquad \lambda_3 = C_1. \tag{4.66}$$

Note that the only difference from before is that the rôles of λ_2 and λ_3 have been interchanged. One could perform a change of symplectic basis to have the same matching as (4.33). This can be be done by acting on the periods with the matrix $\mathcal{T}_{\theta} = \begin{pmatrix} 1 & \theta \\ 0 & 1 \end{pmatrix} \in Sp(4,\mathbb{Z})$ with $\theta = \begin{pmatrix} -1 & 2 \\ 2 & -4 \end{pmatrix}$.²⁶ This would however also change the Rosenhain form, and we therefore prefer to continue with the identification in (4.66).

We will proceed by deriving the relations satisfied by the couplings τ_{IJ} on the locus u = 0.

4.4.1 Algebraic relations

To determine the algebraic relations among the theta constants, we assume that v is real, large and positive. In this limit we find that $s_{\pm} = \sqrt[3]{v \pm 1}$ and

²⁶Note that there is an ambiguity in the choice of \mathcal{T}_{θ} . The λ_j are invariant under a subgroup of Sp(4, \mathbb{Z}). Multiplying \mathcal{T}_{θ} with an element of this group thus gives the same result.

 $s_{\pm} = 0$. The cross-ratios (4.34) simplify, and one finds

$$C_{1} = -\alpha^{2} \frac{s_{++} - \alpha s_{+-}}{s_{++} - \alpha^{2} s_{+-}},$$

$$C_{4} = -\frac{\alpha^{2}}{3} \frac{(s_{++} - \alpha s_{+-})^{2}}{s_{++} s_{+-}},$$

$$C_{5} = +\frac{1}{3} \frac{(s_{++} - \alpha s_{+-})(s_{++} - \alpha^{2} s_{+-})}{s_{++} s_{+-}}.$$
(4.67)

From this we find two algebraic relations between the cross-ratios, namely

$$C_1 C_5 - C_4 = 0,$$

$$C_5^2 + C_4^2 - C_5 C_4 - C_4 = 0.$$
(4.68)

Writing these in terms of the theta constants, we have

$$0 = \Theta_1^4 - \Theta_2^4,$$

$$0 = \Theta_2^4 \Theta_3^2 \Theta_8^4 + \Theta_1^4 \Theta_3^2 \Theta_{10}^4 - \Theta_1^2 \Theta_2^2 \Theta_3^2 \Theta_8^2 \Theta_{10}^2 - \Theta_2^4 \Theta_8^2 \Theta_{10}^2.$$
(4.69)

4.4.2 Modular expression for v

Our next aim is to determine a modular expression for v on this elliptic locus. The first relation in (4.69) implies $\tau_{11} = \tau_{22} + 2\mathbb{Z} + 1$, while the second one implies $\tau_{12} = \pm \frac{1}{2}\tau_{11} + \mathbb{Z}$. We claim that these are all the solutions. As in the case v = 0, the PF solution (4.24) fixes these relations,

$$\tau_{11} = \tau_{22} + 1, \quad \tau_{12} = -\frac{\tau_{11}}{2} + 1.$$
 (4.70)

In contrast to the locus \mathcal{E}_u , these linear relations between the τ_{11} , τ_{22} and τ_{12} are exact on \mathcal{E}_v . Using the first equation in (4.67), we can solve for v,

$$v = -\frac{i}{\sqrt{27}} \frac{(C_1 - 2)(C_1 + 1)(2C_1 - 1)}{C_1(C_1 - 1)}.$$
(4.71)

This can again be written as a rational function of Siegel theta functions,

$$v = -\frac{i}{\sqrt{27}} \frac{(\Theta_8^2 - 2\Theta_{10}^2)(\Theta_8^2 + \Theta_{10}^2)(2\Theta_8^2 - \Theta_{10}^2)}{\Theta_8^2 \Theta_{10}^2(\Theta_8^2 - \Theta_{10}^2)}.$$
 (4.72)

As a function of $\tau_{-} = \tau_{11} - \tau_{12}$, one finds $(q_{-} = e^{2\pi i \tau_{-}})$

$$v = \frac{i}{2\sqrt{27}} \left(\alpha \, q_{-}^{-\frac{1}{6}} - 33 \, \alpha^2 \, q_{-}^{\frac{1}{6}} - 153 \, q_{-}^{\frac{1}{2}} - 713 \, \alpha \, q_{-}^{\frac{5}{6}} + \mathcal{O}(q_{-}^{\frac{7}{6}}) \right). \tag{4.73}$$

The expansion in terms of $\tau_{+} = \tau_{11} + \tau_{12}$ is very similar. One can recognise these series as

$$v = \frac{i}{2\sqrt{27}} m(\frac{\tau_{+}}{2}),$$

$$v = \frac{i}{2\sqrt{27}} m(\frac{\tau_{-}}{6} + \frac{2}{3}),$$
(4.74)

where

$$m(\tau) = \left(\frac{\eta(2\tau)}{\eta(6\tau)}\right)^6 - 27 \left(\frac{\eta(6\tau)}{\eta(2\tau)}\right)^6$$

$$= q^{-1} - 33 q - 153 q^3 - 713 q^5 - 2550 q^7 - 7479 q^9 + \mathcal{O}(q^{11}).$$
(4.75)

The function m is known in the literature as the completely replicable function of class 6a [56–58]. The perturbative expansion (4.73) can also be verified from the Picard-Fuchs solution by starting from Eq. (4.24) and setting u = 0. Then, expand $q = e^{2\pi i (\tau_{11}(v) - \tau_{12}(v))}$ as a series in v and invert it to find (4.73).

4.4.3 The \mathbb{Z}_3 vacua

Let us study the solution (4.74) near the strong coupling vacua. To this end, we eliminate the phases in (4.73) by substitution of $\tau := \tau_{-} + 1$ in (4.74). In the new variable τ , the solution reads

$$v = -\frac{i}{2\sqrt{27}} m\left(\frac{\tau}{6}\right). \tag{4.76}$$

It can be shown that the values of τ at the Argyres-Douglas (AD) vacua $v_{\text{AD},1} = 1$ and $v_{\text{AD},2} = -1$ are $(\omega = e^{\pi i/6})$

$$\tau_{\text{AD},1} = -\frac{3}{2} + \frac{\sqrt{3}i}{2} = \sqrt{3}\,\omega^5,$$

$$\tau_{\text{AD},2} = +\frac{3}{2} + \frac{\sqrt{3}i}{2} = \sqrt{3}\,\omega,$$
(4.77)

and the origin (u, v) = (0, 0) is located at $\tau_0 = \sqrt{3}i$. This can be rigorously using the properties of m.

The solutions to v = 1 and v = -1 are not straightforward to obtain. Let us start with the point $(\underline{u}, v) = (0, -1)$. In the following, all arguments are those of m. Due to the prefactor in (4.76), v = -1 is in fact a quadratic equation with zero discriminant and therefore satisfied if and only if

$$\left(\frac{\eta(2\tau)}{\eta(6\tau)}\right)^6 = -\sqrt{27}\,i.\tag{4.78}$$

A solution to this equation can be found to be

$$\tau_{-1} = \frac{\omega}{2\sqrt{3}} = \frac{1}{4} + \frac{i}{4\sqrt{3}} = \frac{\tau_{\text{AD},2}}{6}, \qquad (4.79)$$

with $\omega = e^{\pi i/6}$ as before and $\tau_{AD,2}$ the argument of v in (4.77). The other AD point can be found using the symmetry of m, and it is given by

$$\tau_{+1} = \frac{\omega^5}{2\sqrt{3}} = -\frac{1}{4} + \frac{i}{4\sqrt{3}} = \frac{\tau_{\text{AD},1}}{6}.$$
(4.80)

The zero of m (and therefore of v) is given by

$$\tau_0 = \frac{i}{2\sqrt{3}}.\tag{4.81}$$

Note that all these numbers have the same absolute value $\frac{1}{2\sqrt{3}}$.

Let us prove (4.79) first: In order to compute both the numerator and the denominator, we can resort to the S- and T-transformations of η as given in A.1,

$$\eta(2\tau_{-1}) \stackrel{S}{=} 3^{\frac{1}{4}} \omega e^{-\frac{\pi i}{12}} \eta(-\frac{1}{2} + \frac{\sqrt{3}}{2}i) \stackrel{T}{=} 3^{\frac{1}{4}} e^{\frac{\pi i}{12}} \eta(\alpha)$$

$$\eta(6\tau_{-1}) = \eta(\frac{3}{2} + \frac{\sqrt{3}}{2}i) \stackrel{T}{=} e^{\frac{\pi i}{6}} \eta(\alpha).$$
(4.82)

Equation (4.78) follows immediately.

In order to find the point where v = +1, we can make the observation that $m(-\frac{1}{\tau}) = -m(\frac{\tau}{12})$. This implies that under the Fricke involution $\begin{pmatrix} 0 & -1 \\ 12 & 0 \end{pmatrix}$, the solution receives a minus sign,

$$m\left(-\frac{1}{12\tau}\right) = -m(\tau). \tag{4.83}$$

Using the *T*-transformation of η , one also finds that $m\left(\tau \pm \frac{1}{2}\right) = -m(\tau)$. We can use either of those maps, $\tau_{+1} = \tau_{-1} - \frac{1}{2} = -\frac{1}{12\tau_{-1}}$ to obtain (4.80).

We can also study the zeros of v. Every root of $m(\tau)$ is given by the equation $\eta(2\tau)^{12} = 27 \eta(6\tau)^{12}$. A solution to this equation is (4.81), which we can prove: Using the S-transformation, we find

$$\eta(2\tau_0) = \eta(\frac{i}{\sqrt{3}}) = 3^{\frac{1}{4}}\eta(\sqrt{3}i) = 3^{\frac{1}{4}}\eta(6\tau_0).$$
(4.84)

The result follows immediately. Another proof follows simply from the fact that τ_0 is the fixed point under (4.83).

The modular group of v is closely related to the duality group of the SU(3) theory on this locus. It can be shown that v is a modular form for the principal congruence subgroup $\Gamma(6)$, as defined in Appendix A.1. However, the fundamental domain of this group has twelve cusps, and v diverges at all of them. This suggests that we found strongly coupled vacua in the region of the moduli space where v is large. But from the discriminant $\Delta_{\Lambda}|_{\mathcal{E}_v} = v^2 - 1$ we expect the only singularities to be at $v \in \{1, -1, \infty\}$.

To resolve this problem, let us study the function m in more detail. It is a linear combination of eta quotients, whose modular properties have been studied extensively [224,225]. Applying Theorem 1 in Appendix A.1, one finds that m is a modular function for the Hecke congruence subgroup $\Gamma_0(12)$. In addition, it satisfies the following non-SL(2, \mathbb{Z}) transformations

$$m\left(\tau - \frac{1}{2}\right) = -m(\tau), \qquad (4.85a)$$
$$m\left(-\frac{1}{2}\right) = -m(\tau) \qquad (4.85b)$$

$$m\left(-\frac{1}{12\tau}\right) = -m(\tau). \tag{4.85b}$$

The transformation (4.85b) is also known as a *Fricke involution*. Translating both equations to the argument of v, we find that v picks up a minus sign under both T^{-3} and $F = \begin{pmatrix} 0 & -3 \\ 1 & 0 \end{pmatrix}$. Taking products, we find that v is properly invariant under $FT^{-3} = \begin{pmatrix} 0 & -3 \\ 1 & -3 \end{pmatrix}$ and T^{-6} . Let us normalise the former to

 $X = \frac{1}{\sqrt{3}} \begin{pmatrix} 0 & -3 \\ 1 & -3 \end{pmatrix}$, and denote the subgroup of $PSL(2, \mathbb{R})$ generated by these two elements as

$$\Gamma_v = \left\langle X, T^{-6} \right\rangle. \tag{4.86}$$

This group is a proper subgroup of the modular group $\Gamma^0(6|2) + 3$ of Atkin-Lehner type, in the notation of [58]. This Atkin-Lehner group extends the ordinary congruence subgroup $\Gamma^0(\frac{6}{2})$ by elements in PSL(2, \mathbb{R}). See Appendix A.1 for the precise definition. If we allow for a non-trivial multiplier system, the modular group associated with m is $\Gamma^0(6|2) + 3$ [58]. The latter contains for example T^{-3} , under which we have shown that v is anti-invariant. We can write a similar set of matrices as (4.65),

$$M_{1} = \begin{pmatrix} -3 & -3 \\ 1 & 0 \end{pmatrix}, \quad M_{2} = \begin{pmatrix} 0 & 3 \\ -1 & 3 \end{pmatrix}, \quad M_{\infty} = \begin{pmatrix} 1 & -6 \\ 0 & 1 \end{pmatrix} = T^{-6},$$
(4.87)

under which $v \sim m(\tau/6)$ is invariant. If we consider their normalisation to unit determinant, $\Pi(M_j) := |\det(M_j)|^{-1/2} M_j$, they lie in the group Γ_v (4.86), and furthermore satisfy

$$\Pi(M_1)\Pi(M_2) = M_{\infty}.$$
(4.88)

A fundamental domain for Γ_v can be drawn using the algorithm given in [58], and it is shown in Figure 31. The element T^6 contains the domain to $|\text{Re }\tau| < 3$. X identifies the interior of the circle with radius $\sqrt{3}$ centered at 0, with a region inside the blue domain in Figure 31. Similarly, the interior of the circles centered at ± 3 is identified with a region of the blue domain. We conclude,

$$\Gamma_v \setminus \mathbb{H} = \{ z \in \mathbb{H} \mid |\text{Re } z| < 3 \} \setminus \bigcup_{\ell = -1}^{1} \overline{D}_{\sqrt{3}}(3\ell).$$
(4.89)

where $\overline{D}_r(c)$ is the closed disc of radius r and center c.

The Argyres-Douglas vacua v = 1 and v = -1 correspond to the special points $\tau_{\text{AD},j}$ (4.77). They are stabilised by M_1 and M_2 , respectively. This makes the AD vacua elliptic points of Γ_v . They are in fact expected to not get mapped to cusps of v, since their coupling matrix lies inside the Siegel upper half-space \mathbb{H}_2 [1]. This is a familiar property of superconformal points [104,155]. It is different from the \mathbb{Z}_2 points where the coupling matrices are located on the boundary $\partial \mathbb{H}_2$ and therefore mapped to the real line $\partial \mathbb{H}_1$. The origin $\tau_0 = \sqrt{3}i$ is mapped under FT^{-3} to $\tau_0 - 3$, which is identified with τ_0 since v = 0 is a fixed point under $T^{-3} : v \mapsto -v$. The anti-invariance under T^{-3} is in fact directly derived from the \mathbb{Z}_2 symmetry $\rho : v \mapsto e^{\pi i}v$ [1]. The large v monodromy ρ^2 acts on τ as T^{-6} , under which v is invariant. The origin of the Fricke involution can therefore be understood from the global structure on the u = 0 plane.

The discussion is similar for the parameter $\tau_{+} = \tau_{11} + \tau_{12}$. If we introduce here $\tau = \tau_{+} - 1$, v equals $-\frac{i}{2\sqrt{27}}m(\tau/2)$, which is again invariant under $\Gamma(6)$.



Figure 31: Fundamental domain $\Gamma_v \setminus \mathbb{H}$ for the group Γ_v . The values of the special points are: $\tau_{AD,1} = \sqrt{3} \omega^5$ and $\tau_{AD,2} = \sqrt{3} \omega$.

It is multiplied by a sign under T as well as under the Fricke involution $\tilde{F} = \begin{pmatrix} 0 & -1 \\ 3 & 0 \end{pmatrix}$. This means that it is invariant under T^2 together with the involution $\tilde{X} := \tilde{F}T^{-1} = \begin{pmatrix} 0 & -1 \\ 3 & -3 \end{pmatrix}$, which again generate an Atkin-Lehner type group. The fundamental domain of this group equals that in Figure 31, but with all points divided by 3.

4.5 Elliptic curves for \mathcal{E}_u and \mathcal{E}_v

It is natural to expect that the complexified couplings τ_{\pm} for both loci \mathcal{E}_u and \mathcal{E}_v have an interpretation as complex structures of elliptic curves. Moreover, these elliptic curves are expected to be related to the geometry of the genus two Seiberg-Witten curve (4.8). We will make these expectations precise in this section.

Recall that the moduli space \mathcal{M}_2 of genus two curves is complex threedimensional. The moduli space \mathcal{M}_2 contains two-dimensional loci $\mathcal{L}_2 \subset \mathcal{M}_2$, for which the genus two curves can be mapped to genus one with a map of degree 2 [226]. The map can be lifted to a map of the Jacobians of the curves. The Jacobian of the genus two curve is a four-torus, while the Jacobian of a genus one curve is a two-torus. For the curves contained in \mathcal{L}_2 , there is a degree two map from the genus two Jacobian to the genus one Jacobian. The Jacobian of a curve in \mathcal{L}_2 factors, $T^4 \equiv T^2 \times T^2$, which demonstrates that for a generic curve in \mathcal{L}_2 , there are two distinct maps $\varphi_j : \Sigma_2 \to \Sigma_{1,j}, j = 1, 2$ to two elliptic curves $\Sigma_{1,j}$. We will see in this section that these elliptic curves $\Sigma_{1,j}$ have precisely the complex structures τ_{\pm} introduced above.

The locus \mathcal{L}_2 of genus 2 fields with elliptic subfields of degree 2 reads

[211, 212, 227]

$$8748J_{10}J_{2}^{4}J_{6}^{2} + 507384000J_{10}^{2}J_{4}^{2}J_{2} - 19245600J_{10}^{2}J_{4}J_{2}^{3} - 592272J_{10}J_{4}^{4}J_{2}^{2} + 77436J_{10}J_{4}^{3}J_{2}^{4} \\ - 81J_{2}^{3}J_{6}^{4} - 3499200J_{10}J_{2}J_{6}^{3} + 4743360J_{10}J_{4}^{3}J_{2}J_{6} - 870912J_{10}J_{4}^{2}J_{2}^{3}J_{6} + 3090960J_{10}J_{4}J_{2}^{2}J_{6}^{2} \\ - 78J_{2}^{5}J_{4}^{5} - 125971200000J_{10}^{3} + 384J_{6}^{6}J_{6} + 41472J_{10}J_{4}^{5} + 159J_{6}^{4}J_{2}^{3} - 236196J_{10}^{2}J_{2}^{5} - 80J_{4}^{7}J_{2} \\ - 47952J_{2}J_{4}J_{6}^{4} + 104976000J_{10}^{2}J_{2}^{2}J_{6} - 1728J_{4}^{5}J_{2}^{2}J_{6} + 6048J_{4}^{4}J_{2}J_{6}^{2} - 9331200J_{10}J_{4}^{2}J_{6}^{2} \\ + 12J_{2}^{6}J_{4}^{3}J_{6} + 29376J_{2}^{2}J_{4}^{2}J_{6}^{3} - 8910J_{2}^{3}J_{4}^{3}J_{6}^{2} - 2099520000J_{10}^{2}J_{4}J_{6} + 31104J_{6}^{5} - 6912J_{4}^{3}J_{6}^{3} \\ - J_{2}^{7}J_{4}^{4} - 5832J_{10}J_{2}^{5}J_{4}J_{6} - 54J_{2}^{5}J_{4}^{2}J_{6}^{2} + 108J_{2}^{4}J_{4}J_{6}^{3} + 972J_{10}J_{2}^{6}J_{4}^{2} + 1332J_{2}^{4}J_{4}^{4}J_{6} = 0. \\ (4.90)$$

It is the zero locus of a weight 30 polynomial in the genus two Igusa invariants J_2 , J_4 , J_6 , J_{10} . The Igusa invariants of a generic sextic curve can be found in [228], for example.

Additionally, the SU(3) vacuum moduli space also corresponds to a twodimensional locus \mathcal{U} in \mathcal{M}_2 . On \mathcal{U} the weight 30 polynomial factors in three terms, such that \mathcal{U} and \mathcal{L}_2 intersect in three one-dimensional loci:

$$\mathcal{E}_{1} = \mathcal{E}_{u}: \quad v = 0, \mathcal{E}_{2} = \mathcal{E}_{v}: \quad u = 0, \mathcal{E}_{3}: \quad 784u^{9} - 24u^{6} \left(297v^{2} + 553\right) - 15u^{3} \left(729v^{4} + 5454v^{2} - 4775\right) + 8 \left(27v^{2} - 25\right)^{3} = 0.$$

$$(4.91)$$

Not suprisingly, we have seen the first two of these loci before. The latter is a cubic equation in v^2 as well as in u^3 , which does not reduce further. It does not include special points of the SU(3) theory. For v = 0, the equation reduces to the points $u^3 = 8$ and $u^3 = \frac{125}{28}$ in the *u*-plane, and for u = 0 it intersects in $v^2 = \frac{25}{27}$ on the *v*-plane.

The locus \mathcal{L}_2 can also be characterised in terms of Rosenhain invariants of the curve [211, Equation (18)]. By plugging in the cross-ratios we can check that the SU(3) Seiberg-Witten curve is not in \mathcal{L}_2 for generic u, v. For v = 0we rediscover the first algebraic relation (4.39), while for u = 0 we find both relations (4.68). This arises from an additional symmetry of the u = 0 curve, which we will comment on below.

4.5.1 Elliptic curves for locus \mathcal{E}_u

In this subsection we will establish two elliptic curves corresponding to the two modular parameters τ_{\pm} in section 4.3. The curves described by the locus \mathcal{L}_2 can be written in the form [211]

$$Y^{2} = X^{6} - s_{1}X^{4} + s_{2}X^{2} - 1, (4.92)$$

with s_1 and s_2 complex coordinates for \mathcal{L}_2 . This family of curves is left invariant by a non-trivial automorphism group, which contains the Klein four-group V_4 [229]. Namely, the curve (4.92) is left invariant by $(X, Y) \mapsto (-X, Y)$ and $(X, Y) \mapsto (X, -Y)$, which generate the dihedral group $D_4 \cong V_4 \cong \mathbb{Z}_2 \times \mathbb{Z}_2$. We interpret this group as the symmetry group of BPS/anti-BPS spectrum, and more precisely the central charges of the W-bosons Z_j (4.4) and their charge conjugates. For v = 0, Eq. (4.20) shows that $a_1 = a_2 = a$, such that $Z_1 = Z_2 = a$, and $Z_3 = 2a$. One $\mathbb{Z}_2 \subset D_4$ corresponds to the charge conjugation symmetry, while the other \mathbb{Z}_2 corresponds to the $a_1 \leftrightarrow a_2$ symmetry on \mathcal{E}_u . Note that the automorphism group of a generic genus two curve is \mathbb{Z}_2 , which is consistent with the charge conjugation symmetry for arbitrary (u, v).

For v = 0, the Seiberg-Witten curve $Y^2 = (X^3 - uX)^2 - 1$ is of the form (4.92), with $s_1 = 2u$ and $s_2 = u^2$. The degree two map to an elliptic curve is

$$(x,y) = (X^2, Y),$$
 (4.93)

which maps the algebraic equation (4.92) to

$$y^{2} = x(x-u)^{2} - 1. (4.94)$$

We can determine u in terms of the complex structure τ of the curve from the j-invariant, $j = 256u^6/(4u^3 - 27)$. This gives

$$4u(\tau) = q^{-1/3} + 104 q^{2/3} - 7396 q^{5/3} + \mathcal{O}(q^{8/3}).$$
(4.95)

We immediately recognise this function as the function u_+ (4.58), which was obtained from the Picard-Fuchs solution for the modular parameter $\tau_+ = \tau_{11} + \tau_{12}$. The curve (4.94) is exactly the Seiberg-Witten curve for the SU(2) theory with one massless hypermultiplet in the fundamental representation and scales related by $\Lambda_{SU(2)} = 2\Lambda_{SU(3)}$ [46], which clarifies the observation in section 4.3.3.

The curve that corresponds to $\tau_{-} = \tau_{11} - \tau_{12}$ can be constructed as follows. On the curve (4.92), the transformation $(X, Y) \mapsto (\frac{1}{X}, \frac{iY}{X^3})$ interchanges s_1 and s_2 . Interchanging those coefficients, $s_1 = u^2$ and $s_2 = 2u$, and setting again $(x, y) = (X^2, Y)$, we obtain

$$y^{2} = x(x^{2} - u^{2}x + 2u) - 1.$$
(4.96)

One finds $j = 256u^3(u^3 - 6)^3/(4u^3 - 27)$, which reproduces the solution u_- for the $\Gamma^0(9)$ curve (4.49). Note that the equation for j shows that u_- is the root of a degree 12 polynomial, which matches with the number of copies of \mathcal{F} in Figure 29. Another way to obtain this curve is to set $x = X^2$ and y = XY, from which one gets a quartic curve with the same j-invariant.

We have thus demonstrated that the two natural choices τ_{\pm} of the modular parameter indeed correspond to the complex structures of two elliptic curves covering the hyperelliptic curve. The physical u is given in terms of two different functions $u_{\pm} : \mathbb{H} \to \mathbb{C}$ with arguments τ_{\pm} .

4.5.2 Elliptic curves for locus \mathcal{E}_v

The Seiberg-Witten curve $Y^2 = (X^3 - v)^2 - 1$ for u = 0 is not in form (4.92) for a curve of \mathcal{L}_2 . However, the discussion around (4.91) suggests that it can be written in this form. We can achieve this by comparing the invariants of

the u = 0 hyperelliptic curve and (4.92), and solving for s_1, s_2 . Just as two elliptic curves are isomorphic if and only if their *j*-functions are equal, genus two curves are isomorphic if and only if their absolute invariants

$$x_1 = 144 \frac{J_4}{J_2^2}, \quad x_2 = -1728 \frac{J_2 J_4 - 3J_6}{J_2^3}, \quad x_3 = 486 \frac{J_{10}}{J_2^5},$$
(4.97)

are equal [212, 218, 219]. On \mathcal{L}_2 , there are only two independent invariants. For the curve (4.92), we find [212]

$$x_{1} = \frac{9}{4} \frac{405 + (a - 126)a + 12b}{(15 + a)^{2}},$$

$$x_{2} = \frac{27}{8} \frac{(a^{3} + 729a^{2} + 4131a - 36(39 + a)b - 3645)}{(15 + a)^{3}},$$

$$x_{3} = \frac{243}{8192} \frac{(27 - a(18 + a) + 4b)^{2}}{(15 + a)^{5}},$$
(4.98)

where $a = s_1 s_2$, $b = s_1^3 + s_2^3$ are the dihedral invariants. Comparing these absolute invariants with those of the SU(3) curve for u = 0, we arrive at

$$s_1 s_2 = 9 \left(25 - 24v^2 \right), \qquad s_1^3 + s_2^3 = 54 \left(216v^4 - 340v^2 + 125 \right).$$
 (4.99)

To solve the two equations in (4.99), let us denote

$$\mathcal{Q}^{\pm}(v) = 27 \left(216v^4 - 340v^2 + 125 \pm 8v \left(27v^2 - 25 \right) \sqrt{v^2 - 1} \right).$$
(4.100)

Then, one of the six solutions is given by

$$s_1^{\pm} = \sqrt[3]{\mathcal{Q}^{\mp}(v)}, \quad s_2^{\pm} = 9 \frac{25 - 24v^2}{\sqrt[3]{\mathcal{Q}^{\mp}(v)}}.$$
 (4.101)

In order to get an elliptic curve, we again take the map $(x, y) = (X^2, Y)$. This gives us the two curves

$$y^{2} = x^{3} - s_{1}^{\pm}x^{2} + s_{2}^{\pm}x - 1$$
(4.102)

with j-function

$$j^{\pm} = -432 \left(1458v^6 - 2673v^4 + 1340v^2 - 125 \mp 2v \left(729v^4 - 972v^2 + 275 \right) \sqrt{v^2 - 1} \right)$$

$$(4.103)$$

and discriminant $\Delta = v^2 - 1$. By inverting (4.103), the resulting function v matches precisely with (4.74) in section 4.4.2. Note that j^{\pm} vanish at the AD points $v = \pm 1$ and the curve (4.101) becomes a cusp $y^2 = x^3$. This implies that the AD points are elliptic fixed points and are in the SL(2, Z) orbit of α , which is easy to check from (4.77): We have that $\tau_{AD,1} = \alpha - 1$ and $\tau_{AD,2} = \alpha + 2$. See also Figure 31. They do however not fall into the (classical) Kodaira classification of singular fibers, since the Weierstraß invariants of (4.101) are not polynomials in v and their order of vanishing is half-integer rather than integer.

In general, the *j*-invariants of the two elliptic curves for (4.92) are the two solutions of [212]

$$j^{2} + 2^{8} \frac{2a^{3} - 54a^{2} + 9ab - b^{2} + 27b}{a^{2} + 18a - 4b - 27}j + 2^{16} \frac{(a^{2} + 9a - 3b)^{3}}{(a^{2} + 18a - 4b - 27)^{2}} = 0.$$
(4.104)

Since the equations for s_1 and s_2 always have solutions, one elliptic curve is found by substituting $(x, y) = (X^2, Y)$, such that it becomes

$$y^{2} = x^{3} - s_{1}x^{2} + s_{2}x - 1.$$
(4.105)

The other elliptic curve is found by relating $(x, y) = (X^2, XY)$, such that

$$y^{2} = x(x^{3} - x_{1}x^{2} + s_{2}x - 1).$$
(4.106)

Returning to the curve for \mathcal{E}_v , we notice that $Y^2 = X^6 - 2vX^3 + v^2 - 1$ for u = 0 has enhanced symmetry compared to the Klein four-group for (4.92). Since $v^2 - 1$ is the discriminant, we can divide and rescale X to find

$$Y^{2} = X^{6} - \frac{2v}{\sqrt{v^{2} - 1}}X^{3} + 1.$$
(4.107)

It is easy to show that any curve of the form $Y^2 = X^6 - aX^3 + 1$ is invariant under $(X, Y) \mapsto (\frac{1}{X}, \frac{Y}{X^3})$ and $(X, Y) \mapsto (\alpha X, -Y)$, where again $\alpha = e^{2\pi i/3}$. These order 2 and 6 elements generate the dihedral group D_{12} . Similarly to the enhanced automorphism group for \mathcal{E}_u , we interpret this group as a symmetry group of the BPS/anti-BPS spectrum. On the locus \mathcal{E}_v , we find that the periods a_1 and a_2 are related as $a_2 = -\alpha a_1$. The central charges Z_j (4.4) of the W-bosons, together with their charge conjugates, span therefore a regular 6-gon, whose symmetry group is D_{12} .

Hyperelliptic curves $C \in \mathcal{L}_2$ with $\operatorname{Aut}(C) \cong D_{12}$ satisfy an additional constraint, it is given by the zero loci of a weight 12 and a weight 20 polynomial in the Igusa invariants [227, Eq. (24)],

$$0 = -J_4 J_2^4 + 12 J_2^3 J_6 - 52 J_4^2 J_2^2 + 80 J_4^3 + 960 J_2 J_4 J_6 - 3600 J_6^2,$$

$$0 = 864 J_{10} J_2^5 + 3456000 J_{10} J_4^2 J_2 - 43200 J_{10} J_4 J_2^3 - 2332800000 J_{10}^2 \quad (4.108)$$

$$-J_4^2 J_2^6 - 768 J_4^4 J_2^2 + 48 J_4^3 J_2^4 + 4096 J_4^5.$$

Moreover, the elliptic subcovers of hyperelliptic curves with $\operatorname{Aut}(C) \cong D_{12}$ are 3-isogenous [211]. We can check explicitly that the u = 0 curve is of this form. Another check on the D_{12} symmetry is [211]

$$0 = a^2 - 110a - 4b + 1125. (4.109)$$

This explains why the elliptic curves for the two complex structures produce a single modular function (4.74), rather than the two independent functions u_{\pm} for \mathcal{E}_u . On \mathcal{E}_u the first algebraic relation in (4.39) holds and places the curve in \mathcal{L}_2 . On \mathcal{E}_v both relations (4.68) hold, where the first one projects into \mathcal{L}_2 and the second one gives the augmented D_{12} symmetry. This is consistent with the argument of Section 4.3.2 that the maps φ_j should exist as long as $\operatorname{Im}(\tau_{11}) = \operatorname{Im}(\tau_{22})$, such that it is possible to define $\tau_{\pm} = \tau_{11} \pm \tau_{12} \in \mathbb{H}$. The first relations in both (4.39) and (4.68) are equivalent to this condition.

4.6 SU(3) theory with matter

With the technology set up in section 4.5, it is in principle straightforward to study other theories characterised by genus two hyperelliptic curves. One such class of theories are the SU(3) theories with $N_f \leq 6$ hypermultiplets in the fundamental representation.

When $N_f \leq 5$ and all the masses are equal to m, the curves are given by [128, 230]

$$Y^{2} = C(X)^{2} - \Lambda^{2N_{c}-N_{f}}(X+m)^{N_{f}},$$

$$C(X) = X^{3} - uX - v + \frac{1}{4}\Lambda^{2N_{c}-N_{f}}\sum_{k=0}^{N_{f}-N_{c}}X^{N_{f}-N_{c}-k}\binom{N_{f}}{k}m^{k}.$$
(4.110)

These theories are also studied in $[8, 178, 205, 231, 232]^{27}$.

Let us thus study the SU(3) curve for $\mathcal{N} = 2$ gauge theory with $N_f \geq 1$ hypermultiplets in the fundamental representation.

$$N_f = 1$$

The locus \mathcal{L}_2 intersects with the $m = 0, N_f = 1$ curve in

$$0 = -6165504u^{12}v^2 - 73809792u^9v^4 - 111484512u^8v^3 + 109220400u^7v^2 + 1796349312u^6v^6 - 8196945984u^5v^5 + 9010440000u^4v^4 - 6598371456u^3v^8 - 4626787500u^3v^3 + 3826375200u^2v^7 + 1261406250u^2v^2 + 2581632u^{11}v - 27737500u^6v + 2048u^{15} - 271784u^{10} + 2162500u^5 - 492075000uv^6 - 175781250uv - 1836660096v^{10} - 59231250v^5 + 9765625, (4.111)$$

and in particular neither u = 0 nor v = 0 are in \mathcal{L}_2 . This is also the case for generic masses, for which the equation because much more complicated.

$$N_f = 2$$

For $N_f = 2$, v = 0 is in fact a singular surface. It intersects with the singular locus $\Delta_{\Lambda} = 0$ in the points $u^2 = 1$. In fact, v = 0 is a sublocus of \mathcal{L}_2 . However, since it is singular, the usual maps $\phi_j : \Sigma_2 \to \Sigma_{1,j}$ are degenerate. Thus the map to the elliptic subcover is ill-defined.

 $N_f = 3$

For massless $N_f = 3$ we find that only u = 0 is in \mathcal{L}_2 . The discriminant is

$$\Delta_{\Lambda} = (4v-1)^3 \left(-3456u^3v^2 + 3888u^3v + 256u^6 - 729u^3 + 11664v^4 - 2916v^3 \right),$$
(4.112)

and therefore u = 0 is not a singular locus. Rather, u = 0 intersects with $\Delta_{\Lambda} = 0$ in the two vacua v = 0 and $v = \frac{1}{4}$. We can compute the absolute

 $[\]overline{^{27}}$ The article [205] clashes with both [231] and [230].



Figure 32: Fundamental domain of $\Gamma_0(3)$ (Fig. 32a) and $\Gamma^0(3)$ (Fig. 32b), both the domains are for $(u = 0) \in \mathcal{L}_2$ for $N_f = 3$.

invariants of the u = 0 curve and compare with (4.98). This gives the dihedral invariants

$$a = -\frac{9(48v^2 - 26v + 3)}{2v}, \quad b = \frac{27(6912v^4 - 3968v^3 + 1152v^2 - 248v + 27)}{16v^2}.$$
(4.113)

One can then either proceed to compute s_1 and s_2 , which gives fractional powers of v, or insert (4.113) into (4.104) and compute the *j*-invariants of the two elliptic curves. This gives

$$j_1 = -\frac{27(4v-9)^3(4v-1)}{64v^3}, \quad j_2 = -\frac{27(4v-1)(36v-1)^3}{4v}, \quad (4.114)$$

demonstrating that the underlying elliptic surfaces are rational. From this, one finds

$$v_{1} = -\frac{f_{3B}(3\tau)}{108},$$

$$v_{2} = -\frac{f_{3B}(\tau)}{108},$$
(4.115)

where f_{3B} is defined in (2.75). This proves that v_1 is a Hauptmodul for $\Gamma_0(3)$, while v_2 is a Hauptmodul for $\Gamma^0(3)$. The fundamental domains are drawn in Fig. 32. We can see that they are topologically equivalent.

The singular point v = 0 translates to a root of f_{3B} for both v_1 and v_2 , which holds at the cusp $\tau = 0$. For v_1 , we have that $v_1 = \frac{1}{4}$ if and only if $f_{3B}(3\tau) = -27$, and therefore $\tau = \frac{1}{\sqrt{3}}\omega$: This is an AD point. For v_2 we get $f_{3B}(\tau) = -27$, such that $\tau = \sqrt{3}\omega$. It is precisely the same parametrisation as the *II* AD theory in SU(2) with $N_f = 1$ and $m = \frac{3}{4}\Lambda_1$, where we get a $\Gamma^0(3)$ curve, or $N_f = 3$ with $m = -\frac{1}{64}\Lambda_3$, where the curve is $\Gamma_0(3)$ (see section 2.6.6). We can see from (4.114) that both curves are quartic modular functions, which agrees with Fig. 32 as both have index 4 in PSL(2, Z).

By solving (4.113) for s_1 and s_2 , we can find the two elliptic curves from (4.105) and (4.106) with *j*-invariants (4.114). It allows to check that at the AD point $u_{AD} = \frac{1}{4}$, the Kodaira signature is ord $(g_2, g_3, \Delta) = (1, 1, 2)$ for both

curves, such that it is a type II singularity. This explains the similarity to the II AD theories found in rank 1.

The curve for u = 0 reads

$$Y^{2} = X^{6} - (\frac{1}{2} + 2v)X^{3} + (v - \frac{1}{4})^{2}.$$
(4.116)

By a rescaling of X, we can bring it to the form $Y^2 = X^6 - fX^3 + 1$, which is known to have D_{12} symmetry. This can easily be checked with (4.108) and (4.109). The D_{12} symmetry implies that the two curves (4.114) are 3isogenous [211]. The 3-isogeny is related to the fact that there is a determinant 3 Fricke involution that acts on the Hauptmoduln,

$$f_{3B}(-\frac{3}{\tau}) = \frac{3^6}{f_{3B}(\tau)}.$$
(4.117)

It relates the *j*-invariants (4.114) as

$$j_1(\frac{1}{16v_2}) = j_2(v_2). \tag{4.118}$$

 $N_f = 4, 5, 6$

For $N_f = 4$ and $N_f = 5$, there are again no simple elliptic loci in the massless case. The \mathcal{L}_2 locus for generic masses becomes increasingly lengthy, compared to (4.111).

The massless $N_f = 6$ theory is superconformal. It is studied in [178, 180, 205–208, 230, 233, 234]²⁸, and it necessarily includes modular forms in the curve itself. It is argued by Minahan and Nemeschansky that for $m_i = 0$ and u = 0 the curve can be expressed as [178, 205]

$$Y^{2} = (f_{-}X^{3} - v)^{2} + (f_{+}^{2} - f_{-}^{2})X^{6}, \qquad (4.119)$$

where

$$f_{\pm}(\tau) = \left(\frac{\eta^{3}(\tau)}{\eta(3\tau)}\right)^{3} \pm 27 \left(\frac{\eta^{3}(3\tau)}{\eta(\tau)}\right)^{3}.$$
 (4.120)

The q-expansions read

$$f_{+}(\tau) = 1 + 18q + 108q^{2} + 234q^{3} + 234q^{4} + 864q^{5} + \mathcal{O}(q^{6}),$$

$$f_{-}(\tau) = 1 - 36q - 54q^{2} - 252q^{3} - 468q^{4} - 432q^{5} + \mathcal{O}(q^{6}).$$
(4.121)

Both f_{\pm} are weight 3 modular forms for $\Gamma_1(3)$. We find that

$$f_{+}(\tau) = b_{3,0}(\tau)^{3},$$

$$f_{-}(\tau) = b_{3,0}(\tau)^{3} - 2b_{3,1}(\tau)^{3}.$$
(4.122)

This allows to define $f_1 := \sqrt[3]{f_+} = b_{3,0}$, which together with f_- is claimed to generate the ring of modular forms on $\Gamma_0(3)$.²⁹ From (4.119), we can compute

 $^{^{28}\}mathrm{Some}$ of these references contradict each other.

²⁹Without this, it is not obvious as in [178, 205] why f_1 should be modular.

the absolute invariants of the curve,

$$x_{1} = \frac{81 \left(4f_{-}^{2}f_{+}^{2} + 5f_{+}^{4}\right)}{\left(f_{-}^{2} - 10f_{+}^{2}\right)^{2}},$$

$$x_{2} = -\frac{729 \left(-26f_{-}^{2}f_{+}^{4} + 4f_{-}^{4}f_{+}^{2} - 5f_{+}^{6}\right)}{\left(f_{-}^{2} - 10f_{+}^{2}\right)^{3}},$$

$$x_{3} = \frac{729f_{+}^{4} \left(f_{-}^{2} - f_{+}^{2}\right)^{3}}{256 \left(f_{-}^{2} - 10f_{+}^{2}\right)^{5}}.$$
(4.123)

They do not depend on v. Since (x_1, x_2, x_3) labels isomorphism classes of hyperelliptic curves, in fact the curve (4.119) does not depend on v at all. This is reminiscent of the massless SU(2) $N_f = 4$ curve [46].

There is also a proposed curve for nonzero u,

$$Y^{2} = (f_{-}X^{3} - f_{1}uX - v)^{2} + (f_{+}^{2} - f_{-}^{2})X^{6}.$$
 (4.124)

Aside from u = 0 and v = 0, the loci $0 = 27f_-v^2 - 2f_1^3u^3$ and

$$0 = 15f_1^6 f_+^2 u^6 v^4 \left(512f_-^2 + 325f_+^2\right) - 96f_1^3 f_- u^3 v^6 \left(2f_-^2 + 25f_+^2\right)^2 - 1160f_1^9 f_- f_+^2 u^9 v^2 + 48f_1^{12} f_+^2 u^{12} + 8v^8 \left(2f_-^2 + 25f_+^2\right)^3$$
(4.125)

are in \mathcal{L}_2 .

Other gauge groups

We could also run an analysis for other gauge groups. The curves for $SO(N_c)$ for instance are found in [97, 235, 236], see also [237] for a review. However, for $SO(N_c)$ the genus of the hyperelliptic curve is g = 2l - 1, where $l = \frac{N_c}{2}$ for N_c even and $l = \frac{N_c-1}{2}$ for N_c odd. So in fact there are no genus 2 curves for $SO(N_c)$ theories: For SO(3) of course one finds the same curve $\Gamma^0(4)$ as for SU(2). For $SO(N_c > 3)$ on the other hand the curve has genus g > 2.

4.7 Discussion

In this section, we have discussed the modular properties of $\mathcal{N} = 2$ Yang-Mills theory in four dimensions with gauge group SU(3). For the pure theory, on the two loci \mathcal{E}_u and \mathcal{E}_v , where v = 0 and u = 0 respectively, we express the parameters u and v of the moduli space as modular functions for discrete subgroups of SL(2, \mathbb{R}). See (4.49) and (4.76). To this end, we formulate the genus two SU(3) SW curve in Rosenhain form in terms of Siegel theta series. The parameters of the theory are then found by relating the Rosenhain form to the PF solution of [96]. We provide an explicit fundamental domain for the effective coupling on the two elliptic loci \mathcal{E}_u and \mathcal{E}_v . The relation between cross-ratios of the curve and theta constants suggests that the full moduli space can be parametrised by higher genus modular forms. It would be interesting

to find a general solution to (4.36) by expressing u and v as algebraic functions of theta constants.

On \mathcal{E}_u , we established a nice generalisation of the structure appearing in the SU(2) case. In rank one, the parameter u is a weakly holomorphic modular function for the congruence subgroup $\Gamma^0(4)$. For SU(3), we instead found that on \mathcal{E}_u the parameter u is a weakly holomorphic modular function of τ_- for the group $\Gamma^0(9) \subset \text{SL}(2,\mathbb{Z})$. The structure of the moduli space near the special points of this locus also seems to generalise the rank one picture: We find that u maps the \mathbb{Z}_2 singularities to the cusps of its fundamental domain. Furthermore, the duality group is generated by the nontrivial monodromies on \mathcal{E}_u . For the other choice of modular parameter $\tau_+ = \tau_{11} + \tau_{12}$, we find that u is not invariant under a congruence subgroup, but is rather a *sextic modular function*, which is the same function as appears for rank 1 $N_f = 1$ SQCD. Nevertheless, we are able to show that the monodromies can be viewed as paths in a new fundamental region, which we propose.

On the other locus \mathcal{E}_v where u = 0, we find that v can be expressed as a modular function for a subgroup $\Gamma_v \subset SL(2,\mathbb{R})$ of Atkin-Lehner type. The AD points are mapped to the elliptic fixed points of the quotient $\Gamma_v \setminus \mathbb{H}$. The group Γ_v includes a Fricke involution, which can be viewed as a manifestation of Sduality [33, 178, 205]. We derive it from the monodromy group on \mathcal{E}_v . On the locus \mathcal{E}_v , the genus two hyperelliptic curve splits into two *elliptic* curves with complex structures $\tau_{\pm} = \tau_{11} \pm \tau_{12}$. The appearance of the Fricke involution is a consequence of the two families of elliptic curves being isogenous [223, 238]. Fricke dualities also appear in String theory, where they have been shown to play an important rôle in the web of dualities of CHL models, i.e. orbifolds of heterotic string theory on T^6 or type II on $K3 \times T^2$ [239,240]. They are also the natural generalisation of S-duality in the context of Montonen–Olive duality in $\mathcal{N} = 4$ super-Yang–Mills theory for non-simply laced gauge groups [241,242] and the geometric Langlands program [30]. Moreover, Fricke involutions are familiar in topological string theory where they act on higher genus amplitudes, which are described by quasi modular forms. They exchange the large complex structure of the Calabi-Yau threefold with the conifold loci, which gives an analogue of the action of electric-magnetic duality or $\mathcal{N} = 2$ S-duality in topological string theory [223, 238]. They have also appeared recently in the context of string compactifications and the swampland program [243].

It would be interesting to extend this work to other theories, such as those with gauge group SU(N), including matter multiplets, theories of class \mathcal{S} [113], or gravitational couplings to these theories [103, 148]. For theories with SU(N > 2), one can for example consider to turn on only the bottom Casimir u_2 and setting u_3, \ldots, u_N to zero. Our analysis naively suggests that it should be parametrised by a modular function for $\Gamma^0(N^2)$. The discriminant of the SU(N) curve [94]

$$y^{2} = \left(x^{N} - \sum_{j=2}^{N} u_{j} x^{N-j}\right)^{2} - 1$$
(4.126)

intersects with this locus in $u_2^N = N^N (N-2)^{2-N}/4$, confirming that there are N singularities at strong coupling. However, it is easy to show that $\Gamma^0(N^2)$ has N cusps aside from $i\infty$ if and only if N is prime. Note that this worked for N = 2, 3. It is furthermore not obvious how the modular parameter would relate to the coupling matrix, and the map to elliptic subcovers is more subtle in the higher rank case [213].

We would like to finish by mentioning a few potential applications and directions for further research:

- We observe that the functions parametrising the SU(2) and SU(3) moduli spaces are all *replicable* [36,56–58] modular functions. See Appendix A.4 for a definition. The SU(2) order parameter u is of class 4C, u₋ of class 9B, and v of class 6a. It would be interesting to explore whether there is an underlying reason for the functions to have this property.
- This work motivates exploring subloci of Coulomb branches for theories with other gauge groups and including matter multiplets. This could provide a better understanding of the modularity of these theories. Moreover, it would be interesting to understand whether the solution of the theory on a sublocus is equivalent to the solution of another theory, such as we found for \mathcal{E}_u and the massless $N_f = 1$, SU(2) theory for example.
- The elliptic loci we consider are somewhat analogous to the special Kähler strata of Coulomb branches being studied in the recent work [244–246]. The latter aims to classify higher rank $\mathcal{N} = 2$ SCFTs by decomposing the singular locus into a nested series of one-dimensional building blocks. It would be interesting to see if our methods find applications in this programme.
- The last application which we would like to mention, is topological quantum field theory [61]. Evaluation of the path integral or correlation functions for a compact four-manifold X involves the integration over the Coulomb branch (the so-called u-plane integral) of the theory [73,87,144]. For gauge group SU(2), the integral becomes an integral over the modular fundamental domain $\Gamma^0(4) \setminus \mathbb{H}$ [73,77,81,84]. A better understanding of the modularity of SU(N > 2) Seiberg-Witten theory could possibly allow further progress in this direction for theories with N > 2.

5 Topological twists of massive SQCD

In this section, we study an infinite family of topological twists of massive $\mathcal{N} = 2$ supersymmetric QCD on a compact four-manifold, and the formulation of topological correlation functions. This section is based on [5].

5.1 Introduction

We consider topologically twisted $\mathcal{N} = 2$ supersymmetric Yang–Mills theories with additional matter multiplets on a compact four-manifold, which were introduced in [247-252]. After the work by Seiberg and Witten on the full non-perturbative solution [45, 46, 71], these theories have received much attention in physics [65, 66, 73, 74, 82, 85, 88, 89, 98, 99, 188, 253–256] and mathematics [196, 257-267]. For reviews, see for example [52, 53, 268, 269]. More specifically, we consider in this section topological twists of $\mathcal{N} = 2$ QCD with gauge group SU(2) and matter multiplets in the fundamental representation of the gauge group. By including background fluxes for the flavour group, we obtain an infinite family of topological theories [255]. The choice of a background flux makes it possible to formulate topologically twists for $\mathcal{N} = 2$ SQCD for arbitrary 't Hooft fluxes, or first Chern classes of the gauge bundle. This is similar to the topological twist of $\mathcal{N} = 2^* \operatorname{SU}(2)$ gauge theory, which requires a non-vanishing background flux on a non-spin four-manifold [85]. We moreover develop techniques to determine correlation functions for arbitrary values of the masses of the hypermultiplets.

The starting point of our approach is the low-energy effective field theory on the Coulomb branch. This phase of the theory contributes for a compact fourmanifold X with the topological condition that $b_2^+(X) = 1$ [73]. In this way, the classical Donaldson invariants can be derived starting from the Seiberg-Witten (SW) solution to $\mathcal{N} = 2$ supersymmetric Yang–Mills theory with gauge group SU(2). The Coulomb branch integral (or *u*-plane integral) reduces to an integral over zero modes [73], and reads schematically

$$\Phi = \int_{\mathcal{B}} da \wedge d\bar{a} \,\rho(a) \,\Psi(a,\bar{a}),\tag{5.1}$$

where \mathcal{B} is the Coulomb branch with local coordinates a and \bar{a} , $\rho(a)$ contains the couplings to the background and $\Psi(a, \bar{a})$ is a sum over fluxes of the unbroken U(1) gauge group. For simplicity, we have suppressed the dependence on the metric and not included observables here. For the pure SU(2) theory, the Coulomb branch integral can be formulated and evaluated for arbitrary four-manifolds, without a requirement for Kähler or toric properties.

Recently, progress has been made on evaluating these *u*-plane integrals using a change of variables from *a* to the running coupling τ . As a result, the integration domain becomes a fundamental domain $\mathcal{F} \subseteq \mathbb{H}$ in the upper half-plane \mathbb{H} for the running coupling [4,77–86]. The integral then takes the form

$$\Phi = \int_{\mathcal{F}} d\tau \wedge d\bar{\tau} \,\nu(\tau) \,\Psi(\tau,\bar{\tau}), \qquad (5.2)$$

where the measure factor $\nu(\tau)$ further contains the Jacobian for the change of variables from a to τ . The domain \mathcal{F} is a modular fundamental domain in previous analyses, corresponding to the duality group $\Gamma^{0}(4)$ for the pure SU(2) theory [73,81,84], $\Gamma(2)$ for the $\mathcal{N} = 2^{*}$ theory [85] and similarly $\Gamma(2)$ and $\Gamma_{0}(4)$ for the theories with two and three massless flavours [77].

As mentioned above, we aim to apply this approach to $\mathcal{N} = 2$ supersymmetric SU(2) theories with $N_f \leq 3$ hypermultiplets in the fundamental representation. Topological correlators of these asymptotically free theories have been considered in various papers before, in particular the formulation of the low energy path integral in [73, 74], SW contributions for four-manifolds with $b_2^+ > 1$ [65, 66, 73, 88], the *u*-plane integral for \mathbb{P}^2 [77], and the calculation of the partition function of the AD theory within the $N_f = 1$ theory [82]. Since no background fluxes are included in these works, the 't Hooft flux necessarily matches the second Stiefel-Whitney class of the four-manifold, $w_2(E) = w_2(X)$, since the twisted hypermultiplets are not well-defined otherwise.

Extending to generic 't Hooft fluxes, and application of the above approach (5.2) to fundamental hypermultiplets with generic masses, gives rise to several new aspects. In particular:

- The fundamental domain of the effective coupling constant becomes more intricate for massive theories, and does for generic masses not correspond to a modular fundamental domain for a subgroup of PSL(2, Z). As demonstrated at length in Sections 2 and 3, the domain contains generically a set of branch points, and branch cuts starting from these points. These aspects have to be dealt with appropriately.
- 2. We couple the hypermultiplets to background fluxes k_j for the flavour group to formulate the theories for arbitrary 't Hooft fluxes. This gives rise to additional couplings in (5.1) and (5.2),

$$\prod_{j,k=1}^{N_f} \exp\left(-2\pi i \frac{\partial^2 F}{\partial m_j \partial m_k} B(\boldsymbol{k}_j, \boldsymbol{k}_k)\right), \qquad (5.3)$$

where F is the prepotential of the massive theory, and $B(\cdot, \cdot)$ is the quadratic form associated to the intersection form on the middle homology $H_2(X,\mathbb{Z})$ of X. Such couplings were suggested by Shapere and Tachikawa [144], and are also essential for the formulation of the $\mathcal{N} = 2^*$ Yang-Mills theory on a non-spin four-manifold [85]. Similarly to [85], we also deduce a non-holomorphic coupling to \mathbf{k}_j . Moreover, for arriving at a single-valued integrand, we fix an ambiguity in the quadratic terms of the prepotential. These terms have appeared earlier in the literature in the context of singularities of the SW differential and winding numbers [46, 141].

3. Special points on the Coulomb branch give rise to superconformal theories, such as the Argyres-Douglas (AD) theories [104, 105] and the massless $N_f = 4$ theory [46]. Their topological partition functions and correlators can be found by considering them in certain mass deformations. The case of $N_f = 1$ is analysed in [269].

The section is organised as follows. In section 5.2, we present the Seiberg-Witten solution of SU(2) $\mathcal{N} = 2$ SQCD in flat space, focusing on the fundamental domains for the effective coupling, which we illustrate in several interesting examples. In section 5.3, we formulate the topological twist by coupling the hypermultiplets to external fluxes, such that the topological field theory is well-defined for arbitrary 't Hooft flux and non-spin manifolds. The topological low-energy effective theory coupled to N_f background fluxes is then modelled in section 5.4 as a SU(2) × U(1)^{N_f} theory, with the matter fields corresponding to frozen U(1) factors. This allows to compute the path integral explicitly as an integral over the *u*-plane. In section 5.5, we formulate the *u*-plane integral as an integral over the fundamental domains. We prove that the single-valuedness under monodromies holds for a specific choice of magnetic winding numbers. Finally, in section 5.6 we demonstrate that such integrals may be evaluated using mock modular forms, and we show that they localise at the cusps, elliptic points and interior singularities of the fundamental domains.

5.2 Special geometry and SW theories

In this Section, we review aspects of the non-perturbative solution for the low energy effective theory of $\mathcal{N} = 2$ SQCD with gauge group SU(2) and $0 \leq N_f \leq 3$ fundamental hypermultiplets [45, 46]. See [47] for a review. Throughout, we let Λ_{N_f} denote the scale of the theory with N_f hypermultiplets having masses m_j , $j = 1, \ldots, N_f$, and a the mass of the W-boson on the Coulomb branch.

5.2.1 Field content

The $\mathcal{N} = 2$ theories we consider contain a vector multiplet and $N_f \leq 3$ hypermultiplets. The fields in these multiplets form representations of Spin(4) = $SU(2)_+ \times SU(2)_-$ and $SU(2)_R$, which we denote by $(\boldsymbol{k}, \boldsymbol{l}, \boldsymbol{m})$, with $\boldsymbol{k}, \boldsymbol{l}$ and \boldsymbol{m} dimensions of the representations.

The vector multiplet consists of a gauge field A_{μ} , complex scalar field ϕ , and a pair of Weyl fermions Ψ^{I}_{α} , $\bar{\Psi}^{I}_{\dot{\alpha}}$. This multiplet transforms under the adjoint representation of the gauge group G. The representation of $\mathrm{SU}(2)_{+} \times \mathrm{SU}(2)_{-} \times \mathrm{SU}(2)_{R}$ formed by the bosonic fields is,

$$(2, 2, 1) \oplus (1, 1, 1) \oplus (1, 1, 1),$$
 (5.4)

while the representation for the fermions is

$$(1, 2, 2) \oplus (2, 1, 2).$$
 (5.5)

The hypermultiplet consist of a pair of complex scalar fields, q and \tilde{q} , and Weyl fermions, λ_{α} , $\bar{\lambda}_{\dot{\alpha}}$, χ_{α} and $\bar{\chi}_{\dot{\alpha}}$. We fix the gauge group $G = \mathrm{SU}(2)$, and let the hypermultiplets transform under the fundamental representation of this group. With the same notation as above, the bosonic fields of this multiplet form the representation,

$$(1, 1, 2) \oplus (1, 1, 2),$$
 (5.6)

while the fermions form the representation

$$(2, 1, 1) \oplus (1, 2, 1) \oplus (2, 1, 1) \oplus (1, 2, 1).$$
 (5.7)

5.2.2 Seiberg-Witten geometry

The Seiberg-Witten geometry underlies the Coulomb branch of $\mathcal{N} = 2$ gauge theory. The Coulomb branch is the phase of the theory where SU(2) is broken to U(1) by a vacuum expectation value (vev) of the vector multiplet scalar ϕ . The vev is semi-classically parametrised by a complex parameter (1.21), up to gauge transformations. In particular, $a \to -a$ is a gauge transformation. The gauge invariant order parameter is the Coulomb branch expectation value of the theory in \mathbb{R}^4 , (1.22). The non-perturbative effective action of $\mathcal{N} = 2$ SQCD is characterised by the prepotential $F(a, \mathbf{m})$, with \mathbf{m} the mass vector $\mathbf{m} = (m_1, \ldots, m_{N_f})$. The semi-classical part of F reads [141, 143, 191, 270]

$$F(a, \boldsymbol{m}) = \frac{2i}{\pi} a^2 \log(a/\Lambda_{N_f}) - \frac{1}{2} \sum_{j=1}^{N_f} \left(n_j \frac{m_j}{\sqrt{2}} a + \frac{3}{8} \frac{i}{\pi} m_j^2 \right) - \frac{i}{4\pi} \sum_{j=1}^{N_f} \left(a + \frac{m_j}{\sqrt{2}} \right)^2 \log((a + \frac{m_j}{\sqrt{2}})/\Lambda_{N_f}) + \left(a - \frac{m_j}{\sqrt{2}} \right)^2 \log((a - \frac{m_j}{\sqrt{2}})/\Lambda_{N_f}) + \dots,$$
(5.8)

where the ... indicate further non-perturbative corrections.

The $n_j \in \mathbb{Z}$ in (5.8) are the magnetic winding numbers of the periods $a_D \coloneqq \frac{\partial F}{\partial a}$ dual to a [141, 143, 271]. These numbers seem to be only rarely discussed in the literature beyond these references.³⁰ Generally, the theory allows for N_f electric winding numbers for a and N_f magnetic winding numbers for a_D . These appear in the massive $N_f > 0$ theories since the Seiberg-Witten differentials now have poles with nonzero residues [143]. The choice (5.8) of the prepotential corresponds to fixing the electric winding numbers to be zero, or equivalently fixing the monodromy at infinity to map $a \to e^{\pi i}a$. Compare

³⁰Nekrasov's partition function gives a specific choice upon expanding the function $\gamma_{\hbar}(x;\Lambda)$ in the perturbative part [191,272].

for example with [143, Eq. (2.17)]. In section 5.5, we will discuss that the single-valuedness of the *u*-plane integral requires $n_j \equiv -1 \mod 4$.

We introduce the period a_D dual to a, and the parameters $m_{D,j}$ dual to m_j by

$$a_D = \frac{\partial F}{\partial a}, \qquad m_{D,j} = \sqrt{2} \frac{\partial F}{\partial m_j}.$$
 (5.9)

These parameters are further combined into the $(2+2N_f)$ -dimensional vector Π ,

$$\Pi = \begin{pmatrix} a_D \\ a \\ m_{D,1} \\ \frac{m_1}{\sqrt{2}} \\ \vdots \\ m_{D,N_f} \\ \frac{m_{N_f}}{\sqrt{2}} \end{pmatrix}.$$
 (5.10)

This vector forms a local system over the u-plane. The elements of the vector form the symplectic form,

$$\omega_{N_f} = da_D \wedge da + \frac{1}{\sqrt{2}} \sum_{j=1}^{N_f} dm_{D,j} \wedge dm_j.$$
 (5.11)

The effective gauge coupling is related to the prepotential through (1.23). We also introduce the couplings v_j and w_{jk} with $j, k \in 1, \ldots, N_f$.

$$v_j = \sqrt{2} \frac{\partial^2 F}{\partial a \partial m_j}, \qquad w_{jk} = 2 \frac{\partial^2 F}{\partial m_j \partial m_k}.$$
 (5.12)

If we consider F as a function of $(a, \frac{1}{\sqrt{2}}m_1, \ldots, \frac{1}{\sqrt{2}}m_{N_f})$, then the dual parameters are encoded in the Jacobian $J_F = (a_D, m_D)$, while the couplings are the elements of the Hessian $H_F = \begin{pmatrix} \tau & v^T \\ v & w \end{pmatrix}$. The derivative of the prepotential with respect to the scale Λ_{N_f} provides the order parameter u (1.22) on the Coulomb branch,

$$u = -\frac{4\pi i}{4 - N_f} \Lambda_{N_f} \frac{\partial F}{\partial \Lambda_{N_f}}.$$
(5.13)

The weak-coupling limit in our convention is given by $\tau \to i\infty$, $a \to \infty$ and $u \to -\infty$.³¹

The Seiberg-Witten (SW) solution provides a family of elliptic curves parametrised by the order parameter u and the masses m_i , whose complex structure corresponds to the running coupling $\tau = \frac{\theta}{\pi} + \frac{8\pi i}{g^2}$. For the theories of interest in this section, the curves are given by (2.3).

³¹Note that this differs slightly from some of the previous literature. However, it is shown in section 2 to be the unique limit consistent with the RG flow.

5.2.3 Monodromies

This section determines the monodromies around the $N_f + 2$ monodromies. We define the physical discriminant Δ_{N_f} as the monic polynomial $\Delta_{N_f} = \prod_{j=1}^{N_f+2} (u - u_j)$, where u_j for $j = 1, \ldots, N_f + 2$ are the singular points of the effective theory. We let $j = 1, \ldots, N_f$ label the singular points where one of the matter hypermultiplets becomes massless; and $j = N_f + 1, N_f + 2$ denote the strong coupling singularities where a monopole and a dyon, respectively, becomes massless.

We leave the winding numbers n_j , $j = 1, ..., N_f$, for a_D generic. Starting with the monodromy around infinity, $a \to e^{\pi i} a$, we deduce from the (5.8) that the vector Π transforms as $\Pi \to \mathbf{M}_{\infty} \Pi$, with \mathbf{M}_{∞} given by

$$\mathbf{M}_{\infty} = \begin{pmatrix} -1 & 4 - N_f & 0 & -n_1 & \cdots & 0 & -n_{N_f} \\ 0 & -1 & 0 & 0 & \cdots & 0 & 0 \\ 0 & n_1 & 1 & 1 & \cdots & 0 & 0 \\ 0 & 0 & 0 & 1 & \cdots & 0 & 0 \\ \vdots & \vdots & & \ddots & & \\ 0 & n_{N_f} & 0 & 0 & \cdots & 1 & 1 \\ 0 & 0 & 0 & 0 & \cdots & 0 & 1 \end{pmatrix}.$$
 (5.14)

The monodromy matrix \mathbf{M}_{∞} is in $\mathrm{SL}(2+2N_f,\mathbb{Z})$, while it acts on the couplings by a symplectic transformation, i.e. it preserves the symplectic form (5.11). This can be checked by requiring that any monodromy \mathbf{M}_{∞} satisfies $\mathbf{M}^{\mathsf{T}}\mathbf{J}\mathbf{M} = \mathbf{J}$, with

$$\mathbf{J} = \begin{pmatrix} 0 & 1\\ -1 & 0 \end{pmatrix}^{\oplus N_f + 1}.$$
 (5.15)

The action on the couplings τ (1.23), v_j and w_{jk} (5.12) is thus

$$\mathbf{M}_{\infty} : \begin{cases} \tau \to \tau + N_f - 4, \\ v_j \to -v_j - n_j, \\ w_{jk} \to w_{jk} + \delta_{jk}, \end{cases}$$
(5.16)

with δ_{jk} the Kronecker delta.

If we assume that the mass m_j is large, we can also deduce the monodromies around $a = \frac{m_j}{\sqrt{2}}, j = 1, \ldots, N_f$ from the perturbative prepotential (5.8). For a encircling $\frac{m_1}{\sqrt{2}}$ counterclockwise, $\Pi \to \mathbf{M}_1 \Pi$, we find for the monodromy matrix \mathbf{M}_1 ,

$$\mathbf{M}_{1} = \begin{pmatrix} 1 & 1 & 0 & -1 & \cdots & 0 & 0 \\ 0 & 1 & 0 & 0 & \cdots & 0 & 0 \\ 0 & -1 & 1 & 1 & \cdots & 0 & 0 \\ 0 & 0 & 0 & 1 & \cdots & 0 & 0 \\ \vdots & \vdots & & \ddots & & \\ 0 & 0 & 0 & 0 & \cdots & 1 & 0 \\ 0 & 0 & 0 & 0 & \cdots & 0 & 1 \end{pmatrix},$$
(5.17)

while the \mathbf{M}_{j} for other values of j are given by permutations. Its action on the couplings is

$$\mathbf{M}_{j}: \begin{cases} \tau \to \tau + 1, \\ v_{k} \to v_{k} - \delta_{jk}, \\ w_{kl} \to w_{kl} + \delta_{kl}\delta_{jl}. \end{cases}$$
(5.18)

Besides the monodromies \mathbf{M}_{∞} and \mathbf{M}_{j} , there are monodromies \mathbf{M}_{m} and \mathbf{M}_{d} around the points where a monopole and a dyon becomes massless, respectively. By requiring that the electro-magnetic charges of the massless particles are $(n_m, n_e) = (1, 0)$ and (1, -2), respectively, we can fix the upper left blocks of the monodromies. We fix the remaining entries by assuming that the masses remains invariant, $m_j \to m_j$, and that the other periods only change by a multiple of the vanishing cycle at the corresponding cusp, together with the requirement that

$$\mathbf{M}_{\infty} = \mathbf{M}_m \mathbf{M}_d \prod_{j=1}^{N_f} \mathbf{M}_j.$$
 (5.19)

For $N_f = 1$ and $n_1 = n$, this gives for \mathbf{M}_m ,

$$\mathbf{M}_{m} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ -1 & 1 & 0 & -(n+1)/2 \\ (n+1)/2 & 0 & 1 & (n+1)^{2}/4 \\ 0 & 0 & 0 & 1 \end{pmatrix}.$$
 (5.20)

This acts on the couplings as

$$\mathbf{M}_{m}: \begin{cases} \tau \to \frac{\tau}{-\tau+1}, \\ v \to \frac{v+(n+1)\tau/2}{-\tau+1}, \\ w \to w + \frac{(v+(n+1)/2)^{2}}{-\tau+1}. \end{cases}$$
(5.21)

The monodromy \mathbf{M}_d around the dyon singularity for $N_f = 1$ is

$$\mathbf{M}_{d} = \begin{pmatrix} -1 & 4 & 0 & -n-1 \\ -1 & 3 & 0 & -(n+1)/2 \\ (n+1)/2 & -n-1 & 1 & (n+1)^{2}/4 \\ 0 & 0 & 0 & 1 \end{pmatrix},$$
 (5.22)

This acts on the couplings as

$$\mathbf{M}_{d}: \begin{cases} \tau \mapsto \frac{-\tau + 4}{-\tau + 3}, \\ v \mapsto \frac{v + (n+1)\tau - n - 1}{-\tau + 3}, \\ w \mapsto w + \frac{(v + (n+1)/2)^{2}}{-\tau + 3}. \end{cases}$$
(5.23)

We can note that all the above monodromy matrices leave the symplectic form (5.11) invariant and are independent of the masses.

We get similar monodromies for $N_f = 2, 3$. The action on the running couplings τ are the same for all N_f , by construction. The transformations of v_j and w_{jk} also take the same form for all N_f and can be summarised as

$$\mathbf{M}_{m} : \begin{cases} v_{j} & \to \frac{v_{j} + (n_{j} + 1)\tau/2}{-\tau + 1}, \\ w_{jk} & \to w_{jk} + \frac{(v_{j} + (n_{j} + 1)/2)(v_{k} + (n_{k} + 1)/2)}{-\tau + 1}, \\ \mathbf{M}_{d} : \begin{cases} v_{j} & \to \frac{v_{j} + (n_{j} + 1)\tau - n_{j} - 1}{-\tau + 3}, \\ w_{jk} & \to w_{jk} + \frac{(v_{j} + (n_{j} + 1)/2)(v_{k} + (n_{k} + 1)/2)}{-\tau + 3}. \end{cases}$$
(5.24)

5.3 The UV theory on a four-manifold

We review various aspects of the formulation of the UV theory on a compact smooth four-manifold.

5.3.1 Aspects of four-manifolds

We let X be a smooth, compact, oriented Riemannian four-manifold, with Euler number $\chi = \chi(X)$ and signature $\sigma = \sigma(X) = b_2^+ - b_2^-$. The *u*-plane integral is non-vanishing only for four-manifolds X with $b_2^+ \leq 1$. In this article, we consider manifolds with $b_2^+ = 1$. Such four-manifolds admit a linear complex structure \mathcal{J} on the tangent space TX_p at each point p of X. The complex structure varies smoothly on X, such that TX is a complex bundle. We introduce furthermore the canonical class $K_X = -c_1(TX)$ of X, with $c_1(TX)$ the first Chern class of TX. For a manifold X with $(b_1, b_2^+) = (0, 1)$, we have that

$$K_X^2 = 8 + \sigma(X). \tag{5.25}$$

The middle cohomology $H^2(X, \mathbb{Z})$ of X gives rise to the uni-modular lattice L. More precisely, we identify L with the natural embedding of $H^2(X, \mathbb{Z})$ in $H^2(X, \mathbb{Z}) \otimes \mathbb{R}$, which modes out the torsion of $H^2(X, \mathbb{Z})$. A characteristic element $K \in L$ is an element which satisfies $l^2 + B(K, l) \in 2\mathbb{Z}$ for all $l \in L$. The Riemann-Roch theorem demonstrates that the canonical class K_X of X is a characteristic element of L. The Wu formula furthermore shows that any characteristic vector K of L is a lift of $w_2(X)$.

The quadratic form Q of the lattice L for a four-manifold with $(b_1, b_2^+) = (0, 1)$ can be brought to a simple standard form depending on whether Q is even or odd [273]. This divides such manifolds into two classes, for which the evaluation of their *u*-plane integrals needs to be done separately [84]. The period point $J \in H^2(X, \mathbb{R})$ is defined as the unique class in the forward light cone of $H^2(X, \mathbb{R})$ that satisfies J = *J and $J^2 = 1$.

All four-manifolds without torsion and even intersection form admit a Spin structure. More generally, for any oriented four-manifold one can define a Spin^c-structure. The group $\text{Spin}^{c}(4)$ can be defined as pairs of unitary 2×2 matrices with coinciding determinant,

$$\operatorname{Spin}^{c}(4) = \{ (u_1, u_2) \in \operatorname{U}(2) \times \operatorname{U}(2) | \det u_1 = \det u_2 \}.$$
 (5.26)

There exists a short exact sequence

$$1 \longrightarrow U(1) \longrightarrow \operatorname{Spin}^{c}(4) \longrightarrow \operatorname{SO}(4) \longrightarrow 1.$$
 (5.27)

A Spin^c-structure \mathfrak{s} on a four-manifold X is then a reduction of the structure group of the tangent bundle on X, i.e. SO(4), to the group Spin^c(4). The different Spin^c-structures correspond to the inequivalent ways of choosing transition functions of the tangent bundle such that the cocycle condition is satisfied. The Spin^c-structure defines two rank two hermitian vector bundles W^{\pm} . We let $c(\mathfrak{s})$ be the first Chern class of the determinant bundles, $c(\mathfrak{s}) \coloneqq c_1(\det W^{\pm}) \in H^2(X, \mathbb{Z}).$

If \mathfrak{s} is the canonical Spin^c structure associated to an almost complex structure on X, then $c(\mathfrak{s})^2 = 2\chi + 3\sigma$. More generally,

$$c_1(\mathfrak{s})^2 \equiv \sigma \mod 8. \tag{5.28}$$

5.3.2 Topological twisting with background fluxes

We discuss in this section topological twisting of theories with fundamental hypermultiplets including background fluxes. The discussion is parallel to the case of $\mathcal{N} = 2^*$ [85], where the hypermultiplet is in the adjoint representation of the gauge group.

We let $(E \to X, \nabla)$ be a principal $\mathrm{SU}(2)/\mathbb{Z}_2 \cong \mathrm{SO}(3)$ -bundle with connection ∇ . The second Stiefel-Whitney class $w_2(E) \in H^2(X, \mathbb{Z}_2)$ measures the obstruction to lift E to an SU(2) bundle, which will exist locally but not globally if $w_2(E) \neq 0$. We denote a lift of $w_2(E)$ to the middle cohomology lattice L by $\bar{w}_2(E) \in L$, and define the 't Hooft flux $\boldsymbol{\mu} = \bar{w}_2(E)/2 \in L/2$. The instanton number of the principal bundle is defined as $k = -\frac{1}{4} \int_X p_1(E)$ and satisfies $k \in -\boldsymbol{\mu}^2 + \mathbb{Z}$, where p_1 is the first Pontryagin class.

To formulate the theories with N_f fundamental hypermultiplets on a compact four-manifold, we perform a topological twist. Coupling the four-dimensional $\mathcal{N} = 2$ SU(2) theory to background fields means choosing two sets of data:

- A principal $SU(2)_R$ R-symmetry bundle, with connection ∇_R ,
- and a principal bundle \mathcal{L} with connection for global symmetries (the flavour symmetries) [85].

The relevant twist for the $\mathcal{N} = 2$ supersymmetry algebra in four dimensions is the Donaldson-Witten twist. This twist is the local identification of the $\mathrm{SU}(2)_+$ with the diagonal subgroup of the $\mathrm{SU}(2)_+ \times \mathrm{SU}(2)_R$ factor of the spin lift of the local spin group $\mathrm{Spin}(4) \cong \mathrm{SU}(2)_+ \times \mathrm{SU}(2)_-$ [61]. Alternatively, one can view the fields as sections of a non-trivial R-symmetry bundle, isomorphic to the spin bundle S^+ . Application of this to the representations of the vector multiplet (5.4) and (5.7) gives:

bosons:
$$(2, 2) \oplus (1, 1) \oplus (1, 1),$$

fermions: $(2, 2) \oplus (3, 1) \oplus (1, 1).$ (5.29)

Thus the bosons remain unchanged, a vector and a complex scalar, while the fermions reorganise to a vector, self-dual two-form and real scalar, which we denote as ψ , χ and η , respectively. We note that none of these fields are spinors, and can thus be considered on a non-spin four-manifold. The original supersymmetry generators also transform in the representations for the fermions above. Thus the theory contains a scalar fermionic supercharge $Q = \epsilon^{\dot{A}\dot{B}} \overline{Q}_{\dot{A}\dot{B}}$, whose cohomology provides the operators in the topological theory [61].

For the fields of a hypermultiplet, (5.6) and (5.7), one finds

bosons:
$$(1, 2) \oplus (1, 2),$$

fermions: $(2, 1) \oplus (1, 2) \oplus (2, 1) \oplus (1, 2).$ (5.30)

Thus hypermultiplet bosons become spinors, i.e. sections of the spin bundle S^+ , while the fermions are sections of S^+ and S^- . Thus the twisted hypermultiplets can in this case only be formulated on four-manifolds which are spin, i.e. $w_2(X) = 0$ [73, 255].

However, if the hypermultiplets are charged under a gauge field or flux, the product of these bundles with S^{\pm} may be a Spin^c bundle, W^+ or W^- [52,85,255]. The latter are defined for arbitrary four-manifolds. For example, an almost complex structure on X determines two canonical Spin^c bundles $W^{\pm} \simeq S^{\pm} \otimes K_X^{-1/2}$ with K_X the canonical class determined by the almost complex structure. Since the hypermultiplets are in the fundamental, twodimensional representation of SU(2), the topologically twisted hypermultiplets are well-defined on a non-spin four-manifold if $\mu = -K_X/2$ [73].

Let us state this also in terms of the gauge bundle E. To this end, we label the two components of the fundamental, two-dimensional representation of SU(2) by \pm . The two components are sections of a line bundle $\mathcal{L}_E^{\pm 1/2}$ with $c_1(\mathcal{L}_E) = \bar{w}_2(E)$. Of course, the square root $\mathcal{L}_E^{1/2}$ only exists if $w_2(E) \in 2L$. On the other hand, the physical requirement is that $S^+ \otimes \mathcal{L}_E^{1/2}$ is well defined, or $\bar{w}_2(X) + \bar{w}_2(E) \in 2L$. Therefore, the obstructions can cancel each other for a suitable choice of $w_2(E)$. Thus the topological twisted theory is not welldefined for an arbitrary choice of 't Hooft flux $\boldsymbol{\mu} \coloneqq \frac{1}{2}\bar{w}_2(E)$; but rather $\boldsymbol{\mu}$ has to satisfy $\boldsymbol{\mu} = \frac{1}{2}\bar{w}_2(X) \mod L$ [73], or

$$\bar{w}_2(X) = \bar{w}_2(E) \mod 2L.$$
 (5.31)

To consider more general 't Hooft fluxes $\boldsymbol{\mu}$ or equivalently $w_2(E)$, we can couple the *j*'th hypermultiplet to a background flux or line bundle \mathcal{L}_j , with \mathcal{L}_j possibly different for each *j*. We let $\mathcal{E}_j = \mathcal{L}_E \otimes \mathcal{L}_j$. Then the requirement that $S^{\pm} \otimes \mathcal{E}_j^{\pm 1/2}$ is globally well-defined is that

$$c_1(\mathcal{E}_j) \in \bar{w}_2(X) + 2L, \tag{5.32}$$

which can be satisfied for any $\bar{w}_2(E)$ for a suitable choice of \mathcal{L}_j . Thus we can formulate the *u*-plane integral for arbitrary $\bar{w}_2(E)$, if we require that the

background fluxes satisfy

$$c_1(\mathcal{L}_j) \in \bar{w}_2(X) + \bar{w}_2(E) + 2L,$$
 (5.33)

for each j. This is consistent with (5.31) for $c_1(\mathcal{L}_j) = 0$.

The Chern classes $c_1(\mathcal{L}_j)$ can also be seen as the splitting classes of the $\text{Spin}(2N_f)$ principal bundle \mathcal{L} . The Chern class of \mathcal{L} reads

$$c(\mathcal{L}) = \sum_{l=0}^{2} c_l(\mathcal{L}) = \prod_{j=1}^{N_f} (1 + c_1(\mathcal{L}_j)).$$
(5.34)

The scalar generators of the equivariant cohomology of $\text{Spin}(2N_f)$ are the masses m_j , which generate the N_f -dimensional Cartan subalgebra of $\text{Spin}(2N_f)$. The gauge bundle E_k is also $\text{Spin}(2N_f)$ equivariant. For generic masses, the flavour group is $U(1)^{N_f}$, and is enhanced for special loci of the masses, for example to $U(N_f)$ for equal masses [46].

The Q-fixed equations are the non-Abelian monopole equations with N_f matter fields in the fundamental representation. For generic gauge group G and with representation R, these equations read [254]

$$\left(F^{a}_{\dot{\alpha}\dot{\beta}}\right)^{+} + \frac{i}{2}\sum_{j=1}^{N_{f}}\bar{M}^{j}_{(\dot{\alpha}}T^{a}M^{j}_{\dot{\beta})} = 0,$$

$$\not D M^{j} = \sum_{\mu}\sigma^{\mu}D_{\mu}M^{j} = 0,$$
(5.35)

where T^a is a generator of the Lie algebra in the representation R. Including the sum over matrix elements, we have

$$M^{j}_{(\alpha}T^{a}M^{j}_{\beta)} = \sum_{k,l} (M^{j})^{k}_{(\alpha}(T^{a})^{kl}(M^{j})^{l}_{\beta)}.$$
 (5.36)

We denote the moduli space of solutions to (5.35) by $\mathcal{M}_{k,\mathcal{L}_{j}}^{Q,N_{f}}$, and leave the dependence on the 't Hooft flux $\boldsymbol{\mu}$ and the metric J implicit. For $N_{f} = 4$ on $X = \mathbb{CP}^{2}$, such moduli spaces are studied in [196].

The moduli spaces $\mathcal{M}_{k,\mathcal{L}_j}^{Q,N_f}$ is non-compact for vanishing masses [65, 265, 274]. This is improved upon turning on masses and localizing with respect to the U(1)^{N_f} flavour symmetry, $M_{\alpha}^j \to e^{i\varphi_j}M_{\alpha}^j$, which leave invariant the \mathcal{Q} -fixed equations (5.35). There are two components:

- the instanton component, with $F^+ = 0$ and $M^j = 0, j = 1, ..., N_f$. The moduli space for this component is denoted \mathcal{M}_k^i . Since the hypermultiplet fields vanish, this component is associated to the Coulomb branch.
- the abelian or monopole component, for which a U(1) subgroup of the flavour group acts as pure gauge. Here the connection is reducible, and a U(1) subgroup of the SU(2) gauge group is preserved. For generic masses, there are N_f such components, where M^{ℓ} is upper or lower triangular for
some ℓ , and $M^j = 0$ for all $j \neq \ell$. The moduli space of this component is denoted $\mathcal{M}_k^{\mathbf{a},j}$, $j = 1, \ldots, N_f$. Since some of the hypermultiplet fields are non-vanishing, this component is associated to the Higgs branch [89,274].

The instanton component \mathcal{M}_k^i is non-compact due to point-like instantons. This can be cured using the Uhlenbeck compactification or algebraic-geometric compactifications. We assume that the physical path integral chooses a specific compactification, whose details are however not manifest at the level of the low energy effective field theory other than that the compactification must be in agreement with the correlation functions.

5.3.3Correlation functions and moduli spaces

The Q-fixed equations (5.35) include a Dirac equation for each hypermultiplet $j = 1, \ldots, N_f$ in the fundamental representation. The corresponding index bundle W_k^j defines an element of the K-group of \mathcal{M}_k^i . Its virtual rank $\operatorname{rk}(W_k^j)$ is the formal difference of two infinite dimensions. It is given by an index theorem and reads

$$\operatorname{rk}(W_k^j) = -k + \frac{1}{4}(c_1(\mathcal{L}_j)^2 - \sigma) \in \mathbb{Z},$$
(5.37)

where $c_1(\mathcal{L}_i)$ is the first Chern class of the bundle \mathcal{L}_i . Note that the rhs is not an integer for an arbitrary $c_1(\mathcal{L}_j) \in H^2(X,\mathbb{Z})$. To verify that the rhs is integral for the $c_1(\mathcal{L}_j)$'s satisfying (5.33), we rewrite $\operatorname{rk}(W_k^j)$ as

$$\operatorname{rk}(W_k^j) = -(k + \mu^2) - c_1(\mathcal{L}_j) \cdot \mu + \frac{1}{4} \left((c_1(\mathcal{L}_j) + 2\mu)^2 - \sigma \right).$$
(5.38)

Then the first term on the rhs is an integer since $k \in -\frac{1}{4}w_2(E)^2 + \mathbb{Z}$ for an SO(3) bundle. The second term is an integer because $c_1(\mathcal{L}_i) \cdot \boldsymbol{\mu} = (\bar{w}_2(X) - 2\boldsymbol{\mu}) \cdot \boldsymbol{\mu}$ mod $\mathbb{Z} \in \mathbb{Z}$, and the third term is an integer using (5.28) and the fact that $c_1(\mathcal{L}_j) + 2\boldsymbol{\mu}$ equals the characteristic class of a Spin^c-structure \mathfrak{s}_j by (5.33),

$$c_1(\mathcal{L}_j) + 2\boldsymbol{\mu} = c(\mathfrak{s}_j), \qquad (5.39)$$

for each j.

The mass m_i is the equivariant parameter of the U(1) flavour symmetry associated to the j'th hypermultiplet. The equivariant Chern class of W_k^j reads in terms of the splitting class x_l ,

$$c(W_k^j) = \prod_{l=0}^{-\mathrm{rk}(W_k^j)} (x_l + m_j) = m_j^{-\mathrm{rk}(W_k^j)} \sum_l \frac{c_l(W_k^j)}{m_j^l}.$$
 (5.40)

We abbreviate $c_l(W_k^j)$ to $c_{l,j}$, and let $c(W_k) = \prod_{j=1}^{N_f} c(W_k^j)$. The moduli space $\mathcal{M}_{k,\mu,\mathcal{L}_j}^Q$ for N_f hypermultiplets corresponds to the vanishing locus of the obstructions for the existence of N_f zero modes of the Dirac operator. As a result, the virtual complex dimension of the moduli

space $\mathcal{M}_{k,\mu,\mathcal{L}_j}^Q$ is that of the instanton moduli space plus the sum of (typically negative) ranks of the index bundles W_k^j , $\operatorname{vdim}(\mathcal{M}_{k,\mathcal{L}_j}^{Q,N_f}) = \operatorname{vdim}(\mathcal{M}_k^Q)_{N_f=0} + \sum_{j=1}^{N_f} \operatorname{rk}(W_k^j)$ [85,99,261,265]. This gives

$$\operatorname{vdim}(\mathcal{M}_{k,\mathcal{L}_{j}}^{Q,N_{f}}) = (4 - N_{f})k + \frac{1}{4} \left(-3\chi - (3 + N_{f})\sigma + \sum_{j=1}^{N_{f}} c_{1}(\mathcal{L}_{j})^{2} \right). \quad (5.41)$$

5.4 The effective theory on a four-manifold

We consider in this section the low energy effective field theory on a fourmanifold. We derive the semi-classical action of the theory coupled to background U(1) fields. As in previous cases [72, 73, 85, 144], the final expression takes the form of a Siegel-Narain theta series multiplied by a measure factor.

5.4.1 Hypermultiplets and background fields

The effective theory coupled to N_f background fluxes can be modelled as that of a theory with gauge group $SU(2) \times U(1)^{N_f}$, where the fields of the U(1) factors have been frozen in a special way [85, 275]. To derive the precise form, we recall the low-energy effective Lagrangian for the r multiplets $(\phi^J, \eta^J, \chi^J, \psi^J, F^J)$ of the topologically twisted $U(1)^r$ SYM theory [87]. Since the u-plane integral reduces to an integral over zero-modes [73], it suffices to only include the zero-modes in the Lagrangian. For simply connected fourmanifolds, there is no contribution from the one-form fields ψ^J . The Lagrangian is then given in terms of the prepotential $F(\{a^J\})$ and its derivatives to the vevs $\langle \phi^J \rangle = a^J$, as

$$\mathcal{L} = \frac{i}{16\pi} (\bar{\tau}_{JK} F_{+}^{J} \wedge F_{+}^{K} + \tau_{JK} F_{-}^{J} \wedge F_{-}^{K}) - \frac{1}{8\pi} y_{JK} D^{J} \wedge D^{K} + \frac{i\sqrt{2}}{16\pi} \bar{\mathcal{F}}_{JKL} \eta^{J} \chi^{K} \wedge (D + F_{+})^{L},$$
(5.42)

with $y_{JK} = \text{Im}(\tau_{JK}), \tau_{JK} = \partial_J \partial_K F(\{a^J\})$ and $\mathcal{F}_{JKL} = \partial_J \partial_K \partial_L F(\{a^J\})$. It is left invariant by the BRST operator Q, which acts on the zero modes as

$$[Q, A^{J}] = \psi^{J} = 0, \qquad [Q, \psi^{J}] = 4\sqrt{2}da^{J},$$

$$[Q, a^{J}] = 0, \qquad [Q, \bar{a}^{J}] = \sqrt{2}i\eta^{J},$$

$$[Q, \eta^{J}] = 0, \qquad [Q, \chi^{J}] = i(F_{+} - D_{+})^{J},$$

$$[Q, D^{J}] = (d\psi^{J})_{+} = 0.$$
(5.43)

Using this operator, we can write \mathcal{L} as the sum of a topological, holomorphic term and a Q-exact term,

$$\mathcal{L} = \frac{i}{16\pi} \tau_{JK} F^J \wedge F^K + \{Q, W\}, \qquad (5.44)$$

with $W = -\frac{i}{8\pi} y_{JK} \chi^J (F_+ + D)^K$.

The low-energy theory of SU(2) gauge theory with N_f hypermultiplets coupled to N_f background fluxes can then be modelled by the above rank rdescription with $r = N_f + 1$. We identify $F(\{a^J\})$ with $F(a, \mathbf{m})$. We let the indices J, K run from 0 to N_f and identify the index 0 with the unbroken U(1) of the SU(2) gauge group and the indices $j, k, l = 1, \ldots, N_f$ with that of the frozen U(1)^{N_f} factors. We further set $\phi^0 \coloneqq \phi$ for any field ϕ . We will proceed by using lower indices for j, k, l, except where the summation convention is explicitly used, to avoid confusion with powers of the fields.

The masses of the hypermultiplets are the vevs of the frozen scalar fields of the corresponding vector multiplets, $\frac{m_j}{\sqrt{2}} = \langle \phi_j \rangle = a_j$ [275]. We set $[F_j] = 4\pi k_j$ with

$$\boldsymbol{k}_j = c_1(\mathcal{L}_j)/2 \in L/2. \tag{5.45}$$

To make the BRST variations of the fields from the frozen U(1) factors vanish, we set $\eta_j = \chi_j = 0$, as well as $D^j = F^j_+$. With these identifications, the Lagrangian becomes

$$\mathcal{L} = \frac{i}{16\pi} \tau_{JK} F^J \wedge F^K + \frac{1}{8\pi} y_{00} F_+ \wedge F_+ - \frac{1}{8\pi} y_{00} D \wedge D + \frac{i\sqrt{2}}{16\pi} \bar{\mathcal{F}}_{000} \eta \chi \wedge (D + F_+) + \frac{i\sqrt{2}}{8\pi} \bar{\mathcal{F}}_{00j} \eta \chi \wedge F_+^j + \frac{1}{4\pi} y_{0j} (F_+ - D) \wedge F_+^j.$$
(5.46)

Integrating over D, η and χ in the standard way [73, 85, 87], we end up with

$$\int dD d\eta d\chi \, e^{-\int_X \mathcal{L}}$$

$$= \frac{\partial}{\partial \bar{a}} \left(i \sqrt{y_{00}} \, B \left(F + \frac{y_{0j}}{y_{00}} F^j, J \right) \right) e^{-\int_X \mathcal{L}_0},$$
(5.47)

where

$$\mathcal{L}_{0} = \frac{i}{16\pi} \tau_{JK} F^{J} \wedge F^{K} + \frac{1}{8\pi} y_{00} F_{+} \wedge F_{+} + \frac{y_{0j}}{4\pi} F_{+} \wedge F_{+}^{j} + \frac{1}{8\pi} \frac{y_{0j} y_{0k}}{y_{00}} F_{+}^{j} \wedge F_{+}^{k}$$

$$= \frac{i}{16\pi} (\bar{\tau} F_{+} \wedge F_{+} + \tau F_{-} \wedge F_{-}) + \frac{i}{8\pi} (v_{j} F_{-} \wedge F_{-}^{j} + \bar{v}_{j} F_{+} \wedge F_{+}^{j})$$

$$+ \frac{i}{16\pi} w_{jk} F^{j} \wedge F^{k} + \frac{y}{8\pi} \mathrm{Im}(v_{j}) \mathrm{Im}(v_{k}) F_{+}^{j} \wedge F_{+}^{k},$$
(5.48)

and we identified $\tau \coloneqq \tau_{00}$, $y = \text{Im}(\tau) = y_{00}$, $v_j \coloneqq \tau_{0j}$ and $w_{jk} \coloneqq \tau_{jk}$. Thus the coupling w_{jk} is holomorphic, but the coupling v_j is non-holomorphic. This is similar to the couplings for $\mathcal{N} = 2^*$ [85].

5.4.2 Sum over fluxes

The path integral includes a sum over fluxes $\mathbf{k} = [F]/4\pi \in L/2$. After summing the exponentiated action (5.47) over the fluxes \mathbf{k} and multiplying by $\frac{d\bar{a}}{d\bar{\tau}}$, we find that this takes the form

$$\sum_{\boldsymbol{k}\in L+\boldsymbol{\mu}}\int dDd\eta d\chi \, e^{-\int_{X}\mathcal{L}} = \left(\prod_{j,k=1}^{N_{f}} C_{jk}^{B(\boldsymbol{k}_{j},\boldsymbol{k}_{k})}\right) \, \Psi_{\boldsymbol{\mu}}^{J}(\tau,\bar{\tau},\boldsymbol{z},\bar{\boldsymbol{z}}). \tag{5.49}$$

The couplings C_{jk} are given in terms of w_{jk} (5.12) by

$$C_{jk} = e^{-\pi i w_{jk}}, (5.50)$$

for $j, k = 1, ..., N_f$. Such couplings were first put forward in [144], and were also crucial in [85].

The term Ψ^J_{μ} is an example of a Siegel-Narain theta function. It reads explicitly

$$\Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},\boldsymbol{z},\bar{\boldsymbol{z}}) = e^{-2\pi y \boldsymbol{b}_{+}^{2}} \sum_{\boldsymbol{k}\in L+\boldsymbol{\mu}} \partial_{\bar{\tau}} \left(4\pi i \sqrt{y} B(\boldsymbol{k}+\boldsymbol{b},J)\right) \\ \times (-1)^{B(\boldsymbol{k},K)} q^{-\boldsymbol{k}_{-}^{2}/2} \bar{q}^{\boldsymbol{k}_{+}^{2}/2} e^{-2\pi i B(\boldsymbol{z},\boldsymbol{k}_{-})-2\pi i B(\bar{\boldsymbol{z}},\boldsymbol{k}_{+})},$$
(5.51)

and discussed in more detail in Appendix A.7. The elliptic variable reads in terms of v_j and k_j ,

$$\boldsymbol{z} = \sum_{j=1}^{N_f} v_j \boldsymbol{k}_j, \quad \text{and} \quad \boldsymbol{b} = \frac{\text{Im}(\boldsymbol{z})}{y},$$
 (5.52)

thus inducing a non-holomorphic dependence on v_j . Furthermore, K appearing in the fourth root of unity $(-1)^{B(\mathbf{k},K)}$ is a characteristic vector of L. Note that $\Psi^J_{\boldsymbol{\mu}}$ changes by the sign $(-1)^{B(\boldsymbol{\mu},K-K')}$ upon replacing K by a different characteristic vector K' [73,85,276].

For $N_f = 0$, this phase can be understood as arising from integrating out massive fermionic modes [72]. It also appears naturally in decoupling the adjoint hypermultiplet in the analogous function for $\mathcal{N} = 2^*$ [85]. For $N_f > 0$, the constant part of the couplings v_j (5.12) effectively contribute to the phase, such that the total phase reads

$$e^{\pi i B(\mathbf{k},K)} \prod_{j=1}^{N_f} e^{\pi i n_j B(\mathbf{k}_j,\mathbf{k})},$$
 (5.53)

with n_j the magnetic winding numbers. For arbitrary $n_j \in \mathbb{Z}$, the phase is an eighth root of unity. It would be interesting to understand this phase from integrating out massive modes.

We deduce from (5.53) that the summand of Ψ^J_{μ} changes by a phase

$$e^{\pi i (n_j' - n_j) B(\boldsymbol{k}_j, \boldsymbol{k})} \tag{5.54}$$

if the winding numbers n_j are replaced by n'_j . Since $\mathbf{k}_j \in K/2 - \boldsymbol{\mu} \mod L$ (see (5.33)) and $\mathbf{k} \in L + \boldsymbol{\mu}$, this phase is 1 if $n'_j - n_j = 0 \mod 4$. We can therefore restrict to $n_j \in \mathbb{Z}_4$. For specific choices of $\boldsymbol{\mu}$ and \boldsymbol{k}_j , the n_j can lie in a subgroup of \mathbb{Z}_4 .

The modular transformations of Ψ^J_{μ} are discussed in Appendix A.7, which are crucial input for single-valuedness of the *u*-plane integrand. We will demonstrate in section 5.5.2 that the *u*-plane integrand is single-valued if we impose further constraints on the winding numbers n_i .

Finally, if the theory is considered on a curved background, topological couplings arise in the effective field theory [72]. These terms couple to the Euler characteristic and the signature of the four-manifold X, respectively denoted A and B. These take the form [72, 73],

$$A = \alpha \left(\frac{du}{da}\right)^{1/2}, \qquad B = \beta \Delta_{N_f}^{1/8}.$$
(5.55)

Here, Δ_{N_f} is the physical discriminant incorporating the singularities of the effective theory, while $\frac{du}{da}$ is the (reciprocal of) the periods of the SW curves as introduced in section 5.2. Both can be determined directly from the SW curve, as described in section 2.3.1. The prefactors α and β are independent of u, but can be functions of other moduli such as the masses \boldsymbol{m} , the dynamical scale Λ_{N_f} or the UV coupling τ_{UV} . However, it turns out that for the theories with fundamental matter they are independent of the masses and only depend on the scale [66, 111]. They satisfy several constraints from holomorphy, RG flow, homogeneity and dimensional analysis, and can in principle be fixed for any Lagrangian theory from a computation in the Ω -background [66, 82, 111, 129].

5.4.3 Observables and contact terms

The observables in the topologically twisted theories are the point observable or 0-observable u, as well as d-observables supported on a d-dimensional submanifold of X. The d-observables are only non-vanishing if the submanifold corresponds to a non-trivial homology class. For $b_1 = 0$, the d-observables with d odd therefore do not contribute.

To introduce the surface observable, let $\boldsymbol{x} \in H_2(X, \mathbb{Q})$. Then the surface observable reads in terms of the UV fields,

$$I(\boldsymbol{x}) = \frac{1}{4\pi^2} \int_{\boldsymbol{x}} \operatorname{Tr} \left[\psi \wedge \psi - \frac{1}{\sqrt{2}} \phi F \right].$$
 (5.56)

In the effective infrared theory, this operator becomes,

$$\tilde{I}(\boldsymbol{x}) = \frac{i}{\sqrt{2}\pi} \int_{\boldsymbol{x}} \frac{1}{32} \frac{d^2 u}{da^2} \psi \wedge \psi - \frac{\sqrt{2}}{4} \frac{du}{da} (F_- + D).$$
(5.57)

Generating functions of correlation functions are obtained by inserting

$$e^{p \, u/\Lambda_{N_f}^2 + I(\boldsymbol{x})/\Lambda_{N_f}} \tag{5.58}$$

in the path integral. The surface observable leads to a change in the argument of the sum over fluxes (5.51),

$$z \to z + \frac{x}{2\pi \Lambda_{N_f}} \frac{du}{da}, \qquad \bar{z} \to \bar{z}.$$
 (5.59)

and to analytically continue **b** (5.52) to the complex number by setting **b** = $(\mathbf{z} - \bar{\mathbf{z}})/(2iy)$.

The inclusion of the surface observable also gives rise to a contact term [70, 73, 277], which in particular ensures that the *u*-plane integrand is single-valued. For $0 \leq N_f \leq 3$, the contact term is $\exp(\boldsymbol{x}^2 G_{N_f})$ with [74, 98, 278]

$$G_{N_f} = -\frac{1}{24\Lambda_{N_f}^2} E_2 \left(\frac{du}{da}\right)^2 + \frac{1}{3\Lambda_{N_f}^2} \left(u + \frac{\Lambda_3^2}{64}\delta_{N_f,3}\right), \qquad (5.60)$$

while for $N_f = 4$ it is given by [98, 174]

$$G_{N_f=4} = -\frac{1}{24\Lambda_4^2} E_2 \left(\frac{du}{da}\right)^2 + \frac{u}{3\Lambda_4^2} E_2(\tau_{\rm UV}) + \frac{1}{18\Lambda_4^2} \left[\!\left[m_1^2\right]\!\right] E_4(\tau_{\rm UV}).$$
(5.61)

A more general scheme to fix the contact terms is proposed in [74]. Contact terms can also be derived from the corresponding Whitham hierarchies [278, 279]. In the presence of surface observables, there are additional mixed contact terms $\frac{\partial^2 \mathcal{F}}{\partial \tau_0 \partial m}$ for the external fluxes $\{k_j\}$ as encountered in [85] for the $\mathcal{N} = 2^*$ theory.

5.5 The *u*-plane integral

In this section, we set up the *u*-plane integral schematically given in (5.2), and demonstrate that it is well-defined on the integration domain for any $\boldsymbol{\mu}$ with appropriate background fluxes. The case $\boldsymbol{\mu} = \bar{w}_2(X)/2$ and $\boldsymbol{k}_j = 0$ was analysed in [73].

5.5.1 Definition of the integrand

As discussed in the previous sections, the *u*-plane integral on a closed fourmanifold X with $(b_1, b_2^+) = (0, 1)$ depends on many parameters. We summarise:

- The scale Λ_{N_f} and masses $\boldsymbol{m} = (m_1, \dots, m_{N_f})$ of the theory. See section 5.2.
- The magnetic winding numbers n_j , $j = 1, ..., N_f$. See section 5.2.2.
- The four-manifold X, in particular its signature $\sigma = \sigma(X)$, Euler characteristic $\chi = \chi(X)$, period point J and intersection form Q. See section 5.3.1.
- The 't Hooft flux μ , and the external fluxes $\{k_j\} = (k_1, \ldots, k_{N_f})$. See section 5.3.3.

• The fugacities for the point and surface observables p and x. See section 5.4.3.

In terms of these parameters, the *u*-plane path integral reduces to the following finite dimensional integral over $\mathcal{F}_{N_f}(\boldsymbol{m})$,

$$\Phi^{J}_{\boldsymbol{\mu},\{\boldsymbol{k}_{j}\}}(\boldsymbol{p},\boldsymbol{x},\boldsymbol{m},\Lambda_{N_{f}}) = \\ \mathcal{K}_{N_{f}} \int_{\mathcal{F}_{N_{f}}(\boldsymbol{m})} d\tau \wedge d\bar{\tau} \,\nu(\tau;\{\boldsymbol{k}_{j}\}) \,\Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},\boldsymbol{z},\bar{\boldsymbol{z}}) \,e^{2pu+\boldsymbol{x}^{2}G_{N_{f}}}.$$
(5.62)

We summarise the different elements on the rhs:

• \mathcal{K}_{N_f} is an overall normalisation factor. For $N_f = 0$, it is fixed by matching to known Donaldson invariants. Due to $\chi + \sigma = 4$, there is an ambiguity [82]

$$(\mathcal{K}_{N_f}, \alpha, \beta) \sim (\zeta^{-4} \mathcal{K}_{N_f}, \zeta \alpha, \zeta \beta),$$
 (5.63)

with α and β the *u*-independent prefactors in (5.55).

- The integration domain $\mathcal{F}_{N_f}(\boldsymbol{m})$ in (5.62) is crucially the fundamental domain of the effective gauge coupling. As discussed in section 2, this domain requires new aspects compared to integration domains for earlier discussions of *u*-plane integrals. The evaluation of integrals over $\mathcal{F}_{N_f}(\boldsymbol{m})$ will be discussed in more detail in section 5.6.
- ν is the "measure factor" [66, 72–74, 85]

$$\nu(\tau; \{\boldsymbol{k}_j\}) = \frac{da}{d\tau} A^{\chi} B^{\sigma} \prod_{i,j=1}^{N_f} C_{ij}^{B(\boldsymbol{k}_i, \boldsymbol{k}_j)}.$$
(5.64)

It combines the topological couplings (5.55) and the couplings to the background fluxes (5.50) with the Jacobian $\frac{da}{d\tau}$ of the change of variables from a to τ .

• The function Ψ^J_{μ} arises from the sum over U(1) fluxes. It is a Siegel-Narain theta function (5.51) and discussed in detail in section 5.4.2. The elliptic parameter \boldsymbol{z} of the Siegel-Narain theta function reads

$$\boldsymbol{z} = \frac{\boldsymbol{x}}{2\pi \Lambda_{N_f}} \frac{d\boldsymbol{u}}{d\boldsymbol{a}} + \sum_{j=1}^{N_f} v_j \boldsymbol{k}_j,$$

$$\bar{\boldsymbol{z}} = \sum_{j=1}^{N_f} \bar{v}_j \boldsymbol{k}_j.$$
 (5.65)

• Finally, G_{N_f} is the contact term, discussed in more detail in section 5.4.3.

While the path integral set up in section 5.4 integrates the exponentiated action over the local coordinates a and \bar{a} , in (5.62) we have changed variables to τ and $\bar{\tau}$. This change of variables $(a, \bar{a}) \to (\tau, \bar{\tau})$ is valid as long as the Jacobian is nonsingular in the integration region. Since the coordinates a and \bar{a} are holomorphic and anti-holomorphic respectively, the Jacobian is diagonal and the functional determinant accordingly reads $\frac{da}{d\tau} \frac{d\bar{a}}{d\bar{\tau}}$. We thus need to show that $\frac{da}{d\tau}$ is not singular away from isolated points in $\mathcal{F}_{N_f}(\boldsymbol{m})$, which in (5.62) we remove implicitly from the integration domain.

Using $\frac{da}{du} = \frac{da}{du}\frac{du}{d\tau}$, we can study the singular points in detail. First, it is shown in section 2 that the singularities of $\frac{du}{d\tau}$ are in one-to-one correspondence with the branch points. In fact, both $\frac{du}{d\tau} = 0$ and $\frac{du}{d\tau} = \infty$ are realised as branch points in $\mathcal{N} = 2$ SQCD. In the following section 5.6, we remove a small circle in $\mathcal{F}_{N_f}(\boldsymbol{m})$ around the branch points, and show that they do not give an extra contribution. Furthermore, the solutions to $\frac{du}{da} = 0$ are shown to be the Argyres-Douglas (AD) points. We exclude them from the integration region, and study their contribution also in section 5.6. Finally, we know that $\eta^{24} \propto \left(\frac{da}{du}\right)^{12} \Delta_{N_f}$ (2.48), with η the Dedekind eta-function as defined in (A.18). Since $\eta \neq 0$ and Δ_{N_f} does not have poles, we find that $\frac{da}{du}$ never vanishes. This agrees with the fact that $\frac{da}{du}$ is the period of a holomorphic differential and therefore is never zero.

We conclude that the functional determinant is singular in \mathbb{H} precisely at the branch points and AD points, however with the proper exclusion of those as done in the following Section, it is non-singular and the change of variables is well-defined. This furthermore conveniently solves the problem that there is no natural integration region in (a, \bar{a}) space [73].

5.5.2 Monodromy transformations of the integrand

We continue by explicitly verifying that the *u*-plane integral is single-valued around the singular points of the moduli space. We find that this puts a constraint on the magnetic winding numbers n_j , in addition to the constraints on the background fluxes k_j discussed in section 5.3.3.

Monodromy around infinity

Let us determine how the *u*-plane integrand transforms under the monodromy around infinity. As a function of the effective coupling τ , the measure factor (5.64) is proportional to $\frac{da}{d\tau} \left(\frac{du}{da}\right)^{\frac{\chi}{2}} \Delta^{\frac{\sigma}{8}}$ times the product over the couplings C_{ij} . We take the monodromy at infinity to be oriented as $u \to e^{2\pi i}u$ and $a \to e^{\pi i}a$, as in section 5.2.3. Then this path also encircles all singularities u_j , which are the roots of the physical discriminant, $\Delta = \prod_{j=1}^{N_f+2} (u - u_j)$. We thus have that $\Delta \to e^{2\pi i (N_f+2)}\Delta$, and hence

$$\Delta^{\frac{\sigma}{8}} \to e^{\pi i (N_f + 2)\sigma/4} \Delta^{\frac{\sigma}{8}} \tag{5.66}$$

Next, since $u \to e^{2\pi i}$ and $a \to e^{\pi i}a$ we find $\frac{du}{da} \to e^{\pi i}\frac{du}{da}$, and therefore

$$\left(\frac{du}{da}\right)^{\frac{\chi}{2}} \to e^{\pi i \chi/2} \left(\frac{du}{da}\right)^{\frac{\chi}{2}}.$$
(5.67)

For $\frac{da}{d\tau}$ we have that $a \to e^{\pi i}a$, while $d\tau \to d\tau$, and thus

$$\frac{da}{d\tau} \to -\frac{da}{d\tau}.$$
(5.68)

From (5.16) we recall that $w_{ij} \to w_{ij} + \delta_{ij}$, such that with the definition (5.50) we find $C_{ij} \to e^{-\pi i \delta_{ij}} C_{ij}$. The couplings C_{ij} transform in the measure factor as

$$\prod_{i,j=1}^{N_f} C_{ij}^{B(\boldsymbol{k}_i, \boldsymbol{k}_j)} \to e^{-\pi i \sum_j \boldsymbol{k}_j^2} \prod_{i,j=1}^{N_f} C_{ij}^{B(\boldsymbol{k}_i, \boldsymbol{k}_j)}.$$
(5.69)

Combining (5.66), (5.67), (5.68), (5.69), and using $\chi = 4 - \sigma$, we obtain

$$\nu \to -e^{\pi i N_f \sigma/4} e^{-\pi i \sum_j k_j^2} \nu. \tag{5.70}$$

This phase for $\mathbf{k}_j = 0$ can be checked directly by taking *q*-expansions from the SW curves, for generic masses.

From (5.16) we recall that under the monodromy around infinity $v_j \rightarrow -v_j - n_j$, and thus $\boldsymbol{z} \rightarrow -\boldsymbol{z} - \sum_{j=1}^{N_f} n_j \boldsymbol{k}_j$. Recall from (5.33) that $c_1(\mathcal{L}_j) \equiv K - 2\boldsymbol{\mu} \mod 2L$. For the sum over fluxes, in [5] we show that

$$\mathbf{M}_{\infty}: \qquad \nu(\tau; \{\boldsymbol{k}_j\}) \, \Psi^{J}_{\boldsymbol{\mu}}(\tau, \boldsymbol{z}) \to e^{2\pi i \boldsymbol{\mu} \sum_j (n_j+1) \boldsymbol{k}_j} \, \nu(\tau; \{\boldsymbol{k}_j\}) \, \Psi^{J}_{\boldsymbol{\mu}}(\tau, \boldsymbol{z}), \quad (5.71)$$

and the *u*-plane integrand is invariant under T^{N_f-4} if and only if $\boldsymbol{\mu} \sum_j (n_j + 1) \boldsymbol{k}_j \in \mathbb{Z}$. Using the fact that K is a characteristic vector of L, we find

$$n_j \equiv 1 \mod 2 \tag{5.72}$$

for all $j = 1, ..., N_f$, which implies the above constraint.

Monodromy M_j

Let us determine how the integrand transforms under the monodromy M_j around the mass singularity $m_j/\sqrt{2}$. Since the mass singularity corresponds to a singularity u_j on the *u*-plane, we have that $(u - u_j) \rightarrow e^{2\pi i}(u - u_j)$. This implies that $\Delta = (u - u_j) \prod_{i \neq j}^{2+N_f} (u - u_i) \rightarrow e^{2\pi i} \Delta$, such that $\Delta^{\frac{\sigma}{8}} \rightarrow e^{\pi i \sigma/4} \Delta^{\frac{\sigma}{8}}$. The transformation of $\frac{da}{du}$ can be determined from (2.46): While $u \rightarrow u_j$, both g_2 and g_3 remain finite and nonzero (otherwise u_j would be an AD point). This implies that $\frac{g_3}{g_2}$ contains no factors of $(u - u_j)$, and thus $\frac{du}{da} \rightarrow \frac{du}{da}$. Similarly, we have that $\frac{da}{d\tau} \rightarrow \frac{da}{d\tau}$. From (5.18) we finally have that $w_{ik} \rightarrow w_{ik} + \delta_{ij}\delta_{ik}$. We combine

$$\mathbf{M}_j: \quad \nu \to e^{\pi i \sigma/4} e^{-\pi i \mathbf{k}_j^2} \nu. \tag{5.73}$$

For the monodromy around the mass singularity $m_j/\sqrt{2}$, we find for Ψ^J_{μ} with (A.58)

$$\Psi^{J}_{\boldsymbol{\mu}}(\tau+1,\boldsymbol{z}-\boldsymbol{k}_{j}) = e^{-\pi i\sigma/4 + \pi i\boldsymbol{k}_{j}^{2}} \Psi^{J}_{\boldsymbol{\mu}}(\tau,\boldsymbol{z}).$$
(5.74)

The phases thus cancel precisely,

$$\mathbf{M}_{j}: \quad \nu(\tau+1) \Psi_{\boldsymbol{\mu}}^{J}(\tau+1, \boldsymbol{z}-\boldsymbol{k}_{j}) = \nu(\tau) \Psi_{\boldsymbol{\mu}}^{J}(\tau, \boldsymbol{z}), \quad (5.75)$$

without any constraints.

Monodromy M_m

For the monopole singularity in $N_f = 1$ we find a further constraint. Since Ψ^J_{μ} is required to transform to itself up to an overall factor, we must demand that $(n+1)\mathbf{k}_1/2 \in L$. Therefore for $\mathbf{k}_1 \in L/2$, we find the requirement that $n = -1 \in \mathbb{Z}_4$. This simplifies the transformations considerably, and we find

$$\Psi^{J}_{\mu}(\tau, \boldsymbol{z}) \to (-\tau + 1)^{b_{2}/2} (-\bar{\tau} + 1)^{2} e^{-\pi i \sigma/4} e^{\pi i \boldsymbol{k}_{1}^{2} \frac{v^{2}}{-\tau + 1}} \Psi^{J}_{\mu}(\tau, \boldsymbol{z}).$$
(5.76)

The k_j -independent part of the measure factor transforms precisely as under \mathbf{M}_j (see (5.73)), as the same argument holds. However, due to the transformation $\tau \to \frac{\tau}{-\tau+1}$, the measure also picks up its modular weight $\frac{\sigma}{2}+1$. From (5.21) we furthermore find the transformation of C_{11} , such that

$$\nu(\tau, \mathbf{k}_1) \to e^{\pi i \sigma/4} e^{-\pi i \mathbf{k}_1^2 \frac{v^2}{-\tau+1}} (-\tau+1)^{\frac{\sigma}{2}+1} \nu(\tau, \mathbf{k}_1), \qquad (5.77)$$

where we have already used $n = -1 \in \mathbb{Z}_4$. If we multiply (5.76) and (5.77) with $d\tau \wedge d\bar{\tau}$ (which has modular weight (-2, -2)), then

$$\mathbf{M}_m: \quad d\tau \wedge d\bar{\tau}\nu(\tau, \boldsymbol{k}_1) \,\Psi^J_{\boldsymbol{\mu}}(\tau, \boldsymbol{z}) \to d\tau \wedge d\bar{\tau}\nu(\tau, \boldsymbol{k}_1) \,\Psi^J_{\boldsymbol{\mu}}(\tau, \boldsymbol{z}), \tag{5.78}$$

where we have used $\sigma + b_2 = 2$. Thus, the *u*-plane integrand is also invariant under \mathbf{M}_m .

For $N_f > 1$ we find the same condition, namely that $n_j = -1 \mod 4$ for all j.

Monodromy M_d

Given the relation (5.19), it is not necessary to explicitly check single-valuedness of the integrand under this monodromy, as it is a product of the above monodromies.

To conclude this section, let us stress the constraints for the winding number n_j , such that the *u*-plane integral is invariant under all monodromies in $N_f \leq 3$. To this end, we need to satisfy the constraints $n_j = 1 \mod 2$ (5.72) from \mathbf{M}_{∞} , and $n_j = -1 \mod 4$ for \mathbf{M}_m . Since the latter is the stronger constraint, we require

$$n_j = -1 \mod 4,\tag{5.79}$$

for all $j = 1, \ldots, N_f$.

5.6 Integration over fundamental domains

As discussed in Sections 5.2 and 5.5, *u*-plane integrals for massive $\mathcal{N} = 2$ theories with fundamental hypermultiplets include new aspects. This section discusses how to evaluate such integrals (5.62). More concretely, we aim to define and evaluate integrals of the form

$$\mathcal{I}_f = \int_{\mathcal{F}(\boldsymbol{m})} d\tau \wedge d\bar{\tau} \, y^{-s} \, f(\tau, \bar{\tau}), \qquad (5.80)$$

with $s \leq 1$. The domain $\mathcal{F}(\boldsymbol{m})$ is the fundamental domain for the effective coupling constant as discussed in section 2, and f a non-holomorphic function of weight (2 - s, 2 - s) arising from the topologically twisted Yang–Mills theory. For $\mathcal{F}(\boldsymbol{m})$ a fundamental domain of a congruence subgroup, such integrals (5.80) have been studied in the context of theta lifts of weakly holomorphic modular forms and harmonic Maass forms [280–282] as well as one-loop amplitudes in string theory [283–285].

We assume that the integrand $y^{-s} f(\tau, \bar{\tau})$ can be expressed as

$$\partial_{\bar{\tau}}h(\tau,\bar{\tau}) = y^{-s} f(\tau,\bar{\tau}), \qquad (5.81)$$

for a suitable function $\hat{h}(\tau, \bar{\tau})$ using mock modular forms. This was indeed the case in [81, 83, 85], and will be demonstrated for massive $\mathcal{N} = 2$ theories with fundamental hypermultiplets. The integral \mathcal{I}_f then reads

$$\mathcal{I}_f = -\int_{\partial \mathcal{F}(\boldsymbol{m})} d\tau \, \widehat{h}(\tau, \bar{\tau}), \qquad (5.82)$$

with $\partial \mathcal{F}(\boldsymbol{m})$ the boundary of $\mathcal{F}(\boldsymbol{m})$. We will carry this out evaluation in Part II [286].

There are a number of aspects to be addressed in order to evaluate integrals over $\mathcal{F}(\boldsymbol{m})$:

- 1. Identifications of boundary components of $\mathcal{F}(\boldsymbol{m})$ due to monodromies on the *u*-plane.
- 2. Contributions from the cusps, that is $\tau \to i\infty$ or $\tau \to \gamma(i\infty) \in \mathbb{Q}$ for an element $\gamma \in PSL(2,\mathbb{Z})$.
- 3. Contributions from a singular point in the interior of $\mathcal{F}(\boldsymbol{m})$.
- 4. Contributions from an elliptic point $p \in \mathbb{H}$ of $PSL(2,\mathbb{Z})$.
- 5. Branch points and branch cuts.

We will discuss these aspects 1.–5. in the following.

1. Identifications

The modular transformation induced by monodromies identify components

of the boundary of the fundamental domain $\partial \mathcal{F}(\boldsymbol{m})$ pairwise. Their contributions to the integral (5.82) vanish, which is, for example, familiar from deriving valence formulas for modular forms [287, Fig. 2]. See Fig. 1 for an example.

2. Cusps

Contributions near the cusps require a regularisation [73,83]. Such regularisations have been developed in the context of string amplitudes [283–285] and analytic number theory [287–289].

Let us first consider the cusp $\tau \to i\infty$. To regularise the divergence, one introduces a cut-off Im $\tau = Y \gg 1$, and takes the limit $Y \to \infty$ after evaluation. We require that f near $i\infty$ has a Fourier expansion of the form³²

$$f(\tau,\bar{\tau}) = \sum_{m \gg -\infty, n \ge 0} c(m,n) q^m \bar{q}^n.$$
(5.83)

Then the function \hat{h} has the form,

$$\widehat{h}(\tau,\bar{\tau}) = h(\tau) + 2^s \int_{-\bar{\tau}}^{i\infty} \frac{f(\tau,-v)}{(-i(v+\tau))^s} dv,$$
(5.84)

where $h(\tau)$ is a weakly holomorphic q-series, with expansion

$$h(\tau) = \sum_{m \gg -\infty} d(m) q^m.$$
(5.85)

The cusp $\tau \to i\infty$ then contributes

$$\left[\mathcal{I}_f\right]_{\infty} = w_{\infty} \, d(0), \tag{5.86}$$

with d(0) the constant term of $h(\tau)$ (5.85), and w_{∞} the width of the cusp $\mathcal{F}(\boldsymbol{m})$ at $i\infty$. For $N_f \leq 3$, w_{∞} is $4 - N_f$ (see section 2).

The other cusps can be treated in a similar fashion using modular transformations. We label the n_c cusps in $\mathcal{F}(\mathbf{m})$ by $j = 1, \ldots, n_c$. If the cusp is on the horizontal axis at $-\frac{d_j}{c_j} \in \mathbb{Q}$ with relative prime $(c_j, d_j) \in \mathbb{Z}^2$, we can map the cusp to $i\infty$ by a modular transformation

$$\gamma_j = \begin{pmatrix} a_j & b_j \\ c_j & d_j \end{pmatrix}. \tag{5.87}$$

We let $\tau_j = \gamma_j \tau$. Then the holomorphic part $h_j(\tau_j)$ of $(c_j \tau + d_j)^{-2} \hat{h}(\gamma \tau_j, \gamma \bar{\tau}_j)$ can be expanded for τ near $-\frac{d_j}{c_j}$ as

$$h_j(\tau_j) = \sum d_j(n) q_j^n, \qquad q_j = e^{2\pi i \tau_j}.$$
 (5.88)

As a result, the cusp j contributes

$$\left[\mathcal{I}_{f}\right]_{j} = w_{j} \, d_{j}(0). \tag{5.89}$$

³²Also if f does not satisfy this requirement, the integral can be regularised as explained in [83,289]. We do not need this regularisation for the correlators in this section.

3. Singular points in the interior of $\mathcal{F}(\boldsymbol{m})$

The integrand can be singular at a point τ_s in the interior of $\mathcal{F}(\boldsymbol{m})$. Such singularities appear typically for deformations of superconformal theories, such as the $\mathcal{N} = 2^*$ theory and the $N_f = 4$ theory, where the UV coupling τ_{UV} gives rise to such a singularity [3,85]. See Fig. 26 for an example. We require that the expansion of f near such a singularity reads,

$$f(\tau, \bar{\tau}) = \sum_{m \gg -\infty, n \ge 0} c_s(m, n) \, (\tau - \tau_s)^m \, (\bar{\tau} - \bar{\tau}_s)^n.$$
(5.90)

Then, the anti-derivative $\hat{h}(\tau, \bar{\tau})$ has similar expansion,

$$\widehat{h}(\tau,\bar{\tau}) = \sum_{m \gg -\infty, n \ge 0} d_s(m,n) \left(\tau - \tau_s\right)^m (\bar{\tau} - \bar{\tau}_s)^n.$$
(5.91)

The contour integral for a small contour around $\tau_{\rm s}$,

$$C_{\varepsilon}(\tau_{\rm s}) = \left\{ \tau = \tau_{\rm s} + \varepsilon \, e^{i\varphi}, \varphi \in [0, 2\pi) \right\},\tag{5.92}$$

is bounded for such a function. Moreover, in the limit $\varepsilon \to 0$, the contour integral is finite. We define the "residue" of a non-holomorphic function $g(\tau, \bar{\tau})$

$$\operatorname{nRes}_{\tau=\tau_{\rm s}}\left[g(\tau,\bar{\tau})\right] = \frac{1}{2\pi i} \lim_{\varepsilon \to 0} \oint_{C_{\varepsilon}(\tau_{\rm s})} g(\tau,\bar{\tau}) \, d\tau.$$
(5.93)

For the expansion (5.90) this evaluates to

$$\left[\mathcal{I}_f\right]_s = 2\pi i \operatorname{nRes}_{\tau=\tau_s} \left[\widehat{h}(\tau,\bar{\tau})\right] = d_s(-1,0), \tag{5.94}$$

with $d_s(-1,0)$ the coefficient in the expansion (5.91).

4. *Elliptic points*

For $\mathcal{N} = 2$ QCD, AD points are the elliptic points of the duality group, and lie on the boundary of $\mathcal{F}(\boldsymbol{m})$. See Fig. 7 for an example. The elliptic points are $\alpha = e^{\pi i/3}$ and i, and their images under PSL(2, Z). Contour integrals around such points can be regularised using a cut-off ε . We assume that the anti-derivative \hat{h} has the following expansion near an elliptic point τ_e ,

$$\widehat{h}(\tau,\bar{\tau}) = \sum_{m \gg -\infty, n \ge 0} d_{\mathbf{e}}(m,n) \left(\tau - \tau_{\mathbf{e}}\right)^m \left(\bar{\tau} - \bar{\tau}_{\mathbf{e}}\right)^n.$$
(5.95)

As a result, the boundary arc around τ_{AD} in \mathbb{H} is a fraction of 2π , which needs to be properly accounted for. These neighbourhoods have an angle $\frac{2\pi}{k_e}$, with $k_e = 2$ for $\tau_e = i$, and $k_e = 6$ for $\tau_e = \alpha$ [145]. Furthermore, it is important how many images of \mathcal{F} in $\mathcal{F}(\mathbf{m})$ coincide at the elliptic point. We denote this number by n_e . For $\mathcal{N} = 2$ SQCD, we found examples with $n_e = 2$ and 4 for $\tau_e \sim \alpha$, while for $\tau_e \sim i$, $n_e = 1$ [2]. The contribution from an elliptic point is then,

$$\left[\mathcal{I}_f\right]_e = 2\pi i \frac{n_e}{k_e} \operatorname{nRes}_{\tau=\tau_e} \left[\widehat{h}(\tau,\bar{\tau})\right] = \frac{n_e}{k_e} d_e(-1,0), \qquad (5.96)$$

5. Branch points and cuts

Branch points and cuts are a new aspect compared to previous analyses (see for instance Fig. 8 and 15). We will demonstrate that their contribution vanishes for the integrands of interest.

We assume that the integrand f satisfies

$$\hat{h}(\tau,\bar{\tau}) = (\tau - \tau_{\rm bp})^n g(\tau,\bar{\tau}), \qquad (5.97)$$

with $n \in \mathbb{Z}/2$ and $n \geq -1/2$, $g(\tau, \bar{\tau})$ being a real analytic function near $\tau_{\rm bp}$. This assumption is satisfied for the twisted Yang–Mills theories [286]. To treat this type of singularity, we remove a δ neighbourhood and analyse the $\delta \to 0$ limit. Let C_{δ} be the contour

$$C_{\delta} = \{ \tau_{\rm bp} + \delta \, e^{i\theta} \, | \, \theta \in (0, 2\pi) \} \tag{5.98}$$

around $\tau_{\rm bp}$ with radius $\delta > 0$. Therefore, on the contour $|y^{-s}f|$ is bounded by

$$|\hat{h}| \le \delta^n \, K \tag{5.99}$$

for some K > 0. The integral around the branch point therefore vanishes in the limit,

$$\mathcal{I}_{f}^{\mathrm{bp}} = \lim_{\delta \to 0} \int_{C_{\delta}} \widehat{h} |d\tau| \leq \lim_{\delta \to 0} \int_{0}^{2\pi} \delta^{n} K \,\delta \,d\theta$$

$$= \lim_{\delta \to 0} 2\pi K \delta^{n+1} = 0.$$
 (5.100)

The branch points necessarily give rise to branch cuts. For the purpose of integration, we remove a neighbourhood with distance r from the cut, and take the limit $r \to 0$ after determining the integral. Since the value of the integrand is finite near the branch cut, the contribution to the integral vanishes.

Summary

Combining all the contributions discussed above, we find

$$\mathcal{I}_f = \sum_{j=1}^n w_j \, d_j(0) + \sum_s d_s(-1,0) + \sum_e \frac{n_e}{k_e} \, d_e(-1,0).$$
(5.101)

This formula generalises [73] for the pure $N_f = 0$ theory on a smooth fourmanifold X that admits a metric of positive scalar curvature, [84, Equation (5.10)] for the pure SU(2) theory on generic X, [85, Equation (4.88)] for the $\mathcal{N} = 2^*$ theory on X, and [77] for the massless $N_f = 2$ and $N_f = 3$ theories on $X = \mathbb{CP}^2$.

6 The *u*-plane integral for non-simply connected manifolds

In this final section, we study the *u*-plane integral on compact four-manifolds X with $b_1(X) > 0$, which are non-simply connected. This section is based on [4].

6.1 Introduction

Recently, interest in DW theory, and in particular the *u*-plane integral Z_u , was revived due to observations relating the latter for special four-manifolds to the theory of mock theta functions and harmonic Maass forms [77, 78]. For more generic, but simply connected, compact four-manifolds, it was later reformulated in terms of the modular completion of a mock modular form [81,83,84]. In this series of papers, the possibility of adding Q-exact operators to the action without affecting the correlation functions was studied in detail. In particular, a specific new Q-exact operator related to the 2-cycles of the background geometry was added to the action of the low-energy U(1) theory, which makes the connection to mock modular forms apparent and elegant [81, 84]. This technique circumvents the cumbersome method of lattice reduction, and allows to evaluate correlation functions efficiently. Previous results relating Z_u and mock modular forms are restricted to the case where the low-energy U(1) theory U(1) theory is formulated on simply connected four-manifolds only.

Taking inspiration from [76], we ask the natural question how these recent results carry over to the case when the four-manifold X has a non-trivial fundamental group and non-zero first Betti number $b_1(X)$. When the fourmanifold is non-simply connected, the theory is more complicated. This is due to the fact that the manifold now admits more structures, in the form of 1form fields and 1- and 3-cycles, which are not present in the simply connected case. These cycles give rise to further contact terms in the low-energy U(1) action [74, 76]. As a result, we will consider more general Q-exact operators related to these cycles.

We present a natural extension of the recent results [81,84] to the case of non-simply connected four-manifolds with $b_2^+ = 1$. Specifically, we introduce a number of new Q-exact operators in the low-energy effective U(1) theory that allow us to express the integrand of the *u*-plane integral elegantly as the non-holomorphic completion of a mock modular form. This further allows us to derive a closed-form expression for the *u*-plane integral for any such fourmanifold and for arbitrary period point J, as is evident from the result (6.57). This solution depends on $H_1(X)$, a fact that is easily seen in the case of product ruled surfaces where it manifests as a genus dependence, while when $H_1(X)$ is trivial, (6.57) reduces to Equation (4.10) of [81].

DW theory on product ruled surfaces $X = \mathbb{CP}^1 \times \Sigma_g$, with Σ_g a genus gRiemann surface, in the limit of vanishing volume for Σ_g has been argued to be equivalent to an $\mathcal{N} = (2, 2)$ 2d topological A-model on \mathbb{CP}^1 with target space the moduli space $\mathcal{M}_{\text{flat}}(\Sigma_g)$ of flat SU(2) connections on Σ_g [74,90,290– 294]. Reference [4] presents a concrete derivation of this equivalence, and in turn shows that due to the relation between DW theory and its low-energy U(1) effective theory as given by Eq. (1.39), a connection between Gromov-Witten (GW) theory (realised physically by the A-model) and mock modular forms (appearing in the low-energy effective action) exists, such that one can compute GW invariants using modular data originating from the 4d theory. The above equivalence is expected to hold in the limit small Σ_g , since the twisted $\mathcal{N} = 2$ gauge theory is topological and we are thus free to shrink Σ_g . As a consequence, flat SU(2) connections along the directions tangent to Σ_g are required to prevent the effective 2d action from blowing up when the limit of small Σ_g is taken.

6.2 The effective theory for $b_1 > 0$

DW theory is the topologically twisted formulation of the pure $\mathcal{N} = 2$ supersymmetric Yang–Mills theory with gauge group G of rank $r_G = 1$ on a smooth four-manifold X [61]. In the IR, the theory becomes a U(1) gauge theory that depends on the complexified effective gauge coupling $\tau = \frac{\theta}{\pi} + \frac{8\pi i}{g^2} \in \mathbb{H}$, where \mathbb{H} denotes the Poincaré half-plane. DW theory contains a scalar fermionic BRST operator $\mathcal{Q} := \epsilon^{\dot{A}\dot{B}}\overline{\mathcal{Q}}_{\dot{A}\dot{B}}$ that obeys $\mathcal{Q}^2 = 0.^{33}$ The field content of the theory is a collection of bosonic and fermionic degree 0, 1 and 2 operator valued differential forms on X, where the degree of the differential form is equal to the ghost number of the physical operator. In Table 2 we summarise the field content of the DW theory. The BRST transformations on these fields are

Bosons	Fermions	Form degree	
$a, ar{a}$	η	0	
A	ψ	1	
D	χ	2	

Table 2: Field content of DW theory. The a, \bar{a} fields originate from the vacuum expectation value of the scalar field of the UV theory. The D field is an auxiliary field.

given in (5.43), where the label J can be removed since we are studying a rank 1 theory. The physical observables of the theory belong to the Q-cohomology. We are interested in computing the path integral of the theory, the *u*-plane integral or Coulomb branch integral, when evaluated on a non-simply connected four-manifold. To this end, let us first introduce some notation.

Let $b_j \coloneqq b_j(X) = \dim H^j(X)$ be the Betti numbers of the smooth, closed and oriented four-manifold X with $b_2(X) = b_2^+(X) + b_2^-(X)$, where the first (second) summand corresponds to the number of positive (negative) eigenvalues of the quadratic form Q of X. For $a \in H^i(X)$ and $b \in H^{4-i}(X)$ we

³³Across the literature, this operator is often denoted as $\overline{\mathcal{Q}}$ instead.

define $B(a,b) = \int_X a \wedge b$. For $a \in H^2(X)$ the quadratic form Q of X corresponds to Q(a) := B(a, a). Furthermore, the signature of X is defined as $\sigma(X) = b_2^+(X) - b_2^-(X)$. Hereafter we consider four-manifolds with $b_2^+(X) = 1$. By Poincaré duality, we have that $b_0 = b_4$, $b_1 = b_3$. We can assume b_1 to be even, since the correlation function of the theory are known to vanish unless $1 - b_1 + b_2^+$ is even [73].

The Coulomb branch integral is the path integral of the low-energy U(1) theory with the insertion of the observables arising from the descent formalism as well as contact terms and Q-exact operators. It takes the form

$$Z_u(p,\gamma,S,Y) = \int [\mathcal{D}\Phi]\nu(\tau)e^{-\int_X \mathcal{L}' + I(S,Y) + I_{\mathcal{O}} + I_{\cap}}, \qquad (6.1)$$

where $\Phi = \{a, \bar{a}, A, \eta, \psi, \chi, D\}$ is the collection of fields of the theory (as in Table 2).

6.2.1 Ingredients of the *u*-plane integrand

In this section, we give explicit modular expressions for the ingredients of the *u*-plane integrand. The integrand transforms under the duality group $\Gamma^{0}(4)$, which is generated by T^{4} and $S^{-1}T^{-1}S$. Let us introduce the shorthand $f = (\phi_{1}, \phi_{2})^{(k,l)}$ if a function f is a non-holomorphic modular form of weight (k, l) for $\Gamma^{0}(4)$ with multipliers, i.e. transforms as

$$f(\tau + 4, \bar{\tau} + 4) = \phi_1 f(\tau, \bar{\tau}),$$

$$f\left(\frac{\tau}{\tau + 1}, \frac{\bar{\tau}}{\bar{\tau} + 1}\right) = \phi_2 (\tau + 1)^k (\bar{\tau} + 1)^l f(\tau, \bar{\tau}).$$
(6.2)

It is clear that

$$(\phi_1, \phi_2)^{(k_1, l_1)} (\varphi_1, \varphi_2)^{(k_2, l_2)} = (\phi_1 \varphi_1, \phi_2 \varphi_2)^{(k_1 + k_2, l_1 + l_2)},$$

$$\overline{(\phi_1, \phi_2)^{(k, l)}} = (\bar{\phi}_1, \bar{\phi}_2)^{(l, k)},$$

$$\frac{1}{(\phi_1, \phi_2)^{(k, l)}} = (\bar{\phi}_1, \bar{\phi}_2)^{(-k, -l)},$$

(6.3)

since $|\phi_i| = 1$. The functions³⁴

$$\frac{d^2u}{da^2} = 4\frac{2E_2 + \vartheta_2^4 + \vartheta_3^4}{3\vartheta_4^8}, \quad \frac{u}{\Lambda^2} = \frac{\vartheta_2^4 + \vartheta_3^4}{2\vartheta_2^2\vartheta_3^2}, \quad \frac{a}{\Lambda} = \frac{2E_2 + \vartheta_2^4 + \vartheta_3^4}{6\vartheta_2\vartheta_3}, \\
\frac{du}{d\tau} = \frac{\pi\Lambda^2}{4i}\frac{\vartheta_4^8}{\vartheta_2^2\vartheta_3^2}, \quad \frac{da}{du} = \frac{1}{2\Lambda}\vartheta_2\vartheta_3, \quad \frac{d\tau}{da} = \frac{8i}{\pi\Lambda}\frac{\vartheta_2\vartheta_3}{\vartheta_4^8}$$
(6.4)

transform as

$$u = (1,1)^{(0,0)}, \quad \frac{du}{d\tau} = (1,1)^{(2,0)}, \quad \frac{da}{du} = (-1,1)^{(1,0)},$$

$$\rho = (-1,1)^{(-1,0)}, \quad \frac{d\tau}{da} = (-1,1)^{(-3,0)}, \quad y = (1,1)^{(-1,-1)}$$
(6.5)

³⁴In this section, we choose the traditional convention where $\frac{u}{\Lambda^2} \to +\infty$ for $\tau \to i\infty$.

6.2.2 Effective Lagrangian

The low-energy U(1) effective Lagrangian \mathcal{L} of the twisted theory is given in [73, (2.15)]. The \mathcal{Q} -exact terms as well as the kinetic terms do not contribute since the zero modes are constant in DW theory on a four-manifold X with $b_2^+(X) = 1$. For such manifolds there is a useful fact stating that for any $\beta_1, \beta_2, \beta_3, \beta_4 \in H^1(X, \mathbb{Z})$, we have [295]

$$\beta_1 \wedge \beta_2 \wedge \beta_3 \wedge \beta_4 = 0. \tag{6.6}$$

We will make extensive use of this fact below.

Let us define \mathcal{L}' as the part of the zero-mode low-energy U(1) effective Lagrangian that contributes to the *u*-plane integral. It is given by [73]

$$\mathcal{L}' = \pi i \bar{\tau} k_+^2 + \pi i \tau k_-^2 - \frac{y}{8\pi} D \wedge *D + \frac{i\sqrt{2}}{16\pi} \frac{d\bar{\tau}}{d\bar{a}} \eta \chi \wedge (F_+ + D) - \frac{i\sqrt{2}}{2^7 \pi} \frac{d\tau}{da} \psi \wedge \psi \wedge (F_- + D),$$
(6.7)

where $F_{\pm} = 4\pi k_{\pm}$ and for any two-form x we abbreviate $B(x, x) = x^2$. In \mathcal{L}' , we disregard any summands of \mathcal{L} containing \mathcal{Q} -exact terms, exact differential forms and \wedge -products of four 1-forms. Here and throughout the rest of the section we use units where the dynamical scale Λ of the low-energy effective U(1) theory is equal to one. The gravitational contributions to \mathcal{L}' are described in the following section.

6.2.3 Measure factors

Assuming X is connected, the (holomorphic) measure factor [52, 73] is³⁵

$$\nu(\tau) \coloneqq -\left(2^{7/2}\pi\right)^{\frac{b_1}{2}} \frac{2^{\frac{3\sigma(X)}{4}+1}}{\pi} (u^2 - 1)^{\frac{\sigma(X)}{8}} \left(\frac{da}{du}\right)^{\frac{\sigma(X)}{2}+b_1-2}.$$
 (6.8)

Here we used $\chi(X) + \sigma(X) = 4 - 2b_1$ to eliminate the Euler character of X, $\chi(X)$. This expression reduces to Eq. (2.9) in [81] if we take $b_1 = 0$. For the simply connected theory one can use the microscopic definition of the theory to determine the effective gravitational couplings (e.g. by considering expansions of the Nekrasov partition function) [53, 111].

The zero modes of the one-forms ψ live in the tangent space of a b_1 dimensional torus $\mathbb{T}^{b_1} = H^1(X, \mathbb{R})/H^1(X, \mathbb{Z}) = H^1(X, \mathcal{O}_X^*)$ which corresponds to isomorphism classes of invertible sheaves (for X a smooth complex variety, that is, holomorphic line bundles) on X which are topologically trivial. We can expand ψ in zero-modes as $\psi = \sum_{i=1}^{b_1} c_i \beta_i$ with β_i an integral basis of harmonic

³⁵Note that this differs from the notation ν used in Section 5, in particular (5.64), by the factor $\frac{da}{d\tau}$, which we insert later, and we do not consider any external fluxes \mathbf{k}_j , since we study the $N_f = 0$ theory in this section.

one-forms, and c_i Grassmann variables. We then have the measure

$$\prod_{i=1}^{b_1} \frac{dc_i}{\sqrt{y}} = y^{-\frac{b_1}{2}} \prod_{i=1}^{b_1} dc_i.$$
(6.9)

The photon partition function will also include an integration over b_1 zero modes of the gauge field corresponding to flat connections [76]. These zero modes span the tangent space of \mathbb{T}^{b_1} . As a consequence of this, the photon partition function will have an overall factor of $y^{\frac{1}{2}(b_1-1)}$ [72]. Combining this with the measure factor (6.9) we see that in the end there will only be a factor of $y^{-1/2}$ surviving.

We can also consider the c_j in the expansion of ψ as a basis of one-forms $\beta_j^{\#} \in H^1(\mathbb{T}^{b_1}, \mathbb{Z})$, dual to β_j , such that

$$\psi = \sum_{j=1}^{b_1} \beta_j \otimes \beta_j^{\#}.$$
(6.10)

A useful fact about four-manifolds with $b_2^+ = 1$ is that the image of the map

 $\wedge: \quad H^1(X,\mathbb{Z}) \otimes H^1(X,\mathbb{Z}) \to H^2(X,\mathbb{Z}) \tag{6.11}$

is generated by a single rational cohomology class, which we denote as W [295].³⁶ This means that we can write $\beta_i \wedge \beta_j = a_{ij}W$, $i, j = 1, \ldots, b_1$, where a_{ij} is an anti-symmetric matrix. This further implies that the two-form on \mathbb{T}^{b_1} can be written as

$$\Omega = \sum_{i < j} a_{ij} \beta_i^{\#} \wedge \beta_j^{\#}, \qquad (6.12)$$

where $\beta_i^{\#} \in H^1(\mathbb{T}^{b_1}, \mathbb{Z})$, such that

$$\operatorname{vol}(\mathbb{T}^{b_1}) = \int_{\mathbb{T}^{b_1}} \frac{\Omega^{b_1/2}}{(b_1/2)!}.$$
 (6.13)

Using the analysis above we can now write $\psi \wedge \psi = 2(W \otimes \Omega)$ [76]. This is useful when one performs the integral over \mathbb{T}^{b_1} for the product ruled surfaces, for instance.

6.2.4 Observables

Q-invariant observables can be constructed using the celebrated descent formalism. By starting with the zero-form operator $\mathcal{O}^{(0)} = 2u$, we find all k-form valued observables $\mathcal{O}^{(k)}$ for k = 1, 2, 3, 4 that are Q-invariant modulo exact forms by solving the descent equations

$$\mathrm{d}\mathcal{Q}^{(j)} = \{\mathcal{Q}, \mathcal{O}^{(j+1)}\}$$
(6.14)

³⁶This class is denoted Σ in [76] and Λ in [90]. However, since we want to reserve Σ for the Riemann surfaces studied below and Λ for the dynamical scale of the theory we choose to call the class W.

inductively. This ensures that for a k-cycle $\Sigma^{(k)} \in H_k(X)$ in X, the integrals $\int_{\Sigma^{(k)}} \mathcal{O}^{(k)}$ are \mathcal{Q} -invariant and only depend on $\Sigma^{(k)}$. Fortunately, there is a canonical solution to the descent equations: Due to the fact that the translation generator is \mathcal{Q} -exact, there is the one-form valued *descent operator* K, which satisfies $d = \{\mathcal{Q}, K\}$ [73]. This implies that (6.14) can be solved by $\mathcal{O}^{(j)} = K^j \mathcal{O}^{(0)}$, where the iterated (anti)-commutators are implicit. The action of the operator K can be inferred from the BRST transformations (5.43) as [73]

$$[K, a] = \frac{1}{4\sqrt{2}}\psi, \quad [K, \bar{a}] = 0, \quad [K, \psi] = -2(F_{-} + D), \quad [K, A] = -2i\chi,$$
$$[K, \eta] = -\frac{i}{\sqrt{2}}d\bar{a}, \quad [K, \chi] = -\frac{3\sqrt{2}i}{4} * d\bar{a}, \quad [K, D] = \frac{3i}{4}\left(2d\chi - *d\eta\right).$$
(6.15)

Let us study the insertion of all possible observables. For ease of notation, let us denote $p = \Sigma^{(0)}$ a point class, $\gamma = \Sigma^{(1)}$ a 1-cycle, $S = \Sigma^{(2)}$ a 2-cycle and $Y = \Sigma^{(3)}$ a 3-cycle. The cycles γ , S and Y can be expanded in formal sums as

$$\gamma = \sum_{i=1}^{b_1} \zeta_i \gamma_i, \qquad S = \sum_{i=1}^{b_2} \lambda_i S_i \qquad Y = \sum_{i=1}^{b_3} \theta_i Y_i,$$
(6.16)

where γ_i , S_i and Y_i are a basis of one-, two- and three-cycles respectively, λ_i are complex numbers, while ζ_i and θ_i are Grassmann variables. By the common abuse of notation, we use the same notation for the 3-, 2-, and 1-forms Poincaré dual to the cycles, and use the convention

$$\int_{\gamma} \omega_1 = \int_X \omega_1 \wedge \gamma, \qquad \int_S \omega_2 = \int_X \omega_2 \wedge S, \qquad \int_Y \omega_3 = \int_X \omega_3 \wedge Y. \quad (6.17)$$

The most general Q-invariant observable we can add is then

$$I_{\mathcal{O}} = 2pu + a_1 \int_{\gamma} Ku + a_2 \int_{S} K^2 u + a_3 \int_{Y} K^3 u, \qquad (6.18)$$

where $a_2 = \frac{i}{\sqrt{2}\pi}$ is fixed from matching with the mathematical literature [73] and

$$Ku = \frac{1}{4\sqrt{2}} \frac{du}{da} \psi,$$

$$K^{2}u = \frac{1}{32} \frac{d^{2}u}{da^{2}} \psi \wedge \psi - \frac{\sqrt{2}}{4} \frac{du}{da} (F_{-} + D),$$

$$K^{3}u = \frac{1}{2^{7}\sqrt{2}} \frac{d^{3}u}{da^{3}} \psi \wedge \psi \wedge \psi - \frac{3}{16} \frac{d^{2}u}{da^{2}} \psi \wedge (F_{-} + D) - \frac{3\sqrt{2}i}{16} \frac{du}{da} (2d\chi - *d\eta).$$
(6.19)

6.2.5 Contact terms

The existence of the canonical solution to the descent equations allows to map an observable of the UV theory to the low-energy U(1) effective theory on the *u*-plane. For instance, the operator $I(S) = \int_S K^2 u$ of the UV theory is mapped to the same observable $\tilde{I}(S) = \int_S K^2 u$ in the IR. This is not quite true for products $I(S_1)I(S_2)\ldots I(S_n)$ of such operators for distinct Riemann surfaces $S_i \in H_2(X,\mathbb{Z})$. At the intersection of the surfaces, contact terms will appear [73,74]. When mapping a product of surface operators to the IR, the product is corrected by a sum over the intersection points. Due to the Q-invariance, the inserted operator is holomorphic and the point at which it is inserted is irrelevant.

Such contact terms appear for all cycles in X that can intersect. They have been classified and the corresponding contact terms have been found in [76, Equations (2.8)-(2.12)],

$$I_{\cap} = \int_{S \cap S} T + a_{13} \int_{Y \cap \gamma} T + a_{32} \int_{Y \cap S} KT + a_{33} \int_{Y \cap Y} K^2 T + a_{332} \int_{S \cap Y \cap Y} \frac{\partial^3 \mathcal{F}}{\partial \tau_0^3} + a_{333} \int_{Y \cap Y \cap Y} K \frac{\partial^3 \mathcal{F}}{\partial \tau_0^3} + a_{3333} \int_{Y \cap Y \cap Y \cap Y} \frac{\partial^4 \mathcal{F}}{\partial \tau_0^4}.$$
(6.20)

Here τ_0 is the deformation parameter of the prepotential, related to the dynamical scale by $\Lambda^4 = e^{\pi i \tau_0}$. The coefficient functions can all be expressed as quasi-modular functions on the *u*-plane. For instance, the contact term for $S \cap S$ is

$$T = \frac{u}{2} - \frac{a}{4}\frac{du}{da} = \frac{\vartheta_2^4 + \vartheta_3^4 - E_2}{6\vartheta_2^2\vartheta_3^2}.$$
 (6.21)

In terms of the prepotential \mathcal{F} , it is given by $T(\tau) = \frac{4}{\pi i} \frac{\partial^2 \mathcal{F}}{\partial \tau_0^2}$ [98]. We can use the action (6.15) to find

$$KT = \frac{1}{4\sqrt{2}} \frac{dT}{da} \psi,$$

$$K^{2}T = \frac{1}{32} \frac{d^{2}T}{da^{2}} \psi \wedge \psi - \frac{1}{2\sqrt{2}} \frac{dT}{da} (F_{-} + D).$$
(6.22)

The intersection constants can be obtained from duality invariance [76]. Due to the identity (6.6), the two last terms in (6.20) vanish and we can disregard them. Thus, from (6.20) and (6.22) we see that all terms in I_{\cap} except for one are only integrated over ψ and τ , which we we do in a later step. The remaining term

$$-\frac{\sqrt{2}a_{33}}{4}\frac{dT}{da}B(F_{-}+D,Y\wedge Y).$$
(6.23)

is to be integrated over D, χ and η .

6.2.6 *Q*-exact operators

As we will later see, the photon path integral combines with the insertion of the surface observable to a Siegel-Narain theta function $\Psi^J_{\mu}(\tau, z)$. See (6.52) for the definition. This function can be expressed as a total derivative to a non-holomorphic modular completion of an indefinite theta function, as has been previously shown in the simply connected case [81, 84]. To facilitate the calculation further the authors of those papers add the Q-exact operator I_S .³⁷ In this section we will generalise this operator insertion to simplify the calculations also in the case of non-simply connected manifolds. This then allows us to evaluate correlation functions efficiently using mock modular forms [81, 83, 84].

Since our computations should be valid for any $b_1 \ge 0$ and in particular $b_1 = 0$, this suggests that it is instructive to add the same Q-exact operator [81, (2.11) and (2.12)]

$$I_{S} = -\frac{1}{4\pi} \int_{S} \left\{ \mathcal{Q}, \frac{d\bar{u}}{d\bar{a}} \chi \right\}$$

$$= -\frac{\sqrt{2}i}{4\pi} \frac{d^{2}\bar{u}}{d\bar{a}^{2}} \int_{S} \eta \chi - \frac{i}{4\pi} \frac{d\bar{u}}{d\bar{a}} \int_{S} (F_{+} - D).$$
 (6.24)

The *u*-plane integrand (6.1) with I_S inserted can also in the case where $b_1 \neq 0$ be written as an anti-holomorphic derivative. However, it does not give the same kind of Siegel-Narain theta function as in the simply-connected case. The reason is that the elliptic argument z of Ψ^J_{μ} does not couple to $H^2_{-}(X)$ symmetrically to how its conjugate \bar{z} couples to $H^2_{+}(X)$. The insertion of I_S in the case $b_1 = 0$ can be viewed as the unique correction to the path integral that symmetrises the couplings to $H^2_{\pm}(X)$. Without such an insertion, the resulting theta functions are not symmetric, see for instance [73, Equation (3.18)].

As we demonstrate below, for $b_1 \neq 0$ this issue can be cured by introducing additional Q-exact operators. More precisely, the new observables and related contact terms require three new Q-exact terms. The first two

$$I_Y = -\frac{3i\bar{a}_3}{16} \int_Y \left[\mathcal{Q}, \frac{d^2\bar{u}}{d\bar{a}^2}\chi \wedge \psi \right] + \frac{\sqrt{2}}{2^7\pi} \int_X \left\{ \mathcal{Q}, \frac{d\bar{\tau}}{d\bar{a}}\chi \wedge \psi \wedge \psi \right\}$$
$$= \frac{3\sqrt{2}\bar{a}_3}{2^4} \frac{d^3\bar{u}}{d\bar{a}^3} B(\eta\chi, \psi \wedge Y) + \frac{3\bar{a}_3}{2^4} \frac{d^2\bar{u}}{d\bar{a}^2} B(F_+ - D, \psi \wedge Y)$$
$$+ \frac{i}{2^6\pi} \frac{d^2\bar{\tau}}{d\bar{a}^2} B(\eta\chi, \psi \wedge \psi) + \frac{\sqrt{2}i}{2^7\pi} \frac{d\bar{\tau}}{d\bar{a}} B(F_+ - D, \psi \wedge \psi)$$
(6.25)

compensate the observables (6.18). From the collection of contact terms (6.20), only the one from the intersection $Y \cap Y$ gives a term (6.23) that is integrated over D, η and χ . This term requires the addition of the Q-exact operator

$$I_{Y \cap Y} = -\frac{\sqrt{2}i\bar{a}_{33}}{4} \int_{Y \cap Y} \left\{ \mathcal{Q}, \frac{d\bar{T}}{d\bar{a}}\chi \right\}$$

= $\frac{\bar{a}_{33}}{2} \frac{d^2\bar{T}}{d\bar{a}^2} B(\eta\chi, Y \wedge Y) + \frac{\sqrt{2}\bar{a}_{33}}{4} \frac{d\bar{T}}{d\bar{a}} B(F_+ - D, Y \wedge Y).$ (6.26)

We can note that $\bar{a}_{33} = -a_{33}$ [76]. The sum of these additional Q-exact terms can be compactly written as

$$I_Y + I_{Y \cap Y} = -\sqrt{2}\eta B(\chi, \partial_{\bar{a}}(y\bar{\omega})) - yB(F_+ - D, \bar{\omega}), \qquad (6.27)$$

³⁷This term is called $\tilde{I}_{+}(\boldsymbol{x})$ in [81,83]. For ease of notation, we remove the tilde from such expressions since we only discuss operators in the IR.

where we introduced the 2-form

$$\omega \coloneqq \frac{\sqrt{2}i}{2^7 \pi y} \frac{d\tau}{da} \psi \wedge \psi - \frac{3a_3}{2^4 y} \frac{d^2 u}{da^2} \psi \wedge Y - \frac{\sqrt{2}a_{33}}{4y} \frac{dT}{da} Y \wedge Y.$$
(6.28)

This 2-form has the property that $y\omega$ is holomorphic and thus $y\bar{\omega}$ is antiholomorphic. The form of (6.25) is derived in Appendix 6.3, where we furthermore show that its one-point function evaluates to zero, such that it is safe to include it into the path integral, following the analysis in [83,84]. We furthermore find it useful to follow [81] and introduce the notation

$$\rho = \frac{S}{2\pi} \frac{du}{da}, \qquad b = \frac{\operatorname{Im}(\rho)}{y}.$$
(6.29)

Anticipating the result as a Siegel-Narain theta function, the elliptic variable will turn out to be $z = \rho + 2iy\omega$, which is a 2-form with holomorphic coefficients. In terms of this variable, the sum of all Q-exact insertions (6.24), (6.25) and (6.26) combine nicely as

$$I(S,Y) \coloneqq I_S + I_Y + I_{Y \cap Y}$$

= $-\frac{i}{2} \left(\sqrt{2}B(\eta\chi, \partial_{\bar{a}}\bar{z}) + B(F_+ - D, \bar{z}) \right).$ (6.30)

It is clear that this is purely anti-holomorphic. The operator I(S, Y) is then included into the path integral, as in (6.1).

6.3 *Q*-exact operators

In this section, we explain the construction of the Q-exact operator I(S, Y) in (6.30), which aids the evaluation of the *u*-plane integral using mock modular forms. A constructive approach is to classify all Q-exact operators in DW theory, add all of them to the path integral, evaluate the path integral and solve for all coefficient functions that lead to the desired properties. For two reasons, this is fortunately not necessary. First, it is convenient that most such operators do not even alter the *u*-plane integrand after integrating out the fermions and the auxiliary field. Second, the path integral can be performed without insertions of any additional operators, or with the insertion of just I_S as was done in the case that $b_1(X) = 0$ [81]. Such calculations lead to integrands that do not contain the symmetric Siegel-Narain theta function $\Psi^J_{\mu}(\tau, z)$ for any z, however only a few terms are missing with an educated guess of z. Only very specific Q-exact operators can provide the necessary terms for the new integrands to complete into $\Psi^J_{\mu}(\tau, z)$.

In section 6.3.1, we classify all possible Q-exact operators that contribute to the *u*-plane *integrand*. In section 6.3.2 we demonstrate how the correct Q-exact operators can be selected, for the simplified example where the intersection $Y \cap Y$ is empty (such that there is no intersection term for $Y \cap Y$).

6.3.1 Construction of Q-exact operators

Let us complete the result of [83] by computing the all Q-exact observables on a four-manifold X with $\pi_1(X) \neq 0$. Let us first collect

These are all $1 \dots 4$ -forms that can be constructed out of the field content in Table 2. Since any operator must be gauge invariant, we do not use the 1form A to construct operators but only F = dA. Furthermore, some operators are identically zero due to fermion saturation. The sets C_k are then generating sets for the spaces of k-form observables [83],

$$\mathcal{O}_{k} = \sum_{j=0}^{1} \sum_{X \in \mathcal{C}_{k}} f_{X,j}(a,\bar{a})\eta^{j}X.$$
(6.32)

Here, $f_{X,j}(a, \bar{a})$ are real-analytic functions without singularities away from strong and weak coupling. The most generic 0-form observable is $\mathcal{O}_0 = f_0(a, \bar{a}) + f_1(a, \bar{a})\eta$. Let us restrict now to the \mathcal{Q} -exact k-observables $[\mathcal{Q}, \mathcal{O}_k]^{\int}$ that survive integration. These do in particular either contain $\eta\chi$ or neither, since otherwise they would not survive the fermionic integration, and they do not contain any derivative term dX, as we consider $b_2^+(X) = 1$ and thus their zero modes vanish. By the notation $[\mathcal{Q}, \mathcal{O}]$ we furthermore mean either $\{\mathcal{Q}, \mathcal{O}\}$ or $[\mathcal{Q}, \mathcal{O}]$, depending on whether \mathcal{O} is Grassmann odd or even.

Recall the action (5.43) of the supersymmetry generator \mathcal{Q} . It follows that $[\mathcal{Q}, F_{\pm}] = (\mathrm{d}\psi)_{\pm}$. The action of \mathcal{Q} on functions $f(a, \bar{a})$ is given by

$$[\mathcal{Q}, f(a,\bar{a})] = \partial_{\bar{a}} f(a,\bar{a})[\mathcal{Q},\bar{a}] = \sqrt{2}i\partial_{\bar{a}} f(a,\bar{a})\eta.$$
(6.33)

The general \mathcal{Q} -exact observable $[\mathcal{Q}, \mathcal{O}_k]$ from (6.32) is very tedious to compute, luckily in $[\mathcal{Q}, \mathcal{O}_k]^{\int}$ generally not many terms survive. Furthermore, (6.32) has $2|\mathcal{C}_k|$ terms, however due to (5.43) we have $[\mathcal{Q}, \eta \mathcal{O}] = -\eta[\mathcal{Q}, \mathcal{O}]$ and for the terms with j = 1 it remains to multiply the j = 0 term by $-\eta$. Then only one of $[\mathcal{Q}, \mathcal{O}]$ and $-\eta[\mathcal{Q}, \mathcal{O}]$ is Grassmann even in the variables η and χ , such that only one of those can contribute to $[\mathcal{Q}, \mathcal{O}]^{\int}$. Lastly, if $\mathcal{O} = \prod_l \tilde{\mathcal{O}}_l$ is a composite operator, its \mathcal{Q} -action $[\mathcal{Q}, \mathcal{O}]$ is an alternating sum $\sim \sum_l [\mathcal{Q}, \tilde{\mathcal{O}}_l] \prod_{k \neq l} \tilde{\mathcal{O}}_k$, and so if all such summands do not contribute (which can be easily checked) then the whole commutator does not either. With this, it is now slightly less work to extract those summands \mathcal{O}_k^{\int} of (6.32) that contribute, $[\mathcal{Q}, \mathcal{O}_k]^{\int} = [\mathcal{Q}, \mathcal{O}_k^{\int}]$. They are

$$\mathcal{O}_{2}^{f} = f_{1}\chi,$$

$$\mathcal{O}_{3}^{f} = f_{2}\psi \wedge \chi,$$

$$\mathcal{O}_{4}^{f} = f_{3}\psi \wedge \psi \wedge \chi + f_{4}D \wedge \chi + f_{5}F_{+} \wedge \chi,$$
(6.34)

while $[\mathcal{Q}, \mathcal{O}_0]^{\int} = [\mathcal{Q}, \mathcal{O}_1]^{\int} = 0$. Their \mathcal{Q} -commutators give

$$\{\mathcal{Q}, \mathcal{O}_{2}^{J}\} = +\sqrt{2}i\partial_{\bar{a}}f_{1}\eta\chi + if_{1}(F_{+} - D),$$

$$[\mathcal{Q}, \mathcal{O}_{3}^{J}] = -\sqrt{2}i\partial_{\bar{a}}f_{2}\eta\chi \wedge \psi - if_{2}(F_{+} - D) \wedge \psi,$$

$$\{\mathcal{Q}, \mathcal{O}_{4}^{J}\} = +\sqrt{2}i\partial_{\bar{a}}f_{3}\eta\chi \wedge \psi \wedge \psi + if_{3}(F_{+} - D) \wedge \psi \wedge \psi$$

$$+\sqrt{2}i\partial_{\bar{a}}f_{4}\eta\chi \wedge D + if_{4}D \wedge (F_{+} - D)$$

$$+\sqrt{2}i\partial_{\bar{a}}f_{5}\eta\chi \wedge F_{+} + if_{5}F_{+} \wedge (F_{+} - D).$$

(6.35)

These are all Q-exact operators in DW theory. The following Q-exact terms can then be added to the action

$$I_2 = \int_S \{\mathcal{Q}, \mathcal{O}_2^f\}, \quad I_3 = \int_Y [\mathcal{Q}, \mathcal{O}_3^f], \quad I_4 = \int_X \{\mathcal{Q}, \mathcal{O}_4^f\}.$$
(6.36)

6.3.2 Solution for I_Y

By adding only I_S as suggested in [81, 83, 86], the *u*-plane integrand can be written as a total derivative, however it does not complete to a Siegel-Narain theta function. Let us construct the operator I_Y such that this becomes true. For simplicity, we ignore the contact terms I_{\cap} . This is possible since all contact terms other than the $Y \cap Y$ are integrated over ψ and τ only and therefore do not affect the path integral calculation. For simplicity and only in this section, we take the intersection $Y \cap Y$ to be empty.

We therefore aim to find the functions $f_1, \ldots f_5$. In the case $\pi_1(X) = 0$, the total integrand must go back to (6.45). If f_4 and f_5 are nonzero, this is not the case since they alter the integral. ³⁸ We therefore set $f_4 = f_5 = 0$. Thus, in the simply connected case, we have $I_2 = I_S$, which implies $f_1 = -\frac{1}{4\pi} \frac{d\bar{u}}{d\bar{a}}$. We shall therefore consider adding the correction

$$I_Y = -\sqrt{2}i\partial_{\bar{a}}f_2\eta B(\chi,\psi\wedge Y) - if_2B(F_+ - D,\psi\wedge Y) + \sqrt{2}i\partial_{\bar{a}}f_3\eta B(\chi,\psi\wedge\psi) + if_3B(F_+ - D,\psi\wedge\psi).$$
(6.37)

to the exponential in (6.45). The terms $\psi \wedge Y$ and $\psi \wedge \psi$ are precisely the terms that lead to the problems if only I_S is added. We can organise $h := f_3 \psi \wedge \psi - f_2 \psi \wedge Y$, such that

$$I_Y = \sqrt{2}i\eta B(\chi, \partial_{\bar{a}}h) + iB(F_+ - D, h).$$
(6.38)

³⁸This is certainly true if f_4 and f_5 can be varied. It is possible in principle that for specific functions f_4 and f_5 the $\pi_1(X) = 0$ integral does not change.

Inserting it into the path integral we find

$$D = \frac{\sqrt{2}i}{4y} \frac{d\bar{\tau}}{d\bar{a}} \eta \chi - 4\pi (b_+ + \omega_+) + \frac{4\pi i}{y} h_+.$$
(6.39)

After integrating out D, this produces new terms

$$4\pi i B(k_+ + b_+, h) + \frac{\sqrt{2}}{4y} \frac{d\bar{\tau}}{d\bar{a}} \eta B(\chi, h) + \sqrt{2} i \eta B(\chi, \partial_{\bar{a}} h)$$
(6.40)

to (6.48) (notice that $\omega \wedge h = h \wedge h = 0$). The first term is only integrated over ψ and τ , so it will not play a role immediately. The second and third term yield after the fermionic integration,

$$\frac{d\bar{\tau}}{d\bar{a}} \left(-\frac{\sqrt{2}}{4y} h - \sqrt{2}i\partial_{\bar{\tau}}h \right). \tag{6.41}$$

In view of (6.50) and the above discussion, we can aim this new contribution to give the missing factor

$$\sqrt{y}\frac{d\bar{\tau}}{d\bar{a}}\partial_{\bar{\tau}}\sqrt{2y}\,\bar{\omega} = \frac{d\bar{\tau}}{d\bar{a}}\left(\frac{\sqrt{2}i}{4}\bar{\omega} + \sqrt{2}y\partial_{\bar{\tau}}\bar{\omega}\right),\tag{6.42}$$

such that the Siegel-Narain theta function has an elliptic variable $z = \rho + 2iy\omega$ and $\beta = b + \omega + \bar{\omega}$.³⁹ Motivated by the computation

$$\partial_{\bar{\tau}}y = \frac{i}{2}, \quad \partial_{\bar{\tau}}\sqrt{2y} = \frac{\sqrt{2}i}{4\sqrt{y}}, \quad \partial_{\bar{\tau}}\frac{1}{y} = \frac{1}{2iy^2}, \quad \partial_{\bar{\tau}}b = \frac{b - \partial_{\bar{\tau}}\bar{\rho}}{2iy}, \quad \partial_{\bar{\tau}}\omega = \frac{1}{2iy}\omega, \tag{6.43}$$

we make the ansatz $h = icy\bar{\omega}$, with $c \in \mathbb{C}$ some number. From this it follows that $y\partial_{\bar{\tau}}\bar{\omega} = -\frac{i}{c}\partial_{\bar{\tau}}h - \frac{i}{2}\bar{\omega}$. Notice that h is purely anti-holomorphic, while $\bar{\omega}$ is not. We find that (6.41) equals (6.42) precisely for c = 1. From this, it is easy to find

$$f_2 = \frac{3ia_3}{16} \frac{d^2 \bar{u}}{d\bar{a}^2}, \qquad f_3 = \frac{\sqrt{2}}{2^7 \pi} \frac{d\bar{\tau}}{d\bar{a}}.$$
 (6.44)

In the simply connected case, the correction $I_2 = I_S$ is necessary in order for the surface observable $\tilde{I}_-(S)$ to combine into a Siegel-Narain function such that the *u*-plane integral is a total derivative. In the case $\pi_1(X) \neq 0$, an analogous procedure is required for the 3-cycle Y, which combines to a 2-form as $\psi \wedge Y$. In the $\pi_1(X) \neq 0$ Lagrangian (6.7) there is a new term $\psi \wedge \psi$ that is integrated over η , χ and D, such that the *u*-plane integral *is* a total derivative but does not contain a SN theta function. After the insertion of an anti-holomorphic Q-exact 4-form operator, the integrand indeed becomes a Siegel-Narain theta function.

³⁹Another possibility would be to chose h to be holomorphic and cancelling the ω inside the derivative. This is possible, however, the ω dependence does not drop from the SN theta function.

6.4 The *u*-plane integral for $b_1 > 0$

The *u*-plane integral (6.1) can be expressed as

$$Z_u(p,\gamma,S,Y) = \int [dad\bar{a}d\eta d\chi dD] \int_{\operatorname{Pic}(X)} d\psi \,\nu(\tau) \frac{1}{\sqrt{y}} e^{-\int_X \mathcal{L}' + I_{\mathcal{O}} + I_{\cap} + I(S,Y)},$$
(6.45)

where $\int_{\text{Pic}(X)}$ denotes a sum over isomorphism classes of line bundles, equivalent to a sum over $H^2(X, \mathbb{Z})$, followed by an integration over \mathbb{T}^{b_1} . The ψ zero modes are tangent to Pic(X), so the integral over these modes is understood as the integral of a differential form on Pic(X) [73]. At this point let us make a remark. The \mathcal{Q} -exact operator I(S, Y) is not strictly required in order to derive our end result (6.57). As a matter of fact, as shown in [84] this operator can be added freely as $\alpha I(S, Y)$, with α any number. In particular, we can have $\alpha = 0$. However, the case of $\alpha = 1$ makes the analysis simpler and more elegant, why we choose to include it.

Let us perform the integrals above in steps, using an economical notation. We integrate first over the auxiliary field D, and then over the fermionic 0and 2-forms, η and χ .

6.4.1 Integration over D, η and χ

Using (6.28) and (6.29), we can expand the terms in the exponential of (6.45) that are affected by the integrals over D, η and χ as (ignoring the remaining terms for now)

$$-\int_{X} (\mathcal{L}' + a_{2}K^{2}u + a_{3}K^{3}u) + I(S,Y) - \frac{\sqrt{2}a_{33}}{4}\frac{dT}{da}B(F_{-} + D,Y \wedge Y)$$

$$= -\pi i\bar{\tau}k_{+}^{2} - \pi i\tau k_{-}^{2} + \frac{y}{8\pi}D^{2} - \frac{\sqrt{2}i}{4}\frac{d\bar{\tau}}{d\bar{a}}B(\eta\chi,k_{+}) - \frac{\sqrt{2}i}{16\pi}\frac{d\bar{\tau}}{d\bar{a}}B(\eta\chi,D)$$

$$-\frac{i}{\sqrt{2}}B(\eta\chi,\frac{d\bar{\rho}}{d\bar{a}}) - 2\pi iB(k_{-},\rho) - 2\pi iB(k_{+},\bar{\rho}) + yB(D,b_{+}) + \frac{\sqrt{2}i}{2^{5}}B(\psi \wedge \psi,\frac{d\rho}{da})$$

$$-\sqrt{2}\eta B(\chi,\partial_{\bar{a}}(y\bar{\omega})) + 4\pi yB(k_{-},\omega_{-}) - 4\pi yB(k_{+},\bar{\omega}) + yB(D,\omega_{+}) + yB(D,\bar{\omega}_{+})$$
(6.46)

At any point we discard terms that vanish identically, such as 4-fermion terms or any instance of (6.6) such as $\psi \wedge \psi \wedge \psi$ or $\omega \wedge \omega$. The exponential (6.46) is Gaussian in D with saddle point

$$D = \frac{\sqrt{2}i}{4y} \frac{d\bar{\tau}}{d\bar{a}} \eta \chi - 4\pi (b_{+} + \omega_{+} + \bar{\omega}_{+}).$$
(6.47)

This can be found by differentiating (6.46) with respect to D and setting it to zero. Inserting D in (6.46) gives ⁴⁰

$$+\frac{\sqrt{2}i}{2^{5}}B(\psi \wedge \psi, \frac{d\rho}{da}) - 2\pi y(b_{+} + \omega_{+} + \bar{\omega}_{+})^{2} - \pi i\bar{\tau}k_{+}^{2} - \pi i\tau k_{-}^{2} - 2\pi iB(k_{-}, \rho) - 2\pi iB(k_{+}, \bar{\rho}) + 4\pi yB(k_{-}, \omega) - 4\pi yB(k_{+}, \bar{\omega}) - \frac{\sqrt{2}i}{4}\frac{d\bar{\tau}}{d\bar{a}}B(\eta\chi, k_{+} - b_{+} - \omega_{+} - \bar{\omega}_{+}) - \frac{i}{\sqrt{2}}B(\eta\chi, \frac{d\bar{\rho}_{+}}{d\bar{a}}) - \sqrt{2}\eta B(\chi, \partial_{\bar{a}}(y\bar{\omega})).$$
(6.48)

The third line are the only terms involving η and χ , which we will integrate over next. Before, we can combine those terms in the expression

$$-\frac{\sqrt{2}i}{4}\frac{d\bar{\tau}}{d\bar{a}}B\left(\eta\chi,k-b-\omega+\bar{\omega}-4iy\partial_{\bar{\tau}}\bar{\omega}+2\partial_{\bar{\tau}}\bar{\rho}\right).$$
(6.49)

Integrating over η and χ , we can rewrite this in a compact way as a total anti-holomorphic derivative times an overall factor that, as we discuss below, cancels with contributions from the rest of the measure,

$$\frac{\sqrt{2}i}{4}\frac{d\bar{\tau}}{d\bar{a}}B\left(k-b-\omega+\bar{\omega}-4iy\partial_{\bar{\tau}}\bar{\omega}+2\partial_{\bar{\tau}}\bar{\rho},\underline{J}\right) = \sqrt{y}\frac{d\bar{\tau}}{d\bar{a}}\partial_{\bar{\tau}}\sqrt{2y}B(k+b+\omega+\bar{\omega},\underline{J}),$$
(6.50)

where $\partial_{\bar{\tau}}$ acts on everything to its right and $\underline{J} = J/\sqrt{Q(J)} \in H^2_+(X)$ is the normalised self-dual harmonic form on X. This result follows directly from the the identities (6.43).

As previously discussed, the photon path integral together with the measure for the zero modes of ψ contains a sum over all fluxes times a factor of $1/\sqrt{y}$, and additionally contributes $(-1)^{B(k,K)}$, where K is the canonical class of X [72]. The $1/\sqrt{y}$ factor is thus absorbed by the \sqrt{y} on the rhs of (6.50).

Using the change of variables $u : \Gamma^0(4) \setminus \overline{\mathbb{H}} \xrightarrow{\sim} \mathbb{CP}^1$ provided by (6.4), we can further integrate over $d\tau \wedge d\bar{\tau}$ rather than over $da \wedge d\bar{a}$. This motivates the definition of the transformed measure

$$\tilde{\nu} = \nu \frac{da}{d\tau},\tag{6.51}$$

such that $da \wedge d\bar{a} \nu = d\tau \wedge d\bar{\tau} \frac{d\bar{a}}{d\bar{\tau}} \tilde{\nu}$. The factor $\frac{d\bar{a}}{d\bar{\tau}}$ cancels with the $\frac{d\bar{\tau}}{d\bar{a}}$ of (6.50).

6.4.2 Siegel-Narain theta function

Let us demonstrate that the *u*-plane integrand for $\pi_1(X) \neq 0$, as in the simplyconnected case [81], evaluates to a Siegel-Narain theta function. To this end,

⁴⁰If we integrate over D instead of inserting the equations of motion we get an additional factor of $2\pi i \sqrt{\frac{2}{y}}$ in front of the integral [83]. The result of the integration does not change otherwise since both methods agree for Gaußian integrals up to an overall factor of $\sqrt{\frac{\pi}{a}}i$, if we integrate e^{aD^2} over D. It is the same factor as in the simply-connected case because the quadratic D-term is the same. According to [73, p. 68] the D determinant should be ignored because it cancels in any case with the fermionic determinants.

let us define

$$\Psi^{J}_{\mu}(\tau,z) = e^{-2\pi y \beta_{+}^{2}} \sum_{k \in L+\mu} \partial_{\bar{\tau}} \left(\sqrt{2y} B(k+\beta,\underline{J}) \right) \times (-1)^{B(k,K)} q^{-k_{-}^{2}/2} \bar{q}^{k_{+}^{2}/2} e^{-2\pi i B(z,k_{-}) - 2\pi i B(\bar{z},k_{+})}$$
(6.52)

with $q = e^{2\pi i \tau}$ and $\beta = \frac{\text{Im}z}{y} \in L \otimes \mathbb{R}$, where $L = H^2(X, \mathbb{Z})$. For the elliptic variable $z = \rho + 2iy\omega$, we have $\beta = b + \omega + \bar{\omega}$ (here, we use that $y\omega$ is holomorphic). Both variables appear naturally in (6.48) and (6.50). In fact, we can combine everything to find

$$Z_u(p,\gamma,S,Y) = \int_{\Gamma^0(4)\backslash\mathbb{H}} d\tau \wedge d\bar{\tau} \int_{\mathbb{T}^{b_1}} [d\psi] \,\tilde{\nu} \,\Psi^J_\mu(\tau,\rho+2iy\omega) e^{I'_\mathcal{O}+I'_\mathcal{O}}.$$
(6.53)

Here,

$$I'_{\cap} = \int_{S \cap S} T + a_{13} \int_{Y \cap \gamma} T + a_{332} \int_{S \cap Y \cap Y} \frac{\partial^3 \mathcal{F}}{\partial \tau_0^3} + \frac{a_{32}}{4\sqrt{2}} \frac{dT}{da} \int_{Y \cap S} \psi \qquad (6.54)$$

and

$$I'_{\mathcal{O}} = 2pu + \frac{\sqrt{2}a_1}{8} \frac{du}{da} \int_{\gamma} \psi + \frac{\sqrt{2}i}{2^6 \pi} \frac{d^2 u}{da^2} \int_{S} \psi \wedge \psi, \qquad (6.55)$$

are the (holomorphic) remainders of the collections of $0, \ldots, 3$ -form observables and their contact terms that has not yet been integrated over, and we eliminated all terms that do not contribute.

Let us check that (6.53) is indeed true from the computations in section 6.4.1. Aside from the $\psi \wedge \psi$ term, the exponential of the first two lines in (6.48) immediately combine into the definition (6.52) with said parameters, $z = \rho + 2iy\omega$ and $\bar{z} = \bar{\rho} - 2iy\bar{\omega}$. Everything not exponentiated is given by the $\bar{\tau}$ derivative term in (6.50), which precisely gives the derivative term in (6.52). This proves (6.53).

The expression (6.53) generalises the result of the *u*-plane integral [84, (4.32)] to four-manifolds X with $b_1(X) > 0$ by giving a decomposition of the integrand into a holomorphic and metric-independent measure $\tilde{\nu} e^{I'_{\mathcal{O}}+I'_{\cap}}$ and a metric-dependent, non-holomorphic component $\Psi^J_\mu(\tau,z).$ Therefore, the evaluation techniques of [84] apply. Namely, we can express the integrand of the *u*-plane integral as an anti-holomorphic derivative,

$$\frac{d}{d\bar{\tau}}\widehat{\mathcal{H}}^{J}_{\mu}(\tau,\bar{\tau}) = \tilde{\nu}\,\Psi^{J}_{\mu}(\tau,z)e^{I'_{\mathcal{O}}+I'_{\cap}}.$$
(6.56)

The holomorphic exponential $e^{I'_{\mathcal{O}}+I'_{\cap}}$ does not affect the anti-holomorphic derivative, and thus the extension to $\pi_1(X) \neq 0$ is simply through the elliptic argument $z = \rho + 2iy\omega$.

Once $\widehat{\mathcal{H}}^{J}_{\mu}(\tau, \bar{\tau})$ is found, we can use coset representatives of $\mathrm{SL}(2,\mathbb{Z})/\Gamma^{0}(4)$ to map the six images of $\mathcal{F} = \mathrm{SL}(2,\mathbb{Z}) \setminus \mathbb{H}$ back to \mathcal{F} (see Fig. 2). The regularisation and renormalisation of such integrals originating from insertions of Q-exact operators has been rigorously established in [83]. This then allows to evaluate the partition function as

$$Z_u(p,\gamma,S,Y) = 4 \mathcal{I}_\mu(\tau) \big|_{q^0} + \mathcal{I}_\mu(-\frac{1}{\tau}) \big|_{q^0} + \mathcal{I}_\mu\left(\frac{2\tau-1}{\tau}\right) \big|_{q^0}, \qquad (6.57)$$

where by $|_{q^0}$ we denote the q^0 coefficient of the resulting Fourier expansion, and the τ -integrand of (6.53) is given by ⁴¹

$$\mathcal{I}_{\mu}(\tau) = \int_{\mathbb{T}^{b_1}} [d\psi] \widehat{\mathcal{H}}^J_{\mu}(\tau, \bar{\tau}).$$
(6.58)

The prefactors in (6.57) can be recognised as the widths of the cusps $i\infty$, 0 and 1 of the modular curve $\Gamma^{0}(4) \setminus \mathbb{H}$.

To derive a suitable anti-derivative $\widehat{\mathcal{H}}^{J}_{\mu}(\tau, \bar{\tau})$, it is auxiliary to choose a convenient period point J. The u-plane integral for a different choice J' is then related to the one for J by a wall-crossing formula, given explicitly in [76]. It is shown in [84] that for convenient choices of J, $\Psi^{J}_{\mu}(\tau, z)$ factors into holomorphic and anti-holomorphic terms, and the anti-derivative $\widehat{\mathcal{H}}^{J}_{\mu}$ can be found for both L even and odd. Furthermore, the u-plane integral can be evaluated using mock modular forms for point observables $p \in H_0(X)$ and Appell-Lerch sums for surface observables $z \in H_2(X)$ [84].

In [83] it is furthermore shown that in the above mentioned renormalisation, any Q-exact operator (such as I(S, Y)) decouples in DW theory. However, it is clear that the insertion of I(S, Y) crucially changes the *integrand*, making the Siegel-Narain theta function symmetric. Instead of inserting I(S, Y), we can contemplate adding $\alpha I(S, Y)$ for an arbitrary constant α . It was noticed in [84] that the Siegel-Narain theta function $\Psi^J_{\mu,\alpha}$ for $b_1 = 0$ with the insertion αI_S remains finite at weak coupling $(\text{Im}\tau \to \infty)$ if and only if $\alpha = 1$. This can be seen from the exponential prefactor in (6.52), whose exponent is negative definite if and only if \bar{z} (which we suppress in the notation) is the complex conjugate of z.

6.4.3 Single-valuedness of the integrand

An essential requirement, for the consistency of the theory, is that the path integral (6.53) is single-valued. For this it is advantageous to first change variables in the ψ -integral as

$$\psi' = \psi + \frac{12\pi i a_3}{\sqrt{2}} \frac{da}{d\tau} \frac{d^2 u}{da^2} Y.$$
 (6.59)

This is because the coefficient function of $\psi \wedge \psi$ in $y\omega$ is modular, while the $\psi \wedge Y$ and $Y \wedge Y$ coefficients of $y\omega$ are only quasi-modular. Such shifts (6.59)

⁴¹One could also contemplate switching the order of integration, and integrate over ψ first. This would however not necessarily result in a similar function to (6.57), and it might not be possible to use the results of [84].

leave the measure of $\int [d\psi]$ invariant, as $d\psi = d\psi'$. Due to the order of integration in (6.53), the change of variables (6.59) is well-defined. Since Y is also Grassmann-odd, ψ and Y \wedge -commute. This gives

$$\omega = \frac{\sqrt{2}i}{2^7 \pi y} \frac{d\tau}{da} \psi' \wedge \psi' + \frac{9\sqrt{2}\pi^2 a_3^2}{16y} \frac{du}{da} Y \wedge Y.$$
(6.60)

Let us use the notation of section 6.2.1. It is argued in [76] that ψ' transforms as $(-1,1)^{(1,0)}$. Using (6.5), one then finds that $y\omega = (-1,1)^{(-1,0)}$ transforms precisely as $\rho = (-1,1)^{(-1,0)}$, such that $z = \rho + 2iy\omega = (-1,1)^{(-1,0)}$ is a modular form and transforms exactly as in the $\pi_1(X) = 0$ case.

Furthermore, it is auxiliary to define [76, (2.14)]

$$S' = S + 4\pi i y \frac{da}{du} \omega. \tag{6.61}$$

It is well-defined, as $S' = (1,1)^{(0,0)}$ is fully invariant. In contrast to (6.59), this is not a change of variables or a redefinition, but rather a substitution to simplify some expressions. For instance, the elliptic variable now reads

$$z = \frac{S'}{2\pi} \frac{du}{da},\tag{6.62}$$

which takes the same form (6.29) as in the simply-connected case.

By incorporating the shift of $\psi \to \psi'$ together with (6.61) we find that the contact terms and observables in (6.54) and (6.55) can be written as

$$I_{\mathcal{O}+\cap} = 2pu + S'^{2}T + \frac{\sqrt{2}a_{1}}{8}\frac{du}{da}\int_{\gamma}\psi' - 3\pi^{2}a_{1}a_{3}u\int_{\gamma}Y + \frac{\sqrt{2}}{32}\frac{d\tau}{du}u\int_{S'}\psi'\wedge\psi' - \frac{3\pi i}{8}a_{3}\frac{du}{da}\int_{Y}S'\wedge\psi' + \frac{3\sqrt{2}}{4}i\pi^{3}a_{3}^{2}u\int_{S'}Y\wedge Y.$$
(6.63)

All terms but S'^2T are modular functions with trivial multipliers. Due to (6.62), the quasi-modular shift of T combines precisely with the one of $\Psi(\tau, z)$.

Measure factor

Since $\Delta \propto \frac{\vartheta_4^8}{\vartheta_2^4 \vartheta_3^4}$, $\frac{da}{d\tau} = \frac{\pi}{8i} \frac{\vartheta_4^8}{\vartheta_2 \vartheta_3}$ and $\frac{da}{du} = \frac{1}{2} \vartheta_2 \vartheta_3$, from (6.8) we have that $\nu \propto \frac{\vartheta_4^{\sigma}}{(\vartheta_2 \vartheta_3)^{2-b_1}}$ and therefore

$$\tilde{\nu} \propto \frac{\vartheta_4^{8+\sigma}}{(\vartheta_2\vartheta_3)^{3-b_1}}.$$
(6.64)

We find that under the generators of $\Gamma^0(4)$, $\tilde{\nu} = (-1, e^{-\pi i \sigma/4})^{(2-\frac{b_2}{2}+b_1,0)}$. For this we have used that $\sigma + b_2 = 2$ and that b_1 is even.

We also need to consider the fermion measure. As we have discussed earlier, this comes with an overall factor of $y^{-\frac{b_1}{2}}$ which gets absorbed by a similar factor coming from the photon partition function. This leaves us with $\prod_{i=1}^{b_1} dc_i$, which has weight $(-b_1, 0)$, since ψ has weight (1, 0) [72]. So after the integration over

 D, η and χ , and after changing integration variables from $da \wedge d\bar{a}$ to $d\tau \wedge d\bar{\tau}$ the measure of the integral will have weight $(-2 - b_1, -2)$, and we thus need the rest of the integrand to have weight $(2+b_1, 2)$. Finally, the transformations of the Siegel-Narain theta function $\Psi^J_{\mu}(\tau, z)$ can be found in Appendix A.7.

The integrand of the u-plane integral (6.53) reads

$$\mathcal{J}^{J}_{\mu} = d\tau \wedge d\bar{\tau} \int_{\mathbb{T}^{b_1}} [d\psi] \,\tilde{\nu} \, \Psi^{J}_{\mu}(\tau, z) e^{I'_{\mathcal{O}} + I'_{\cap}}.$$
(6.65)

Since it is integrated over the fundamental domain of $\Gamma^0(4)$, in order to check whether the integral is well-defined \mathcal{J}^J_{μ} must transform as a modular function for $\Gamma^0(4)$ with no phases. In Table 3 we collect the phases and weights of the individual factors as discussed above. This shows that the integral is indeed well-defined.

object	$d\tau \wedge d\bar{\tau}$	$\int_{\mathbb{T}^{b_1}} [d\psi]$	$\tilde{ u}$	$\Psi^J_\mu(au, z)$	$e^{I'_{\mathcal{O}}+I_{\cap}}$	\mathcal{J}_{μ}^{J}
weight	(-2, -2)	$(-b_1, 0)$	$(2 - \frac{b_2}{2} + b_1, 0)$	$(\frac{b_2}{2}, 2)$	(0, 0)	(0, 0)
T^4	1	1	-1	-1	1	1
$S^{-1}T^{-1}S$	1	1	$e^{-\frac{\pi i\sigma}{4}}$	$e^{\frac{\pi i\sigma}{4}}e^{-\frac{\pi iz^2}{\tau+1}}$	$e^{\frac{\pi i z^2}{\tau+1}}$	1

Table 3: Modular weights and phases of the *u*-plane integrand (6.65) under $\Gamma^{0}(4)$ transformations. This proves that $\mathcal{J}^{J}_{\mu}(\gamma \tau) = \mathcal{J}^{J}_{\mu}(\tau)$ for any $\gamma \in \Gamma^{0}(4)$.

7 Discussion and conclusion

In this thesis, we have studied in detail the modularity of a class of $\mathcal{N} = 2$ supersymmetric field theories. Being a manifestation of duality, modularity gives a precise description of duality and thus provides nontrivial insight into the dynamics. For the asymptotically free SU(2) theories with fundamental matter, we have found that the modularity is mildly obstructed by branch points, which give rise to the superconformal Argyres-Douglas points. In the superconformal $N_f = 4$ theory, for specific masses the order parameters are enhanced to bimodular forms, as they are modular in two parameters. For more general mass configurations however, branch points will also appear.

When the gauge group is enlarged to SU(3), we rather found that there exist elliptic loci which allow a modular parametrisation. With the inclusion of fundamental matter, such elliptic loci become more complicated, but in certain examples can be found explicitly.

These results vastly generalise earlier work by Seiberg and Witten [45, 46], Nahm [101], Malmendier and Ono [77], Klemm, Lerche, Theisen [96] and others on the modularity of the Coulomb branch for $\mathcal{N} = 2$ supersymmetric Yang-Mills theory and $\mathcal{N} = 2$ supersymmetric QCD. They open up several new directions for further research, as will be discussed in the following.

The fundamental domains obtained in this thesis are used as integration domains for topological correlation functions, when the theory is formulated on a four-manifold. We found that by coupling the theory to background fluxes, the constraint on the gauge bundle can be lifted. This allows to study a larger class of topological correlation functions. We investigated the definition, analysis and evaluation of such *u*-plane integrals for SU(2) $\mathcal{N} = 2$ supersymmetric QCD with $N_f \leq 4$ arbitrary masses on a compact four-manifold. The change of variables from the *u*-plane to fundamental domains for the effective coupling allows to express the integrands as generalisations of mock modular forms.

Another application of the fundamental domains is the study of BPS spectra. Many four-dimensional $\mathcal{N} = 2$ theories have the property that their spectrum of BPS states can be encoded in a quiver, the *BPS quiver* [296, 297]. Recently, it has been proposed that BPS quivers can be obtained from the fundamental domain of a particular 4d $\mathcal{N} = 2$ theory through a simple map [1,129,298]. This is due to the fact that fundamental domains encode the monodromy matrices, whose eigenvectors are the charge vectors of BPS states. It would be interesting to expand these ideas to non-modular configurations.

As mentioned in section 4.7, it would be interesting to extend our work on SU(3) to higher rank gauge groups, such as SU(N). While the elliptic loci are more subtle in this case, an approach similar to the one for the rank one theories could be worth exploring. In rank one, we match the \mathcal{J} -invariant of the elliptic curve with the modular *j*-invariant. When \mathcal{J} is a rational function of the order parameter, then this relation can be equivalently expressed as a polynomial equation (such as in Sections 2 and 3). For rank two, genus two curves are isomorphic if and only if their absolute invariants x_1 , x_2 and x_3 (as defined in (4.97)) coincide. The absolute invariants themselves are rational functions of Siegel modular forms of weight 0, which are meromorphic on the genus two Siegel upper-half plane $\mathbb{H}_2 \ni \Omega$. Thus from $x_i = x_i(\Omega)$, i = 1, 2, 3 we find three equations for the two variables u and v. For pure SU(3) for example, these equations are necessarily equivalent to the system (4.36). By multiplying with the denominators of $x_i(\Omega)$, we obtain three two-variable polynomial equations over the field of meromorphic Siegel modular forms of weight 0. Such a reformulation however likely runs into the same problem as the map to Rosenhain form, essentially since it does not circumvent the Schottky problem [299, 300]. It would be interesting to find a general solution to this problem.

It would furthermore be very interesting and challenging to extend our techniques and results to other $\mathcal{N} = 2$ supersymmetric theories. Many $\mathcal{N} = 2$ field theories in four dimensions can be obtained from geometric engineering [20, 21] in Type IIA string theory on local Calabi-Yau threefold singularities [22, 23]. They are also closely related to five-dimensional theories compactified on a circle, as a consequence of the Type-IIA/M-theory duality [301, 302]. Arguably the simplest such five-dimensional theories are the rank one theories with E_n flavour symmetry [129, 166, 298, 303]. Their circle compactification gives a class of four-dimensional $\mathcal{N} = 2$ supersymmetric theories, which are related by mass deformations, starting from E_8 to E_n with n < 8. The E_8 theory itself can be obtained from the *E*-string theory [304].

Another motivation to study $\mathcal{N} = 2$ theories in four dimensions is to explore indirectly the non-trivial six-dimensional $\mathcal{N} = (2,0)$ superconformal theories [305–307]. It can be realised as the worldvolume theory of a stack of N parallel M5-branes [308]. By a twisted compactification of the 6d superconformal theory on a Riemann surface, the so-called class \mathcal{S} of 4d $\mathcal{N} = 2$ supersymmetric theories are obtained [20,21,112,113]. The class \mathcal{S} theories include the rank 1 SCFTs [155,309,310], general non-Lagrangian theories [64,82,109] and the above mentioned circle compactifications of 5d SCFTs. This six-dimensional construction gives a geometric perspective on the S-duality of Seiberg-Witten theory. It furthermore gives rise to the AGT correspondence, which is a conjectured equality of many class \mathcal{S} observables with 2d CFT correlation functions [172, 311, 312].

It would be interesting to generalise our results on the topologically twisted theories and study the Coulomb branch integrals of other $\mathcal{N} = 2$ theories, such as those of class \mathcal{S} . While the topological twist of Donaldson-Witten theory can be implemented in any such theory, the formulation and evaluation of correlation functions as integrals over fundamental domains has been achieved to date only for a selection of $\mathcal{N} = 2$ theories and four-manifolds [73, 74, 77– 80, 84, 85]. It would further be fruitful to study the relations to 2d $\mathcal{N} = (0, 2)$ theories [63], vertex operator algebras [65], topological modular forms [62], and investigate the possibility to establish new four-manifold invariants [63, 85]. In view of the arguments given in section 1.1 on the complexity for the dynamics of supersymmetric quantum field theory with decreasing \mathcal{N} , it would be interesting to explore whether our results can be used to learn about the dynamics of theories with less supersymmetry, for instance $\mathcal{N} = 1$. Such supersymmetry breaking has already been studied in the 1990s [45,149,150,251, 313,314], with revived interest due to relations to integrable systems [121,315] as well as line defects and 't Hooft anomalies [276, 316].

A Automorphic forms and elliptic curves

In this Appendix, we collect some properties of modular and automorphic forms, as well as elliptic curves. For further reading see for example [59, 145, 154, 217, 224, 225, 287, 317–321].

A.1 Elliptic modular forms

We make use of modular forms for the congruence subgroups $\Gamma_0(n)$ and $\Gamma^0(n)$ of $SL(2,\mathbb{Z})$. They are defined as

$$\Gamma_0(n) = \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{SL}(2, \mathbb{Z}) \middle| c \equiv 0 \mod n \right\},$$

$$\Gamma^0(n) = \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{SL}(2, \mathbb{Z}) \middle| b \equiv 0 \mod n \right\},$$
(A.1)

and are related by conjugation with the matrix $\operatorname{diag}(n, 1)$. We furthermore define the *principal congruence subgroup* $\Gamma(n)$ as the subgroup of $\operatorname{SL}(2, \mathbb{Z}) \ni A$ with $A \equiv 1 \mod n$. A subgroup Γ of $\operatorname{SL}(2, \mathbb{Z})$ is called a congruence subgroup if there exists an integer $n \in \mathbb{N}$ such that it contains $\Gamma(n)$. The smallest such n is then called the *level* of Γ .

We furthermore make use of the theta group [322]

$$\Gamma_{\theta} \coloneqq \langle T^2, S \rangle \subseteq \mathrm{SL}(2, \mathbb{Z}). \tag{A.2}$$

It is a congruence subgroup of $SL(2,\mathbb{Z})$, as $[321,323]^{42}$

$$\Gamma_{\theta} = \{ A \in \mathrm{SL}(2, \mathbb{Z}) \, | \, A \equiv \mathbb{1} \text{ or } S \mod 2 \} \,. \tag{A.3}$$

Eisenstein series

We let $\tau \in \mathbb{H}$ and define $q = e^{2\pi i \tau}$. Then the Eisenstein series $E_k : \mathbb{H} \to \mathbb{C}$ for even $k \geq 2$ are defined as the q-series

$$E_k(\tau) = 1 - \frac{2k}{B_k} \sum_{n=1}^{\infty} \sigma_{k-1}(n) q^n,$$
 (A.4)

with $\sigma_k(n) = \sum_{d|n} d^k$ the divisor sum. For $k \ge 4$ even, E_k is a modular form of weight k for SL(2, \mathbb{Z}). On the other hand E_2 is a quasi-modular form, which means that the SL(2, \mathbb{Z}) transformation of E_2 includes a shift in addition to the weight,

$$E_2\left(\frac{a\tau+b}{c\tau+d}\right) = (c\tau+d)^2 E_2(\tau) - \frac{6i}{\pi}c(c\tau+d).$$
 (A.5)

From the S-transformation, we find that

$$E_4(e^{\pi i/3}) = 0, \qquad E_6(i) = 0,$$
 (A.6)

⁴²It can also be written as the group of matrices $\begin{pmatrix} a & b \\ c & d \end{pmatrix}$ with $a + b + c + d \equiv 0 \mod 2$, or $ab \equiv cd \equiv 0 \mod 2$.
and the zeros are unique in $SL(2,\mathbb{Z})\setminus\mathbb{H}$ according to the valence formula for modular forms on $SL(2,\mathbb{Z})$. Any modular form for $SL(2,\mathbb{Z})$ can be related to the Jacobi theta functions (1.14) by

$$E_4 = \frac{1}{2}(\vartheta_2^8 + \vartheta_3^8 + \vartheta_4^8), \qquad E_6 = \frac{1}{2}(\vartheta_2^4 + \vartheta_3^4)(\vartheta_3^4 + \vartheta_4^4)(\vartheta_4^4 - \vartheta_2^4).$$
(A.7)

All quasi-modular forms for $SL(2,\mathbb{Z})$ can be expressed as polynomials in E_2 , E_4 and E_6 . The derivatives of the Eisenstein series are quasi-modular,

$$E_2' = \frac{2\pi i}{12}(E_2^2 - E_4), \quad E_4' = \frac{2\pi i}{3}(E_2 E_4 - E_6), \quad E_6' = \frac{2\pi i}{2}(E_2 E_6 - E_4^2).$$
(A.8)

These equations give the differential ring structure of quasi-modular forms on $PSL(2,\mathbb{Z})$. With our normalisation (1.7) the *j*-invariant can be written as

$$j = 1728 \frac{E_4^3}{E_4^3 - E_6^2} = 256 \frac{(\vartheta_3^8 - \vartheta_3^4 \vartheta_4^4 + \vartheta_4^8)^3}{\vartheta_2^8 \vartheta_3^8 \vartheta_4^8}.$$
 (A.9)

Theta functions

The Jacobi theta functions $\vartheta_j : \mathbb{H} \to \mathbb{C}, j = 2, 3, 4$, are defined as

$$\vartheta_2(\tau) = \sum_{r \in \mathbb{Z} + \frac{1}{2}} q^{r^2/2}, \quad \vartheta_3(\tau) = \sum_{n \in \mathbb{Z}} q^{n^2/2}, \quad \vartheta_4(\tau) = \sum_{n \in \mathbb{Z}} (-1)^n q^{n^2/2}, \quad (A.10)$$

with $q = e^{2\pi i \tau}$. These functions transform under $T, S \in SL(2, \mathbb{Z})$ as

$$S: \quad \vartheta_2(-1/\tau) = \sqrt{-i\tau}\vartheta_4(\tau), \quad \vartheta_3(-1/\tau) = \sqrt{-i\tau}\vartheta_3(\tau), \quad \vartheta_4(-1/\tau) = \sqrt{-i\tau}\vartheta_2(\tau)$$
$$T: \quad \vartheta_2(\tau+1) = e^{\frac{\pi i}{4}}\vartheta_2(\tau), \qquad \vartheta_3(\tau+1) = \vartheta_4(\tau), \qquad \qquad \vartheta_4(\tau+1) = \vartheta_3(\tau).$$
(A.11)

They furthermore satisfy the Jacobi abstruse identity

$$\vartheta_2^4 + \vartheta_4^4 = \vartheta_3^4. \tag{A.12}$$

Derivatives of modular functions are described by Ramanujan's differential operator. It increases the holomorphic weight by 2 and it can be explicitly constructed using the theory of Hecke operators. For the derivatives of the Jacobi theta functions, one finds

$$D\vartheta_{2}^{4} = \frac{1}{6}\vartheta_{2}^{4} \left(E_{2} + \vartheta_{3}^{4} + \vartheta_{4}^{4}\right),$$

$$D\vartheta_{3}^{4} = \frac{1}{6}\vartheta_{3}^{4} \left(E_{2} + \vartheta_{2}^{4} - \vartheta_{4}^{4}\right),$$

$$D\vartheta_{4}^{4} = \frac{1}{6}\vartheta_{4}^{4} \left(E_{2} - \vartheta_{2}^{4} - \vartheta_{3}^{4}\right),$$

(A.13)

where $D := \frac{1}{2\pi i} \frac{d}{d\tau} = q \frac{d}{dq}$ and E_2 is the quasi-modular Eisenstein series (1.7) of weight 2, transforming as (A.5).

The modular lambda function is defined as $\lambda = \frac{\vartheta_2^4}{\vartheta_3^4}$, and is a Hauptmodul for $\Gamma(2)$.

Another class of theta series is provided by the one of the A_2 root lattice, $b_{3,j} : \mathbb{H} \to \mathbb{C}$,

$$b_{3,j}(\tau) = \sum_{k_1, k_2 \in \mathbb{Z} + \frac{j}{3}} q^{k_1^2 + k_2^2 + k_1 k_2}, \quad j \in \{-1, 0, 1\}.$$
 (A.14)

It is clear that $b_{3,-1} = b_{3,1}$. The transformation properties under $SL(2,\mathbb{Z})$ are

$$S: \quad b_{3,j}\left(-\frac{1}{\tau}\right) = -\frac{i\tau}{\sqrt{3}} \sum_{l \mod 3} \omega_3^{2jl} b_{3,l}(\tau),$$

$$T: \quad b_{3,j}(\tau+1) = \omega_3^{j^2} b_{3,j}(\tau),$$
(A.15)

with $\omega_3 = e^{2\pi i/3}$. The $b_{3,j}$ series can be expressed through the Dedekind eta function (A.19) as

$$b_{3,0}(\tau) = \frac{\eta(\frac{\tau}{3})^3 + 3\eta(3\tau)^3}{\eta(\tau)}, \qquad b_{3,1}(\tau) = 3\frac{\eta(3\tau)^3}{\eta(\tau)}.$$
 (A.16)

A relation to the Jacobi theta functions is given by

$$b_{3,0}(\tau) = \vartheta_3(2\tau)\vartheta_3(6\tau) + \vartheta_2(2\tau)\vartheta_2(6\tau).$$
(A.17)

Dedekind eta function

The Dedekind eta function $\eta : \mathbb{H} \to \mathbb{C}$ is defined as the infinite product

$$\eta(\tau) = q^{\frac{1}{24}} \prod_{n=1}^{\infty} (1-q^n), \quad q = e^{2\pi i \tau}.$$
 (A.18)

It transforms under the generators of $SL(2, \mathbb{Z})$ as

$$S: \quad \eta(-1/\tau) = \sqrt{-i\tau} \,\eta(\tau),$$

$$T: \quad \eta(\tau+1) = e^{\frac{\pi i}{12}} \,\eta(\tau),$$
(A.19)

and relates to the Jacobi theta series as $\eta^3 = \frac{1}{2}\vartheta_2\vartheta_3\vartheta_4$. The derivative of η is given by $\eta' = \frac{\pi i}{12}\eta E_2$.

Quotients of η -functions are frequently used to generate bases for the spaces of modular functions for congruence subgroups of $SL(2,\mathbb{Z})$. We use the following

Theorem 1 ([224, 225]). Let $f(\tau) = \prod_{\delta \mid N} \eta(\delta \tau)^{r_{\delta}}$ be an η -quotient with $k = \frac{1}{2} \sum_{\delta \mid N} r_{\delta} \in \mathbb{Z}$ and $\sum_{\delta \mid N} \delta r_{\delta} \equiv \sum_{\delta \mid N} \frac{N}{\delta} r_{\delta} \equiv 0 \mod 24$. Then, f is a weakly holomorphic modular form for $\Gamma_0(N)$ with weight k.

Atkin-Lehner involutions

The modular groups of n|h-type are defined in the following way [58]. Consider matrices of the form

$$\begin{pmatrix} ae & b/h \\ cn & de \end{pmatrix}$$
(A.20)

with determinant e, where $a, b, c, d, e, h, n \in \mathbb{Z}$, and h is the largest integer for which $h^2|N$ and h|24 with n = N/h. These matrices are also referred to as Atkin-Lehner involutions.

In the case that n is a positive integer and h|n, we define $\Gamma_0(n|h)$ as the set of above matrices with e = 1. For any positive integer e which satisfies e|n/hand (e, n/eh) = 1 (e is called an *exact divisor* of n/h), one can include also matrices of the above form with e > 1, forming a group denoted by $\Gamma_0(n|h) + e$. In fact, this construction works for any choice $\{e_1, e_2, \ldots\}$ of exact divisors of n/h, resulting in the group $\Gamma_0(n|h) + e_1, e_2, \ldots$ If h = 1, the |h| is omitted in the notation, and in case that all the possible e_i are included, the group is simply denoted by $\Gamma_0(n|h)+$.

In the Γ^0 convention the notation simplifies, since $\Gamma^0(n|h) = \Gamma^0(\frac{n}{h})$. This can be checked by conjugating (A.20) with diag(n, 1). The extension by non-unity determinant matrices follows by analogy.

Fundamental domains

A key concept of the theory of modular forms is the fundamental domain. A fundamental domain for a group $\Gamma \subset SL(2, \mathbb{R})$ is an open subset $\mathcal{F} \subset \mathbb{H}$ with the property that no two distinct points of \mathcal{F} are equivalent under the action of Γ and every point in \mathbb{H} is mapped to some point in the closure of \mathcal{F} by the action of an element in Γ . The quotient $\Gamma \setminus \mathbb{H}$ can be compactified by adding finitely many points called *cusps*. Cusps are Γ -equivalence classes of $\mathbb{Q} \cup \{i\infty\}$. Special points in the fundamental domain are the *elliptic fixed points*, which are points in \mathbb{H} that have a non-trivial Γ -stabiliser. There, the quotient $\Gamma \setminus \mathbb{H}$ becomes singular. Elliptic points can always be mapped to the boundary of the fundamental domain. They furthermore contribute non-trivially to the order of vanishing, which determines the dimension of the spaces of modular forms for fixed weight.

A.2 Siegel modular forms

Ordinary modular forms are constructed by the action of an $SL(2,\mathbb{Z})$ Möbius transformation on the upper half-plane \mathbb{H} . Siegel modular forms [287,319,324] generalise this notion by introducing an action of $Sp(2g,\mathbb{Z})$ on the so-called Siegel upper half-plane \mathbb{H}_g , which works for any genus $g \in \mathbb{N}$.

Define the Siegel modular group of genus g as

$$Sp(2g,\mathbb{Z}) = \{ M \in \operatorname{Mat}(2g;\mathbb{Z}) \mid M^{\mathsf{T}}JM = J \} \quad \text{with } J = \begin{pmatrix} 0 & \mathbb{I}_g \\ -\mathbb{I}_g & 0 \end{pmatrix}. \quad (A.21)$$

The group $\operatorname{Sp}(4, \mathbb{Z})$ can be generated [287] by the elements J and $T = \begin{pmatrix} \mathbb{I}_g & s \\ 0 & \mathbb{I}_g \end{pmatrix}$ with $s = s^{\mathsf{T}}$. The Siegel upper half-plane

$$\mathbb{H}_g = \{ \Omega \in \operatorname{Mat}(g; \mathbb{C}) \, | \, \Omega^{\mathsf{T}} = \Omega, \, \operatorname{Im}\Omega > 0 \}$$
(A.22)

consists of complex symmetric $g \times g$ matrices whose (componentwise) imaginary part is positive definite. This generalises the ordinary upper half-plane $\mathbb{H} = \mathbb{H}_1$. For example, for g = 2 this means that

$$\Omega = \begin{pmatrix} \tau_{11} & \tau_{12} \\ \tau_{12} & \tau_{22} \end{pmatrix}, \quad \mathrm{Im}\tau_{11} > 0, \quad \mathrm{Im}\tau_{11}\mathrm{Im}\tau_{22} - (\mathrm{Im}\tau_{12})^2 > 0.$$
(A.23)

An element $\gamma = \begin{pmatrix} A & B \\ C & D \end{pmatrix} \in Sp(2g, \mathbb{Z})$ acts on the Siegel upper half-plane by

$$\Omega \longmapsto \gamma(\Omega) = (A\Omega + B)(C\Omega + D)^{-1}.$$
 (A.24)

A (classical) Siegel modular form of weight k and genus g is then a holomorphic function $f : \mathbb{H}_g \to \mathbb{C}$ satisfying

$$f(\gamma(\Omega)) = \det(C\Omega + D)^k f(\Omega) \qquad \forall \gamma = \begin{pmatrix} A & B \\ C & D \end{pmatrix} \in Sp(2g, \mathbb{Z}),$$
 (A.25)

where for g = 1 holomorphicity at $i\infty$ is required in addition.

Theta series provide an explicit class of classical Siegel modular forms. For $a, b \in \mathbb{Q}^2$ and $\Omega \in \mathbb{H}_2$, define

$$\Theta\begin{bmatrix}a\\b\end{bmatrix}(\Omega) = \sum_{k\in\mathbb{Z}^2} \exp\left(\pi i(k+a)^{\mathsf{T}}\Omega(k+a) + 2\pi i(k+a)^{\mathsf{T}}b\right).$$
(A.26)

We are especially interested in the case where the entries of these column vectors take values in the set $\{0, \frac{1}{2}\}$. The corresponding theta functions are usually referred to as the theta characteristics. We call $\gamma = \begin{bmatrix} a \\ b \end{bmatrix}$ an even (odd) characteristic if $4a^{\mathsf{T}}b$ is even (odd). In the case of genus two there are ten even theta constants [318],

$$\Theta_{1} = \Theta \begin{bmatrix} 0 & 0 \\ 0 & 0 \end{bmatrix}, \quad \Theta_{2} = \Theta \begin{bmatrix} 0 & 0 \\ \frac{1}{2} & \frac{1}{2} \end{bmatrix}, \quad \Theta_{3} = \Theta \begin{bmatrix} 0 & 0 \\ \frac{1}{2} & 0 \end{bmatrix}, \quad \Theta_{4} = \Theta \begin{bmatrix} 0 & 0 \\ 0 & \frac{1}{2} \end{bmatrix}, \quad \Theta_{5} = \Theta \begin{bmatrix} \frac{1}{2} & 0 \\ 0 & 0 \end{bmatrix},$$

$$\Theta_{6} = \Theta \begin{bmatrix} \frac{1}{2} & 0 \\ 0 & \frac{1}{2} \end{bmatrix}, \quad \Theta_{7} = \Theta \begin{bmatrix} 0 & \frac{1}{2} \\ 0 & 0 \end{bmatrix}, \quad \Theta_{8} = \Theta \begin{bmatrix} \frac{1}{2} & \frac{1}{2} \\ 0 & 0 \end{bmatrix}, \quad \Theta_{9} = \Theta \begin{bmatrix} 0 & \frac{1}{2} \\ \frac{1}{2} & 0 \end{bmatrix}, \quad \Theta_{10} = \Theta \begin{bmatrix} \frac{1}{2} & \frac{1}{2} \\ \frac{1}{2} & \frac{1}{2} \end{bmatrix}.$$

$$(A.27)$$

All even theta constants can be related through algebraic identities to four fundamental ones, Θ_1 , Θ_2 , Θ_3 , Θ_4 [318].

The above theta functions are weight $\frac{1}{2}$ Siegel modular forms for a subgroup of Sp(4, Z). Their transformation properties under the Siegel modular group can be found in [319].

A.3 Fundamental domains

Here we present an algorithm to draw fundamental domains for discrete groups $G \subset SL(2, \mathbb{R})$ acting on \mathbb{H} by $\tau \mapsto \frac{a\tau+b}{c\tau+d}$, based on [58]. Let us assume that G contains T^s for some $s \in \mathbb{Q}$, and define $\mathcal{O}(T^s) = \{\tau \in \mathbb{H} \mid \operatorname{Re} \tau < \frac{s}{2}\}$. For all other elements X where $c \neq 0$, define

$$\mathcal{O}(X) = \mathcal{O}\left(\frac{a}{c}, \frac{\sqrt{\det X}}{c}\right)$$
 (A.28)

where $\mathcal{O}(c, r) = \{\tau \in \mathbb{H} \mid |\tau - c| > r\}$ is the exterior of a half-circle in \mathbb{H} around c with radius r > 0. A fundamental domain for G is then given by

$$\mathcal{R} = \bigcap_{X \in G \setminus \mathbb{I}} \mathcal{O}(X). \tag{A.29}$$

If a set of generators of G is known, (A.29) is an intersection of finitely many regions: for products of generators the circles becomes smaller. In practise it is enough to run (A.29) over the generators and their inverses. We denote by

$$\mathcal{C}_r(c) = \partial \mathcal{O}(c, r) = \{ \tau \in \mathbb{H} \, | \, |\tau - c| = r \}$$
(A.30)

the half-circles around c with radius r. For any $X \in G$ they are given analogously to (A.28). Then by (A.29), the boundary of \mathcal{R} which does not have constant real part (generated by $\partial \mathcal{O}(T^s)$) are piecewise arcs of circles $\mathcal{C}_r(c)$.

A simple Mathematica code to draw fundamental domains from a set of generators is the following:

```
g1 = MatrixPower[T, 4]; g2 = S.T.S; g3 = g1; g4 = g2;
1
    w = 4;
                (*width at infinity*)
2
3
    left = -1/2; (*left boundary*)
    p = 2; (*depth of algorithm*)
4
                 (*number of generators*)
    prod = 4:
5
    T = \{\{1,1\}, \{0,1\}\}; S = \{\{0,-1\}, \{1,0\}\};
6
   Flatten[Table[MatrixPower[g1, k1].MatrixPower[g2, k2].MatrixPower[g3, k3].
7
        MatrixPower[g4, k4], {k1,-p,p}, {k2,-p,p}, {k3,-p,p}, {k4,-p,p}], prod-1];
8
    list = DeleteCases[%, x_ /; x[[2, 1]] == 0];
9
    t = DeleteDuplicates[Flatten[{list, Inverse /@ list}, 1]];
10
   circ[x_?MatrixQ] := {x[[1, 1]]/x[[2, 1]], 1/Abs[x[[2, 1]]]};
11
   CR = Thread[circ[t]];
12
   lines = ContourPlot[Product[(x - (left+k)), {k, 0, w}] == 0,{x, left-0.01,
13
14
        left+w+0.01}, {y, 0, w}, ContourStyle -> {Orange}, ImageSize -> Large];
    Table[p1[k]=ContourPlot[(x-CR[[k,1]])<sup>2</sup>+y<sup>2</sup>==CR[[k,2]]<sup>2</sup>,{x,left,left+w},{y,0,w},
15
        Axes->True, ImageSize->Large, AspectRatio->Automatic], {k,1, Length[CR]}];
16
17
    Show[{Table[pl[k], {k, 1, Length[CR]}], lines}]
```

For standard subgroups, fundamental domains can be drawn using Verrill's Java applet [325] or the FareySymbol class in Sage [326].

Modular curves

A subgroup Γ of $SL(2,\mathbb{Z})$ is a congruence subgroup if $\Gamma \supset \Gamma(N)$ for some $N \in \mathbb{N}$, which is called the *level* of Γ . The (projective) *index* of a congruence subgroup Γ is defined as

$$\operatorname{ind} \Gamma = [\operatorname{PSL}(2,\mathbb{Z}):\Gamma], \tag{A.31}$$

and it is finite for all N. By $SL(2,\mathbb{Z})$ we strictly mean $PSL(2,\mathbb{Z})$ in the following. In fact, one can prove [145]

$$\operatorname{ind} \Gamma(N) = N^3 \prod_{p|N} \left(1 - \frac{1}{p^2} \right), \qquad \operatorname{ind} \Gamma^0(N) = N \prod_{p|N} \left(1 + \frac{1}{p} \right), \qquad (A.32)$$

where the sum is over all prime divisors of N. It can also be computed in the following way. The volume of the curve $\Gamma \setminus \mathbb{H}$ is defined as

$$\operatorname{vol}(\Gamma \backslash \mathbb{H}) = \int_{\Gamma \backslash \mathbb{H}} \mathrm{d}\mu, \qquad (A.33)$$

where $d\mu = y^{-2} dx dy$ is the hyperbolic metric on \mathbb{H} , with $\tau = x + \text{Im}y$. Since $\operatorname{vol}(\operatorname{SL}(2,\mathbb{Z})\backslash\mathbb{H}) = \frac{\pi}{3}$ can easily be computed, the index of any $\Gamma \subseteq \operatorname{SL}(2,\mathbb{Z})$ is then given by

ind
$$\Gamma = \frac{3}{\pi} \operatorname{vol}(\Gamma \setminus \mathbb{H}).$$
 (A.34)

Let Γ be a congruence subgroup of $\mathrm{SL}(2,\mathbb{Z})$. Cusps of Γ are Γ -equivalence classes of $\mathbb{Q} \cup \{\infty\}$. Adjoining coordinate charts to the cusps and compactifying gives the modular curve $X(\Gamma) \coloneqq \Gamma \setminus (\mathbb{H} \cup \mathbb{Q} \cup \{i\infty\})$. The isotropy (stabiliser) group of ∞ in $\mathrm{SL}(2,\mathbb{Z})$ is the abelian group of translations,

$$\operatorname{SL}(2,\mathbb{Z})_{\infty} = \left\{ \begin{pmatrix} 1 & m \\ 0 & 1 \end{pmatrix} : m \in \mathbb{Z} \right\}.$$
(A.35)

For each cusp $s \in \mathbb{Q} \cup \{i\infty\}$ some $\delta_s \in \mathrm{SL}(2,\mathbb{Z})$ maps $s \mapsto \infty$. The width of s is defined as

$$h_{\Gamma}(s) = \left| \mathrm{SL}(2, \mathbb{Z})_{\infty} / (\delta_s \Gamma \delta_s^{-1})_{\infty} \right|.$$
 (A.36)

It can be proven that this definition is independent of δ_s . For a fixed group Γ it can be viewed as a well-defined function $\mathbb{Q} \cup \{i\infty\} \to \mathbb{N}_0$. It is straightforward to show that the sum over the widths of all inequivalent cusps \mathcal{C} is equal to the index [327]

$$\sum_{s \in \mathcal{C}_{\Gamma}} h_{\Gamma}(s) = \operatorname{ind} \Gamma.$$
(A.37)

The width of 0 (which is the level) is the lcm of all widths, and the width of ∞ is the gcd of all widths.

Other invariants of modular curves are the elliptic fixed points. A point $\tau \in \mathbb{H}$ is an *elliptic point* for Γ if its isotropy (stabiliser) group is nontrivial. The *period* of τ is defined as the order of the isotropy group. It can be shown that any congruence subgroup of $SL(2, \mathbb{Z})$ has only finitely many elliptic points, and the period for any point $\tau \in \mathbb{H}$ is 1, 2 or 3.

Riemann-Hurwitz formula

Let $f: X \to Y$ be a nonconstant holomorphic map between compact Riemann surfaces X, Y. It has a degree $n \in \mathbb{N}$, such that $|f^{-1}(y)| = n$ for all but finitely many $y \in Y$. More precisely, for each point $x \in X$ let $e_x \in \mathbb{N}$ be the ramification degree of f at x, i.e. the multiplicity with which f takes 0 to 0 as a map in local coordinates, making f an e_x -to-1 map around x. Then there exists a positive integer n such that

$$\sum_{x \in f^{-1}(y)} e_x = n \tag{A.38}$$

for all $y \in Y$. If g_X and g_Y are the genera of X and Y, the Riemann-Hurwitz formula

$$2g_X - 2 = n(2g_Y - 2) + \sum_{x \in X} (e_x - 1)$$
(A.39)

states that the Euler characteristic of X is that of Y multiplied by the degree n of the cover, corrected by contributions from the ramification points. It is obvious that $g_X \ge g_Y$, otherwise f is not holomorphic.

This allows to compute the genus of a modular curve $X(\Gamma)$ for any congruence subgroup $\Gamma \subseteq \mathrm{SL}(2,\mathbb{Z})$. For this, let $X = X(\Gamma)$ and Y = X(1). Let $y_2 = \mathrm{SL}(2,\mathbb{Z}) \cdot i$, $y_3 = \mathrm{SL}(2,\mathbb{Z}) \cdot e^{\frac{\pi i}{3}}$ and ε_2 and ε_3 be the number of elliptic fixed points of period 2 and 3 for $X(\Gamma)$, and finally $y_{\infty} = \mathrm{SL}(2,\mathbb{Z}) \cdot \infty$ and ε_{∞} be the number of cusps $X(\Gamma)$. Then $e_x - 1$ is only nonzero when $x \in f^{-1}(y_h)$ for $h = 2, 3, \infty$. Since $g_{X(1)} = 0$, it follows from (A.39) that

$$g = 1 + \frac{n}{12} - \frac{\varepsilon_2}{4} - \frac{\varepsilon_3}{3} - \frac{\varepsilon_\infty}{2}, \qquad (A.40)$$

where $g = g_{X(\Gamma)}$ and $n = \text{ind }\Gamma$. See, for example, [145] for a more detailed derivation of this formula.

Conjugacy classes of subgroups of $SL(2,\mathbb{Z})$ can be classified by the data $(n, \varepsilon_{\infty}, \varepsilon_2, \varepsilon_3)$ together with the set of widths of the ε_{∞} cusps. For instance, the subgroups of $SL(2,\mathbb{Z})$ of index n = 6 have been completely classified [154,328]. There are precisely 22 subgroups that fall into 8 conjugacy classes, and they are listed in Table 4. It has been shown in [329] that every subgroup of $SL(2,\mathbb{Z})$ with index $n \leq 6$ is a congruence subgroup. Examples of *noncongruence* subgroups of $SL(2,\mathbb{Z})$ of index 7 have been constructed already in the 19th century by Fricke. Hauptmoduln for low index subgroups of $SL(2,\mathbb{Z})$ are studied in more detail in [115, 330].

A.4 McKay-Thompson series

For discrete genus 0 subgroups of $PSL(2, \mathbb{R})$, Hauptmoduln corresponding to 174 Atkin-Lehner groups are known [331]. Conway and Norton [36] conjectured a natural \mathbb{Z} -graded representation V_n of the Fischer-Griess monster group \mathbb{M} . For every element $g \in \mathbb{M}$, we can collect the characters in a q-series

$$T_g(\tau) = \sum_{n=-1}^{\infty} \chi_{V_n}(g) q^n, \qquad (A.41)$$

which are called the *McKay-Thompson* series (see also [240, 332-336]). For the identity element $g = 1 \in \mathbb{M}$, the character counts the dimension $\chi_{V_n}(1) = \dim V_n$,

$$T_1(\tau) = \sum_{n=-1}^{\infty} \dim V_n q^n = J(\tau), \qquad (A.42)$$

which famously gives the Klein *j*-function J = j - 744. Since the character of a representation is a class function, there are as many distinct McKay-Thompson

$(\varepsilon_{\infty}, \varepsilon_2, \varepsilon_3)$	cusps	Hauptmodul	conj
(1, 0, 0)	6	g = 1	1
(1, 0, 3)	6	$j = -27x^3(x^3 + 16)$	2
(1, 4, 0)	6	$j = 27(x^2 + 4)^3$	3
(2, 2, 0)	3 + 3	$j = \frac{x^3(x+12)^3}{(x+9)^3}$	3
(2, 2, 0)	4 + 2	$j = \frac{x^{3}(x+8)^{3}}{(x+4)^{2}}$	3
(2, 2, 0)	5 + 1	$j = \frac{(x^2 + 10x + 5)^3}{x}$	6
(3, 0, 0)	2 + 2 + 2	$j = \frac{(x^2 + 192)^3}{(x^2 - 64)^2}$	1
(3, 0, 0)	4 + 1 + 1	$j = \frac{(x^2 + 48)^3}{x^2 + 64}$	3

Table 4: Classification of conjugacy classes of index 6 subgroups of $SL(2,\mathbb{Z})$ [154]. The first column gives the number of cusps and elliptic fixed points of order 2 and 3. According to the Riemann-Hurwitz formula (A.40) all but the (1,0,0) one induce genus 0 modular curves, whose Hauptmodul x is expressed in terms of the j-invariant. In fact the group with signature (1,0,0) is the only subgroup of $SL(2,\mathbb{Z})$ of nonzero genus and index ≤ 7 . The second column gives the indices of the ε_{∞} number of cusps, and the last column counts the number of conjugate groups with the same data. From the expression of $j = \frac{p(x)}{q(x)}$ as a rational function in x we learn that deg p = n gives the index and the difference deg $p - \deg q = h(\infty)$ gives the width of ∞ . The roots of q(x) = 0 then give the cusps in \mathbb{Q} with widths provided by their corresponding multiplicity.

series as there are conjugacy classes of the Monster, namely 194. If g is not the identity element, the series (A.41) are not modular for $PSL(2,\mathbb{Z})$ but rather for a discrete subgroup Γ_g of $PSL(2,\mathbb{R})$. In fact, the Γ_g are all genus 0 groups and T_g is a Hauptmodul on $\Gamma_g \setminus \mathbb{H}$ [280]. It turns out that all but 3 of the 174 Atkin-Lehner groups of genus zero correspond to McKay-Thompson series. The series (A.41) are all examples of *replicable* functions [56–58, 337–339]

Let f be a q-series

$$f(q) = \frac{1}{q} + \sum_{n=1}^{\infty} a_n q^n$$
 (A.43)

with vanishing constant term. For every $n \in \mathbb{N}_0$ there exists a unique monic polynomial F_n such that

$$F_n(f(q)) = \frac{1}{q^n} + \mathcal{O}(q) \tag{A.44}$$

as $q \to 0$. The F_n are called *Faber polynomials*. For any f as in (A.43), they can be defined as $F_n(z) = \det(z\mathbb{1} - A_n)$, where

$$A_{n} = \begin{pmatrix} a_{0} & 1 & & \\ 2a_{1} & a_{0} & 1 & & \\ \vdots & \vdots & \vdots & \ddots & \\ (n-2)a_{n-3} & a_{n-4} & a_{n-5} & \cdots & 1 & \\ (n-1)a_{n-2} & a_{n-3} & a_{n-4} & \cdots & a_{0} & 1 \\ na_{n-1} & a_{n-2} & a_{n-3} & \cdots & a_{1} & a_{0} \end{pmatrix}.$$
 (A.45)

This allows to write

$$F_n(f(q)) = \frac{1}{q^n} + n \sum_{m=1}^{\infty} h_{m,n} q^m.$$
 (A.46)

The q-series f is then said to be replicable if $h_{m,n} = h_{r,s}$ whenever mn = rsand gcd(m,n) = gcd(r,s). This gives a relatively sparse list of equivalences between the entries of the infinite matrix $(h_{m,n})_{m,n\in\mathbb{N}}$. In practise one can truncate the matrix and compare $h_{m,n}$ and $h_{r,s}$ for pairs with the given constraints. They can be worked out before. For instance, if all series $F_n(f(q))$ are computed up to q^{14} , there are 8 constraints

A.5 Algebraic modular functions

In this Appendix, we discuss aspects of roots of polynomials over the field of modular functions for $PSL(2,\mathbb{Z})$, which describe order parameters found throughout this thesis. In the following, by Γ we mean $PSL(2,\mathbb{Z})$, unless specified otherwise. We begin with a

Lemma 1 (Proposition 18 of [59]). Let $f \in M_0(\Gamma')$ be a modular form of weight 0 for a congruence subgroup $\Gamma' \subset \Gamma$. Then f is constant.

Proof. Let $a = f(\tau_0)$ for some fixed $\tau_0 \in \mathbb{H}$. Let $\Gamma = \bigcup_j \alpha_j \Gamma'$, where the disjoint union is over $j = 1, \ldots, \text{ind } \Gamma' = [\Gamma : \Gamma']$. Consider the function

$$g: \mathbb{H} \to \mathbb{C}, \quad \tau \longmapsto \prod_{j=1}^{\mathrm{ind}\,\Gamma'} (f(\alpha_j^{-1}\tau) - a),$$
 (A.48)

where $\alpha_j^{-1}\tau$ is the usual action of Γ on \mathbb{H} . One finds that $g(\gamma^{-1}\tau) = \prod_j (f((\gamma\alpha_j)^{-1}\tau) - a)$ for $\gamma \in \Gamma$. This however only permutes the cosets, such that the terms in the product (A.48) are merely rearranged. Therefore, $g(\gamma\tau) = g(\tau)$ and g is a holomorphic modular form for Γ and thus constant. As $g(\tau_0) = 0$, we have g = 0 identically. This means that one of the factors $f(\alpha_j^{-1}\tau) - a$ in (A.48) is zero, such that $f(\tau) = a$ for all $\tau \in \mathbb{H}$.

Lemma 2 (Exercise III §3 7(a) of [59]). Let f be a modular function for a congruence subgroup $\Gamma' \subset \Gamma$. Then f satisfies a polynomial of degree $ind\Gamma' = [\Gamma : \Gamma']$ over the field $\mathbb{C}(\Gamma)$ of modular functions on Γ .

Proof. We can recycle most parts of the proof of Lemma 1. Let $X \in \mathbb{C}$ be a constant, and let f be a modular function for $\Gamma' \subset \Gamma$, which can be written as $\Gamma = \bigcup_{i} \alpha_{i} \Gamma'$. The function

$$\tilde{g} : \mathbb{H} \to \mathbb{C}, \quad \tau \longmapsto \prod_{j=1}^{\mathrm{ind}\,\Gamma'} (f(\alpha_j^{-1}\tau) - X)$$
(A.49)

is meromorphic on \mathbb{H} and admits a Fourier series, since f does. For the same reason as before, one finds that $\tilde{g}(\gamma \tau) = \tilde{g}(\tau)$ for all $\gamma \in \Gamma$ and $\tau \in \mathbb{H}$. Therefore, \tilde{g} is a modular function for Γ . The coefficients of the polynomial \tilde{g} in X are the elementary symmetric polynomials in $\{f \circ \alpha_1^{-1}, \ldots, f \circ \alpha_{\operatorname{ind} \Gamma'}^{-1}\}$. Since X is arbitrary, the coefficients are also modular functions for Γ . Therefore,

$$P(X) = \prod_{j=1}^{\operatorname{ind} \Gamma'} (f \circ \alpha_j^{-1} - X) \in \mathbb{C}(\Gamma)[X]$$
(A.50)

is a polynomial of degree ind Γ' in X over $\mathbb{C}(\Gamma)$ with f as a solution, as demanded.

It is clear from the above proof that some properties of the coefficients of the modular polynomial are inherited from those of f. For instance, instead of $\Gamma = \text{PSL}(2, \mathbb{Z})$ we can take $\Gamma = \text{PSL}(2, \mathbb{R})$ and Γ' to be an arithmetic (i.e. finite index) Fuchsian group. Another possibility is to study modular polynomials for weakly holomorphic functions f on $\text{PSL}(2, \mathbb{Z})$, for which the coefficients of P are polynomials in j. This motivates the following

Definition 7 (Algebraic modular form). Let Γ' be an arithmetic Fuchsian group, i.e. a discrete finite index subgroup of $PSL(2, \mathbb{R})$. An algebraic modular form of degree d for Γ' is a root of a polynomial of degree d over the field of automorphic functions on Γ' . The degree is taken to be minimal in the obvious sense.

According to Lemma 2, every congruence subgroup Γ' of $PSL(2,\mathbb{Z})$ is an algebraic modular form for $PSL(2,\mathbb{Z})$ of degree ind Γ' . This is because congruence subgroups of $PSL(2,\mathbb{Z})$ have finite index. In some cases, algebraic modular forms form Galois extensions of the field over which the polynomial is defined:

Lemma 3 (Section 2.1 in [118]). Let Γ be a Fuchsian group of the first kind, and let Γ' be a subgroup of Γ of finite index. Identify $A_0(\Gamma)$ (resp. $A_0(\Gamma')$) with the field of all meromorphic functions on $\Gamma \setminus \mathbb{H}$ (resp. $\Gamma' \setminus \mathbb{H}$). Then, $\Gamma' \setminus \mathbb{H}$ is a covering of $\Gamma \setminus \mathbb{H}$ of degree [$\Gamma : \Gamma'$]. If Γ' is a normal subgroup of Γ , one can consider the automorphisms of $\Gamma' \setminus \mathbb{H}$, or equivalently of $A_0(\Gamma')$. Then $A_0(\Gamma')$ is a Galois extension of $A_0(\Gamma)$, and $Gal(A_0(\Gamma')/A_0(\Gamma))$ is isomorphic to Γ/Γ' .

The principal congruence subgroup $\Gamma(N)$ is normal in $\mathrm{PSL}(2,\mathbb{Z})$, however neither $\Gamma_0(N)$ nor $\Gamma_1(N)$ are for $N \geq 2$. The extension $\mathbb{C}(X(N))/\mathbb{C}(X(1))$ of the modular curve $\Gamma(N)$ is Galois with Galois group $\mathrm{SL}(2,\mathbb{Z}/N\mathbb{Z})\{\pm 1\}$. Modular function field extensions are further studied in [340] and [145, Section 7.5].

Application to the SW curves

The above allows us to find a general algebraic description of the $N_f = 0, 1, 2, 3, 4$ SW curves. From section 2 it is clear that g_2 and g_3 are polynomials in u of degree 2 and 3. In $N_f \leq 3$, the coefficients of u contain the

masses \boldsymbol{m} and the dynamical scale Λ_{N_f} . In $N_f = 4$, they contain the masses \boldsymbol{m} and the UV-coupling τ_{UV} . We can consider them to be fixed complex numbers.

Consider now the polynomial P(X) (2.13) of degree 6 in X with coefficients in the field $\mathbb{C}(\Gamma)$ of modular functions on $\Gamma \backslash \mathbb{H}$. Any solution to

$$P(X) = 0 \tag{A.51}$$

is called an order parameter for the SW curve. It is important to note that the polynomial (2.13) is not an invariant of the SW curve. However, since it is a relative invariant, its roots (A.51) are absolute invariants.

Examples

It is instructive to study some examples. For the pure SU(2) case, up to \mathbb{C} -proportionality one finds⁴³

$$P(X) = (-1728 + j) + (6912 - j)X^2 - 9216X^4 + 4096X^6.$$
 (A.52)

Such a relation also gives a recurrence relation on the coefficients of the Fourier expansion of any solution u = X. The fact that the coefficients of P(X) can be normalised to be in \mathbb{Z} does however not imply an integer q-expansion of u (it can for instance also be done for some $\mathbf{m} \in \mathbb{Q}^{N_f}$ in $N_f > 0$).

Another example is $N_f = 2$ with $\boldsymbol{m} = (m, m)$ and $m = m_{AD} = \frac{1}{2}$. The polynomial factors over $\mathbb{C}(\Gamma)$ as

$$P(X) = (X - \frac{3}{8})^3 \left((1728 - 5j) + 8(1728 - j)X + 36864X^2 + 32768X^3 \right).$$
(A.53)

Three solutions are $X = \frac{3}{8}$, which is a constant function in $\mathbb{C}(\Gamma)$. If we assume that any solution to above equation has index 6, then Lemma 2 gives a contradiction since if $f \circ \alpha_j^{-1}$ is constant for some coset representative α_j , then since α_j is a bijection on \mathbb{H} also f is constant. But $j \in \mathbb{C}(\Gamma)$ is transcendental and therefore does not satisfy a polynomial equation over \mathbb{Q} such as (A.53). This is true in general, any linear factor of P(X) over \mathbb{C} needs to be divided from P(X). Such constant solutions signal the appearance of an AD point. The remaining degree 3 polynomial in (A.53) does not factor over $\mathbb{C}(\Gamma)$. However, it does factor over $\mathbb{C}(\Gamma^0(2))$, which is genus 0 and generated by $\mathbb{C}(\Gamma^0(2)) \ni h : \tau \mapsto f_{2\mathrm{B}}(\frac{\tau}{2})$ (as demonstrated in section 2.5.3). In order to see this, let us use (2.92) to write $j = \frac{(h+16)^3}{h}$. Then one solution to the degree equation in (A.53) is X = -(h + 40)/64, which is precisely what is found in (2.89). The other solutions are related by $\mathrm{SL}(2,\mathbb{Z})$ images of h.

One also finds the following

Fact 1. For $N_f = 0, 1, 2$, the sum over all 6 solutions to (2.13) is zero.

⁴³We set $\Lambda_{N_f} = 1$ in this subsection.

Proof. The relative invariants g_2 and g_3 are polynomials of degree 2 and 3 in u. For $N_f = 0, 1, 2$, by direct inspection one finds that the coefficients of u in g_2 and the coefficient of u^2 in g_3 are zero. Therefore, the coefficient of u^5 in (2.13) is zero. By Vieta's theorem, the sum over the roots of a polynomial $\sum_{k=0}^{n} a_k x^k$ is $-\frac{a_{n-1}}{a_n}$. Since the u^6 coefficient is never zero, the claim follows.

This does not hold for any mass in $N_f = 3, 4$. Field extensions have been studied in the context of SW theory in [121–123].

Geometric description

The above algebraic construction has a geometric description in terms of Riemann surfaces. We again first lay out the mathematical structure and then apply it to the family of SW curves.

Let Y be a Riemann surface, i.e. a connected complex one-dimensional manifold. Furthermore, let

$$T = y^{n} + a_{1}y^{n-1} + \dots + a_{n} \in K(Y)[y]$$
(A.54)

be a polynomial in y of degree n over the field K(Y) of meromorphic functions $Y \to \mathbb{P}^1(\mathbb{C})$. In case that in the factorisation of T every irreducible factor occurs with multiplicity one, the discriminant D of T is nonzero. This does however not need to be the true. Denote by O the discrete subset of Y containing all poles of the coefficients $a_i \in K(Y)$ and all zeros of the discriminant D. Then, for every point $x_0 \in Y \setminus O$ the polynomial $T_{x_0} =$ $y^n + a_1(x_0)q^{n-1} + \ldots a_n(x_0)$ has exactly n distinct roots over \mathbb{C} . We have the

Theorem 2 (Thm. 2.2.12 in [119]). The Riemann surface of the equation T = 0 is an n-fold ramified covering $\pi : M \to Y$ together with a meromorphic function $y : M \to \mathbb{P}^1(\mathbb{C})$, such that for every point $x_0 \in Y \setminus O$ the set of roots of the polynomial T_{x_0} coincides with the set of values of the function y on the preimage $\pi^{-1}(x_0)$ of the point x_0 under the projection π .

The set O of critical values of the ramified covering $\pi : M \to Y$ associated with the Riemann surface of the equation T = 0 is strictly contained in O. It is called the *ramification set* of the equation T = 0. In the field of germs of meromorphic functions at a point $a \in Y \setminus \tilde{O}$, the equation T = 0 has only roots of multiplicity 1, and their number is equal to the degree of the polynomial T. Each of the meromorphic germs at a satisfying the equation T = 0 corresponds to a point over a in the Riemann surface of the equation.

A.6 Kodaira classification

Let us consider an elliptic curve in Weierstraß form,

$$y^2 = 4x^3 - g_2x - g_3, \tag{A.55}$$

F_v	au	$\operatorname{ord}(g_2)$	$\operatorname{ord}(g_3)$	$\operatorname{ord}(\Delta)$	M	g
I_k	$i\infty$	0	0	k	T^k	$\mathfrak{su}(k)$
I_k^*	$i\infty$	2	3	k+6	PT^k	$\mathfrak{so}(2k+8)$
I_0^*	$ au_0$	≥ 2	≥ 3	6	P	$\mathfrak{so}(8)$
II	$e^{\frac{2\pi i}{3}}$	≥ 1	1	2	$(ST)^{-1}$	-
II^*	$e^{\frac{2\pi i}{3}}$	≥ 4	5	10	(ST)	\mathfrak{e}_8
III	i	1	≥ 2	3	S^{-1}	$\mathfrak{su}(2)$
III^*	i	3	≥ 5	9	S	\mathfrak{e}_7
IV	$e^{\frac{2\pi i}{3}}$	≥ 2	2	4	$(ST)^{-2}$	$\mathfrak{su}(3)$
IV^*	$e^{\frac{2\pi i}{3}}$	≥ 3	4	8	$(ST)^2$	\mathfrak{e}_6

Table 5: Kodaira classification of singular fibers based on the orders of vanishing of g_2 , g_3 and Δ in the Weierstraß model. The value of the complex structure is denoted by τ , while \mathbb{M} is the monodromy and \mathfrak{g} the associated flavour algebra.

where g_2 and g_3 are functions of some parameter u. This is the case for all SW curves (2.3), where g_2 and g_3 are polynomials of degree 2 and 3 in u. The discriminant of the Weierstraß curve is $\Delta = g_2^3 - 27g_3^2$. The study of the elliptic curves on the discriminant divisor $\Sigma = \{u \mid \Delta(u) = 0\}$ is the famous Kodaira classification. At a generic point in Σ , g_2 and g_3 do not vanish simultaneously. If we denote by ord f the order of vanishing (order of zero) of a polynomial or power series f at a given point, then on a generic point in Σ we have $\operatorname{ord}(g_2, g_3, \Delta) = (0, 0, 1)$. The various combinations of orders of vanishing of the invariants g_2 and g_3 are classified [41,42], see for a review [341]. At special points in Σ both $g_2 = g_3 = 0$. From (A.55) we see that the curve becomes cuspidal, $y^2 = x^3$. However the shape of the curve depends on the orders of vanishing of the respective quantities. The full classification is given in Table 5. The singular fibres of the Kodaira classification are all realised as curves of physical theories [245, 246].

A.7 Siegel-Narain theta function

Let *L* be an *n*-dimensional uni-modular lattice with signature (1, n-1). For the application to the *u*-plane integral, $n = b_2(X)$. Let *K* be a characteristic vector of *L*. Its defining property is $l^2 = l \cdot K \mod 2$ for every $l \in L$. Furthermore, we have that $\mu \in L/2$.

We consider the Siegel-Narain theta function $\Psi^J_{\mu} : \mathbb{H} \times \mathbb{C} \to \mathbb{C}$ defined in the main text in (5.51). We repeat it here for convenience,

$$\Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},\boldsymbol{z},\bar{\boldsymbol{z}}) = e^{-2\pi y \boldsymbol{b}_{+}^{2}} \sum_{\boldsymbol{k}\in L+\boldsymbol{\mu}} \partial_{\bar{\tau}} \left(4\pi i \sqrt{y} B(\boldsymbol{k}+\boldsymbol{b},\underline{J})\right) \times (-1)^{B(\boldsymbol{k},K)} q^{-\boldsymbol{k}_{-}^{2}/2} \bar{q}^{\boldsymbol{k}_{+}^{2}/2} e^{-2\pi i B(\boldsymbol{z},\boldsymbol{k}_{-})-2\pi i B(\bar{\boldsymbol{z}},\boldsymbol{k}_{+})},$$
(A.56)

where $\boldsymbol{b} = \text{Im}(\boldsymbol{z})/y$. The transformations under the generators S and T of $\text{PSL}(2,\mathbb{Z})$ are most easily determined if we shift $\boldsymbol{\mu} \to \boldsymbol{\mu} + K/2$. One finds

[81, 84]

$$S: \qquad \Psi^{J}_{\mu+K/2}(-1/\tau, -1/\bar{\tau}, \boldsymbol{z}/\tau, \bar{\boldsymbol{z}}/\bar{\tau}) = -i(-i\tau)^{n/2}(i\bar{\tau})^{2} \\ \times e^{-\pi i \boldsymbol{z}^{2}/\tau + \pi i K^{2}/2} (-1)^{B(\boldsymbol{\mu},K)} \Psi^{J}_{K/2}(\tau, \bar{\tau}, \boldsymbol{z} - \boldsymbol{\mu}, \bar{\boldsymbol{z}} - \boldsymbol{\mu}),$$

$$T: \qquad \Psi^{J}_{\boldsymbol{\mu}+K/2}(\tau + 1, \bar{\tau} + 1, \boldsymbol{z}, \bar{\boldsymbol{z}}) = \\ e^{\pi i (\boldsymbol{\mu}^{2} - K^{2}/4)} \Psi^{J}_{\boldsymbol{\mu}+K/2}(\tau, \bar{\tau}, \boldsymbol{z} + \boldsymbol{\mu}, \bar{\boldsymbol{z}} + \boldsymbol{\mu}).$$

(A.57)

Using these transformations, one finds for the periodicity in $\tau,$

$$\Psi_{\mu}^{J}(\tau+1,\bar{\tau}+1,\boldsymbol{z},\bar{\boldsymbol{z}}) = e^{\pi i (\boldsymbol{\mu}^{2} - B(\boldsymbol{\mu},K))} \Psi_{\mu}^{J}(\tau,\bar{\tau},\boldsymbol{z}+\boldsymbol{\mu}-K/2,\bar{\boldsymbol{z}}+\boldsymbol{\mu}-K/2)$$
(A.58)

and for $S^{-1}T^{-k}S = \begin{pmatrix} 1 & 0 \\ k & 1 \end{pmatrix}$,

$$\Psi^{J}_{\mu}\left(\frac{\tau}{k\tau+1}, \frac{\bar{\tau}}{k\bar{\tau}+1}, \frac{z}{k\tau+1}, \frac{z}{k\bar{\tau}+1}\right) = (k\tau+1)^{\frac{n}{2}}(k\bar{\tau}+1)^{2}e^{-\frac{\pi i k z^{2}}{k\tau+1}}e^{\frac{\pi i}{4}kK^{2}}\Psi^{J}_{\mu}(\tau, \bar{\tau}, \boldsymbol{z}, \bar{\boldsymbol{z}}).$$
(A.59)

We furthermore list the following transformations for z:

• For the reflection $\boldsymbol{z} \rightarrow -\boldsymbol{z}$,

$$\Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},-\boldsymbol{z},-\bar{\boldsymbol{z}}) = -e^{2\pi i B(\boldsymbol{\mu},K)} \Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},\boldsymbol{z},\bar{\boldsymbol{z}}).$$
(A.60)

• For shifting $\boldsymbol{z} \to \boldsymbol{z} + \boldsymbol{\nu}$ with $\boldsymbol{\nu} \in L$,

$$\Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},\boldsymbol{z}+\boldsymbol{\nu},\bar{\boldsymbol{z}}+\boldsymbol{\nu}) = e^{-2\pi i B(\boldsymbol{\nu},\boldsymbol{\mu})} \Psi^{J}_{\boldsymbol{\mu}}(\tau,\bar{\tau},\boldsymbol{z},\bar{\boldsymbol{z}}).$$
(A.61)

• For shifting $\boldsymbol{z} \to \boldsymbol{z} + \boldsymbol{\nu} \tau$ with $\boldsymbol{\nu} \in L \otimes \mathbb{R}$,

$$\Psi^{J}_{\mu}(\tau, \boldsymbol{z} + \boldsymbol{\nu}\tau) = e^{2\pi i B(\boldsymbol{z}, \boldsymbol{\nu})} q^{\boldsymbol{\nu}^{2}/2} (-1)^{-B(\boldsymbol{\nu}, K)} \Psi^{J}_{\mu+\boldsymbol{\nu}}(\tau, \bar{\tau}, \boldsymbol{z}, \bar{\boldsymbol{z}}).$$
(A.62)

We can restrict to $\boldsymbol{\nu} \in L/2$, if the characteristic $\boldsymbol{\mu} + \boldsymbol{\nu}$ is required to be in L/2.

References

- J. Aspman, E. Furrer and J. Manschot, *Elliptic Loci of SU(3) Vacua*, *Annales Henri Poincare* 22 (2021) 2775–2830, [2010.06598].
- [2] J. Aspman, E. Furrer and J. Manschot, *Cutting and gluing with running couplings in* $\mathcal{N} = 2 \ QCD, Phys. Rev. D 105 (2022) 025021, [2107.04600].$
- [3] J. Aspman, E. Furrer and J. Manschot, Four flavours, triality and bimodular forms, Phys. Rev. D 105 (2022) 025017, [2110.11969].
- [4] J. Aspman, E. Furrer, G. Korpas, Z.-C. Ong and M.-C. Tan, The u-plane integral, mock modularity and enumerative geometry, Lett. Math. Phys. 112 (2022) 30, [2109.04302].
- [5] J. Aspman, E. Furrer and J. Manschot, Topological twists of massive SQCD, Part I, 2206.08943.
- S. P. Martin, A Supersymmetry primer, Adv. Ser. Direct. High Energy Phys. 18 (1998) 1–98, [hep-ph/9709356].
- [7] J. Wess and J. Bagger, Supersymmetry and supergravity. Princeton University Press, Princeton, NJ, USA, 1992.
- [8] E. D'Hoker and D. H. Phong, Lectures on supersymmetric Yang-Mills theory and integrable systems, in Theoretical physics at the end of the twentieth century. Proceedings, Summer School, Banff, Canada, June 27-July 10, 1999, pp. 1–125, 1999. hep-th/9912271.
- [9] N. Seiberg, The Power of holomorphy: Exact results in 4-D SUSY field theories, in Particles, Strings, and Cosmology (PASCOS 94), pp. 0357-369, 5, 1994. hep-th/9408013.
- [10] K. A. Intriligator and N. Seiberg, Lectures on supersymmetric gauge theories and electric-magnetic duality, Nucl. Phys. B Proc. Suppl. 45BC (1996) 1–28, [hep-th/9509066].
- S. Elitzur, A. Forge, A. Giveon and E. Rabinovici, Summary of results in N=1 supersymmetric SU(2) gauge theories, Nucl. Phys. B Proc. Suppl. 49 (1996) 174-182, [hep-th/9512140].
- [12] Y. Tachikawa, Lectures on 4d N=1 dynamics and related topics, 1812.08946.
- [13] C. Cordova, T. T. Dumitrescu and K. Intriligator, Multiplets of Superconformal Symmetry in Diverse Dimensions, JHEP 03 (2019) 163, [1612.00809].
- [14] J. Kaidi, M. Martone and G. Zafrir, *Exceptional moduli spaces for exceptional* $\mathcal{N} = 3$ theories, 2203.04972.
- [15] O. Aharony and M. Evtikhiev, On four dimensional N = 3 superconformal theories, JHEP 04 (2016) 040, [1512.03524].
- [16] I. n. García-Etxebarria and D. Regalado, $\mathcal{N} = 3$ four dimensional field theories, JHEP 03 (2016) 083, [1512.06434].
- [17] O. Aharony and Y. Tachikawa, S-folds and 4d N=3 superconformal field theories, JHEP 06 (2016) 044, [1602.08638].
- [18] J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Adv. Theor. Math. Phys. 2 (1998) 231-252, [hep-th/9711200].
- [19] C. Montonen and D. I. Olive, Magnetic Monopoles as Gauge Particles?, Phys. Lett. B 72 (1977) 117–120.
- [20] E. Witten, Solutions of four-dimensional field theories via M theory, Nucl. Phys. B 500 (1997) 3-42, [hep-th/9703166].
- [21] D. Gaiotto, N=2 dualities, JHEP 08 (2012) 034, [0904.2715].

- [22] M. Bershadsky, K. A. Intriligator, S. Kachru, D. R. Morrison, V. Sadov and C. Vafa, Geometric singularities and enhanced gauge symmetries, Nucl. Phys. B 481 (1996) 215-252, [hep-th/9605200].
- [23] S. H. Katz, A. Klemm and C. Vafa, Geometric engineering of quantum field theories, Nucl. Phys. B 497 (1997) 173-195, [hep-th/9609239].
- [24] A. Kapustin, "Langlands duality and topological field theory." https://people.math.harvard.edu/~gaitsgde/Jerusalem_2010/ MathPhysicsSeminar/barcelona.pdf, 2010.
- [25] D. I. Olive, Exact electromagnetic duality, Nucl. Phys. B Proc. Suppl. 45 (1996) 88-102, [hep-th/9508089].
- [26] K. Hori and C. Vafa, Mirror symmetry, hep-th/0002222.
- [27] K. Hori, S. Katz, C. Vafa, R. Thomas, A. M. Society, C. M. Institute et al., *Mirror Symmetry*. Clay mathematics monographs. American Mathematical Society, 2003.
- [28] K. A. Intriligator and N. Seiberg, Mirror symmetry in three-dimensional gauge theories, Phys. Lett. B 387 (1996) 513-519, [hep-th/9607207].
- [29] E. Witten, Mirror manifolds and topological field theory, AMS/IP Stud. Adv. Math. 9 (1998) 121–160, [hep-th/9112056].
- [30] A. Kapustin and E. Witten, Electric-Magnetic Duality And The Geometric Langlands Program, Commun. Num. Theor. Phys. 1 (2007) 1–236, [hep-th/0604151].
- [31] E. Frenkel, Gauge Theory and Langlands Duality, in Bourbaki Seminar, 6, 2009. 0906.2747.
- [32] M. Goresky, "Langlands' conjectures for physicists." https://www.math.ias.edu/~goresky/pdf/Physicists.pdf.
- [33] P. C. Argyres and N. Seiberg, S-duality in N=2 supersymmetric gauge theories, JHEP 12 (2007) 088, [0711.0054].
- [34] D. Gaiotto, Families of N = 2 Field Theories, in New Dualities of Supersymmetric Gauge Theories (J. Teschner, ed.), pp. 31–51. Springer, 2016. 1412.7118. DOI.
- [35] S. Collier and E. Perlmutter, Harnessing S-Duality in $\mathcal{N} = 4$ SYM & Supergravity as $SL(2,\mathbb{Z})$ -Averaged Strings, 2201.05093.
- [36] J. H. Conway and S. P. Norton, Monstrous Moonshine, Bull. London Math. Soc. 11 (1979) 308–339.
- [37] I. Frenkel, J. Lepowsky and A. Meurman, Vertex operator algebras and the Monster, vol. 134 of Pure and Applied Mathematics. Academic Press, United States, 1988.
- [38] C. F. Doran, Picard-Fuchs uniformization: Modularity of the mirror map and mirror moonshine, math/9812162.
- [39] M. Schuett and T. Shioda, *Elliptic Surfaces, arXiv e-prints* (July, 2009) arXiv:0907.0298, [0907.0298].
- [40] T. Shioda, On elliptic modular surfaces, J. Math. Soc. Japan 24 (01, 1972) 20–59.
- [41] K. Kodaira, On Compact Analytic Surfaces: II, Annals of Mathematics 77 (1963) 563–626.
- [42] K. Kodaira, On Compact Analytic Surfaces, III, Annals of Mathematics 78 (1963) 1–40.
- [43] R. Miranda, Persson's list of singular fibers for a rational elliptic surface, Mathematische Zeitschrift 205 (1990) 191–211.
- [44] U. Persson, Configurations of kodaira fibers on rational elliptic surfaces, Mathematische Zeitschrift 205 (1990) 1–47.

- [45] N. Seiberg and E. Witten, Electric magnetic duality, monopole condensation, and confinement in N=2 supersymmetric Yang-Mills theory, Nucl. Phys. B426 (1994) 19-52, [hep-th/9407087].
- [46] N. Seiberg and E. Witten, Monopoles, duality and chiral symmetry breaking in N=2 supersymmetric QCD, Nucl. Phys. B431 (1994) 484-550, [hep-th/9408099].
- [47] A. Klemm, On the geometry behind N=2 supersymmetric effective actions in four-dimensions, in 33rd Karpacz Winter School of Theoretical Physics: Duality -Strings and Fields, 5, 1997. hep-th/9705131.
- [48] A. Bilal, Duality in N=2 SUSY SU(2) Yang-Mills theory: A Pedagogical introduction to the work of Seiberg and Witten, in Quantum fields and quantum space time. Proceedings, NATO Advanced Study Institute, Cargese, France, July 22-August 3, 1996, pp. 21-43, 1997. hep-th/9601007.
- [49] W. Lerche, Introduction to Seiberg-Witten theory and its stringy origin, Nucl. Phys. Proc. Suppl. 55B (1997) 83-117, [hep-th/9611190].
- [50] Y. Tachikawa, N=2 supersymmetric dynamics for pedestrians, vol. 890. Lect.Notes Phys. 890 (2014), 2014, 10.1007/978-3-319-08822-8.
- [51] C. Gomez and R. Hernandez, Electric magnetic duality and effective field theories, in Advanced School on Effective Theories, 6, 1995. hep-th/9510023.
- [52] J. Labastida and M. Marino, Topological quantum field theory and four manifolds. Springer Netherlands, 2005.
- [53] H. Nakajima and K. Yoshioka, Lectures on instanton counting, in CRM Workshop on Algebraic Structures and Moduli Spaces Montreal, Canada, July 14-20, 2003, 2003. math/0311058.
- [54] L. Alvarez-Gaume and S. F. Hassan, Introduction to S duality in N=2 supersymmetric gauge theories: A Pedagogical review of the work of Seiberg and Witten, Fortsch. Phys. 45 (1997) 159–236, [hep-th/9701069].
- [55] F. Ferrari and A. Bilal, The Strong coupling spectrum of the Seiberg-Witten theory, Nucl. Phys. B469 (1996) 387-402, [hep-th/9602082].
- [56] D. Alexander, C. Cummins, J. McKay and C. Simons, "Completely replicable functions" in *Groups, Combinatorics, and Geometry, Durham Symposium*, 1990, London Math. Soc. Lecture Note Ser. 165, Cambridge Univ. Press, Cambridge, 1992.
- [57] D. Ford, J. McKay and S. Norton, More on replicable functions, Communications in Algebra 22 (1994) 5175–5193.
- [58] C. R. Ferenbaugh, The genus-zero problem for n|h -type groups, Duke Math. J. 72 (10, 1993) 31–63.
- [59] N. Koblitz, Introduction to Elliptic Curves and Modular Forms. Graduate Texts in Mathematics. Springer New York, 1993.
- [60] E. Witten, Topological sigma models, Comm. Math. Phys. 118 (1988) 411-449.
- [61] E. Witten, Topological Quantum Field Theory, Commun. Math. Phys. 117 (1988) 353.
- [62] S. Gukov, D. Pei, P. Putrov and C. Vafa, 4-manifolds and topological modular forms, JHEP 05 (2021) 084, [1811.07884].
- [63] A. Gadde, S. Gukov and P. Putrov, Fivebranes and 4-manifolds, Prog. Math. 319 (2016) 155-245, [1306.4320].
- [64] S. Gukov, Trisecting non-Lagrangian theories, JHEP 11 (2017) 178, [1707.01515].
- [65] M. Dedushenko, S. Gukov and P. Putrov, Vertex algebras and 4-manifold invariants, in Nigel Hitchin's 70th Birthday Conference, vol. 1, pp. 249–318, 5, 2017. 1705.01645. DOI.

- [66] M. Mariño, G. Moore and G. Peradze, Superconformal invariance and the geography of four-manifolds, Communications in Mathematical Physics **205** (1999) 691–735.
- [67] M. Atiyah and L. Jeffrey, Topological Lagrangians and cohomology, J. Geom. Phys. 7 (1990) 119–136.
- [68] S. Gukov, P. Putrov and C. Vafa, Fivebranes and 3-manifold homology, JHEP 07 (2017) 071, [1602.05302].
- [69] T. Dimofte, D. Gaiotto and S. Gukov, Gauge Theories Labelled by Three-Manifolds, Commun. Math. Phys. 325 (2014) 367–419, [1108.4389].
- [70] E. Witten, Supersymmetric Yang-Mills theory on a four manifold, J. Math. Phys. 35 (1994) 5101-5135, [hep-th/9403195].
- [71] E. Witten, Monopoles and four manifolds, Math. Res. Lett. 1 (1994) 769-796, [hep-th/9411102].
- [72] E. Witten, On S duality in Abelian gauge theory, Selecta Math. 1 (1995) 383, [hep-th/9505186].
- [73] G. W. Moore and E. Witten, Integration over the u plane in Donaldson theory, Adv. Theor. Math. Phys. 1 (1997) 298–387, [hep-th/9709193].
- [74] A. Losev, N. Nekrasov and S. L. Shatashvili, Issues in topological gauge theory, Nucl. Phys. B534 (1998) 549-611, [hep-th/9711108].
- [75] A. Lossev, N. Nekrasov and S. L. Shatashvili, Testing Seiberg-Witten solution, NATO Sci. Ser. C 520 (1999) 359–372, [hep-th/9801061].
- [76] M. Marino and G. W. Moore, Donaldson invariants for nonsimply connected manifolds, Commun. Math. Phys. 203 (1999) 249, [hep-th/9804104].
- [77] A. Malmendier and K. Ono, SO(3)-Donaldson invariants of P² and Mock Theta Functions, Geom. Topol. 16 (2012) 1767–1833, [0808.1442].
- [78] A. Malmendier and K. Ono, Moonshine and Donaldson invariants of \mathbb{P}^2 , 1207.5139.
- [79] M. Griffin, A. Malmendier and K. Ono, SU(2)-Donaldson invariants of the complex projective plane, Forum Math. 27 (2015) 2003–2023, [1209.2743].
- [80] A. Malmendier, Donaldson invariants of $\mathbb{P}^1 \times \mathbb{P}^1$ and Mock Theta Functions, Commun. Num. Theor. Phys. 5 (2011) 203–229, [1008.0175].
- [81] G. Korpas and J. Manschot, Donaldson-Witten theory and indefinite theta functions, JHEP 11 (2017) 083, [1707.06235].
- [82] G. W. Moore and I. Nidaiev, The Partition Function Of Argyres-Douglas Theory On A Four-Manifold, 1711.09257.
- [83] G. Korpas, J. Manschot, G. Moore and I. Nidaiev, Renormalization and BRST Symmetry in Donaldson-Witten Theory, Annales Henri Poincare 20 (2019) 3229-3264, [1901.03540].
- [84] G. Korpas, J. Manschot, G. W. Moore and I. Nidaiev, Mocking the u-plane integral, Research in the Mathematical Sciences 8 (2021) 43.
- [85] J. Manschot and G. W. Moore, Topological correlators of SU(2), $\mathcal{N} = 2^*$ SYM on four-manifolds, 2104.06492.
- [86] G. Korpas, Mock modularity and surface defects in topological $\mathcal{N} = 2$ super Yang-Mills theory, Phys. Rev. D 105 (2022) 026025.
- [87] M. Marino and G. W. Moore, The Donaldson-Witten function for gauge groups of rank larger than one, Commun. Math. Phys. 199 (1998) 25–69, [hep-th/9802185].
- [88] H. Kanno and S.-K. Yang, Donaldson-Witten functions of massless N=2 supersymmetric QCD, Nucl. Phys. B 535 (1998) 512–530, [hep-th/9806015].
- [89] M. Marino, G. W. Moore and G. Peradze, Four manifold geography and superconformal symmetry, math/9812042.

- [90] C. Lozano and M. Marino, Donaldson invariants of product ruled surfaces and two-dimensional gauge theories, Commun. Math. Phys. 220 (2001) 231-261, [hep-th/9907165].
- [91] A. Sen, Strong weak coupling duality in four-dimensional string theory, Int. J. Mod. Phys. A 9 (1994) 3707-3750, [hep-th/9402002].
- [92] C. Vafa and E. Witten, A Strong coupling test of S duality, Nucl. Phys. B431 (1994) 3-77, [hep-th/9408074].
- [93] E. P. Verlinde, Global aspects of electric magnetic duality, Nucl. Phys. B455 (1995) 211-228, [hep-th/9506011].
- [94] A. Klemm, W. Lerche, S. Yankielowicz and S. Theisen, Simple singularities and N=2 supersymmetric Yang-Mills theory, Phys. Lett. B344 (1995) 169–175, [hep-th/9411048].
- [95] A. Klemm, W. Lerche, S. Yankielowicz and S. Theisen, On the monodromies of N = 2 supersymmetric yang-mills theory, hep-th/9412158.
- [96] A. Klemm, W. Lerche and S. Theisen, Nonperturbative effective actions of N=2 supersymmetric gauge theories, Int. J. Mod. Phys. A11 (1996) 1929–1974, [hep-th/9505150].
- [97] U. H. Danielsson and B. Sundborg, The Moduli space and monodromies of N=2 supersymmetric SO(2r+1) Yang-Mills theory, Phys. Lett. B358 (1995) 273-280, [hep-th/9504102].
- [98] M. Marino and G. Moore, Integrating over the coulomb branch in n = 2 gauge theory, Nuclear Physics B Proceedings Supplements 68 (1998) 336 347.
- [99] S. Hyun, J. Park and J.-S. Park, N=2 supersymmetric QCD and four manifolds: 1. The Donaldson and Seiberg-Witten invariants, hep-th/9508162.
- [100] M. Matone, Koebe 1/4 theorem and inequalities in n=2 supersymmetric qcd, Phys. Rev. D 53 (Jun, 1996) 7354–7358.
- [101] W. Nahm, On the Seiberg-Witten approach to electric magnetic duality, hep-th/9608121.
- [102] K. Ito and S.-K. Yang, Prepotentials in N=2 SU(2) supersymmetric Yang-Mills theory with massless hypermultiplets, Phys. Lett. B 366 (1996) 165-173, [hep-th/9507144].
- [103] M.-x. Huang and A. Klemm, Holomorphicity and Modularity in Seiberg-Witten Theories with Matter, JHEP 07 (2010) 083, [0902.1325].
- [104] P. C. Argyres and M. R. Douglas, New phenomena in SU(3) supersymmetric gauge theory, Nucl. Phys. B448 (1995) 93-126, [hep-th/9505062].
- [105] P. C. Argyres, M. R. Plesser, N. Seiberg and E. Witten, New N=2 superconformal field theories in four-dimensions, Nucl. Phys. B461 (1996) 71-84, [hep-th/9511154].
- [106] T. Eguchi, K. Hori, K. Ito and S.-K. Yang, Study of N=2 superconformal field theories in four-dimensions, Nucl. Phys. B 471 (1996) 430-444, [hep-th/9603002].
- [107] T. Eguchi and K. Hori, N=2 superconformal field theories in four-dimensions and A-D-E classification, in Conference on the Mathematical Beauty of Physics (In Memory of C. Itzykson), 7, 1996. hep-th/9607125.
- [108] J. Seo and K. Dasgupta, Argyres-Douglas Loci, Singularity Structures and Wall-Crossings in Pure N=2 Gauge Theories with Classical Gauge Groups, JHEP 05 (2012) 072, [1203.6357].
- [109] D. Xie, General Argyres-Douglas Theory, JHEP 01 (2013) 100, [1204.2270].
- [110] M. Matone, Instantons and recursion relations in N=2 SUSY gauge theory, Phys. Lett. B357 (1995) 342-348, [hep-th/9506102].

- [111] J. Manschot, G. W. Moore and X. Zhang, Effective gravitational couplings of four-dimensional N = 2 supersymmetric gauge theories, JHEP 06 (2020) 150, [1912.04091].
- [112] A. Klemm, W. Lerche, P. Mayr, C. Vafa and N. P. Warner, Selfdual strings and N=2 supersymmetric field theory, Nucl. Phys. B 477 (1996) 746-766, [hep-th/9604034].
- [113] D. Gaiotto, G. W. Moore and A. Neitzke, Wall-crossing, Hitchin Systems, and the WKB Approximation, 0907.3987.
- [114] A. Malmendier, The Signature of the Seiberg-Witten surface, Surveys Diff. Geom. 15 (2010) 255–278, [0802.1363].
- [115] R. S. Maier, On Rationally Parametrized Modular Equations, arXiv Mathematics e-prints (Nov., 2006) math/0611041, [math/0611041].
- [116] R. Miranda, An overview of algebraic surfaces, Lecture Notes in Pure and Appl. Math. (1997) 157–217.
- [117] T. Eguchi, Seiberg-Witten Theory and S-Duality, vol. Strings, Branes and Dualities, pp. 103–120. Springer Netherlands, Dordrecht, 1999.
- [118] G. Shimura, Introduction to the Arithmetic Theory of Automorphic Functions. Kanô memorial lectures. Princeton University Press, 1971.
- [119] A. Khovanskii, Galois Theory, Coverings, and Riemann Surfaces. Springer Berlin Heidelberg, 2013, 10.1007/978-3-642-38841-5.
- [120] C. Y. Park, Ramification Points of Seiberg-Witten Curves, JHEP 07 (2011) 068, [1102.0288].
- [121] A. Bourget and J. Troost, Permutations of Massive Vacua, JHEP 05 (2017) 042, [1702.02102].
- [122] F. Ferrari, On the Geometry of Super Yang-Mills Theories: Phases and Irreducible Polynomials, JHEP 01 (2009) 026, [0810.0816].
- [123] F. Ferrari, Galois symmetries in Super Yang-Mills Theories, JHEP 03 (2009) 128, [0901.4079].
- [124] Y.-H. He and J. Read, Hecke Groups, Dessins d'Enfants and the Archimedean Solids, Front. in Phys. 3 (2015) 91, [1309.2326].
- [125] S. K. Ashok, F. Cachazo and E. Dell'Aquila, Children's drawings from Seiberg-Witten curves, Commun. Num. Theor. Phys. 1 (2007) 237-305, [hep-th/0611082].
- [126] S. K. Ashok, F. Cachazo and E. Dell'Aquila, Strebel differentials with integral lengths and Argyres-Douglas singularities, hep-th/0610080.
- [127] Y.-H. He and J. McKay, N=2 Gauge Theories: Congruence Subgroups, Coset Graphs and Modular Surfaces, J. Math. Phys. 54 (2013) 012301, [1201.3633].
- [128] T. Eguchi, K. Hori, K. Ito and S.-K. Yang, Study of n = 2 superconformal field theories in 4 dimensions, Nuclear Physics B 471 (1996) 430 – 442.
- [129] C. Closset and H. Magureanu, The U-plane of rank-one 4d $\mathcal{N} = 2$ KK theories, SciPost Phys. 12 (2022) 065, [2107.03509].
- [130] J. Sonnenschein, S. Theisen and S. Yankielowicz, On the relation between the holomorphic prepotential and the quantum moduli in susy gauge theories, *Physics* Letters B 367 (Jan, 1996) 145–150.
- [131] T. Eguchi and S.-K. Yang, Prepotentials of N=2 supersymmetric gauge theories and soliton equations, Mod. Phys. Lett. A11 (1996) 131-138, [hep-th/9510183].
- [132] E. D'Hoker, I. Krichever and D. Phong, The Renormalization group equation in N=2 supersymmetric gauge theories, Nucl. Phys. B 494 (1997) 89-104, [hep-th/9610156].

- [133] R. Flume, F. Fucito, J. F. Morales and R. Poghossian, Matone's relation in the presence of gravitational couplings, Journal of High Energy Physics 2004 (Apr, 2004) 008–008.
- [134] W. He, Matone's relation of super yang—mills and spectrum of toda chain, Communications in Theoretical Physics 56 (Nov, 2011) 905–912.
- [135] K. Ito, S. Kanno and T. Okubo, Quantum periods and prepotential in $\mathcal{N} = 2 \, su(2)$ sqcd, Journal of High Energy Physics **2017** (Aug, 2017).
- [136] R. Donagi and E. Witten, Supersymmetric Yang-Mills theory and integrable systems, Nucl. Phys. B460 (1996) 299-334, [hep-th/9510101].
- [137] A. Gorsky, I. Krichever, A. Marshakov, A. Mironov and A. Morozov, Integrability and Seiberg-Witten exact solution, Phys. Lett. B 355 (1995) 466-474, [hep-th/9505035].
- [138] E. D'Hoker and D. Phong, Calogero-moser systems in su(n) seiberg-witten theory, Nuclear Physics B 513 (Mar, 1998) 405–444.
- [139] N. Nekrasov, Five-dimensional gauge theories and relativistic integrable systems, Nuclear Physics B 531 (Oct, 1998) 323–344.
- [140] H. Kanno and Y. Ohta, Picard-fuchs equation and prepotential of five-dimensional susy gauge theory compactified on a circle, Nuclear Physics B 530 (Oct, 1998) 73–98.
- [141] Y. Ohta, Prepotentials of n=2 su(2) yang-mills theories coupled with massive matter multiplets, Journal of Mathematical Physics 38 (Feb, 1997) 682–696.
- [142] Y. Ohta, Differential equations for scaling relation in n=2 supersymmetric su(2) yang-mills theory coupled with massive hypermultiplet, Journal of Mathematical Physics 40 (Apr, 1999) 1891–1900.
- [143] Y. Ohta, Prepotential of N=2 SU(2) Yang-Mills gauge theory coupled with a massive matter multiplet, J. Math. Phys. 37 (1996) 6074-6085, [hep-th/9604051].
- [144] A. D. Shapere and Y. Tachikawa, Central charges of N=2 superconformal field theories in four dimensions, JHEP 09 (2008) 109, [0804.1957].
- [145] F. Diamond and J. Shurman, A First Course in Modular Forms, vol. 228 of Graduate Texts in Mathematics. Springer-Verlag New York, 1 ed., 2005.
- [146] A. Brandhuber and S. Stieberger, Periods, coupling constants and modular functions in N=2 SU(2) SYM with massive matter, Int. J. Mod. Phys. A 13 (1998) 1329–1344, [hep-th/9609130].
- [147] T. Eguchi and H. Kanno, Five-dimensional gauge theories and local mirror symmetry, Nucl. Phys. B 586 (2000) 331-345, [hep-th/0005008].
- [148] M.-x. Huang, A.-K. Kashani-Poor and A. Klemm, The Ω deformed B-model for rigid $\mathcal{N} = 2$ theories, Annales Henri Poincare 14 (2013) 425–497, [1109.5728].
- [149] L. Alvarez-Gaume, M. Marino and F. Zamora, Softly broken N=2 QCD with massive quark hypermultiplets. 2., Int. J. Mod. Phys. A 13 (1998) 1847–1880, [hep-th/9707017].
- [150] L. Alvarez-Gaume, M. Marino and F. Zamora, Softly broken N=2 QCD with massive quark hypermultiplets. 1., Int. J. Mod. Phys. A 13 (1998) 403-430, [hep-th/9703072].
- [151] B. P. Dolan, Meromorphic scaling flow of N=2 supersymmetric SU(2) Yang-Mills with matter, Nucl. Phys. B 737 (2006) 153-175, [hep-th/0506088].
- [152] B. P. Dolan, Modular Symmetry and Fractional Charges in N=2 Supersymmetric Yang-Mills and the Quantum Hall Effect, SIGMA 3 (2007) 010, [hep-th/0611282].
- [153] A. Klemm, J. Manschot and T. Wotschke, Quantum geometry of elliptic Calabi-Yau manifolds, 1205.1795.

- [154] D. Schultz, "Notes on modular forms." URL: https://faculty.math.illinois.edu/~schult25/ModFormNotes.pdf.
- [155] P. Argyres, M. Lotito, Y. Lü and M. Martone, Geometric constraints on the space of $\mathcal{N} = 2$ SCFTs. Part I: physical constraints on relevant deformations, JHEP **02** (2018) 001, [1505.04814].
- [156] T. Masuda and H. Suzuki, Periods and prepotential of N=2 SU(2) supersymmetric Yang-Mills theory with massive hypermultiplets, Int. J. Mod. Phys. A 12 (1997) 3413-3431, [hep-th/9609066].
- [157] A. Bilal and F. Ferrari, The BPS spectra and superconformal points in massive N=2 supersymmetric QCD, Nucl. Phys. B 516 (1998) 175-228, [hep-th/9706145].
- [158] J. Juan-Zacarías and A. Verjovsky, Some remarks on equilateral triangulations of surfaces and Belyi functions, arXiv e-prints (Feb., 2020) arXiv:2002.04660, [2002.04660].
- [159] G. Almkvist and J. Guillera, Ramanujan-sato-like series, in Number Theory and Related Fields (J. M. Borwein, I. Shparlinski and W. Zudilin, eds.), (New York, NY), pp. 55–74, Springer New York, 2013.
- [160] H. Chan, S. Chan and Z.-G. Liu, Domb's numbers and ramanujan-sato type series for $1/\pi$, Advances in Mathematics 186 (08, 2004).
- [161] H. H. Chan, Y. Tanigawa, Y. Yang and W. Zudilin, New analogues of clausens identities arising from the theory of modular forms, Advances in Mathematics 228 (2011) 1294 – 1314.
- [162] H. Chan and S. Cooper, Rational analogues of ramanujan's series for 1/π, Mathematical Proceedings of the Cambridge Philosophical Society 153 (09, 2012) 1 – 23.
- [163] V. Tatitscheff, Y.-H. He and J. McKay, Cusps, Congruence Groups and Monstrous Dessins, 1812.11752.
- [164] Y.-H. He and M. van Loon, Gauge Theories, Tessellations \& Riemann Surfaces, JHEP 06 (2014) 053, [1402.3846].
- [165] N. Hajouji and P.-K. Oehlmann, Modular Curves and Mordell-Weil Torsion in F-theory, JHEP 04 (2020) 103, [1910.04095].
- [166] N. Seiberg, Five-dimensional SUSY field theories, nontrivial fixed points and string dynamics, Phys. Lett. B 388 (1996) 753-760, [hep-th/9608111].
- [167] J. G. Russo, $\mathcal{N} = 2$ gauge theories and quantum phases, JHEP 12 (2014) 169, [1411.2602].
- [168] J. G. Russo, Properties of the partition function of $\mathcal{N} = 2$ supersymmetric QCD with massive matter, JHEP 07 (2019) 125, [1905.05267].
- [169] P. C. Argyres and J. R. Wittig, Infinite coupling duals of N=2 gauge theories and new rank 1 superconformal field theories, JHEP 01 (2008) 074, [0712.2028].
- [170] P. C. Argyres, M. Lotito, Y. Lü and M. Martone, Expanding the landscape of $\mathcal{N} = 2$ rank 1 SCFTs, JHEP 05 (2016) 088, [1602.02764].
- [171] P. Argyres and M. Martone, Construction and classification of Coulomb branch geometries, 2003.04954.
- [172] L. F. Alday, D. Gaiotto and Y. Tachikawa, Liouville Correlation Functions from Four-dimensional Gauge Theories, Lett. Math. Phys. 91 (2010) 167–197, [0906.3219].
- [173] F. Ferrari, The Dyon spectra of finite gauge theories, Nucl. Phys. B 501 (1997) 53-96, [hep-th/9702166].
- [174] J. M. F. Labastida and C. Lozano, Duality in twisted N=4 supersymmetric gauge theories in four-dimensions, Nucl. Phys. B537 (1999) 203-242, [hep-th/9806032].

- [175] J. Stienstra and D. Zagier, "Bimodular forms and holomorphic anomaly equation." Workshop on Modular Forms and String Duality, Banff International Research Centre, 2006, report: https://www.birs.ca/workshops/2006/06w5041/report06w5041.pdf.
- [176] Y. Yang, N. Yui et al., Differential equations satisfied by modular forms and k3 surfaces, Illinois Journal of Mathematics 51 (2007) 667–696.
- [177] L. Wang and Y. Yang, Ramanujan-type $1/\pi$ -series from bimodular forms, arXiv e-prints (Mar., 2020) arXiv:2003.02572, [2003.02572].
- [178] J. A. Minahan and D. Nemeschansky, N=2 superYang-Mills and subgroups of SL(2,Z), Nucl. Phys. B468 (1996) 72-84, [hep-th/9601059].
- [179] N. Dorey, V. V. Khoze and M. P. Mattis, On N=2 supersymmetric QCD with four flavors, Nucl. Phys. B 492 (1997) 607–622, [hep-th/9611016].
- [180] P. C. Argyres and S. Pelland, Comparing instanton contributions with exact results in N=2 supersymmetric scale invariant theories, JHEP 03 (2000) 014, [hep-th/9911255].
- [181] J. F. Adams, Lectures on exceptional lie groups. Chicago lectures in mathematics. Chicago Univ. Press, Chicago, IL, 1996.
- [182] W. Fulton and J. Harris, *Representation Theory*. Springer New York, 2004, 10.1007/978-1-4612-0979-9.
- [183] M. Marino and F. Zamora, Duality symmetry in softly broken N=2 gauge theories, Nucl. Phys. B 533 (1998) 373-405, [hep-th/9804038].
- [184] T.-S. Tai, Triality in SU(2) Seiberg-Witten theory and Gauss hypergeometric function, Phys. Rev. D 82 (2010) 105007, [1006.0471].
- [185] N. Nekrasov, BPS/CFT correspondence: non-perturbative Dyson-Schwinger equations and qq-characters, JHEP 03 (2016) 181, [1512.05388].
- [186] S. Jeong and X. Zhang, A note on chiral trace relations from qq-characters, JHEP 04 (2020) 026, [1910.10864].
- [187] N. Nekrasov and V. Pestun, Seiberg-Witten geometry of four dimensional N=2 quiver gauge theories, 1211.2240.
- [188] J. Labastida and C. Lozano, Mass perturbations in twisted n = 4 supersymmetric gauge theories, Nuclear Physics B 518 (05, 1998) 37–58.
- [189] G. Bonelli, F. Del Monte, P. Gavrylenko and A. Tanzini, N = 2^{*} Gauge Theory, Free Fermions on the Torus and Painlevé VI, Commun. Math. Phys. 377 (2020) 1381-1419, [1901.10497].
- [190] J. A. Minahan, D. Nemeschansky and N. P. Warner, Instanton expansions for mass deformed N=4 superYang-Mills theories, Nucl. Phys. B528 (1998) 109-132, [hep-th/9710146].
- [191] N. A. Nekrasov, Seiberg-Witten prepotential from instanton counting, Adv. Theor. Math. Phys. 7 (2003) 831-864, [hep-th/0206161].
- [192] M. Miyamoto, A modular invariance on the theta functions defined on vertex operator algebras, arXiv Mathematics e-prints (Sept., 1998) math/9809168, [math/9809168].
- [193] M. Krauel and G. Mason, Vertex operator algebras and weak jacobi forms, International Journal of Mathematics 23 (2012) 1250024.
- [194] M. C. N. Cheng, T. Gannon and G. Lockhart, Modular Exercises for Four-Point Blocks – I, 2002.11125.
- [195] T. W. Grimm, A. Klemm, M. Marino and M. Weiss, Direct Integration of the Topological String, JHEP 08 (2007) 058, [hep-th/0702187].

- [196] A. L. Gorodentsev and M. I. Leenson, How to calculate the correlation function in twisted SYM N=2, $N_f = 4$ QFT on projective plane, alg-geom/9604011.
- [197] G. Moore, "On four-manifolds and N = 2 supersymmetric field theory." Talk at String-Math, Tohoku University, Sendai, http://www.physics.rutgers.edu/~gmoore/StringMath2018-Final.pdf, June, 2018.
- [198] A. Sen, F-theory and orientifolds, Nucl. Phys. B 475 (1996) 562-578, [hep-th/9605150].
- [199] M. J. Duff, J. T. Liu and J. Rahmfeld, Four-dimensional string-string-string triality, Nucl. Phys. B 459 (1996) 125-159, [hep-th/9508094].
- [200] B. Bastian, S. Hohenegger, A. Iqbal and S.-J. Rey, *Triality in Little String Theories*, *Phys. Rev. D* 97 (2018) 046004, [1711.07921].
- [201] M. Aganagic, V. Bouchard and A. Klemm, Topological Strings and (Almost) Modular Forms, Commun. Math. Phys. 277 (2008) 771-819, [hep-th/0607100].
- [202] T. Masuda, T. Sasaki and H. Suzuki, Seiberg-Witten theory of rank two gauge groups and hypergeometric series, Int. J. Mod. Phys. A13 (1998) 3121-3144, [hep-th/9705166].
- [203] T. Masuda and H. Suzuki, Periods and prepotential of N=2 SU(2) supersymmetric Yang-Mills theory with massive hypermultiplets, Int. J. Mod. Phys. A12 (1997) 3413-3431, [hep-th/9609066].
- [204] T. Masuda and H. Suzuki, On explicit evaluations around the conformal point in n=2 supersymmetric yang-mills theories, arXiv:hep-th/9612240.
- [205] J. A. Minahan and D. Nemeschansky, Hyperelliptic curves for supersymmetric Yang-Mills, Nucl. Phys. B464 (1996) 3-17, [hep-th/9507032].
- [206] P. C. Argyres and A. Buchel, The Nonperturbative gauge coupling of N=2 supersymmetric theories, Phys. Lett. B442 (1998) 180-184, [hep-th/9806234].
- [207] S. K. Ashok, M. Billò, E. Dell'Aquila, M. Frau, A. Lerda and M. Raman, Modular anomaly equations and S-duality in N = 2 conformal SQCD, JHEP 10 (2015) 091, [1507.07476].
- [208] S. K. Ashok, E. Dell'Aquila, A. Lerda and M. Raman, S-duality, triangle groups and modular anomalies in $\mathcal{N} = 2$ SQCD, JHEP 04 (2016) 118, [1601.01827].
- [209] D. Galakhov, P. Longhi, T. Mainiero, G. W. Moore and A. Neitzke, Wild Wall Crossing and BPS Giants, JHEP 11 (2013) 046, [1305.5454].
- [210] Q. Wang, Wall Crossing Structures and Application to SU(3) Seiberg-Witten Integrable system, 1903.10169.
- [211] T. Shaska and H. Voelklein, Elliptic subfields and automorphisms of genus 2 function fields, arXiv Mathematics e-prints (July, 2001) math/0107142, [math/0107142].
- [212] T. Shaska, Genus two curves covering elliptic curves: a computational approach, 1209.3187.
- [213] J. Gutierrez and T. Shaska, Hyperelliptic curves with extra involutions, arXiv Mathematics e-prints (Jan., 2006) math/0601456, [math/0601456].
- [214] J. Gutierrez, D. Sevilla and T. Shaska, Hyperelliptic curves of genus 3 with prescribed automorphism group, arXiv e-prints (Sept., 2012) arXiv:1209.2938, [1209.2938].
- [215] P. C. Argyres and A. E. Faraggi, The vacuum structure and spectrum of N=2 supersymmetric SU(n) gauge theory, Phys. Rev. Lett. 74 (1995) 3931-3934, [hep-th/9411057].
- [216] M. Abramowitz and I. A. Stegun, Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables. Dover, New York, ninth dover printing, tenth gpo printing ed., 1964.

- [217] G. Rosenhain, Abhandlung über die functionen zweier variablen mit vier perioden welche die inversion sind der ultra-elliptische integrale erster klasse (1851), Translation to German from Latin manuscript published in: Klassiker der Exacten Wissenschaften 65 (1895) 1 – 96.
- [218] J. ichi Igusa, Modular forms and projective invariants, American Journal of Mathematics 89 (1967) 817–855.
- [219] J.-I. Igusa, On siegel modular forms of genus two, American Journal of Mathematics 84 (1962) 175–200.
- [220] M. R. Douglas and S. H. Shenker, Dynamics of SU(N) supersymmetric gauge theory, Nucl. Phys. B447 (1995) 271-296, [hep-th/9503163].
- [221] M. Magro, L. O'Raifeartaigh and I. Sachs, On the uniqueness of the effective Lagrangian for N = 2 SQCD, Nucl. Phys. B 508 (1997) 433-448, [hep-th/9704027].
- [222] B. H. Lian and S.-T. Yau, Mirror maps, modular relations and hypergeometric series 1, hep-th/9507151.
- [223] M. Alim, E. Scheidegger, S.-T. Yau and J. Zhou, Special Polynomial Rings, Quasi Modular Forms and Duality of Topological Strings, Adv. Theor. Math. Phys. 18 (2014) 401–467, [1306.0002].
- [224] K. H. B. Gordon, Multiplicative properties of eta-products, vol. 143 of Contemp. Math. Amer. Math. Soc., 1993.
- [225] K. Ono, The Web of Modularity: Arithmetic of the Coefficients of Modular Forms and q-series, vol. 102. American Mathematical Society, cbms regional conference series in mathematics ed., 2004.
- [226] J.-I. Igusa, Arithmetic variety of moduli for genus two, Annals of Mathematics 72 (1960) 612–649.
- [227] T. Shaska and S. Wijesiri, Theta functions and algebraic curves with automorphisms, NATO Science for Peace and Security Series D: Information and Communication Security 24 (06, 2009) 193–237.
- [228] A. Klemm, M. Poretschkin, T. Schimannek and M. Westerholt-Raum, Direct Integration for Mirror Curves of Genus Two and an Almost Meromorphic Siegel Modular Form, 1502.00557.
- [229] T. Shaska, Determining the automorphism group of a hyperelliptic curve, arXiv Mathematics e-prints (Dec., 2003) math/0312284, [math/0312284].
- [230] A. Hanany and Y. Oz, On the quantum moduli space of vacua of N=2 supersymmetric SU(N(c)) gauge theories, Nucl. Phys. B 452 (1995) 283-312, [hep-th/9505075].
- [231] P. C. Argyres, M. Plesser and A. D. Shapere, The Coulomb phase of N=2 supersymmetric QCD, Phys. Rev. Lett. 75 (1995) 1699–1702, [hep-th/9505100].
- [232] M. R. Abolhasani, M. Alishahiha and A. M. Ghezelbash, The Moduli space and monodromies of the N=2 supersymmetric Yang-Mills theory with any Lie gauge groups, Nucl. Phys. B 480 (1996) 279-295, [hep-th/9606043].
- [233] O. Aharony and S. Yankielowicz, Exact electric magnetic duality in N=2 supersymmetric QCD theories, Nucl. Phys. B 473 (1996) 93-120, [hep-th/9601011].
- [234] J. A. Minahan, Duality symmetries for N=2 supersymmetric QCD with vanishing Beta functions, Nucl. Phys. B 537 (1999) 243-259, [hep-th/9806246].
- [235] A. Brandhuber and K. Landsteiner, On the monodromies of N=2 supersymmetric Yang-Mills theory with gauge group SO(2n), Phys. Lett. B 358 (1995) 73-80, [hep-th/9507008].
- [236] A. Hanany, On the quantum moduli space of vacua N=2 supersymmetric gauge theories, Nucl. Phys. B 466 (1996) 85-100, [hep-th/9509176].

- [237] J. S. Seo, Singularity Structure of N=2 Supersymmetric Yang-Mills Theories: A Review, 1307.2691.
- [238] J. Zhou, Arithmetic Properties of Moduli Spaces and Topological String Partition Functions of Some Calabi-Yau Threefolds. PhD thesis, Harvard U. (main), 2014.
- [239] D. Persson and R. Volpato, Fricke S-duality in CHL models, JHEP 12 (2015) 156, [1504.07260].
- [240] N. M. Paquette, D. Persson and R. Volpato, Monstrous BPS-Algebras and the Superstring Origin of Moonshine, Commun. Num. Theor. Phys. 10 (2016) 433–526, [1601.05412].
- [241] P. C. Argyres, A. Kapustin and N. Seiberg, On S-duality for non-simply-laced gauge groups, JHEP 06 (2006) 043, [hep-th/0603048].
- [242] N. Dorey, C. Fraser, T. J. Hollowood and M. A. C. Kneipp, S duality in N=4 supersymmetric gauge theories with arbitrary gauge group, Phys. Lett. B383 (1996) 422-428, [hep-th/9605069].
- [243] D. Kläwer, Modular curves and the refined distance conjecture, JHEP 12 (2021) 088, [2108.00021].
- [244] P. C. Argyres, C. Long and M. Martone, The Singularity Structure of Scale-Invariant Rank-2 Coulomb Branches, JHEP 05 (2018) 086, [1801.01122].
- [245] M. Martone, Towards the classification of rank-r $\mathcal{N} = 2$ SCFTs. Part I: twisted partition function and central charge formulae, 2006.16255.
- [246] P. C. Argyres and M. Martone, Towards a classification of rank $r \mathcal{N} = 2 SCFTs$ Part II: special Kahler stratification of the Coulomb branch, 2007.00012.
- [247] J. P. Yamron, Topological actions from twisted supersymmetric theories, Physics Letters B 213 (1988) 325 – 330.
- [248] D. Anselmi and P. Fré, Topological twist in four dimensions, r-duality and hyperinstantons, Nuclear Physics B 404 (08, 1993) 288–320.
- [249] D. Anselmi and P. Fré, Topological sigma-models in four dimensions and triholomorphic maps, Nuclear Physics B **416** (03, 1994) 255–300.
- [250] D. Anselmi and P. Frè, Gauged hyperinstantons and monopole equations, Physics Letters B 347 (1995) 247 – 254.
- [251] M. Alvarez and J. Labastida, Breaking of topological symmetry, Physics Letters B 315 (1993) 251 – 257.
- [252] M. Alvarez and J. Labastida, Topological matter in four-dimensions, Nucl. Phys. B 437 (1995) 356-390, [hep-th/9404115].
- [253] J. Labastida and M. Marino, A topological lagrangian for monopoles on four-manifolds, Physics Letters B 351 (03, 1995) 146–152.
- [254] J. M. F. Labastida and M. Marino, NonAbelian monopoles on four manifolds, Nucl. Phys. B 448 (1995) 373-398, [hep-th/9504010].
- [255] S. Hyun, J. Park and J.-S. Park, Spin-c Topological QCD, Nucl. Phys. B453 (1995) 199-224, [hep-th/9503201].
- [256] R. Dijkgraaf, J.-S. Park and B. J. Schroers, N=4 supersymmetric Yang-Mills theory on a Kahler surface, hep-th/9801066.
- [257] L. Göttsche, H. Nakajima and K. Yoshioka, Donaldson = seiberg-witten from mochizuki's formula and instanton counting, 1001.5024.
- [258] P. M. N. Feehan and T. G. Leness, PU(2) monopoles. I. Regularity, Uhlenbeck compactness, and transversality, J. Diff. Geom. 49 (1998) 265–410, [dg-ga/9710032].

- [259] P. M. N. Feehan and T. G. Leness, PU(2) monopoles and relations between four manifold invariants, Topology and its Applications 88 (1998) 111–145, [dg-ga/9709022].
- [260] P. M. N. Feehan and T. G. Leness, PU(2) monopoles. 2. Top level Seiberg-Witten moduli spaces and Witten's conjecture in low degrees, dg-ga/9712005.
- [261] P. M. N. Feehan and T. G. Leness, SO(3) monopoles, level one Seiberg-Witten moduli spaces, and Witten's conjecture in low degrees, Topology Appl. 124 (2002) 221-326, [math/0106238].
- [262] H. Nakajima and K. Yoshioka, Instanton counting on blowup. II. K-theoretic partition function, math/0505553.
- [263] H. Nakajima and K. Yoshioka, Perverse coherent sheaves on blowup, iii: Blow-up formula from wall-crossing, Kyoto Journal of Mathematics 51 (2011) 263–335.
- [264] M. Marcolli, Seiberg-Witten gauge theory. Hindustan Book Agency, 2011.
- [265] J. Bryan and R. Wentworth, The multi-monopole equations for Kähler surfaces, Turkish J. Math. 20 (1996) 119–128.
- [266] V. Pidstrigach and A. Tyurin, Localisation of the donaldson's invariants along seiberg-witten classes, arXiv preprint dg-ga/9507004 (1995).
- [267] C. Okonek and A. Teleman, Recent developments in seiberg-witten theory and complex geometry, arXiv preprint alg-geom/9612015 (1996).
- [268] M. Marino, Topological quantum field theory and four manifolds, in 3rd European Congress of Mathematics: Shaping the 21st Century, 8, 2000. hep-th/0008100.
- [269] G. Moore, "Lectures on the physical approach to donaldson and seiberg-witten invariants of four-manifolds." URL: https: //www.physics.rutgers.edu/~gmoore/SCGP-FourManifoldsNotes-2017.pdf, 2017.
- [270] E. D'Hoker, I. Krichever and D. Phong, The effective prepotential of n = 2 supersymmetric su(nc) gauge theories, Nuclear Physics B 489 (Mar, 1997) 179–210.
- [271] Y. Ohta, Non-perturbative solutions to n=2 supersymmetric yang-mills theories -progress and perspective-, 1999.
- [272] N. Nekrasov and A. Okounkov, Seiberg-Witten theory and random partitions, Prog. Math. 244 (2006) 525-596, [hep-th/0306238].
- [273] S. K. Donaldson and P. B. Kronheimer, The geometry of four-manifolds / S.K. Donaldson and P.B. Kronheimer. Clarendon Press; Oxford University Press Oxford : New York, 1990.
- [274] G. W. Moore, N. Nekrasov and S. Shatashvili, Integrating over Higgs branches, Commun. Math. Phys. 209 (2000) 97–121, [hep-th/9712241].
- [275] A. E. Nelson and N. Seiberg, R symmetry breaking versus supersymmetry breaking, Nucl. Phys. B 416 (1994) 46–62, [hep-ph/9309299].
- [276] C. Córdova and T. T. Dumitrescu, Candidate Phases for SU(2) Adjoint QCD_4 with Two Flavors from $\mathcal{N} = 2$ Supersymmetric Yang-Mills Theory, 1806.09592.
- [277] A. Losev, N. Nekrasov and S. Shatashvili, *Testing seiberg-witten solution*, 1998.
- [278] M. Mariño, The uses of whitham hierarchies, Progress of Theoretical Physics Supplement 135 (1999) 29–52.
- [279] K. Takasaki, Integrable hierarchies and contact terms in u-plane integrals of topologically twisted supersymmetric gauge theories, International Journal of Modern Physics A 14 (Mar, 1999) 1001–1013.
- [280] R. E. Borcherds, Automorphic forms with singularities on Grassmannians, Invent. Math. 132 (1998) 491 doi:10.1007/s002220050232 (1998), [alg-geom/9609022].

- [281] K. Bringmann, N. Diamantis and S. Ehlen, Regularized inner products and errors of modularity, International Mathematics Research Notices 2017 (2017) 7420–7458.
- [282] J. H. Bruinier and J. Funke, On two geometric theta lifts, Duke Mathematical Journal 125 (2004) 45–90.
- [283] W. Lerche, A. N. Schellekens and N. P. Warner, Lattices and Strings, Phys. Rept. 177 (1989) 1.
- [284] L. J. Dixon, V. Kaplunovsky and J. Louis, Moduli dependence of string loop corrections to gauge coupling constants, Nucl. Phys. B355 (1991) 649–688.
- [285] J. A. Harvey and G. W. Moore, Algebras, BPS states, and strings, Nucl. Phys. B463 (1996) 315-368, [hep-th/9510182].
- [286] J. Aspman, E. Furrer and J. Manschot, "The *u*-plane integral of supersymmetric qcd, part ii." to appear, 2022.
- [287] G. H. J.H. Bruinier, G. van der Geer and D. Zagier, The 1-2-3 of Modular Forms. Springer-Verlag Berlin Heidelberg, 2008, 10.1007/978-3-540-74119-0.
- [288] H. Petersson, Konstruktion der Modulformen und der zu gewissen Grenzkreisgruppen gehörigen automorphen Formen von positiver reeller Dimension und die vollständige Bestimmung ihrer Fourierkoeffizienten, S.-B. Heidelberger Akad. Wiss. Math.-Nat. Kl. (1950) 417–494.
- [289] K. Bringmann, N. Diamantis and S. Ehlen, Regularized inner products and errors of modularity, International Mathematics Research Notices 2017 (2017) 7420–7458.
- [290] M. Bershadsky, A. Johansen, V. Sadov and C. Vafa, Topological reduction of 4d sym to 2d σ-models, Nuclear Physics B 448 (1995) 166–186.
- [291] J. A. Harvey, G. W. Moore and A. Strominger, *Reducing S-duality to T-duality*, *Phys. Rev. D* 52 (1995) 7161-7167, [hep-th/9501022].
- [292] V. Muñoz, Quantum cohomology of the moduli space of stable bundles over a riemann surface, Duke mathematical journal 98 (1999) 525–540.
- [293] V. Muñoz, Gromov-Witten invariants of the moduli of bundles on a surface, arXiv Mathematics e-prints (Oct., 1999) math/9910105, [math/9910105].
- [294] S. K. Donaldson, Floer homology and algebraic geometry, pp. 119–138. London Mathematical Society Lecture Note Series. Cambridge University Press, 1995. 10.1017/CBO9780511569319.006.
- [295] V. Muñoz, Wall-crossing formulae for algebraic surfaces with q > 0, arXiv e-prints (Sept., 1997), [alg-geom/9709002].
- [296] M. Alim, S. Cecotti, C. Cordova, S. Espahbodi, A. Rastogi and C. Vafa, N = 2 quantum field theories and their BPS quivers, Adv. Theor. Math. Phys. 18 (2014) 27–127, [1112.3984].
- [297] M. Alim, S. Cecotti, C. Cordova, S. Espahbodi, A. Rastogi and C. Vafa, BPS Quivers and Spectra of Complete N=2 Quantum Field Theories, Commun. Math. Phys. 323 (2013) 1185–1227, [1109.4941].
- [298] H. Magureanu, Seiberg-Witten geometry, modular rational elliptic surfaces and BPS quivers, JHEP 05 (2022) 163, [2203.03755].
- [299] D. Mumford, Tata Lectures on Theta II. Birkhäuser Boston, MA, 01, 2007, 10.1007/978-0-8176-4578-6.
- [300] S. Grushevsky, The Schottky Problem, arXiv e-prints (Sept., 2010) arXiv:1009.0369, [1009.0369].
- [301] O. J. Ganor, D. R. Morrison and N. Seiberg, Branes, Calabi-Yau spaces, and toroidal compactification of the N=1 six-dimensional E(8) theory, Nucl. Phys. B 487 (1997) 93-127, [hep-th/9610251].

- [302] A. E. Lawrence and N. Nekrasov, Instanton sums and five-dimensional gauge theories, Nucl. Phys. B 513 (1998) 239-265, [hep-th/9706025].
- [303] D. R. Morrison and N. Seiberg, Extremal transitions and five-dimensional supersymmetric field theories, Nucl. Phys. B 483 (1997) 229-247, [hep-th/9609070].
- [304] O. J. Ganor and A. Hanany, Small E(8) instantons and tensionless noncritical strings, Nucl. Phys. B 474 (1996) 122–140, [hep-th/9602120].
- [305] E. Witten, Some comments on string dynamics, in STRINGS 95: Future Perspectives in String Theory, pp. 501–523, 7, 1995. hep-th/9507121.
- [306] N. Seiberg and E. Witten, Comments on string dynamics in six-dimensions, Nucl. Phys. B 471 (1996) 121-134, [hep-th/9603003].
- [307] N. Seiberg, Nontrivial fixed points of the renormalization group in six-dimensions, Phys. Lett. B 390 (1997) 169–171, [hep-th/9609161].
- [308] A. Strominger, Open p-branes, Phys. Lett. B 383 (1996) 44-47, [hep-th/9512059].
- [309] P. C. Argyres, M. Lotito, Y. Lü and M. Martone, Geometric constraints on the space of N = 2 SCFTs. Part II: construction of special Kähler geometries and RG flows, JHEP 02 (2018) 002, [1601.00011].
- [310] P. Argyres, M. Lotito, Y. Lü and M. Martone, Geometric constraints on the space of $\mathcal{N} = 2$ SCFTs. Part III: enhanced Coulomb branches and central charges, JHEP 02 (2018) 003, [1609.04404].
- [311] N. Wyllard, A(N-1) conformal Toda field theory correlation functions from conformal N = 2 SU(N) quiver gauge theories, JHEP 11 (2009) 002, [0907.2189].
- [312] B. Le Floch, A slow review of the AGT correspondence, J. Phys. A 55 (2022) 353002, [2006.14025].
- [313] P. C. Argyres, M. Plesser and N. Seiberg, The Moduli space of vacua of N=2 SUSY QCD and duality in N=1 SUSY QCD, Nucl. Phys. B 471 (1996) 159-194, [hep-th/9603042].
- [314] M. A. Luty and R. Rattazzi, Soft supersymmetry breaking in deformed moduli spaces, conformal theories, and N=2 Yang-Mills theory, JHEP 11 (1999) 001, [hep-th/9908085].
- [315] A. Bourget and J. Troost, Duality and modularity in elliptic integrable systems and vacua of $\mathcal{N} = 1^*$ gauge theories, JHEP 04 (2015) 128, [1501.05074].
- [316] T. D. Brennan, C. Cordova and T. T. Dumitrescu, Line Defect Quantum Numbers & Anomalies, 2206.15401.
- [317] D. Zagier, Introduction to modular forms; From Number Theory to Physics. Springer, Berlin (1992), pp. 238-291, 1992.
- [318] E. Previato, T. Shaska and G. S. Wijesiri, *Thetanulls of cyclic curves of small genus*, arXiv e-prints (Jan, 2013) arXiv:1301.4595, [1301.4595].
- [319] E. Freitag, Siegelsche Modulfunktionen, vol. 254 of Grundlehren der mathematischen Wissenschaften. Springer-Verlag Berlin Heidelberg, 1 ed., 1983.
- [320] K. Chandrasekharan, *Elliptic Functions*. Grundlehren der mathematischen Wissenschaften. Springer Berlin Heidelberg, 2012.
- [321] R. A. Rankin, Modular Forms and Functions. Cambridge University Press, 1977, 10.1017/CBO9780511566035.
- [322] R. W. Bruggeman, Families of Automorphic Forms. Birkhäuser Basel, 1994, 10.1007/978-3-0346-0336-2.
- [323] K. Conrad, "sl₂(F)." URL: https://kconrad.math.uconn.edu/blurbs/grouptheory/SL(2,Z).pdf.

- [324] A. Pitale, Siegel Modular Forms: A Classical and Representation-Theoretic Approach. Lecture Notes in Mathematics. Springer International Publishing, 2019.
- [325] "Fundamental domain drawer." https://wstein.org/Tables/fundomain.
- [326] "Fareysymbol, sage." https://doc.sagemath.org/html/en/reference/arithgroup/sage/modular/arithgroup/farey_symbol.html.
- [327] A. Sebbar, Classification of torsion-free genus zero congruence groups, Proceedings of the American Mathematical Society 129 (2001) 2517–2527.
- [328] S. A. Vidal, Sur la classification et le denombrement des sous-groupes du groupe modulaire et de leurs classes de conjugaison, 2007.
- [329] K. Wohlfahrt, An extension of f. klein's level concept, Illinois J. Math. 8 (09, 1964) 529–535.
- [330] R. S. Maier, Nonlinear Differential Equations Satisfied by Certain Classical Modular Forms, arXiv e-prints (July, 2008) arXiv:0807.1081, [0807.1081].
- [331] B. J. Birch and W. Kuyk, eds., Modular Functions of One Variable IV. Springer Berlin Heidelberg, 1975, 10.1007/bfb0097580.
- [332] V. Anagiannis and M. C. Cheng, TASI Lectures on Moonshine, PoS TASI2017 (2018) 010, [1807.00723].
- [333] B. H. Lian and S.-T. Yau, Arithmetic properties of mirror map and quantum coupling, Commun. Math. Phys. 176 (1996) 163–192, [hep-th/9411234].
- [334] S. Kachru, Elementary introduction to Moonshine, in "Mathematics of String Theory", 5, 2016. 1605.00697.
- [335] M. C. Cheng, J. F. Duncan and J. A. Harvey, Umbral Moonshine, Commun. Num. Theor. Phys. 08 (2014) 101–242, [1204.2779].
- [336] N. M. Paquette, D. Persson and R. Volpato, BPS Algebras, Genus Zero, and the Heterotic Monster, J. Phys. A 50 (2017) 414001, [1701.05169].
- [337] J. McKay and A. Sebbar, *Replicable Functions: An Introduction*, pp. 373–386. Springer Berlin Heidelberg, Berlin, Heidelberg, 2007.
- [338] R. E. Borcherds, Monstrous moonshine and monstrous lie superalgebras, Inventiones mathematicae 109 (1992) 405–444.
- [339] T. Gannon, Monstrous moonshine: The First twenty five years, math/0402345.
- [340] C. A. Kurth and L. Long, On modular forms for some noncongruence subgroups of sl2(z), Journal of Number Theory 128 (2008) 1989 – 2009.
- [341] T. Weigand, F-theory, PoS TASI2017 (2018) 016, [1806.01854].